SAFETY ANALYSIS REPORT

for the

HOLTEC INTERNATIONAL STORAGE, TRANSPORT AND REPOSITORY CASK SYSTEM (HI-STAR 100 CASK SYSTEM)

NRC DOCKET NO. 71-9261

HOLTEC REPORT HI-951251



HOLTEC INTERNATIONAL

STORAGE, TRANSPORT, AND REPOSITORY CASK SYSTEM

(HI-STAR CASK SYSTEM)

SAFETY ANALYSIS REPORT

10 CFR 71

DOCKET 71-9261

TABLE OF CONTENTS

CHAPTER 1: GENERAL INFORMATION

1.0	GENERAL INFORMATION 1.0-1
	1.0.1 Engineering Change Orders1.0-1
1.1	INTRODUCTION 1.1-1
1.2	PACKAGE DESCRIPTION 1.2-1
	1.2.1 Packaging 1.2-1 1.2.1.1 Gross Weight 1.2-2 1.2.1.2 Materials of Construction, Dimensions and Fabrication 1.2-3 1.2.1.3 Impact Limiters 1.2-9 1.2.1.4 Shielding 1.2-9 1.2.1.5 Lifting and Tie-Down Devices 1.2-14 1.2.1.6 Heat Dissipation 1.2-15 1.2.1.7 Coolants 1.2-16 1.2.1.8 Pressure Relief Systems 1.2-16 1.2.1.9 Descurity Seal 1.2-17 1.2.1.10 Design Life 1.2-17 1.2.2 Operational Features 1.2-17 1.2.2.1 Applicability of Operating Procedures for the Dual-Purpose HI-STAR 100 System HI-STAR 100 System 1.2-18 1.2.3 Contents of Package 1.2-21 1.2.3.1 Determination of Design Basis Fuel 1.2-21 1.2.3.2 Design Payload for Intact Fuel 1.2-22 1.2.3.3 Design Payload for Damaged Fuel and Fuel Debris 1.2-23 1.2.3.4 Structural Payload Parameters 1.2-25 1.2.3.5 Thermal Payload Parameter
1.3	DESIGN CODE APPLICABILITY 1.3-1
1.4	DRAWINGS 1.4-1
HI-ST REPO	AR SAR Revision 10 RT HI-951251 i

1.5 C	OMPLIANCE WITH 10CFR71 1.5-1
1.6 F	EFERENCES 1.6-1
Appendi Appendi Appendi Appendi	x 1.A:Alloy X Descriptionx 1.B:Holtite™ Material Datax 1.C:Miscellaneous Material Datax 1.D:Comment Resolution Letters
СНАРТ	ER 2: STRUCTURAL EVALUATION
2.0 S	TRUCTURAL EVALUATION
2.1 S	TRUCTURAL DESIGN 2.1-1
2	1.1 Discussion
2	1.2 Design Criteria 2.1-5 2.1.2.1 Loading and Load Combinations 2.1-6 2.1.2.2 Allowables 2.1-11 2.1.2.3 Brittle Fracture Failure 2.1-14 2.1.2.4 Impact Limiter 2.1-16 2.1.2.5 Buckling 2.1-16
2.2 V	EIGHTS AND CENTERS OF GRAVITY
2.3 N	ECHANICAL PROPERTIES OF MATERIALS
2	3.1 Structural Materials 2.3-1 2.3.1.1 Alloy X 2.3-1 2.3.1.2 Carbon Steel, Low-Alloy, and Nickel Alloy Steel 2.3-2 2.3.1.3 Bolting Materials 2.3-2 2.3.1.4 Weld Material 2.3-2 2.3.1.5 Impact Limiter 2.3-2
2	3.2 Nonstructural Materials 2.3-5 2.3.2.1 Neutron Shield 2.3-5 2.3.2.2 Boral [™] Neutron Absorber 2.3-5 2.3.2.3 Aluminum Heat Conduction Elements 2.3-5

HI-STAR SAR REPORT HI-951251 Revision 10

ŧ

2.4	GEN	ERAL STANDARDS FOR ALL PACKAGES 2.4-1
	2.4.1	Minimum Package Size 2.4-1
	2.4.2	Tamperproof Feature 2.4-1
	2.4.3	Positive Closure
	2.4.4	Chemical and Galvanic Reactions 2.4-1
2.5	LIFT	NG AND TIE-DOWN STANDARDS 2.5-1
	2.5.1	Lifting Devices2.5-12.5.1.1 Overpack Trunnion Analysis2.5-32.5.1.2 Stresses in the Overpack Closure Plate, Main Flange, andDevice Plate Device Plate
		Baseplate During Lifting 2.5-4 2.5.1.3 MPC Lifting Analysis 2.5-6
	2.5.2	Tie-Down Devices2.5-72.5.2.1 Discussion2.5-72.5.2.2 Equilibrium Equations to Determine the Tie-Down
		Forces 2.5-10 2.5.2.3 Longitudinal Loading 2.5-11 2.5.2.4 Vertical Load 2.5-11 2.5.2.5 Lateral Load 2.5-12 2.5.2.6 Numerical Results for Tie-Down Reactions 2.5-12
		2.5.2.7 Structural Integrity of Pocket Trunnions 2.5-14
	2.5.3	Failure of Lifting and Tie-Down Devices 2.5-15
	2.5.4	Conclusion
2.6	NORN	MAL CONDITIONS OF TRANSPORT 2.6-1
	2.6.1	Heat2.6-12.6.1.1 Summary of Pressures and Temperatures2.6-12.6.1.2 Differential Thermal Expansion2.6-12.6.1.3 Stress Calculations2.6-32.6.1.3.1MPC Stress Calculations2.6.1.3.2Overpack Stress Calculations2.6.1.3.3Fatigue Considerations2.6.1.3.4Fatigue Considerations
111.07		
HI-ST REPO	AK SAR RT HLQ	Revision 10

2614 Comparison	with Allowable Stresses 26.32
2.0.1.4 Comparison	MCC and Eval Dashet and Englasure
2.0.1.4.1	Mir C and ruci Basket and Enclosure
26142	V CSSCI
2.0.1.4.2	Domit Summer for the Normal Heat
2.0.1.4.5	Condition of Transmost 2.6.29
Cald	
2 6 2 1 Differential	Thermal Expansion 2.6.41
2.0.2.1 Differential	1 denimar Expansion
2.0.2.2 MIPC Suess	$\begin{array}{llllllllllllllllllllllllllllllllllll$
2.0.2.5 Overpack S	uess Analysis
Reduced External H	Pressure
Increased External	Pressure 2.6-47
Vibration	
Water Spray	
Free Drop	
Corner Drop	
Compression	2 6-49
compression	2.0 ()
Penetration	
THETICAL ACCID	ENT CONDITIONS 2.7-1
Free Drop	
2.7.1.1 End Drop .	
2.7.1.2 Side Drop (Load Cases F3 (Table 2.1.6), E3 (Table 2.1.7),
And 3 and 1	1 (Table 2.1.9) 2.7-13
2.7.1.3 Corner Drog	
2.7.1.4 Oblique Dro	ops
2.7.1.5 Comparison	with Allowable Stresses 2.7-24
Puncture	
Thermal	
2.7.3.1 Summary of	Pressures and Temperatures 2.7-27
2	
	2.6.1.4 Comparison 2.6.1.4.1 2.6.1.4.2 2.6.1.4.3 Cold 2.6.1.4.3 Cold 2.6.1.4.3 Cold 2.6.1.4.3 Cold 2.6.1.4.3 Cold 2.6.2.1 Differential 2.6.2.2 MPC Stress 2.6.2.3 Overpack S Reduced External F Increased External F Increased External F Vibration Water Spray Free Drop Compression Compression Penetration THETICAL ACCID Free Drop Free Drop 2.7.1.1 End Drop 2.7.1.2 Side Drop (I And 3 and 1 2.7.1.3 Corner Drop 2.7.1.4 Oblique Drop 2.7.1.5 Comparison Puncture Thermal

-	2.7.3.2 Differential Thermal Expansion
	2.7.3.3 Stress Calculations
	2.7.3.3.1 MPC 2.7-30
	2.7.3.3.2 Overpack
	2.7.3.3.3 Closure Bolts
	2.7.3.3.4 Bounding Thermal Stresses During the
	Fire Transient
	2.7.3.4 Comparison of Fire Accident Results with Allowable
	Stresses
	2.7.4 Immersion - Fissile Material 2.7-36
	2.7.5 Immersion - All Packages 2.7-36
	2.7.6 Summary of Damage 2.7-37
2.8	SPECIAL FORM
2.9	FUEL RODS
2.10	MISCELLANEOUS 2.10-1
	2.10.1 Appendices 2.10-1 2.10.2 Summary of NUREG-1617/10CFR71 Compliance 2.10-1
2.11	REFERENCES 2.11-1
Apper Apper Apper	ndix 2.A: Design, Testing, and Computer Simulation of the AL-STAR [™] Impact Limiter ndix 2.B: Summary of Results for Structural Integrity of Damaged Fuel Canisters ndix 2.C through 2.AO: Deleted
CHA	PTER 3: THERMAL EVALUATION
3.0	INTRODUCTION
3.1	DISCUSSION
3.2	SUMMARY OF THERMAL PROPERTIES OF MATERIALS
3.3	TECHNICAL SPECIFICATIONS FOR COMPONENTS
HI-ST	AR SAR Revision 10
REPOI	RT HI-951251

v

			TABLE OF CONTENTS (continued)
	3.3.1	Deleted	
	3.3.2	Deleted	
	3.3.3	Deleted	
.4	THER	MAL EVALU	JATION FOR NORMAL CONDITIONS
	OF TI	RANSPORT .	
	3.4.1	Thermal Mo	del
		3.4.1.1 Analy	vtical Model - General Remarks
		3.4.1.1.1	Overview of the Thermal Model
		3.4.1.1.2	Fuel Region Effective Thermal Conductivity
			Calculation
		3.4.1.1.3	Effective Thermal Conductivity of
			Sheathing/Boral/Cell Wall Sandwich
		3.4.1.1.4	Modeling of Basket Conductive Heat Transport
		3.4.1.1.5	Heat Transfer in MPC Basket Perinheral
			Regions 3.4-12
		34116	Effective Conductivity of Multilavered
		5	Intermediate Shell Region 34-14
		34117	Heat Rejection from Overnack and
		5.1.1.1.7	Impact Limiter Outside Surfaces 34.15
		34118	Determination of Solar Heat Input 34.16
		34110	Effective Thermal Conductivity of Radial
		5.4.1.1.7	Channels - Holtite Degion 3.4.16
		3 4 1 1 10	Effective Thermal Conductivity of the
		5.4.1.1.10	Encentric MPC to Overnock Con
		2/1111	Effective Thermal Conductivity of Elevible MPC
		5.4.1.1.11	Basket to Shall Aluminum Heat Conduction Elements 2.4.10
		2 4 1 1 1 2	ELUENT Model for LICSTAD
		5.4.1.1.12	Temperature Eigld Computation 2.4.21
		2 4 1 1 12	Temperature Field Computation
		3.4.1.1.13	Effect of Fuel Cladding Crud Resistance
		3.4.1.1.14	Maximum Time Limit During wet Transfer
		3.4.1.1.15	Cask Cooldown and Reflood Analysis
			During Fuel Unloading Operation 3.4-26
		3.4.1.1.16	MPC Temperature Distribution Under
		• • • • •	Vacuum Conditions 3.4-28
		3.4.1.1.17	Effects of Helium Dilution from Fuel
			Rod Gases 3.4-29

Revision 10

		TABLE OF CONTENTS (continued)	
-	<u></u>	3.4.1.1.18 HI-STAR Temperature Field With Low Heat Emitting Fuel	l
		3.4.1.2 Test Model 3.4-32	?
	3.4.2	Maximum Temperatures Under Normal Transport Conditions 3.4-32 3.4.2.1 Maximum Accessible Surface Temperatures 3.4-34	} }
	3.4.3	Minimum Temperatures	5
	3.4.4	Maximum Internal Pressures 3.4-36	j
	3.4.5	Maximum Thermal Stresses 3.4-37	r
	3.4.6	Evaluation of System Performance for Normal Conditions of Transport 3.4-37	,
3.5	НҮРС	OTHETICAL ACCIDENT THERMAL EVALUATION	
	3.5.1	Thermal Model 3.5-2 3.5.1.1 Analytical Model 3.5-2 3.5.1.2 Test Model 3.5-4	•
	3.5.2	System Conditions and Environment	
	3.5.3	System Temperatures 3.5-4	
	3.5.4	Maximum Internal Pressure 3.5-6	
	3.5.5	Maximum Thermal Stresses 3.5-6	
	3.5.6	Evaluation of System Performance for the Hypothetical Accident Thermal Conditions 3.5-6	
3.6	REGU	JLATORY COMPLIANCE	
3.7	REFE	RENCES	
Арре: Арре:	ndix 3.A ndix 3.B	Conservatisms in the Thermal Analysis of the HI-STAR System The Forced Helium Dehydration (FHD) System	
HI-ST	AR SAR	Revision 10	

CHAPTER 4: CONTAINMENT

4.0	INTRODUCTION 4.0-	
4.1	CON	FAINMENT BOUNDARIES 4.1-1
	4.1.1	Containment Vessel 4.1-1
	4.1.2	Containment Penetrations 4.1-2
	4.1.3	Seals and Welds 4.1-2 4.1.3.1 Containment Seals 4.1-2 4.1.3.2 Containment Welds 4.1-3
	4.1.4	Closure 4.1-3 4.1.4.1 Primary Closure 4.1-3 4.1.4.2 Secondary Closure 4.1-4
	4.1.5	Damaged Fuel Container 4.1-4
4.2	REQU ACCI	JIREMENTS FOR NORMAL AND HYPOTHETICAL DENT CONDITIONS OF TRANSPORT
	4.2.1	Containment Criteria 4.2-1
	4.2.2	Containment of Radioactive Material 4.2-1
	4.2.3	Pressurization of Containment Vessel 4.2-2
	4.2.4	Assumptions 4.2-2
	4.2.5	Analysis and Results4.2-44.2.5.1 Volume in the Containment Vessel4.2-44.2.5.2 Source Terms For Spent Nuclear Fuel Assemblies4.2-54.2.5.3 Effective A2 of Individual Contributors (Crud, Fines, Gases, and Volatiles)4.2-94.2.5.4 Releasable Activity4.2-104.2.5.5 Effective A2 for the Total Source Term4.2-104.2.5.6 Allowable Radionuclide Release Rates4.2-104.2.5.7 Allowable Leakage Rates at Operating Conditions4.2-11

HI-STAR SAR REPORT HI-951251 **Revision 10**

	4.2.5.8 Leakage Rate Acceptance Criteria for Test Conditions 4.2-11 4.2.5.9 10CFR71.63(b) Plutonium Leakage Verification 4.2-12 4.2.5.10 Leak Test Sensitivity 4.2-14
4.3	REGULATORY COMPLIANCE 4.3-1
4.4	REFERENCES
Appe Appe	ndix 4.A: Bolt and Plug Torques ndix 4.B: Manufacturer Seal Information
CHA	PTER 5: SHIELDING EVALUATION
5.0	INTRODUCTION
5.1	Discussion and Results 5.1-1
	5.1.1Normal Operations5.1-25.1.2Hypothetical Accident Conditions5.1-5
5.2	SOURCE SPECIFICATION 5.2-1
	5.2.1 Gamma Source
	5.2.2 Neutron Source 5.2-4
	5.2.3 Stainless Steel Clad Fuel Source 5.2-4
	5.2.4 Non-fuel Hardware 5.2-5 5.2.4.1 BPRAs and TPDs 5.2-5 5.2.4.2 RCCAs 5.2-6
	5.2.5Choice of Design Basis Assembly5.2-75.2.5.1 PWR Design Basis Assembly5.2-85.2.5.2 BWR Design Basis Assembly5.2-85.2.5.3 Decay Heat Loads5.2-10
	5.2.6 Thoria Rod Canister
	5.2.7 Fuel Assembly Neutron Sources .5.2-11 5.2.7.1 Dresden Unit 1 Neutron Source Assemblies .5.2-11
HI-S'	AR SAR Revision 10

		TABLE OF CONTENTS (continued)
		5.2.7.2 Trojan Nuclear Plant Neutron Source Assemblies
	5.2.8	Trojan Non-Fuel Bearing Components, Damaged Fuel and Fuel Debris5.2-12
5.3	MOD	EL SPECIFICATIONS
	5.3.1	Description of the Radial and Axial Shielding Configuration5.3-15.3.1.1 Fuel Configuration5.3-35.3.1.2 Streaming Considerations5.3-4
	5.3.2	Regional Densities 5.3-5
5.4	SHIEI	DING EVALUATION
	5.4.1	Streaming Through Radial Steel Fins and Pocket Trunnions
	5.4.2	Damaged Fuel Post-Accident Shielding Evaluation
	5.4.3	Mixed Oxide Fuel Evaluation 5.4-4
	5.4.4	Stainless Steel Clad Fuel Evaluation 5.4-5
	5.4.5	Dresden Unit 1 Antimony-Beryllium Neutron Sources
	5.4.6	Thoria Rod Canister 5.4-7
	5.4.7	Trojan Fuel Contents 5.4-7
	5.4.8	Trojan Antimony-Beryllium Neutron Sources
5.5	REGU	LATORY COMPLIANCE 5.5-1
5.6	REFEI	RENCES
Appen Appen Appen	dix 5.A dix 5.B dix 5.C	Sample Input File for SAS2H Sample Input File for ORIGEN-S Sample Input File for MCNP

Revision 10

CHAPTER 6: CRITICALITY EVALUATION

6.1	DISCUSSION AND RESULTS	
6.2	SPEN	T FUEL LOADING
	6.2.1	Definition of Assembly Classes 6.2-1
	6.2.2	PWR Fuel Assemblies in the MPC-24 6.2-2
	6.2.3	BWR Fuel Assemblies in the MPC-68 6.2-3
	6.2.4	Damaged BWR Fuel Assemblies and BWR Fuel Debris
	6.2.5	Thoria Rod Canister
	6.2.6	PWR Assemblies in the MPC-24E and MPC-24EF
	6.2.7	PWR Intact Fuel, Damaged Fuel And Fuel Debris in the Trojan MPC-24E/EF 6.2-6
	6.2.8	PWR Assemblies in the MPC-32
6.3	MOD	EL SPECIFICATION 6.3-1
	6.3.1	Description of Calculational Model 6.3-1
	6.3.2	Cask Regional Densities
6.4	CRITI	CALITY CALCULATIONS
	6.4.1	Calculational or Experimental Method
	6.4.2	Fuel Loading or Other Contents Loading Optimization 6.4-2
		6.4.2.1 Internal and External Moderation6.4-26.4.2.2 Partial Flooding6.4-56.4.2.3 Clad Gap Flooding6.4-66.4.2.4 Preferential Flooding6.4-66.4.2.5 Hypothetical Accident Conditions of Transport6.4-7

xi

	6.4.3	Criticality Results 6.4-8
	6.4.4	Damaged Fuel Container for BWR Fuel
	6.4.5	Fuel Assemblies with Missing Rods
	6.4.6	Thoria Rod Canister
	6.4.7	Sealed Rods Replacing BWR Water Rods
	6.4.8	Neutron Sources in Fuel Assemblies
	6.4.9	PWR Damaged Fuel and Fuel Debris
	6.4.10	Non-fuel Hardware in PWR Fuel Assemblies
	6.4.11	Reactivity Effect of Potential Boral Damage
6.5	CRITI	CALITY BENCHMARK EXPERIMENTS 6.5-1
6.6	REGU	LATORY COMPLIANCE 6.6-1
6.7	REFE	RENCES
Append Append Append Append Append	lix 6.A lix 6.B: lix 6.C: lix 6.D: lix 6.E:	Benchmark Calculations Distributed Enrichments in BWR Fuel Deleted Sample Input Files Burnup Credit for the MPC-32 (Later)
СНАР	TER 7:	OPERATING PROCEDURES
7.0	INTRO	DUCTION
	7.0.1	Technical and Safety Basis for Loading and Unloading Procedures
HI-STA	AR SAR	Revision 10

7.1	PROCEDURE FOR LOADING THE HI-STAR 100 SYSTEM IN THE SPENT FUEL POOL AND PREPARATION FOR SHIPMENT
	7.1.1 Overview of Loading Operations
	7.1.2 HI-STAR 100 System Receiving and Handling Operations
	7.1.3 HI-STAR 100 Overpack and MPC Receipt Inspection and Loading Preparation
	7.1.4 MPC Fuel Loading
	7.1.5 MPC Closure
	7.1.6 Preparation for Transport
	7.1.7 Placement of the HI-STAR 100 Overpack on the Transport Vehicle
7.2	PROCEDURE FOR UNLOADING THE HI-STAR 100 SYSTEM IN THE SPENT FUEL POOL
	7.2.1 Overview of HI-STAR 100 System Unloading Operations
	7.2.2 HI-STAR 100 Overpack Packaging Receipt
	7.2.3 Preparation for Unloading
	7.2.4 MPC Unloading
	7.2.5 Post-Unloading Operations
7.3	PREPARATION OF AN EMPTY PACKAGE FOR TRANSPORT
	7.3.1 Overview of the HI-STAR 100 System Empty Package Transport
	7.3.2 Preparation for Empty Package Shipment
7.4	PROCEDURE FOR PREPARING THE HI-STAR 100 OVERPACK FOR TRANSPORT FOLLOWING A PERIOD OF STORAGE7.4-1

	7 4 2	Following a	a Period of Storage
	7.4.2	гтератации	
7.5	REGU	JLATORY C	OMPLIANCE
7.6	REFE	RENCES .	
СНА	PTER	8: ACCEPT	ANCE CRITERIA AND MAINTENANCE PROGRAM
8.0	INTR	ODUCTION	
8.1	ACCI	EPTANCE C	RITERIA 8.1-1
	8.1.1	Fabrication	and Nondestructive Examination (NDE)
		8.1.1.1	MPC Lid-to-Shell Weld Volumetric Inspection
	812	Structural a	nd Pressure Tests 8 1-6
	•••••	8.1.2.1	Lifting Trunnions
		8.1.2.2	Pressure Testing
		8.1.2.3	Material Testing
		8.1.2.4	Pneumatic Bubble Testing of the Neutron Shield
			Enclosure Vessel
	8.1.3	Leakage Te	sting
	01110	8.1.3.1	HI-STAR Overnack 8.1-9
		8.1.3.2	MPC Secondary Containment Boundary
	8.1.4	Component	Tests
		8.1.4.1	Valves, Rupture Discs, and Fluid Transport Devices
		8.1.4.2	Seals and Gaskets
		8.1.4.3	Transport Impact Limiter
	8.1.5	Shielding Ir	ntegrity
		8.1.5.1	Fabrication Testing and Controls
		8.1.5.2	Shielding Effectiveness Tests
		8.1.5.3	Neutron Absorber Tests
	8.1.6	Thermal Ac	ceptance Test
HI-S	TAR SA	R	Revision 10
REPO	ORT HI-	951251	
			XIV

-	8.1.7 Cask Identification
8.2	MAINTENANCE PROGRAM
	8.2.1 Structural and Pressure Parts
	8.2.2 Leakage Tests
	8.2.3 Subsystem Maintenance
	8.2.4 Rupture Discs
	8.2.5 Shielding
	8.2.6 In-Service Thermal Test
	8.2.7 Miscellaneous
8.3	REGULATORY COMPLIANCE
8.4	REFERENCES

HI-STAR SAR REPORT HI-951251 Revision 10

LIST OF FIGURES

- 1.1.1 Pictorial View of HI-STAR 100 Package
- 1.1.2 HI-STAR 100 Overpack With MPC Partially Inserted
- 1.1.3 Cross Section Elevation View of MPC
- 1.1.4 Cross Section Elevation View of Overpack
- 1.1.5 Trojan MPC Spacer
- 1.2.1 Overpack Containment Boundary
- 1.2.2 Overpack Mid-Plane Cross Section
- 1.2.3 MPC-68/68F Cross Section View
- 1.2.4 MPC-32 Cross Section View
- 1.2.5 MPC-24/E/EF Cross Section View
- 1.2.6 MPC Lid and Port Details
- 1.2.7 HI-STAR 100 Overpack Shell Layering
- 1.2.8 HI-STAR 100 Transport Configuration
- 1.2.9 Major HI-STAR 100 Loading Operations (3 Sheets)
- 1.2.10 Holtec Damaged Fuel Container for Dresden Unit 1/Humboldt Bay SNF
- 1.2.10A Trojan Failed Fuel Can
- 1.2.10B Trojan Fuel Debris Process Can
- 1.2.10C Trojan Fuel Debris Process Can Capsule
- 1.2.10D Holtec Damaged Fuel Container for Trojan Plant SNF in MPC-24E/24EF
- 1.2.11 TN Damaged Fuel Canister for Dresden Unit -1

HI-STAR SAR REPORT HI-951251 **Revision 10**

xvi

	LIST OF FIGURES (continued)
1.2.11A	TN Thoria Rod Canister for Dresden Unit-1
1.2.12	Deleted
1.2.13	PWR Axial Burnup Profile with Normalized Distribution
1.2.13A	Trojan Plant Fuel Axial Burnup Distribution Profile
1.2.14	BWR Axial Burnup Profile with Normalized Distribution
1.2.15	HI-STAR 100 MPC With Upper and Lower Fuel Spacers
1.2.16	Major HI-STAR 100 Unloading Operations (3 Sheets)
1.2.17	Fuel Debris MPC ("F" Model)
1.A.1	Design Stress Intensity vs. Temperature
1.A.2	Tensile Strength vs. Temperature
1.A.3	Yield Stress vs. Temperature
1.A.4	Coefficient of Thermal Expansion vs. Temperature
1.A.5	Thermal Conductivity vs. Temperature
2.1.1	MPC Fuel Basket Geometry
2.1.2	Schematic of Closure Plate Bolted Joint
2.1.3	0° Drop Orientation for the MPCs
2.1.4	45° Drop Orientation for the MPCs
2.1.5	Free Body Diagram of Overpack - Internal Pressure
2.1.6	Free Body Diagram of Overpack - External Pressure
2.1.7	Free Body Diagram of Overpack - Bottom End Drop
2.1.8	Free Body Diagram of Overpack - Top End Drop
HI-STAR S. REPORT H	AR Revision 10

	LIST OF FIGURES (continued)
2.1.9	Free Body Diagram of Overpack - Side Drop
2.1.10	Free Body Diagram for Bottom CG-Over-Corner Drop
2.1.11	Free Body Diagram for Top CG-Over-Corner Drop
2.1.12	Free Body Diagram for Puncture Drop Onto Bar - Side
2.1.13	Free Body Diagram for Puncture Drop Onto Bar - Top End
2.1.14	Free Body Diagram for Puncture Drop Onto Bar - Bottom End
2.2.1	HI-STAR 100 Datum Definition for Table 2.2.2
2.3.1	Cross Layered Aluminum Honeycomb
2.3.2	Average Crush Strength of Tested Specimens vs. Test Temperature
2.5.1	Free Body Sketch of Lifting Trunnion Threaded Region Showing Moment Balance by Shear Stress
2.5.2	Partial Free Body Diagram of Overpack Showing Applied Transport Load and Tie- Down Reaction Vectors at Saddle Supports
2.5.3 throu	gh 2.5.11: Deleted
2.5.12	Free Body at Tie Down Location with Saddle Support and Tie Down Strap - Vertical Reaction Load
2.5.13	Free Body at Tie Down Location with Saddle Support and Tie Down Strap - Lateral Reaction Load
2.6.1	Temperature Distribution for MPC Thermal Stress Analysis
2.6.2	Temperature Distribution for Overpack Thermal Stress Analysis
2.6.3	Finite Element Model of MPC-24/24E/24EF (Basic Model)
2.6.4	Finite Element Model for MPC-32 (Basic Model)
2.6.5	Finite Element Model of MPC-68 (Basic Model)
HISTAR	SAR Revision 10

	LIST OF FIGURES (continued)
2.6.6	Finite Element Model of MPC-24/24E/24EF (0 Degree Drop Model)
2.6.7	Finite Element Model of MPC-32 (0 Degree Drop Model)
2.6.8	Finite Element Model of MPC-68 (0 Degree Drop Model)
2.6.9	Finite Element Model of MPC-24/24E/24EF (45 Degree Drop Model)
2.6.10	Finite Element Model of MPC-24/24E/24EF (45 Degree Drop Model)
2.6.11	Finite Element Model of MPC-68 (45 Degree Drop Model)
2.6.12	MPC Thermal Load
2.6.13	0 Degree Side Drop of MPC
2.6.14	45 Degree Side Drop of MPC
2.6.15	Detail of Fuel Assembly Pressure Load on MPC Basket
2.6.16	Overpack Finite Element Model
2.6.17	Overpack Bottom Plate
2.6.18	Overpack Top Forging
2.6.19	Overpack Lid
2.6.19A	Overpack Inner and Intermediate Shells
2.6.19B	Overpack Holtite-A Elements
2.6.19C	Overpack Outer Enclosure
2.6.20	Confinement Boundary Model Showing Temperature Data Points
2.6.21	MPC Confinement Boundary Finite Element Grid (Exploded View)
2.6.22	Nodal Coupling in Overpack Finite Element Model
2.6.23	Overpack Internal Pressure Loading

xix

2.6.24	Simulation Model for Fabrication Stresses in the Overpack
2.6.25	Partial Free Body Diagram of a Shell Section
2.7.1	Non-Linear Buckling Analysis for MPC-24 Displacement vs. Impact Acceleration (0° Drop)
2.7.2	Non-Linear Buckling Analysis for MPC-24 Displacement vs. Impact Acceleration (45° Drop)
2.7.3	Non-Linear Buckling Analysis for MPC-32 Displacement vs. Impact Acceleration (0° Drop)
2.7.4	Non-Linear Buckling Analysis for MPC-32 Displacement vs. Impact Acceleration (45° Drop)
2.7.5	Non-Linear Buckling Analysis for MPC-68 Displacement vs. Impact Acceleration (0° Drop)
2.7.6	Non-Linear Buckling Analysis for MPC-68 Displacement vs. Impact Acceleration (45° Drop)
2.7.7	Impact Limiter Loaded Areas on the Overpack
2.7.8	Top Lid Loading - Drop Analysis
2.7.9	Side View of Top Forging Showing End Loads
2.7.10	Baseplate Loading From Impact Limiter - Top End Drop
2.7.11	Loading From MPC On Inner Shell
2.7.12	End View Showing MPC Loading On Inner Shell
2.7.13	Load From Impact Limiter At Support Locations - Side Drop
2.7.14	Oblique Drop Showing Offset of Impact Limiter Reaction Load From Applied Inertia Loads
2.7.15	Inner Shell Stress Intensity Distribution - "Heat" Condition - 30 Degree - Top End Impact

2.7.16	Intermediate Shell - Stress Intensity Distribution - "Heat" Condition - 30 Degree - Top End Impact
2.7.17	Localized Stress Intensity Distribution In Lid - Top End Puncture - (Deformed Shape Expanded For Clarity
2.7.18	Impact Loads For Slapdown Finite Element Analysis
2.7.19 throu	igh 2.7.22: Deleted
2.9.1	Fuel Rod Deformation; $g_1 > 0$
2.9.2	Fuel Rod Deformation; $g_1 = 0$
2.9.3	Fuel Rod Deformation; $g_1 = 0$, $F_2 > F_1$
2.9.4	Inter-Grid Strap Deformation; $F_3 > F_2$
2.9.5	Point Contact at Load F ₄ , Maximum Bending Moment at A'
2.9.6	Extended Region of Contact, $F_5 > F_4$, Zero Bending Moment at A'
2.9.7	Free Body Diagram When Moment at A' = 0, $P_{AX} = F_5/\cos\theta$. Resisting Moment M_R At Grid Strap Not Shown
2.9.8	View C-C
2.9.9	Exaggerated Detail Showing Multiple Fuel Rods Subject to Lateral Deflection with Final Stacking of Rod Column
2.A.1.1	HI-STAR 100 Package With Top and Bottom Impact Limiters
2.A.1.2	Drop From 9 Meters On To An Essentially Unyielding Surface (Hypothetical Accident Condition)
2.A.1.3	Pictorial View of AL-STAR (With a Portion Removed)
2.A.2.1	Pressure-Crush Strain Curve (Static Testing)
2.A.3.1	Illustration of the Force-Crush Model Construction Using the Example of the Side Drop Test
HI-STAR S	AR Revision 10

	LIST OF FIGURES (continued)		
2.A.4.1	Test Fixture		
2.A.4.2	Internal Stiffening Structure in 1/8 Scale Model		
2.A.4.3	1/8th Scale Initial Impact Limiter Configuration - Comparison of Static Force-Crush Data from Test and Simulation - END DROP		
2.A.4.4	1/8th Scale Initial Impact Limiter Configuration - Comparison of Static Force-Crush Data from Test and Simulation - 60 DEGREE CRUSH		
2.A.5.1	Quarter Scale Model of Loaded MPC For Quarter Scale Drop Test Experiment		
2.A.5.2	Overpack Quarter Scale Model (Cross Section View)		
2.A.5.3	Overpack Quarter Scale Model (Top View)		
2.A.5.4	Overpack Quarter Scale Model (Bottom View)		
2.A.5.5	Accelerometer Locations for End and Side Drops		
2.A.5.6	Accelerometer Locations for Slap Down and CGOC Drops		
2.A.5.7	1/4 Scale HI-STAR 100 Packaging at 30 FT (9M) Prior to Side Drop		
2.A.5.8	1/4 Scale Bottom Impact Limiter After Side Drop		
2.A.5.9	1/4 Scale Top Impact Limiter After Side Drop		
2.A.5.10	1/4 Scale HI-STAR 100 Packaging After Slap Down Drop		
2.A.5.11	1/4 Scale Impact Limiter After C.G. Over Corner Drop		
2.A.5.12	1/4 Scale HI-STAR 100 Packaging After Top End Drop		
2.A.5.13	1/4 Scale Impact Limiter Top End Drop		
2.A.5.14	Acceleration Raw Data for Top End Drop		
2.A.5.15	Acceleration Data Filtered at 450 Hz for Top End Drop		
2.A.5.15A	Top End Drop, Cutoff Frequency = 450 Hz		
HI-STAR SA REPORT HI	HI-STAR SAR Revision 10 REPORT HI-951251		

xxii

LIST OF FIGURES (continued)

- 2.A.5.15B Top End Drop, Cutoff Frequency = 550 Hz
- 2.A.5.15C Top End Drop, Cutoff Frequency = 1250 Hz
- 2.A.5.16 Acceleration Raw Data for C.G. Over Corner Drop
- 2.A.5.17 Acceleration Data Filtered at 450 Hz for C.G. Over Corner Drop
- 2.A.5.17.A CG Over Corner, Cutoff Frequency = 450 Hz
- 2.A.5.18 Acceleration Raw Data at Bottom End During Slapdown Drop
- 2.A.5.19 Acceleration Data Filtered at 350 Hz for Slapdown Drop
- 2.A.5.19A Slapdown, Cutoff Frequency = 350 Hz
- 2.A.5.20 Acceleration Raw Data for Side Drop
- 2.A.5.21 Acceleration Data Filtered at 350 Hz for Side Drop
- 2.A.5.21A Side Drop, Cutoff Frequency = 350 Hz
- 2.A.6.1 Impact Limiter Force vs. Crush Depth ($\theta = 90$ degrees)
- 2.A.6.2 Impact Limiter Force vs. Crush Depth ($\theta = 67.5$ degrees)
- 2.A.6.3 Impact Limiter Force vs. Crush Depth ($\theta = 0$ degrees)
- 2.A.6.4 Impact Limiter Force vs. Crush Depth ($\theta = 15$ degrees)
- 2.A.6.5 Dynamic Model for Dual Impact Scenarios
- 2.A.6.6 Free-Body Diagram for Impact Scenarios
- 2.A.7.1 Static Crush Force vs. Crush Depth Impact at 30 Degrees with Horizontal Target
- 2.A.7.2 Static Crush Force vs. Crush Depth Impact at 45 Degrees with Horizontal Target
- 2.A.7.3 Static Crush Force vs. Crush Depth Impact at 60 Degrees with Horizontal Target
- 2.A.10.1 1/8th Scale Impact Limiter Crush Force vs. Crush Depth Side Orientation

HI-STAR SAR REPORT HI-951251

2.A.10.2	1/8th Scale Impact Limiter - Crush Force vs. Crush Depth - Center of Gravity Ove Corner Orientation
2.A.10.3	1/8th Scale Impact Limiter - Crush Force vs. Crush Depth - End Orientation
2.B.1 throug	h 2.G.1: Deleted
2.H.1.1 throu	1gh 2.H.10.3: Relocated to Appendix 2.A
2.I.1 through	2.AN.2: Deleted
3.3.1	Deleted
3.4.1	Homogenization of the Storage Cell Cross-Section
3.4.2	MPC Cross-Section Replaced with an Equivalent Two Zone Axisymmetric Model
3.4.3	Natural Convection in MPC Basket Periphery Region
3.4.4	Neutron Shield Region Resistance Network Analogy for Effective Conductivity Calculation
3.4.5	Typical MPC Basket Parts in a Cross-Sectional View
3.4.6	"Box Wall-Boral-Sheathing" Sandwich
3.4.7	Westinghouse 17x17 OFA PWR Fuel Assembly Model
3.4.8	General Electric 9x9 BWR Fuel Assembly Model
3.4.9	Deleted
3.4.10	MPC-24 Basket Cross-Section ANSYS Finite Element Model
3.4.11	MPC-68 Basket Cross-Section ANSYS Finite Element Model
3.4.12	Illustration of an MPC Basket to Shell Aluminum Heat Conduction Element
3.4.13	Comparison of FLUENT Based Fuel Assembly Effective Conductivity Results with Published Technical Data

REPORT HI-951251

xxiv

3.4.14 3.4.15 3.4.16	HI-STAR 100 System Finite Element Mesh for Thermal Analysis
3.4.15 3.4.16	
3.4.16	Deleted
	HI-STAR 100 System Normal Transport Condition Temperature Contours Plot (MPC- 24 Basket)
3.4.17	HI-STAR 100 System Normal Transport Condition Temperature Contours Plot (MPC- 68 Basket)
3.4.18	Deleted
3.4.19	MPC-24 Hottest Rod Temperature Profile
3.4.20	MPC-68 Hottest Rod Temperature Profile
3.4.21	Deleted
3.4.22	MPC-24 Radial Temperature Profile
3.4.23	MPC-68 Radial Temperature Profile
3.4.24	HI-STAR 100 Package Control Locations Tracked in the Cooldown Event
3.4.25	Containment Boundary Cooldown Temperature Profiles
3.4.26	Multi-Layered Shells Cooldown Temperature Profiles
3.4.27	Overpack Forging Through Thickness Temperature Gradient During Cooldown Transient
3.4.28	Impact Limiters Surface Temperature Map
3.5.1	Location of HI-STAR 100 Package Control Points Monitored During Hypothetical Fire Accident Condition
3.5.2	HI-STAR 100 Package Containment Boundary Components Fire Accident Transient Temperature Response
3.5.3	HI-STAR 100 Package Neutron Shielding Region Fire Accident Transient Temperature Response

	LIST OF FIGURES (continued)
3.5.4	HI-STAR 100 Package MPC Shell and Fuel Basket Fire Accident Transient Temperature Response
3.5.5	Fuel Cladding Peak Axial Temperature Distribution During Post Fire Cooldown
3.5.6	HI-STAR Model for Transport Fire
3.5.7	Transport Fire Condition Containment Boundary and Layered Shells Temperature Distributions (Section A-A)
3.5.8	Transport Fire Condition Overpack Top Forging Temperature Distributions (Section B-B)
3.5.9	Transport Fire Condition Overpack Top Forging Temperature Distributions (Section C-C)
3.B.1	Schematic of the Forced Helium Dehydration System
4.1.1	HI-STAR 100 Overpack Primary Containment Boundary Components
4.1.2	HI-STAR 100 Closure Plate Containment Details
4.1.3	HI-STAR Overpack Vent and Drain Port Details
4.1.4	MPC Secondary Containment Boundary
5.1.1	Cross Section Elevation View of the HI-STAR 100 System with Dose Point Locations During Normal Conditions
5.1.2	Cross Section Elevation View of the HI-STAR 100 System with Dose Point Locations During Accident Conditions
5.3.1	HI-STAR 100 Overpack with MPC-32 Cross Sectional View as Modeled in MCNP
5.3.2	HI-STAR 100 Overpack with MPC-24 Cross Sectional View as Modeled in MCNP
5.3.3	HI-STAR 100 Overpack with MPC-68 Cross Sectional View as Modeled in MCNP
5.3.4	Cross Sectional View of an MPC-32 Basket Cell as Modeled in MCNP
5.3.5	Cross Sectional View of an MPC-24 Basket Cell as Modeled in MCNP
HI-STAR	SAR Revision 10

xxvi

5.3.6	Cross Sectional View of an MPC-68 Basket Cell as Modeled in MCNP
5.3.7	Axial Location of PWR Design Basis Fuel in the HI-STAR 100 System
5.3.8	Axial Location of BWR Design Basis Fuel in the HI-STAR 100 System
5.3.9	HI-STAR 100 Overpack with MPC-24 Cross Sectional View Showing the Thickness of the MPC Shell and Overpack as Modeled in MCNP
5.3.10	Axial View of HI-STAR 100 Overpack and MPC with Axial Dimensions Shown a Modeled in MCNP
5.3.11	Cross Section Elevation View of the HI-STAR 100 System With As-Modele Dimensions for the Impact Limiter
5.3.12	Cross Sectional View of Impact Limiter with As-Modeled Dimensions
5.4.1	Depiction of the Azimuthal Segmentation of the Overpack Used in Analyzing Neutro and Photon Streaming
5.4.2	Cross Sectional View of the Trojan MPC-24E Design Showing the Dose Location Use for the Secondary Source Calculations
5.3.1	Typical Cell in the Calculation Model (Planar Cross-Section) with Representative Fue in the MPC-24 Basket
5.3.1A	Typical Cell in the Calculation Model (Planar Cross-Section) with Representative Fue in the MPC-24E & MPC-24E/EF Trojan Basket
5.3.2	Typical Cell in the Calculation Model (Planar Cross-Section) with Representative Fue in the MPC-32 Basket
5.3.3	Typical Cell in the Calculation Model (Planar Cross-Section) with Representative Fue in the MPC-68 Basket
5.3.4	Calculation Model (Planar Cross-Section) with Fuel Illustrated in One Quadrant of th MPC-24, MPC-24E, and MPC-24E/EF Trojan
.3.5	Calculation Model (Planar Cross-Section) with Fuel Illustrated in One Quadrant of th MPC-32

xxvii

	LIST OF FIGURES (continued)
6.3.6	Calculation Model (Planar Cross-Section) with Fuel Illustrated in One Quadrant of th MPC-68
5.3.7	Sketch of the Calculational Model in the Axial Direction
5.4.1	Failed Fuel Calculation Model (Planar Cross-Section) with 6x6 Array with 4 Missin Rods in the MPC-68 Basket
5.4.2	Failed Fuel Calculation Model (Planar Cross-Section) with 6x6 Array with 8 Missin Rods in the MPC-68 Basket
5.4.3	Failed Fuel Calculation Model (Planar Cross-Section) with 6x6 Array with 12 Missin Rods in the MPC-68 Basket
5.4.4	Failed Fuel Calculation Model (Planar Cross-Section) with 6x6 Array with 18 Missin Rods in the MPC-68 Basket
5.4.5	Failed Fuel Calculation Model (Planar Cross-Section) with 7x7 Array with 8 Missin Rods in the MPC-68 Basket
5.4.6	Failed Fuel Calculation Model (Planar Cross-Section) with 7x7 Array with 13 Missin Rods in the MPC-68 Basket
5.4.7	Failed Fuel Calculation Model (Planar Cross-Section) with 7x7 Array with 24 Missin Rods in the MPC-68 Basket
5.4.8	Failed Fuel Calculation Model (Planar Cross-Section) with Damaged Fuel Collapse into 8x8 Array in the MPC-68 Basket
5.4.9	Calculated k-effective as a Function of Internal Moderator Density
5.4.10	Thoria Rod Canister (Planar Cross-Section) with 18 Thoria Rods in the MPC-68 Baske
5.4.11	Locations of the Damaged Fuel Containers in the MPC-24E/EF
5.4.12	Maximum k _{eff} for the MPC-24E/EF with Generic PWR Damaged Fuel Container, Initia Enrichment of 4.0 wt.% for Damaged and Intact Fuel
i.A.1	MCNP4a Calculated k-eff Values for Various Values of the Spectral Index
5.A.2	KENO5a Calculated k-eff Values for Various Values of the Spectral Index
HI-STAR S	SAR Revision 10

LIST OF FIGURES (continued)				
6.A.3	MCNP4a Calculated k-eff Values at Various U-235 Enrichments			
6.A.4	KENO5a Calculated k-eff Values at Various U-235 Enrichments			
6.A.5	Comparison of MCNP4a and KENO5a Calculations for Various Fuel Enrichments			
6.A.6	Comparison of MCNP4a and KENO5a Calculations for Various Boron-10 Areal Densities			
6.A.7	MCNP4a Calculated k-eff Values for Various values of the Spectral Index			
7.1.1	Loading Operations Flow Diagram			
7.1.2a	Major HI-STAR 100 Loading Operations (Sheet 1 of 3)			
7.1.2b	Major HI-STAR 100 Loading Operations (Sheet 2 of 3)			
7.1.2c	Major HI-STAR 100 Loading Operations (Sheet 3 of 3)			
7.1.3	Lift Yoke Engagement and Vertical HI-STAR Handling			
7.1.4	HI-STAR Upending/Downending			
7.1.5	HI-STAR 100 Overpack Rigging for Horizontal Handling			
7.1.6	MPC Upending in the MPC Upending Frame			
7.1.7	MPC Rigging for Vertical Lifts			
7.1.8	MPC Alignment in HI-STAR			
7.1.9	Personnel Barrier Removal and Installation			
7.1.10	Impact Limiter Handling			
7.1.11	HI-STAR 100 Impact Limiter and Bolting			
7.1.12	MPC Lid and HI-STAR Accessory Rigging (Closure Plate Shown)			
7.1.13	Annulus Seal/Annulus Shield and Seal Surface Protector			
HISTAD	NAP Participa 10			

HI-STAR SAR REPORT HI-951251 Revision 10

xxix

LIST OF FIGURES (continued) 7.1.14 **Fuel Spacers** 7.1.15 **Drain Port Details** 7.1.16 **Drain Line Positioning** 7.1.17 Annulus Overpressure System 7.1.18 HI-STAR 100 Lid Retention System in Exploded View 7.1.19 **Drain Line Installation** 7.1.20 **Temporary Shield Ring** 7.1.21 MPC Vent and Drain Port RVOA Connector 7.1.22 Water Pump-Down for MPC Lid Welding Operations (Example P&I Diagram) 7.1.23 MPC Air Displacement and Pressure Testing (Example P&I Diagram) 7.1.24 MPC Blowdown and Helium Injection for Leak Testing (Example P&I Diagram) 7.1.25 Forced Helium Dehydration System (Example P&I Diagram) 7.1.26 Helium Backfill (Example P&I Diagram) 7.1.27 HI-STAR 100 Overpack Test Cover 7.1.28 HI-STAR 100 Backfill Tool 7.1.29 HI-STAR 100 Closure Plate Test Tool 7.1.30 HI-STAR Closure Plate Bolt Torquing Pattern 7.2.1 Unloading Operations Flow Diagram 7.2.2a Major HI-STAR 100 Unloading Operations (Sheet 1 of 3) 7.2.2b Major HI-STAR 100 Unloading Operations (Sheet 2 of 3) 7.2.2c Major HI-STAR 100 Unloading Operations (Sheet 3 of 3) HI-STAR SAR **Revision 10** REPORT HI-951251

LIST OF FIGURES (continued)

- 7.2.3 HI-STAR Annulus Gas Sampling
- 7.2.4 MPC Gas Sampling in Preparation for Unloading (Example P&I Diagram)
- 7.2.5 MPC Cool-Down (Example P&I Diagram)
- 8.1.1 Deleted
- 8.1.2 Thermocouple Locations
- 8.1.3 Overpack Steam Heated Test Temperature Contours Plot
- 8.1.4 Overpack Surface Temperature History During A Steam Heated Test
- 8.2.1 Temperature Measurement Locations for Periodic Thermal Test

Revision 10

LIST OF EFFECTIVE PAGES FOR PROPOSED REVISION 10

Page	Revision	Page	Revision
<u> </u>	10	XXX	10
ii	10	xxxi	10
iii	10	xxxdi	Deleted
iv	10	xxxiii	Deleted
V	10	xxxiv	Deleted
vi	10		
vii	10	·	·····
viii	10		
ix	10		
x	10	·····	
xi	10		
xii	10		
xiii	10		
xiv	10		
xv	10	· · · · · · · · · · · · · · · · · · ·	
xvi	10		
xvii	10		· · · · · · · · · · · · · · · · · · ·
Xviii	10		
xix	10		
xx	10		
xxi	10		
xxii	10		
xxiii	10		
xxiv	10		
XXV	10		
xxvi	10		
xxvii	10		
xxviii	10		
xxix	10		
XXX	10		
xxvi	10		
xxvii	10		
xxvi	10		
xxvii	10		
	10		
XXIX	10	·	
	10		
	10		
	10		
	10		
	10		
	10		
	10		
	10		
	10		
	10		
	10		
	10		
	10		
	10		
XXIX	10		

HI-STAR 100 SAR

Page	Revision	Page	Revision
1.0-1	10	1.2-37	10
1.0-2	10	1.2-38	10
1.0-3	10	1.2-39	10
1.0-4	10	1.2-40	10
1.0-5	10	1.2-41	10
1.0-6	10	1.2-42	10
1.0-7	10	1.2-43	10
1.0-8	10	1.2-44	10
1.0-9	10	1.2-45	10
1.1-1	10	1.2-46	10
Fig. 1.1.1	7	1.2-47	10
Fig. 1.1.2	10	1.2-48	10
Fig. 1.1.3	10	1.2-49	10
Fig. 1.1.4	10	1.2-50	10
1.2-1	10	1.2-51	10
1.2-2	10	1.2-52	10
1.2-3	10	1.2-53	10
1.2-4	10	1.2-54	10
1.2-5	10	1.2-55	10
1.2-6	10	1.2-56	10
1.2-7	10	1.2-57	10
1.2-8	10	1.2-58	10
1.2-9	10	1.2-59	10
1.2-10	10	1.2-60	10
1.2-11	10	1.2-61	10
1.2-12	10	1.2-62	10
1.2-13	10	1.2-63	10
1.2-14	10	1.2-64	10
1.2-15	10	1.2-65	10
1.2-16	10	1.2-66	10
1.2-17	10	1.2-67	10
1.2-18	10	1.2-68	10
1.2-19	10	1.2-69	10
1.2-20	10	1.2-70	10
1.2-21	10	1.2-71	10
1.2-22	10	1.2-72	10
1.2-23	10	1.2-73	10
1.2-24	10	Fig. 1.2.1	7
1.2-25	10	Fig. 1.2.2	7
1.2-26	10	Fig. 1.2.3	
1.2-27	10	Fig. 1.2.4	10
1.2-28	10	Fig. 1.2.5	10
1.2-29	10	Fig. 1.2.6	7
1.2-30	10	Fig. 1.2.7	7
1.2-31	10	Fig. 1.2.8	10
1.2-32	10	Fig. 1.2.9	10
1.2-33	10	Fig. 1.2.10	10
1.2-34	10	Fig. 1.2.10A	10
1.2-35	10	Fig. 1.2.10B	10
1.2-36	10	Fig. 1.2.10C	10

LIST OF EFFECTIVE PAGES FOR PROPOSED REVISION 10

Page	Revision	1	Page	Revision
Fig. 1 2 10D	10		Fig 1 A 5	
Fig. 1.2.105	9		1 B_1	10
Fig. 1.2.11	9		1 B-2	10
Fig. 1.2.17	Deleted		1 B-3	10
Fig. 1.2.12	Deleted		1 B-4 through 1 B-20	Deleted
Fig. 1.2.13	10	 	1 C-1	Deleted
Fig. 1.2.10A			1 C-2	7
Fig. 1.2.14	10	<u> </u>	1 C-3	7
Fig. 1.2.16	10	┨─────	1.0-5	7
Fig. 1.2.10	10	<u> </u>	1.0-4	7
1 2 1	10		1.0-5	7
13.2	10		1.0-0	7
1.3-2	10			7
134	10		App. 1 D (Total of 82 Barros)	P
125	10	 	App. 1.D (Total of 65 Fages)	O
13.6	10			
127	10		·	·····
1.0-1	10	<u>├</u>		
1.3-0	10			
1.3-9	10			
1.3-10	10			
1.3-11	10	L		
1.3-12	10			
1.3-13	10	<u> </u>		
	10			
1.3-15	10			
1.3-10	10		· · · · · · · · · · · · · · · · · · ·	
1.3-17	10			
1 2 10	10			
1.0-10	10			
1.4-1	10			
Drawings	See Section 1.4			
3 Bills of Material w/ 6 shoets	Deleted			
1 5_1				
1.5-1	10			
16-1	10			
16-2	10			
16-3	10			
1 A-1	10			
1A-2	10		······	
1 A-3	10		··· ··· ··· ··· ······	
1 A-4	10		· · · · · · · · · · · · · · · · · · ·	
1 A-5	10			
1.A-6				
1.A-7	10			
Fig. 1.A.1	A			······
Fig. 1.A.2	4			
Fig. 1 A 3	4			
Fig. 1.A.4				<u> </u>

LIST OF EFFECTIVE PAGES FOR PROPOSED REVISION 10
Page	Revision	Page	Revision
2.0-1	10	Fig. 2.1.1	7
2.0-2	10	Fig. 2.1.2	6
2.0-3	10	Fig. 2.1.3	10
2.0-4	10	Fig. 2.1.4	10
2.0-5	10	Fig. 2.1.5	4
2.0-6	10	Fig. 2.1.6	4
2.1-1	10	Fig. 2.1.7	4
2.1-2	10	Fig. 2.1.8	4
2.1-3	10	Fig. 2.1.9	4
2.1-4	10	Fig. 2.1.10	6
2.1-5	10	Fig. 2.1.11	4
2.1-6	10	Fig. 2.1.12	4
2.1-7	10	Fig. 2.1.13	4
2.1-8	10	Fig. 2.1.14	6
2.1-9	10	2.2-1	10
2.1-10	10	2.2-2	10
2.1-11	10	2.2-3	10
2.1-12	10	2.2-4	10
2.1-13	10	2.2-5	10
2.1-14	10	Fig. 2.2.1	4
2.1-15	10	2.3-1	10
2.1-16	10	2.3-2	10
2.1-17	10	2.3-3	10
2.1-18	10	2.3-4	10
2.1-19	10	2.3-5	10
2.1-20	10	2.3-6	10
2.1-21	10	2.3-7	10
2.1-22	10	2.3-8	10
2.1-23	10	2.3-9	10
2.1-24	10	2.3-10	10
2.1-25	10	2.3-11	10
2.1-26	10	Fig. 2.3.1	4
2.1-27	10	Fig. 2.3.2	7
2.1-28	10	2.4-1	10
2.1-29	10	2.4-2	10
2.1-30	10	2.4-3	10
2.1-31	10	2.4-4	10
2.1-32	10	2.4-5	10
2.1-33	10	2.4-6	10
2.1-34	10	2.4-7	10
2.1-35	10	2.5-1	10
2.1-36	10	2.5-2	10
2.1-37	10	2.5-3	10
2.1-38	10	2.5-4	10
2.1-39	10	2.5-5	10
2.1-40	10	2.5-6	10
2.1-41	10	2.5-7	10

2.1-42

2.1-43

2.5-8

2.5-9

10

10

10

10

Page	Revision	Page	Revision
2.5-10	10	2.6-36	10
2.5-11	10	2.6-37	10
2.5-12	10	2.6-38	10
2.5-13	10	2.6-39	10
2.5-14	10	2.6-40	10
2.5-15	10	2.6-41	10
2.5-16	10	2.6-42	10
2.5-17	10	2.6-43	10
2.5-18	10	2.6-44	10
2.5-19 through 2.5-29	Deleted	2.6-45	10
Fig. 2.5.1	10	2.6-46	10
Fig. 2.5.2	10	2.6-47	10
Fig. 2.5.3 through 2.5.11	Deleted	2.6-48	10
Fig. 2.5.12	10	2.6-49	10
Fig. 2.5.13	10	2.6-50	10
2.6-1	10	2.6-51	10
2.6-2	10	2.6-52	10
2.6-3	10	2.6-53	10
2.6-4	10	2.6-54	10
2.6-5	10	2.6-55	10
2.6-6	10	2.6-56	10
2.6-7	10	2.6-57	10
2.6-8	10	2.6-58	10
2.6-9	10	2.6-59	10
2.6-10	10	2.6-60	10
2.6-11	10	2.6-61	10
2.6-12	10	2.6-62	10
2.6-13	10	2.6-63	10
2.6-14	10	2.6-64	10
2.6-15	10	2.6-65	10
2.6-16	10	2.6-66	10
2.6-17	10	2.6-67	10
2.6-18	10	2.6-68	10
2.6-19	10	2.6-69	10
2.6-20		2.6-70	10
2.0-21		Fig. 2.6.1	4
2.0-22	10	Fig. 2.0.2	
2.0-23	10	Fig. 2.0.3	10
2.0-24	10	Fig. 2.0.4	
2.0-20	10	Fig. 2.0.0	
2.0-20	10	Fig. 2.0.0	10
2.0-21	10	Fig. 2.0.1	10
2.0-20	10	Fig. 2.0.0	D
26.30		Fig. 2.0.9	10
26.31	10	Fig. 2.0.10	10
26.32	10	Fig. 2.0.11	
26.33	10	Fig 26 12	
2 6-34	10	Fig. 2.0.13	
26-35	10	Fig 26 15	
		I.g. 2.0.10	4

Page	Revision	T	Page	Revision
Fig. 2.6.16	8	1	2.7-38	10
Fig. 2.6.17	8		2.7-39	10
Fig. 2.6.18	8		2.7-40	10
Fig. 2.6.19	8		2.7-41	10
Fig. 2.6.19A	8		2.7-42	10
Fig. 2.6.19B	8		2.7-43	10
Fig. 2.6.19C	8	<u> </u>	2.7-44	10
Fig. 2.6.20	10		2.7-45	10
Fig. 2.6.21	7		2.7-46	10
Fig. 2.6.22	6		2.7-47	10
Fig. 2.6.23	8		2.7-48	10
Fig. 2.6.24	10		2.7-49	Deleted
Fig. 2.6.25	10		2.7-50	Deleted
2.7-1	10	<u> </u>	Fig. 2.7.1	10
2.7-2	10		Fig. 2.7.2	10
2.7-3	10		Fig. 2.7.3	10
2.7-4	10		Fig. 2.7.4	10
27-5	10		Fig. 2.7.5	6
27-6	10	<u> </u>	Fig 2.7.6	6
2.7-7	10	<u> </u>	Fig. 2.7.7	8
2.7-8	10		Fig. 2.7.8	
2.7-9	10		Fig. 2.7.9	8
2.7-10	10		Fig. 2.7.10	
2.7-11	10		Fig. 2.7.11	8
2.7-12	10		Fig. 2.7.12	8
2.7-13	10		Fig. 2.7.13	8
2.7-14	10		Fig. 2.7.14	8
2.7-15	10		Fig. 2.7.15	8
2.7-16	10		Fig. 2.7.16	8
2.7-17	10		Fig. 2.7.17	8
2.7-18	10		Fig. 2.7.18	8
2.7-19	10		Fig. 2.7.19 through -22	Deleted
2.7-20	10		2.8-1	7
2.7-21	10		2.9-1	10
2.7-22	10		2.9-2	10
2.7-23	10		2.9-3	10
2.7-24	10		2.9-4	10
2.7-25	10		2.9-5	10
2.7-26	10		2.9-6	10
2.7-27	10		2.9-7	10
2.7-28	10		2.9-8	10
2.7-29	10		2.9-9	10
2.7-30	10		2.9-10	10
2.7-31	10		2.9-11	10
2.7-32	10		2.9-12	10
2.7-33	10		2.9-13	10
2.7-34	10		2.9-14	10
2.7-35	10		2.9-15	10
2.7-36	10		2.9-16	10
2.7-37	10		2.9-17	10

Page	Revision	Page	Revision
2.9-18	10	Fig. 2.A.2.1	10
2.9-19	10	Fig. 2.A.3.1	10
Fig. 2.9.1	8	Fig. 2.A.4.1	10
Fig. 2.9.2	8	Fig. 2.A.4.2	10
Fig. 2.9.3	8	Fig. 2.A.4.3	10
Fig. 2.9.4	8	Fig. 2.A.4.4	10
Fig. 2.9.5	8	Fig. 2.A.5.1	10
Fig. 2.9.6	8	Fig. 2.A.5.2	10
Fig. 2.9.7	8	Fig. 2.A.5.3	10
Fig. 2.9.8	8	Fig. 2.A.5.4	10
Fig. 2.9.9	8	Fig. 2.A.5.5	10
2.10-1	10	Fig. 2.A.5.6	10
2.10-2	10	Fig. 2.A.5.7	10
2.10-3	Deleted	Fig. 2.A.5.8	10
2.10-4	Deleted	Fig. 2.A.5.9	10
2.10-5	Deleted	Fig. 2.A.5.10	10
2.11-1	10	Fig. 2.A.5.11	10
2.11-2	10	Fig. 2.A.5.12	10
2.11-3	10	Fig. 2.A.5.13	10
2.A-1	10	Fig. 2.A.5.14	10
2.A-2	10	Fig. 2.A.5.15	10
2.A-3	10	Fig. 2.A.5.15A	10
2.A-4	10	Fig. 2.A.5.15B	10
2.A-5	10	Fig. 2.A.5.15C	10
2.A-6	10	Fig. 2.A.5.6	
2.A-7	10	Fig. 2.A.5.17	
2.A-8	10	Fig. 2.A.5.17A	
2.A-9	10	Fig. 2.A.5.18	
2.A-10	10	Fig. 2.A.5.19	
2.A-11	10	Fig. 2.a.5.19A	
2.A-12	10	2.A.5.20	
2.A-13	10	Fig. 2.A.5.21	
2.A-14	10	Fig. 2.A.5.21A	
2.A-15	10	Fig. 2.A.6.1	
2.A-16	10	Fig. 2.A.6.2	
2.A-17	10	Fig. 2.A.6.3	
2.A-18	10	Fig. 2.A.6.4	
2.A-19	10	Fig. 2.A.6.5	
2.A-20	10	Fig. 2.A.6.6]
2.A-21	10	Fig. 2.A.7.1	
2.A-22	10	Fig. 2.A.7.2	
2.A-23	10	Fig. 2.A.7.3]
2.A-24	10	Fig. 2.A.10.1	
2.A-25	10	Fig. 2.A.10.2	
2.A-26	10	Fig. 2.A.10.3	
2.A-27	10	2.B-1	10
Z.A-28	10	2.8-2	10
Hig. 2.A.1.1	10	2.8-3	10
Hig. 2.A.1.2	10	Appendices 2.C through 2.AO	Deleted
Fig. 2.A.1.3	10		

Page	Revision		Page	Revision
3.0-1	10		3.4-20	10
3.0-2	10		3.4-21	10
3.0-3	10		3.4-22	10
3.1-1	10		3.4-23	10
3.1-2	10		3.4-24	10
3.1-3	10	<u> </u>	3.4-25	10
3.2-1	10		3.4-26	10
32-2	10		3.4-27	10
32-3	10		3.4-28	10
3 2-4	10		3.4-29	10
3 2-5	10		3.4-30	10
3.2-6	10		3.4-31	10
3 2-7	10		3.4-32	10
3.2_8	10		3 4-33	10
3.2-0	10		3 4-34	10
3 2-10	10		3 4-35	10
3 2 1	10		3 4-36	10
3.3.7	10		3 4-37	10
2 2 2	10		3 4-38	10
334	Deleted		3 4-39	10
225	Deleted	}	3 4-40	10
336	Deleted		3 4-41	10
3 2.7	Deleted		3 4-42	10
338	Deleted		3 4-43	10
3 3 0	Deleted		3 4-44	10
3 2.10	Deleted		3 4-45	10
3.3-10	Deleted		3 4-46	10
3 3 12	Deleted		3 4_47	10
3 2.13	Deleted		3 4-48	10
3 2.14	Deleted		3 4-49	10
3 2.15	Deleted		3 4-50	10
3.4-1	10		3 4-51	10
3.4-2	10		3 4-52	10
3 4 3	10		3 4-53	10
344	10		3.4-54	10
34-5	10		3.4-55	10
3.4-6	10		3.4-56	10
3 4-7	10		3 4-57	10
3 4.8	10		3 4-58	10
3 4 9	10		3 4-59	10
3 4-10	10		3 4-60	10
3 4-11	10		3.4-61	10
3 4-12	10		3.4-62	10
3 4-13	10		3.4-63	10
3 4-14	10		3.4-64	10
3.4-15	10	}	3.4-65	10
3 4-16	10	<u>├</u> };	3 4-66	10
3 4-17	10		3 4-67	10
3 4-18	10	<u>├</u> ;	3 4-68	10
3 /_10	10	;	3 4-70	
0.7-13		<u>├</u> `	<u> </u>	

Page	Revision	Page	Revision
3.4-71	10	3.6-1	10
3.4-72	10	3.6-2	10
Fig. 3.4.1	6	3.6-3	10
Fig. 3.4.2	7	3.6-4	10
Fig. 3.4.3	7	3.7-1	10
Fig. 344	4	37-2	10
Fig. 345	7	37-3	10
Fig. 34.6	10	3 A-1	10
Fig. 3.4.7		3 4-2	10
Fig. 3.4.8		3 A-3	10
Fig. 3.4.9	Deleted	3 4-4	10
Fig. 3.4.10	10	3 4-5	10
Fig. 3.4.11	<u>0</u>	3 4-6	
Fig. 3.4.12	7	3 R-1	10
Fig. 3 A 13		3 B-2	10
Fig. 3.4.16	6	3 B-3	10
Fig 34 15	0 Deleted	3 B_4	10
Fig. 3.4.16	Deleted	5.0-4 Fig 3 P.1	10
Fig. 3 / 17			
Fig. 3.4.17	Deleted		
Fig. 3.4.10	7		
Fig. 3.4.19	7		
Fig. 3.4.20	Deleted		
Fig. 3.4.22	7		
Fig. 3.4.22	7		
Fig. 3.4.25			
Fig. 3.4.24	0		
Fig. 2.4.25	0		
Fig. 3.4.20	0		
Fig. 3.4.27	0		
2 5 4	0		
252	10		
2.5.2	10		
2.5-3 2.5.4	10		
255	10		·
3.5-5	10		
35.7	10		
35.8	10		
35-9	10		
3.5-10	10		
3.5-11	Deleted		
Fig. 3.5.1	7		
Fig 352			
Fig. 3.5.3	2		
Fig. 3.5.4	2		
Fig. 3.5.5	6	·····	
Fig. 3.5.6	7		
Fig. 3.5.7	7		
Fig. 3.5.8	7		
Fig. 3.5.0			
		· · · · · · · · · · · · · · · · · · ·	
	<u> </u>		

Page	Revision	Page	Revision
4.0-1	10	4.A-1	8
4.0-2	10	4.A-2	
4.1-1	10	4.A-3	8
4.1-2	10	4.B-1	10
4.1-3	10	4.B-2	10
4.1-4	10	4.B-3	10
4.1-5	10	4.B-4	
4.1-6	10	4.B-5	
4.1-7	10		
Fig. 4.1.1	8		
Fig. 4.1.2	8		
Fig. 4.1.3	8		
Fig 414	10		
4 2-1	10		
4 2-2	10		
4 2-3	10		
4 2-4	10		
4 2-5	10		
4 2-6	10		
4.2-7	10	· · · · · · · · · · · · · · · · · · ·	
4 2-8	10		
4 2-9	10		
4 2-10	10		
4 2-11	10		
4 2-12	10		
4 2-13	10		
4 2-14	10		
4 2-15	10	_·	
4 2-16	10		
4 2-17	10		
4 2-18	10		
4.2-19	10		
4.2-20	10		
4.2-21	10		
4.2-22	10		
4.2-23	10		
4.2-24	10		
4.2-25	10		
4.2-26	10		
4.2-27	10		
4.2-28	10		
4 2-29	10		
4,2-30	10		
4.2-31	10		
4.2-32	10		
4.3-1	10		
4.3-2	10		
4.4-1	10		
<u></u>			
	<u>├</u> ──────────		
	<u> </u>		
	1		1 I

LIST OF EFFECTIVE PAGES FOR PROPOSED REVISION IU	LIST	OF	EFFE	CTIVE	PAGE	S FOR	PROPOSEI	D REVISION 10
--	------	----	------	-------	------	-------	----------	---------------

6.6-1 10 $5.2.27$ 10 $5.0-2$ 10 $5.2.28$ 10 $5.1-1$ 10 $5.2.30$ 10 $5.1-3$ 10 $5.2.30$ 10 $5.1-3$ 10 $5.2.31$ 10 $5.1-5$ 10 $5.2.33$ 10 $5.1-5$ 10 $5.2.33$ 10 $5.1-5$ 10 $5.2.35$ 10 $5.1-8$ 10 $5.2.35$ 10 $5.1-8$ 10 $5.2.37$ 10 $5.1-8$ 10 $5.2.37$ 10 $5.1-10$ 10 $5.2.33$ 10 $5.1-11$ 10 $5.2.41$ 10 $5.1-12$ 10 $5.2.41$ 10 $5.1-14$ 10 $5.2.44$ 10 $5.1-16$ 10 $5.2.44$ 10 $5.1-10$ 10 $5.2.44$ 10 $5.1-2$ 10 $5.2.45$ 10 <	Page	Revision	Page	Revision
6.0.2 10 $5.2.28$ 10 $5.1.1$ 10 $5.2.30$ 10 $5.1.3$ 10 $5.2.31$ 10 $5.1.3$ 10 $5.2.32$ 10 $5.1.5$ 10 $5.2.33$ 10 $5.1.6$ 10 $5.2.33$ 10 $5.1.7$ 10 $5.2.35$ 10 $5.1.7$ 10 $5.2.35$ 10 $5.1.7$ 10 $5.2.35$ 10 $5.1.9$ 10 $5.2.33$ 10 $5.1.10$ 10 $5.2.38$ 10 $5.1.11$ 10 $5.2.43$ 10 $5.1.12$ 10 $5.2.41$ 10 $5.1.13$ 10 $5.2.41$ 10 $5.1.16$ 10 $5.2.43$ 10 $5.1.16$ 10 $5.2.43$ 10 $5.1.17$ 10 $5.2.46$ 10 $5.1.19$ 10 $5.2.54$ 10 $5.2.51$ 10	5.0-1	10	5.2-27	10
5.1-1 10 $5.2.29$ 10 $5.1-2$ 10 $5.2.30$ 10 $5.1-3$ 10 $5.2.31$ 10 $5.1-4$ 10 $5.2.32$ 10 $5.1-5$ 10 $5.2.33$ 10 $5.1-6$ 10 $5.2.33$ 10 $5.1-7$ 10 $5.2.35$ 10 $5.1-8$ 10 $5.2.33$ 10 $5.1-9$ 10 $5.2.33$ 10 $5.1-10$ 10 $5.2.33$ 10 $5.1-10$ 10 $5.2.40$ 10 $5.1-11$ 10 $5.2.43$ 10 $5.1-12$ 10 $5.2.44$ 10 $5.1-16$ 10 $5.2.44$ 10 $5.1-16$ 10 $5.2.45$ 10 $5.1-16$ 10 $5.2.45$ 10 $5.1-10$ 10 $5.2.44$ 10 $5.1-10$ 10 $5.2.45$ 10 $5.1-20$ 10 $5.2.45$ 10 $5.1-20$ 10 $5.2.45$	5.0-2	10	5.2-28	10
5.1-2 10 5.230 10 $5.1-3$ 10 5.231 10 $5.1-4$ 10 5.232 10 $5.1-5$ 10 5.233 10 $5.1-6$ 10 5.235 10 $5.1-7$ 10 5.235 10 $5.1-9$ 10 5.235 10 $5.1-9$ 10 5.233 10 $5.1-9$ 10 5.233 10 $5.1-10$ 10 5.233 10 $5.1-11$ 10 5.240 10 $5.1-12$ 10 5.244 10 $5.1-12$ 10 5.242 10 $5.1-14$ 10 5.242 10 $5.1-15$ 10 5.243 10 $5.1-17$ 10 5.244 10 $5.1-16$ 10 5.244 10 $5.1-17$ 10 5.246 10 $5.1-19$ 10 5.246 10 $5.1-12$ 10 5.245 10	5.1-1	10	5.2-29	10
5.1-3 10 5.2-31 10 5.1-4 10 5.2-32 10 5.1-5 10 5.2-33 10 5.1-6 10 5.2-33 10 5.1-7 10 5.2-35 10 5.1-8 10 5.2-35 10 5.1-9 10 5.2-38 10 5.1-10 10 5.2-38 10 5.1-11 10 5.2-38 10 5.1-12 10 5.2-33 10 5.1-13 10 5.2-33 10 5.1-14 10 5.2-33 10 5.1-15 10 5.2-42 10 5.1-16 10 5.2-43 10 5.1-16 10 5.2-43 10 5.1-16 10 5.2-43 10 5.1-18 10 5.2-44 10 5.1-19 10 5.2-45 10 5.1-20 10 5.2-45 10 5.121 10 5.2-55 10 5.2-1 10 5.2	5.1-2	10	5,2-30	10
5.1-4 10 $5.2-32$ 10 $5.1-6$ 10 $5.2-33$ 10 $5.1-6$ 10 $5.2-33$ 10 $5.1-7$ 10 $5.2-35$ 10 $5.1-8$ 10 $5.2-35$ 10 $5.1-9$ 10 $5.2-33$ 10 $5.1-10$ 10 $5.2-33$ 10 $5.1-11$ 10 $5.2-33$ 10 $5.1-11$ 10 $5.2-33$ 10 $5.1-11$ 10 $5.2-33$ 10 $5.1-11$ 10 $5.2-41$ 10 $5.1-13$ 10 $5.2-44$ 10 $5.1-16$ 10 $5.2-44$ 10 $5.1-16$ 10 $5.2-44$ 10 $5.1-16$ 10 $5.2-45$ 10 $5.1-18$ 10 $5.2-46$ 10 $5.1-20$ 10 $5.2-43$ 10 $5.1-21$ 10 $5.2-50$ 10 $5.2-3$ 10 $5.2-52$ 10 $5.2-4$ 10 $5.2-55$ <td< td=""><td>5.1-3</td><td>10</td><td>5.2-31</td><td>10</td></td<>	5.1-3	10	5.2-31	10
5.1-5 10 $5.2.33$ 10 5.1-6 10 $5.2.34$ 10 5.1-7 10 $5.2.35$ 10 5.1-8 10 $5.2.35$ 10 5.1-9 10 $5.2.37$ 10 5.1-10 10 $5.2.33$ 10 5.1-12 10 $5.2.33$ 10 5.1-12 10 $5.2.40$ 10 5.1-12 10 $5.2.41$ 10 5.1-13 10 $5.2.41$ 10 5.1-14 10 $5.2.42$ 10 5.1-15 10 $5.2.43$ 10 5.1-16 10 $5.2.45$ 10 5.1-16 10 $5.2.45$ 10 5.1-16 10 $5.2.45$ 10 5.1-16 10 $5.2.45$ 10 5.1-19 10 $5.2.45$ 10 5.1-20 10 $5.2.48$ 10 5.12 6 $5.2.51$ 10 5.2-1 10 $5.2.65$ 10 5.2	5.1-4	10	5.2-32	10
10 12.34 10 $5.1-6$ 10 5.235 10 $5.1-7$ 10 5.235 10 $5.1-8$ 10 5.236 10 $5.1-9$ 10 5.237 10 $5.1-10$ 10 5.239 10 $5.1-11$ 10 5.239 10 $5.1-11$ 10 5.241 10 $5.1-13$ 10 5.241 10 $5.1-16$ 10 5.242 10 $5.1-15$ 10 5.242 10 $5.1-16$ 10 5.244 10 $5.1-17$ 10 5.247 10 $5.1-10$ 10 5.247 10 $5.1-20$ 10 5.247 10 $5.1-21$ 6 5.251 10 $5.2-11$ 10 $5.2-52$ 10 $5.2-12$ 10 $5.2-55$ 10 <td< td=""><td>5 1-5</td><td>10</td><td>5 2-33</td><td>10</td></td<>	5 1-5	10	5 2-33	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5 1-6	10	5 2-34	10
1-1 10 52.36 10 $5.1-6$ 10 52.36 10 $5.1-9$ 10 52.37 10 $5.1-10$ 10 52.38 10 $5.1-11$ 10 52.39 10 $5.1-12$ 10 52.40 10 $5.1-13$ 10 52.41 10 $5.1-16$ 10 52.42 10 $5.1-16$ 10 52.44 10 $5.1-16$ 10 52.44 10 $5.1-16$ 10 52.44 10 $5.1-17$ 10 52.44 10 $5.1-20$ 10 52.44 10 $51-21$ 10 52.49 10 $51-20$ 10 52.49 10 $51-21$ 10 52.50 10 52.51 10 52.52 10 52.44 10 52.55 10	5 1-7	10	5 2 35	10
10 $12, 33$ 10 $51-9$ 10 $52, 37$ 10 $51-10$ 10 $52, 33$ 10 $51-11$ 10 $52, 39$ 10 $51-12$ 10 $52, 40$ 10 $51-13$ 10 $52, 40$ 10 $51-16$ 10 $52, 42$ 10 $51-16$ 10 $52, 43$ 10 $51-16$ 10 $52, 44$ 10 $51-16$ 10 $52, 44$ 10 $51-16$ 10 $52, 44$ 10 $51-17$ 10 $52, 44$ 10 $51-12$ 6 $52, 47$ 10 $51-20$ 10 $52, 48$ 10 $51-21$ 6 $52, 51$ 10 $52-2$ 10 $52, 52$ 10 $52-3$ 10 $52, 55$ 10 $52-4$ 10 $52, 55$ 10	15 1-8	10	5.2-36	10
$10^{-10^{-10^{-10^{-10^{-10^{-10^{-10^{-$	5 1-0	10	5.2-30	10
1-10 10 $5.2-30$ 10 $5.1-11$ 10 $5.2-39$ 10 $5.1-12$ 10 $5.2-40$ 10 $5.1-13$ 10 $5.2-41$ 10 $5.1-14$ 10 $5.2-42$ 10 $5.1-16$ 10 $5.2-43$ 10 $5.1-16$ 10 $5.2-44$ 10 $5.1-16$ 10 $5.2-45$ 10 $5.1-16$ 10 $5.2-45$ 10 $5.1-16$ 10 $5.2-45$ 10 $5.1-20$ 10 $5.2-48$ 10 $5.1-20$ 10 $5.2-48$ 10 $5.1-20$ 10 $5.2-50$ 10 $5.2-5$ 10 $5.2-53$ 10 $5.2-5$ 10 $5.2-53$ 10 $5.2-5$ 10 $5.2-56$ 10 $5.2-6$ 10 $5.2-57$ 10 $5.2-6$ 10 $5.2-59$ 10 $5.2-6$ 10	5.1-5	10	5.2-38	10
1.11 10 5.240 10 $5.1-12$ 10 5.240 10 $5.1-13$ 10 5.241 10 $5.1-16$ 10 5.242 10 $5.1-16$ 10 5.243 10 $5.1-16$ 10 5.244 10 $5.1-18$ 10 5.245 10 $5.1-18$ 10 5.245 10 $5.1-20$ 10 5.248 10 $5.1-21$ 10 5.248 10 5.12 6 5.250 10 $5.2-1$ 10 $5.2-50$ 10 $5.2-1$ 10 $5.2-50$ 10 $5.2-1$ 10 $5.2-50$ 10 $5.2-3$ 10 $5.2-53$ 10 $5.2-4$ 10 $5.2-56$ 10 $5.2-5$ 10 $5.2-56$ 10 $5.2-6$ 10 $5.3-7$ 10 <td>51-10</td> <td>10</td> <td>5.2-30</td> <td></td>	51-10	10	5.2-30	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.1-13	10	5.2-39	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		10	<u> </u>	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		10	5.2-41	
5.1-16 10 5.2-43 10 5.1-16 10 5.2-44 10 5.1-18 10 5.2-45 10 5.1-19 10 5.2-46 10 5.1-20 10 5.2-48 10 5.1-20 10 5.2-49 10 5.1-21 10 5.2-49 10 Fig. 5.1.1 10 5.2-50 10 Fig. 5.1.2 6 5.2-51 10 5.2-4 10 5.2-52 10 5.2-4 10 5.2-52 10 5.2-3 10 5.2-52 10 5.2-4 10 5.2-55 10 5.2-5 10 5.2-56 10 5.2-6 10 5.2-57 10 5.2-7 10 5.2-58 10 5.2-8 10 5.2-59 10 5.2-9 10 5.3-1 10 5.2-10 10 5.3-1 10 5.2-11 10 5.3-5 10 5.2-12 10 <td< td=""><td></td><td>10</td><td>5.2.42</td><td>10</td></td<>		10	5.2.42	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		10	5.2-43	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		10	5.2.44	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		10	5.2-45	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.1-16	10	5.2.40	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.1-19	10	5.2-47	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.1-20	10	5.2-48	10
Fig. 5.1.110 $5.2-50$ 10Fig. 5.1.26 $5.2-51$ 105.2-110 $5.2-52$ 105.2-210 $5.2-52$ 105.2-310 $5.2-55$ 105.2-410 $5.2-55$ 105.2-510 $5.2-56$ 105.2-610 $5.2-57$ 105.2-710 $5.2-58$ 105.2-810 $5.2-59$ 105.2-910 $5.2-59$ 105.2-1010 $5.3-2$ 105.2-1110 $5.3-3$ 105.2-1210 $5.3-4$ 105.2-1310 $5.3-7$ 105.2-1410 $5.3-7$ 105.2-1510 $5.3-7$ 105.2-1610 $5.3-9$ 105.2-1710 $5.3-9$ 105.2-1810 $5.3-7$ 105.2-1910 $5.3-11$ 105.2-2010Fig. 5.3.1105.2-2110Fig. 5.3.2105.2-2210Fig. 5.3.345.2-2310Fig. 5.3.5105.2-2610Fig. 5.3.76	5.1-21	10	5.2-49	10
$Fig. 5.1.2$ 6 $5.2 \cdot 51$ 10 $5.2 \cdot 1$ 10 $5.2 \cdot 52$ 10 $5.2 \cdot 2$ 10 $5.2 \cdot 53$ 10 $5.2 \cdot 3$ 10 $5.2 \cdot 53$ 10 $5.2 \cdot 4$ 10 $5.2 \cdot 55$ 10 $5.2 \cdot 5$ 10 $5.2 \cdot 55$ 10 $5.2 \cdot 6$ 10 $5.2 \cdot 58$ 10 $5.2 \cdot 6$ 10 $5.2 \cdot 59$ 10 $5.2 \cdot 8$ 10 $5.2 \cdot 59$ 10 $5.2 \cdot 9$ 10 $5.3 \cdot 1$ 10 $5.2 \cdot 10$ 10 $5.3 \cdot 2$ 10 $5.2 \cdot 11$ 10 $5.3 \cdot 3$ 10 $5.2 \cdot 11$ 10 $5.3 \cdot 5$ 10 $5.2 \cdot 12$ 10 $5.3 \cdot 5$ 10 $5.2 \cdot 13$ 10 $5.3 \cdot 5$ 10 $5.2 \cdot 14$ 10 $5.3 \cdot 6$ 10 $5.2 \cdot 16$ 10 $5.3 \cdot 1$ 10 $5.2 \cdot 17$ 10 $5.3 \cdot 1$ 10 $5.2 \cdot 19$ 10 $5.3 \cdot 11$ 10 $5.2 \cdot 20$ 10Fig. $5.3.1$ 10 $5.2 \cdot 21$ 10Fig. $5.3.3$ 4 $5.2 \cdot 22$ 10Fig. $5.3.5$ 10 $5.2 \cdot 25$ 10Fig. $5.3.7$ 10 $5.2 \cdot 26$ 10Fig. $5.3.7$ 6	Fig. 5.1.1	10	5.2-50	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Fig. 5.1.2	6	5.2-51	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-1	10	5.2-52	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-2	10	5.2-53	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-3	10	5.2-54	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-4	10	5.2-55	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-5	10	5.2-56	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.2-6	10	5.2-57	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.2-7	10	5.2-58	10
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-8	10	5.2-59	10
5.2-1010 $5.3-2$ 10 $5.2-11$ 10 $5.3-3$ 10 $5.2-12$ 10 $5.3-3$ 10 $5.2-12$ 10 $5.3-4$ 10 $5.2-13$ 10 $5.3-5$ 10 $5.2-14$ 10 $5.3-6$ 10 $5.2-15$ 10 $5.3-6$ 10 $5.2-16$ 10 $5.3-8$ 10 $5.2-16$ 10 $5.3-9$ 10 $5.2-17$ 10 $5.3-9$ 10 $5.2-18$ 10 $5.3-10$ 10 $5.2-19$ 10 $5.3-11$ 10 $5.2-20$ 10Fig. $5.3.1$ 10 $5.2-22$ 10Fig. $5.3.2$ 10 $5.2-22$ 10Fig. $5.3.3$ 4 $5.2-23$ 10Fig. $5.3.5$ 10 $5.2-24$ 10Fig. $5.3.5$ 10 $5.2-26$ 10Fig. $5.3.7$ 6	5.2-9	10	5.3-1	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-10	10	5.3-2	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-11	10	5.3-3	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.2-12	10	5.3-4	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5.2-13	10	5.3-5	10
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5.2-14	10	5.3-6	10
	5.2-15	10	5.3-7	10
	5.2-16	10	5.3-8	10
5.2-18 10 5.3-10 10 5.2-19 10 5.3-11 10 5.2-20 10 Fig. 5.3.1 10 5.2-21 10 Fig. 5.3.2 10 5.2-22 10 Fig. 5.3.3 4 5.2-23 10 Fig. 5.3.4 10 5.2-24 10 Fig. 5.3.5 10 5.2-25 10 Fig. 5.3.6 7 5.2-26 10 Fig. 5.3.7 6	5.2-17	10	5.3-9	10
5.2-19105.3-11105.2-2010Fig. 5.3.1105.2-2110Fig. 5.3.2105.2-2210Fig. 5.3.345.2-2310Fig. 5.3.4105.2-2410Fig. 5.3.5105.2-2510Fig. 5.3.675.2-2610Fig. 5.3.76	5.2-18	10	5.3-10	10
5.2-20 10 Fig. 5.3.1 10 5.2-21 10 Fig. 5.3.2 10 5.2-22 10 Fig. 5.3.3 4 5.2-23 10 Fig. 5.3.4 10 5.2-24 10 Fig. 5.3.5 10 5.2-25 10 Fig. 5.3.6 7 5.2-26 10 Fig. 5.3.7 6	5.2-19	10	5.3-11	10
5.2-21 10 Fig. 5.3.2 10 5.2-22 10 Fig. 5.3.3 4 5.2-23 10 Fig. 5.3.4 10 5.2-24 10 Fig. 5.3.5 10 5.2-25 10 Fig. 5.3.6 7 5.2-26 10 Fig. 5.3.7 6	5.2-20	10	Fig. 5.3.1	10
5.2-22 10 Fig. 5.3.3 4 5.2-23 10 Fig. 5.3.4 10 5.2-24 10 Fig. 5.3.5 10 5.2-25 10 Fig. 5.3.6 7 5.2-26 10 Fig. 5.3.7 6	5.2-21	10	Fig. 5.3.2	10
5.2-23 10 Fig. 5.3.4 10 5.2-24 10 Fig. 5.3.5 10 5.2-25 10 Fig. 5.3.6 7 5.2-26 10 Fig. 5.3.7 6	5.2-22	10	Fig. 5.3.3	4
5.2-24 10 Fig. 5.3.5 10 5.2-25 10 Fig. 5.3.6 7 5.2-26 10 Fig. 5.3.7 6	5.2-23	10	Fig. 5.3.4	10
5.2-25 10 Fig. 5.3.6 7 5.2-26 10 Fig. 5.3.7 6	5.2-24	10	Fig. 5.3.5	10
5.2-26 10 Fig. 5.3.7 6	5.2-25	10	Fig. 5.3.6	7
	5.2-26	10	Fig. 5.3.7	6

Page	Revision	Page	Revision
Fig. 5.3.8	6	Fig. 5.4.2	10
Fig. 5.3.9	10	5.5-1	10
Fig. 5.3.10	10	5.5-2	10
Fig. 5.3.11	8	5.5-3	10
Fig. 5.3.12	8	5.5-4	10
5.4-1	10	5.6-1	10
5.4-2	10	5.6-2	10
5.4-3	10	5.6-3	10
5.4-4	10	5.A-1	7
5.4-5	10	5.A-2	7
5.4-6	10	5.A-3	7
5.4-7	10	5.B-1	7
5.4-8	10	5.B-2	7
5.4-9	10	5.B-3	7
5.4-10	10	5.B-4	7
5.4-11	10	5.B-5	7
5.4-12	10	5.B-6	7
5.4-13	10	5.C-1	
5.4-14	10	5.C-2	10
5.4-15	10	5.C-3	10
5.4-16	10	5.C-4	10
5.4-17	10	5.C-5	10
5.4-18	10	5.C-6	10
5.4-19	10	5.C-7	10
5.4-20	10	5.C-8	10
5.4-21	10	5.C-9	10
5.4-22	10	5.C-10	10
5.4-23	10	5.C-11	10
5.4-24	10	5.C-12	10
5.4-25	10	5.C-13	10
5.4-26	10	5.C-14	10
5.4-27	10	5.C-15	10
5.4-28	10	5.C-16	10
5.4-29	10	5.C-17	10
5.4-30	10	5.C-18	10
5.4-31	10	5.C-19	10
5.4-32	10	5.C-20	10
5.4-33	10	5.C-21	10
5.4-34	10	5.C-22	10
5.4-35	10	5.C-23	10
5.4-36	10	5.C-24	10
5.4-37	10	5.C-25	10
5.4-38	10	5.C-26	10
5.4-39	10	5.C-27	10
5.4-40	10	5.C-28	10
5.4-41	10	5.C-29	10
5.4-42	10	5.C-30	10
5.4-43	10	5.C-31	10
5.4-44	10	5.C-32	10
5.4-45	10	5.C-33	10
Fig. 5.4.1		····	

Page	Revision	Page	Revision
6.1-1	10	6.2-35	10
6.1-2	10	6.2-36	10
6.1-3	10	6.2-37	10
6.1-4	10	6.2-38	10
6.1-5	10	6.2-39	10
6.1-6	10	6.2-40	10
6.1-7	10	6.2-41	10
6.1-8	10	6.2-42	10
6.1-9	10	6.2-43	10
6.1-10	10	6.2-44	10
6.1-11	10	6.2-45	10
6 1-12	10	6 2-46	10
6 1-13	10	6 2-47	10
6 1-14	10	6 2-48	10
6 1-15	10	6 2-49	10
6 1-16	10	6 2-50	
6 2-1	10	6 2-51	
622	10	6 2 52	10
623	10	6.2-53	10
6.2-3	10	6.2-53	10
0.2-4	10	6.2.54	10
0.2-0	10	0.2-00	10
0.2-0	10	0.2-00	10
0.2-7	10	6.2-57	10
6.2-8	10	0.2-36	10
0.2-9	10	0.3-1	10
6.2-10	10	0.3-2	10
6.2-11	10	6.3-3	10
6.2-12	10	6.3-4	10
6.2-13	10	6.3-5	10
6.2-14	10	6.3-6	10
6.2-15	10	6.3-7	10
6.2-16	10	6.3-8	10
6.2-17	10	6.3-9	10
6.2-18	10	6.3-10	10
6.2-19	10	6.3-11	10
6.2-20	10	6.3-12	10
6.2-21	10	6.3-13	10
6.2-22	10	6.3-14	10
6.2-23	10	6.3-15	10
6.2-24	10	6.3-16	10
6.2-25	10	6.3-17	10
6.2-26	10	6.3-18	10
6.2-27	10	Fig. 6.3.1	10
6.2-28	10	Fig. 6.3.1A	10
6.2-29	10	Fig. 6.3.2	10
6.2-30	10	Fig. 6.3.3	10
6.2-31	10	Fig. 6.3.4	10
6.2-32	10	Fig. 6.3.5	10
6.2-33	10	Fig. 6.3.6	4
6.2-34	10	Fig. 6.3.7	10

Page	Revision	Page	Revision
6.4-1	10	6.7-1	10
6.4-2	10	6.7-2	10
6.4-3	10	6.A-1	10
6.4-4	10	6.A-2	10
6.4-5	10	6.A-3	10
6.4-6	10	6.A-4	10
6.4-7	10	6.A-5	10
6.4-8	10	6.A-6	10
6.4-9	10	6.A-7	10
6.4-10	10	6.A-8	10
6.4-11	10	6.A-9	10
6.4-12	10	6.A-10	10
6.4-13	10	6.A-11	10
6.4-14	10	6.A-12	10
6.4-15	10	6.A-13	10
6.4-16	10	6.A-14	10
6.4-17	10	6.A-15	10
6.4-18	10	6.A-16	10
6.4-19	10	6_A-17	10
6.4-20	10	6.A-18	10
6.4-21	10	6.A-19	10
6.4-22	10	6.A-20	10
6.4-23	10	6.A-21	10
6.4-24	10	<u>6.A-22</u>	10
6.4-25	10	Fig. 6.A.1	7
6.4-26	10	Fig. 6.A.2	7
6.4-27	10	Fig. 6.A.3	7
6.4-28	10	Fig. 6.A.4	
6.4-29	10	Fig. 6.A.5	
0.4-30	10	Fig. 6.A.6	
0.4-31	10	Fig. 6.A.7	
0.4-32	10	0.8-1	
0.4-33	10		/
0.4-34 © 4 25	10		
6.4.26	10		10
Eig 6 4 1		6 D-3	10
Fig. 6.4.2	 	6 D-4	10
Fig. 6.4.3	G	6.D-5	10
Fig. 6.4.4	Q	6.D-6	10
Fig. 6.4.5	ā	6.D-7	10
Fig. 6.4.6	g	6.D-8	10
Fig. 6.4.7	a	6.D-9	10
Fig. 6.4.8	9	6.D-10	10
Fig. 6.4.9	10	6.D-11	10
Fig. 6.4.10		6.D-12	
Fig. 6.4.11	10	6.D-13	10
Fig. 6.4.12	10	6.D-14	10
6.5-1	10	6.D-15	10
6.6-1	10	6.D-16	10

Page	Revision	Page	Revision
6.D-17	10		
6.D-18	10		····
6.D-19	10		
6.D-20	10		
6.D-21	10		
6.D-22	10		
6 D-23	10		
6 D-24	10		
6.D-25	10		
6.D-26	10		
6.D-27	10		
6.D-28	10		
6.D-29	10		
6.D-30	10		
6.D-31	10		
6.D-32	10		
6.D-33	10		
6.D-34	10		
6.D-35	10		
6.D-36	10		
6.D-37	10		
6.D-38	10		
6.D-39	10		
6.D-40	10		
6.D-41	10		
6.D-42	10		
6.D-43	10		
6.D-44	10		
6.D-45	10		
6.D-46	10		
6.3-1	10		
	· · ·	<u></u>	
		······	
	·		
·			

Page	Revision	Page	Revision
7.0-1	10	Fig. 7.1.6	10
7.0-2	10	Fig. 7.1.7	10
7.0-3	10	Fig. 7.1.8	7
7.1-1	10	Fig. 7.1.9	10
7.1-2	10	Fig. 7.1.10	10
7.1-3	10	Fig. 7.1.11	10
7.1-4	10	Fig. 7.1.12	10
7.1-5	10	Fig. 7.1.13	10
7.1-6	10	Fig. 7.1.14	7
7. <u>1-7</u>	10	Fig. 7.1.15	7
7.1-8	10	Fig. 7.1.16	7
7.1-9	10	Fig. 7.1.17	10
7.1-10	10	Fig. 7.1.18	7
7.1-11	10	Fig. 7.1.19	7
7.1-12	10	Fig. 7.1.20	10
7.1-13	10	Fig. 7.1.21	7
7.1-14	10	Fig. 7.1.22	10
7.1-15	10	Fig. 7.1.23	10
7.1-16	10	Fig. 7.1.24	10
7.1-17	10	Fig. 7.1.25	10
7.1-18	10	Fig. 7.1.26	10
7.1-19	10	Fig. 7.1.27	7
7.1-20	10	Fig. 7.1.28	
7.1-21	10	Fig. 7.1.29	
7.1-22	10	7.2.4	/
7.1-23	10	7.2-1	
7.1-24	10	7.2.2	10
7.1-20	10	724	10
7.1-20	10	7.2-4	10
7.1-27	10	7 2-6	10
7.1-20	10	7 2-7	10
7.1-25	10	7 2-8	10
7 1-31	10	7 2-9	
7 1-32	10	7.2-10	Deleted
7 1-33	10	Fig. 7.2.1	10
7 1-34	10	Fig. 7.2.2a	7
7.1-35	10	Fig. 7.2.2b	
7,1-36	10	Fig. 7.2.2c	7
7.1-37	10	Fig. 7.2.3	10
7.1-38	10	Fig. 7.2.4	10
7.1-39	10	Fig. 7.2.5	10
7.1-40	Deleted	7.3-1	10
7.1-41	Deleted	7.3-2	10
Fig. 7.1.1	10	7.3-3	10
Fig. 7.1.2a	7	7.4-1	10
Fig. 7.1.2b	7	7.4-2	10
Fig. 7.1.2c	10	7.4-3	10
Fig. 7.1.3	7	7.4-4	10
Fig. 7.1.4	10	7.5-1	8
Fig. 7.1.5	10	7.6-1	10

Page	Revision	Page	Revision
8.1-1	10		
8.1-2	10		
8.1-3	10		
8.1-4	10		
8.1-5	10		
8.1-6	10		
8.1-7	10		
8.1-8	10		
8.1-9	10		
8.1-10	10		
8.1-11	10		
8.1-12	10		
8.1-13	10		
8.1-14	10		
8.1-15	10		
8.1-16	10		
8.1-17	10		
8.1-18	10		
8.1-19	10		
8.1-20	10		
8.1-21	10		
8.1-22	10		
8.1-23	10		
8.1-24	10		
8.1-25	10		
8.1-26	10		
8.1-27	10		
8.1-28	10		
8.1-29	10		
Fig. 8.1.1	Deleted		
Fig. 8.1.2	7		
Fig. 8.1.3	7		
0.2-1	10	·····	
	10		
0.2-3		····	
0.2-4	10		
	10		
	10		
0.3-2	10		
0.4-1	10		

CHAPTER 1: GENERAL INFORMATION

1.0 GENERAL INFORMATION

This Safety Analysis Report (SAR) for Holtec International's HI-STAR 100 packaging is a compilation of information and analyses to support a United States Nuclear Regulatory Commission (NRC) licensing review as a spent nuclear fuel transportation package (Docket No. 71-9261) under requirements specified in 10CFR71 [1.0.1] and 49CFR173 [1.0.2]. This SAR supports NRC approval and issuance of Certificate of Compliance No. 9261, issued under the provisions and definitions in 10CFR71, Subpart D, for the design Model: HI-STAR 100 as an acceptable Type B(U)F-85 packaging for transport by exclusive use shipment (10CFR71.47).

The HI-STAR 100 packaging complies with the requirements of 10CFR71 for a Type B(U)F-85 package. The HI-STAR 100 packaging does not have a maximum normal operating pressure (MNOP) greater than 700 kPa (100 lb/in²). The HI-STAR 100 internal design pressure is specified in Table 2.1.1 as 100 psig to calculate bounding stress values. Section 3.4 calculates the MNOP (reported in Table 3.4.15) and demonstrates that the value remains below the design value specified in Table 2.1.1. No pressure relief device is provided on the HI-STAR 100 containment boundary, as discussed in Subsection 1.2.1.8. Therefore, there is no pressure relief device that would allow the release of radioactive material under the tests specified in 10CFR71.73. Analyses that demonstrate that the HI-STAR 100 packaging complies with the requirements of Subparts E and F of 10CFR71 are provided in this SAR. Specific reference to each section of the SAR that is used to specifically address compliance to 10CFR71 is provided in Table 1.0.2. Therefore, the HI-STAR 100 packaging to transport spent nuclear fuel should be designated B(U)F-85.

The HI-STAR 100 transport index for nuclear criticality control is zero, as an unlimited number of packages is subcritical under the procedures specified in 10CFR71.59(a). Section 6.1 provides the determination of the transport index for nuclear criticality control. The transport index based on radiation is in excess of 10 for the HI-STAR 100 Packaging with design basis fuel contents. Therefore, the HI-STAR 100 Packaging must be transported by exclusive use shipment (10CFR71.47).

The HI-STAR 100 packaging design, fabrication, assembly, and testing shall be performed in accordance with Holtec International's quality assurance program. Holtec International's quality assurance program was originally developed to meet NRC requirements delineated in 10CFR50, Appendix B, and was expanded to include provisions of 10CFR71, Subpart H, and 10CFR72, Subpart G, for structures, systems, and components designated as important to safety. NRC approval of Holtec International's quality assurance program is documented by the Quality Assurance Program Approval for Radioactive material Packages (NRC Form 311), Approval Number 0784, Docket No. 71-0784.

This SAR has been prepared in the format and content suggested in NRC Regulatory Guide 7.9 [1.0.3]. The purpose of this chapter is to provide a general description of the design features and transport capabilities of the HI-STAR 100 packaging including its intended use. This chapter provides a summary description of the packaging, operational features, and contents, and provides

reasonable assurance that the package will meet the regulations and operating objectives. Table 1.0.1 contains a listing of the terminology and notation used in preparing this SAR.

This SAR was initially prepared prior to the issuance of the draft version of NUREG-1617 [1.0.5]. To aid NRC review, additional tables and references have been added to facilitate the location of information needed to demonstrate compliance with 10CFR71 as outlined by NUREG-1617. Table 1.0.2 provides a matrix of the 10CFR71 requirements as outlined in NUREG-1617, the format requirements of Regulatory Guide 7.9, and reference to the applicable SAR section(s) that address(es) each topic.

The HI-STAR 100 System is a dual purpose system, certified under 10 CFR 71 and 10 CFR 72. The HI-STAR 100 Final Safety Analysis Report (FSAR) [1.0.6] supports Certificate of Compliance No. 1008 for HI-STAR 100 to store spent nuclear fuel at an Independent Spent Fuel Storage Installation (ISFSI) facility under requirements of 10CFR72, Subpart L [1.0.4] (Docket Number 72-1008).

Within this report, all figures, tables and references cited are identified by the double decimal system m.n.i, where m is the chapter number, n is the section number, and i is the sequential number. Thus, for example, Figure 1.1.1 is the first figure in Section 1.1 of Chapter 1 (which is the next section in this chapter).

Revision of this document to Revision 10 was made on a section level. Therefore, if any change occurs on a page, the entire section was updated to Revision 10. The locations of specific text changes are indicated by revision bars in the margin of the page. Figures are controlled individually at the latest SAR revision level for that particular figure. Sections and figures unchanged in the latest SAR revision indicate the revision level corresponding to the last changes made in the section/figure. Drawings are also controlled individually within the Holtec International drawing control system. The revisions of drawings included in Revision 10 of this SAR are the same as those incorporated by reference into CoC No. 9261, Amendment 2.

Revision 10 of this SAR includes information pertaining to the MPC-32 basket. However, the MPC-32 is not certified for transportation at this time. MPC-32 is under review by the NRC and will be certified in a future CoC amendment.

1.0.1 Engineering Change Orders

The changes authorized by the following Holtec Engineering Change Orders (ECOs) are reflected in Revision 10 of this SAR:

MPC-68/68F: ECOs 1021-1, 3, 4, 7, 8, 13, 14, 16, 18 through 20, 22, 27, 29, 33, 34, 39, 41, 44, 46 through 54, and 56; and 71188-43.

MPC-24/24E/24EF: ECOs 1022-3, 6, 9, 10 through 13, 16, 19, 21 through 24, 26, 28, 35, 37 through 41, 43 through 46, and 48 through 51; and 1135-24 through 26, 28 through 33, 36, 39, and 40.

MPC-32: ECOs 1023-3, 5, 7, 10 through 14, 16, 18 through 21, and 24.

HI-STAR overpack: ECOs 1020-6, 7, 9, 11, 12, 14, 15, 19, 22, 23, 25, 29, 31, 33, 34 through 36, 38, 41 through 43, and 45 through 47; and 71188-31 and 33.

Ancillary Equipment: ECOs 1027-27, 31 and 50.

General FSAR changes: ECOs 5014-36, 49, 54, 57, 62, 67, 68, 71, 72, 77, 80, 82, 86 through 88, 91, and 94.

Table 1.0.1

TERMINOLOGY AND NOTATION

ALARA is an acronym for As Low As Reasonably Achievable.

AL-STARTM is the trademark name of the HI-STAR 100 impact limiter.

Boral is a generic term to denote an aluminum-boron carbide cermet manufactured in accordance with U.S. Patent No. 4027377. The individual material supplier may use another trade name to refer to the same product.

BoralTM means Boral manufactured by AAR Advanced Structures.

BWR is an acronym for boiling water reactor.

C.G. is an acronym for center of gravity.

Commercial Spent Fuel or CSF refers to nuclear fuel used to produce energy in a commercial nuclear power plant.

Containment System Boundary means the enclosure formed by the overpack inner shell welded to a bottom plate and top flange plus the bolted closure plate with dual seals and the vent and drain port plugs with seals. It is also called the primary containment boundary when used with the inner (secondary) containment boundary of the MPC-68F and MPC-24EF.

Containment System means the HI-STAR 100 overpack that forms the containment boundary of the packaging intended to contain the radioactive material during transport.

Cooling Time (or post-irradiation cooling time) for a spent fuel assembly is the time between reactor shutdown and the time the spent fuel assembly is loaded into the MPC.

Damaged Fuel Assembly is a fuel assembly with known or suspected cladding defects, as determined by a review of records, greater than pinhole leaks or hairline cracks, empty fuel rod locations that are not filled with dummy fuel rods, or those that cannot be handled by normal means. Fuel assemblies which cannot be handled by normal means due to fuel cladding damage are considered Fuel Debris.

Damaged Fuel Container (or Canister) means a specially designed enclosure for damaged fuel assemblies or fuel debris which permits gaseous and liquid media to escape while minimizing dispersal of gross solid particulates.

TERMINOLOGY AND NOTATION

Enclosure Vessel (or MPC Enclosure Vessel) means the pressure vessel defined by the cylindrical shell, baseplate, port cover plates, lid, closure ring, and associated welds that provides confinement for the helium gas contained within the MPC. The Enclosure Vessel (EV) and the fuel basket together constitute the multi-purpose canister.

Exclusive use means the sole use by a single consignor of a conveyance for which all initial, intermediate, and final loading and unloading are carried out in accordance with the direction of the consignor or consignee. The consignor and the carrier must ensure that loading or unloading is performed by personnel having radiological training and resources appropriate for safe handling of the consignment. The consignor must issue specific instructions, in writing, for maintenance of exclusive use shipment controls, and include them with the shipping paper information provided to the carrier by the consignor.

FSAR is an acronym for Final Safety Analysis Report (10CFR72).

Fuel Basket means a honeycomb structural weldment with square openings which can accept a fuel assembly of the type for which it is designed.

Fuel Debris is ruptured fuel rods, severed fuel rods, loose fuel pellets, or fuel assemblies with known or suspected defects which cannot be handled by normal means due to fuel cladding damage. Fuel debris also includes certain Trojan plant-specific fuel material contained in Trojan Failed Fuel Cans.

HI-STAR 100 overpack or overpack means the cask that receives and contains the sealed multipurpose canisters containing spent nuclear fuel. It provides the containment boundary for radioactive materials, gamma and neutron shielding, and a set of lifting trunnions for handling. Certain overpack models also include optional pocket trunnions for upending and downending.

HI-STAR 100 System or HI-STAR 100 Packaging consists of the MPC sealed within the HI-STAR 100 overpack with impact limiters installed.

HoltiteTM is the trade name for all present and future neutron shielding materials formulated under Holtec International's R&D program dedicated to developing shielding materials for application in dry storage and transport systems. The Holtite development program is an ongoing experimentation effort to identify neutron shielding materials with enhanced shielding and temperature tolerance characteristics. Holtite-ATM is the first and only shielding material qualified under the Holtite R&D program. As such, the terms Holtite and Holtite-A may be used interchangeably throughout this SAR.

HoltiteTM-A is a trademarked Holtec International neutron shield material.

TERMINOLOGY AND NOTATION

Impact Limiter means a set of fully-enclosed energy absorbers that are attached to the top and bottom of the overpack during transport. The impact limiters are used to absorb kinetic energy resulting from normal and hypothetical accident drop conditions. The HI-STAR impact limiters are called AL-STAR.

Important to Safety (ITS) means a function or condition required to transport spent nuclear fuel safely; to prevent damage to spent nuclear fuel, and to provide reasonable assurance that spent nuclear fuel can be received, handled, packaged, transported, and retrieved without undue risk to the health and safety of the public.

Intact Fuel Assembly is defined as a fuel assembly without known or suspected cladding defects greater than pinhole leaks and hairline cracks, and which can be handled by normal means. Fuel assemblies without fuel rods in fuel rod locations shall not be classified as Intact Fuel Assemblies unless dummy fuel rods are used to displace an amount of water greater than or equal to that displaced by the original fuel rod(s).

Load-and-Go is a term used in this SAR that means the practice of loading spent fuel into the HI-STAR 100 System packaging and placing the packaging into transportation service under 10 CFR 71, without first deploying the system at an Independent Spent Fuel Storage Installation (ISFSI) under 10 CFR 72.

Maximum Normal Operating Pressure (MNOP) means the maximum gauge pressure that would develop in the containment system in a period of 1 year under the heat condition specified in 10CFR71.71(c)(1), in the absence of venting, external cooling by an ancillary system, or operational controls during transport.

Maximum Reactivity means the highest possible k-effective including bias, uncertainties, and calculational statistics evaluated for the worst-case combination of fuel basket manufacturing tolerances.

MGDS is an acronym for Mined Geological Depository System.

MPC Fuel Basket means the honeycombed composite cell structure utilized to maintain subcriticality of the spent nuclear fuel. The number and size of the storage cells depends on the type of spent nuclear fuel to be transported. Each MPC fuel basket has sheathing welded to the storage cell walls for retaining the Boral neutron absorber. Boral is a commercially-available thermal neutron poison material composed of boron carbide and aluminum.

Multi-Purpose Canister (MPC) means the sealed canister consisting of a honeycombed fuel basket for spent nuclear fuel storage, contained in a cylindrical canister shell (the MPC Enclosure Vessel). There are different MPCs with different fuel basket geometries for storing PWR or BWR fuel. All MPCs except the Trojan plant MPCs have identical exterior dimensions. The Trojan plant MPCs have the same outside diameter, but are approximately nine inches shorter than the generic MPC

TERMINOLOGY AND NOTATION

design. MPC is an acronym for multi-purpose canister. The MPCs used as part of the HI-STAR 100 Packaging are identical to the MPCs authorized for use in the HI-STAR 100 Storage (Docket No. 72-1008) and HI-STORM 100 Storage (72-1014) [1.0.7] CoCs to the extent that the particular MPC models are authorized for use under both CoCs.

Neutron Shielding means Holtite, a material used in the overpack to thermalize and capture neutrons emanating from the radioactive spent nuclear fuel.

Neutron Sources means specially designed inserts for fuel assemblies that produce neutrons for startup of the reactor. The specific types of neutron sources authorized for transportation in the HI-STAR 100 System are discussed in Section 1.2.3.

Non-fuel Hardware, or NFH, means Burnable Poison Rod Assemblies (BPRAs), Thimble Plug Devices (TPDs), Control Rod Assemblies (CRAs), Axial Power Shaping Rods (APSRs), Wet Annular Burnable Absorbers (WABAs), Rod Cluster Control Assemblies (RCCAs), Control Element Assemblies (CEAs), water displacement guide tube plugs, orifice rod assemblies, and vibration suppressor inserts. The specific types of NFH authorized for transportation in the HI-STAR 100 System are discussed in Section 1.2 of this SAR.

Packaging means the HI-STAR 100 System consisting of a single HI-STAR 100 overpack, a set of impact limiters, and a multi-purpose canister (MPC). It excludes all lifting devices, rigging, transporters, saddle blocks, welding machines, and auxiliary equipment (such as the drying and helium backfill system) used during fuel loading operations and preparation for off-site transportation.

Package means the HI-STAR 100 System plus the licensed radioactive contents loaded for transport.

Planar-Average Initial Enrichment is the average of the distributed fuel rod initial enrichments within a given axial plane of the assembly lattice.

PWR is an acronym for pressurized water reactor.

Reactivity is used synonymously with effective multiplication factor or k-effective.

SAR is an acronym for Safety Analysis Report (10CFR71).

Secondary Containment Boundary means the Enclosure Vessel of the "F" model MPC. The secondary containment boundary of the "F" model MPC provides the separate inner container for the transport of fuel debris. The "F" model MPC, in conjunction with the overpack containment system boundary, is designed to meet the double barrier requirement of 10CFR71.63(b) for plutonium shipments.

TERMINOLOGY AND NOTATION

Single Failure Proof means that the handling system is designed so that a single failure will not result in the loss of the capability of the system to safely retain the load.

SNF is an acronym for spent nuclear fuel.

STP is Standard Temperature (298°K) and Pressure (1 atm) conditions.

Transport index means the dimensionless number (rounded up to the next tenth) placed on the label of a package, to designate the degree of control to be exercised by the carrier during transportation. The transport index is determined for fissile material packages as the number determined by multiplying the maximum radiation level in millisievert per hour at one meter (3.3 ft) from the external surface of the package by 100 (equivalent to the maximum radiation level in millirem per hour at one meter (3.3 ft)), or, for criticality control purposes, the number obtained as described in 10CFR71.59, whichever is larger.

Trojan Damaged Fuel Container (or Canister) is a Holtec damaged fuel container customdesigned for Trojan plant damaged fuel and fuel debris. Trojan plant damaged fuel and fuel debris not loaded into a Trojan Failed Fuel Can must be loaded into a Trojan Damaged Fuel Container.

Trojan Failed Fuel Can (FFC) is a non-Holtec designed Trojan plant-specific damaged fuel container that may be loaded with Trojan plant damaged fuel assemblies, Trojan fuel assembly metal fragments (e.g., portions of fuel rods, grid assemblies, bottom nozzles, etc.), a Trojan fuel rod storage container, a Trojan Fuel Debris Process Can Capsule, or a Trojan Fuel Debris Process Can.

Trojan Failed Fuel Can Spacer is a square, structural steel tube with a baseplate designed to be placed inside one Trojan Failed Fuel Can to occupy any space between the top of the contents and the top of the FFC in order to minimize movement of the FFC contents during transportation.

Trojan Fuel Debris Process Can is a Trojan plant-specific canister containing fuel debris (metal fragments) and was used to process organic media removed from the Trojan plant spent fuel pool during cleanup operations in preparation for spent fuel pool decommissioning. Trojan Fuel Debris Process Cans are loaded into Trojan Fuel Debris Process Can Capsules or directly into Trojan Failed Fuel Cans.

Trojan Fuel Debris Process Can Capsule is a Trojan plant-specific canister that contains up to five Trojan Fuel Debris Process Cans and is vacuumed, purged, backfilled with helium, and then seal-welded closed.

ZR means any zirconium-based fuel cladding material authorized for use in a commercial nuclear power plant reactor. Any reference to Zircaloy fuel cladding in this SAR applies to any zirconium-based fuel cladding material.

Table 1.0.2

HI-STAR 100 SAR CORRELATION WITH 10CFR71 AND REGULATORY GUIDE 7.9

RG 7.9 Section	10CFR Part 71 Section	HI-STAR SAR Section
1.1	71.31(a)(1), 71.31(a)(2),	1.0, 1.1, 1.2, 1.5
	71.31(a)(3), 71.31(c),	
	71.33(a)(1), 71.33(a)(3),	
	71.35(b), 71.37, 71.59	
1.2	71.31(a)(1), 71.33(a)(2),	1.2, 1.3, 1.4
	71.33(a)(4), 71.33(a)(5),	
	71.33(a)(6), 71.33(b), 71.43(b)	
None	71.31(a)(2), 71.35(a), 71.41(a)	1.5
1.3	None	Appendices 1.A, 1.B, and 1.C
2.1, 2.2	71.31(a)(1), 71.31(c), 71.33	2.1, 2.2
2.3, 2.4	71.43(d)	2.3, 2.4
2.5	71.45	2.5
2.6, 2.7	71.31(a)(2), 71.35(a), 71.41(a),	2.6, 2.7
	71.61, 71.71, 71.73	
2.6	71.35(a), 71.41(a), 71.43(f),	2.6
	71.51(a)(1), 71.55(d)(4), 71.71	
2.7	71.35(a), 71.41(a), 71.73	2.7
None	71.61	2.7
None	71.85(b)	8.1.2.2
2.10	None	2.10
3.1	71.31(a)(1), 71.31(c),	Chapters 1 & 2,
	71.33(a)(5), 71.33(a)(6),	Sections 3.0, 3.1, 3.4, 3.6
	71.33(b)(1), 71.33(b)(3),	
	71.33(b)(5), 71.33(b)(7),	
	71.33(b)(8), 71.51(c)	
3.2, 3.3	71.31(a)(1), 71.33(a)(5)	Chapters 1 & 2,
		Sections 3.0, 3.1, 3.4, 3.6
None	71.31(a)(2), 71.35(a), 71.41(a)	3.0, 3.1, 3.4, 3.5, 3.6
None	71.43(g)	3.0, 3.4, 3.6
3.4	71.43(f), 71.51(a)(1), 71.71	3.0, 3.4, 3.6
3.5	71.73	3.0, 3.5, 3.6
3.6	None	N/A
4.1	71.31(a)(1), 71.31(c),	4.0, 4.1, 4.2, 4.3
	71.33(a)(4), 71.33(a)(5),	
	71.33(b)(1), 71.33(b)(3),	
	71.33(b)(5), 71.33(b)(7),	
	71.43(c), 71.43(d), 71.43(e)	
4.2	71.31(a)(2), 71.35(a), 71.41(a),	4.2, 4.3
	71.43(f), 71.43(h), 71.51(a)(1),	
	71.51(c)	

HI-STAR 100 SAR CORRELATION WITH 10CFR71 AND REGULATORY GUIDE 7.9

RG 7.9 Section	10CFR Part 71 Section	HI-STAR SAR Section
4.3	71.31(a)(2), 71.35(a), 71.41(a),	4.2, 4.3
	71.51(a)(2), 71.51(c)	
4.4	71.63	4.2, 4.3, 7.1
4.5	None	-
5.1	71.31(a)(1), 71.31(c),	5.1
	71.33(a)(5)	
5.2	71.31(a)(1), 71.33(b)(1),	5.2
	71.33(b)(2), 71.33(b)(3)	
5.3	71.31(a), 71.31(b)	5.3
5.4	71.31(a)(2), 71.35(a), 71.41(a),	5.1, 5.4, 5.5
	71.43(f), 71.47(b), 71.51(a)(1),	
	71.51(a)(2)	
5.5	None	Appendices 5.A, 5.B, 5.C
6.1	71.31(a)(1), 71.31(c),	6.1
	71.33(a)(5), 71.35(b), 71.59(b)	
6.2	71.31(a)(1), 71.33(b)(1),	6.2
	71.33(b)(2), 71.33(b)(3), 71.83	
6.3	71.31(a)(2), 71.35(a), 71.41(a)	6.3
6.4	71.35, 71.43(f), 71.51(a)(1),	6.4
	71.55(b), 71.55(d), 71.55(e),	
	71.59	
6.5	71.31(a)(2), 71.35	6.5, Appendix 6.A
6.6	None	6.2, 6.4, Appendices 6.B, 6.C,
		6.D
7.1	71.31(c), 71.35(c), 71.43(g),	7.4, 7.1.3, 7.1.7
	71.47(b), 71.47(c), 71.47(d),	
	71.87, 71.89	
7.2	71.35(c)	7.1.7
7.3	71.87(i)	7.1.7
None	71.35(c)	7.1.7
7.4	None	-
8.1	71.31(c), 71.37(b), 71.85(a),	8.1
	71.85(b), 71.85(c), 71.87(g),	
	71.93(b)	- -
8.2	71.31(c), 71.37(b), 71.87(b),	8.2
	71.87(g), 71.93(b)	

Notes:

"-" There is no HI-STAR SAR section that addresses this.

1.1 INTRODUCTION

HI-STAR 100 (acronym for <u>Holtec International Storage</u>, <u>Transport and Repository</u>) is a spent nuclear fuel (SNF) packaging designed to be in general compliance with the U.S. Department of Energy's (DOE) original design procurement specifications for multi-purpose canisters and large transportation casks [1.1.1], [1.1.2].

The HI-STAR 100 System consists of a sealed, metal multi-purpose canister, herein abbreviated as the "MPC", contained within an overpack with impact limiters. Figure 1.1.1 provides a pictorial view of the HI-STAR 100 System. The HI-STAR 100 System is designed to accommodate a wide variety of spent fuel assemblies in a single overpack design by utilizing different MPC basket designs. The exterior dimensions of all MPCs (except the custom-designed Trojan plant MPCs) are identical to allow the use of a single overpack design. The Trojan plant MPCs are approximately nine inches shorter than the generic Holtec MPC design and have the same outer diameter. Each of the MPCs has different design features (e.g., fuel baskets) to accommodate distinct fuel characteristics. Each MPC is identified by the maximum quantity of fuel assemblies it is capable of receiving. The MPC-24, -24E, and -24EF each can contain a maximum of 24 PWR assemblies; the MPC-32 can contain up to 32 PWR assemblies; and the MPC-68 and -68F each can contain a maximum of 68 BWR fuel assemblies. Figure 1.1.2 depicts the HI-STAR 100 with two of its major constituents, the MPC and the overpack, in a cutaway view. This view does not include depiction of the spacer required for the Trojan version of the MPC-24E/EF design, which is shorter than the Holtec generic MPC-24E/EF design. The spacer is required for the shorter MPC to ensure the design characteristics of the HI-STAR 100 System (e.g., center-of-gravity, MPC lid-to-overpack closure plate gap, etc.) remain bounded by the supporting analyses. See Figure 1.1.5 for a depiction of the Trojan MPC spacer. A drawing of the Trojan MPC spacer is also included in Section 1.4. A summary of the qualification of the spacer for performing its design function is provided in Section 2.7.1.1.

Figure 1.1.2 also indicates that the overpack pocket trunnions are optional appurtenances. Overpack serial numbers 1020-001 through 1020-007 include the pocket trunnions, while later serial number overpacks do not. The impact of this design change on vehicle tie down methods and qualification analyses are discussed in Section 2.5 of this SAR. The pocket trunnions are not part of the qualified vehicle tie-down system for the package. Figure 1.1.3 provides an elevation cross sectional view of an MPC, and Figure 1.1.4 contains an elevation cross sectional view of the HI-STAR 100 overpack.

The HI-STAR 100 System is designed for both storage and transport. The HI-STAR 100 System's multi-purpose design reduces SNF handling operations and thereby enhances radiological protection. Once SNF is loaded and the MPC and overpack are sealed, the HI-STAR 100 System can be positioned on site for temporary or long-term storage or transported directly off-site. The HI-STAR 100 System's ability to both store and transport SNF eliminates repackaging.











1.2 PACKAGE DESCRIPTION

1.2.1 Packaging

The HI-STAR 100 System consists of an MPC designed for BWR or PWR spent nuclear fuel, an overpack that provides the containment boundary and a set of impact limiters that provide energy absorption capability for the normal and hypothetical accident conditions of transport. Each of these components is described below, including information with respect to component fabrication techniques and designed safety features. This discussion is supplemented by a set of drawings in Section 1.4. Section 1.3 provides the HI-STAR 100 design code applicability and details any alternatives to the ASME Code.

Before proceeding to present detailed physical data on HI-STAR 100, it is contextual to summarize the design attributes that set it apart from the prior generation of spent fuel transportation packages.

There are several features in the HI-STAR 100 System design that increase its effectiveness with respect to the safe transport of spent nuclear fuel (SNF). Some of the principal features of the HI-STAR 100 System that enhance its effectiveness are:

- the honeycomb design of the MPC fuel basket
- the effective distribution of neutron and gamma shielding materials within the system
- the high heat rejection capability
- the structural robustness of the multi-shell overpack construction

The honeycomb design of the MPC fuel baskets renders the basket into a multi-flanged plate weldment where all structural elements (box walls) are arrayed in two orthogonal sets of plates. Consequently, the walls of the cells are either completely coplanar (no offset) or orthogonal with each other. There is complete edge-to-edge continuity between contiguous cells.

Among the many benefits of the honeycomb construction is the uniform distribution of the metal mass over the body of the basket (in contrast to the "box and spacer disk" construction where the support plates are localized mass points). Physical reasoning suggests that a uniformly distributed mass provides a more effective shielding barrier than can be obtained from a nonuniform (box and spacer disk) basket. In other words, the honeycomb basket is a more effective radiation attenuation device.

The complete cell-to-cell connectivity inherent in the honeycomb basket structure provides an uninterrupted heat transmission path, making the HI-STAR 100 MPC an effective heat rejection device.

The multi-layer shell construction in the overpack provides a natural barrier against crack propagation in the radial direction across the overpack structure. If, during a hypothetical

accident (impact) event, a crack was initiated in one layer, the crack could not propagate to the adjacent layer. Additionally, it is highly unlikely that a crack would initiate as the thinner layers are more ductile than a thicker plate.

In this Safety Analysis Report the HI-STAR 100 System design is demonstrated to have predicted responses to accident conditions that are clearly acceptable with respect to certification requirements for post-accident containment system integrity, maintenance of subcriticality margin, dose rates, and adequate heat rejection capability. Table 1.2.18 presents a summary of the HI-STAR 100 System performance against these aspects of post-accident performance at two levels. At the first level, the integrity of the MPC boundary prevents release of radioactive material or helium from the MPC, and ingress of moderator. The integrity of the MPC is demonstrated by the analysis of the response of this high quality, ASME Code, Section III, Subsection NB-designed, pressure vessel to the accident loads while in the overpack. With this demonstration of MPC integrity, the excellent performance results listed in the second column of Table 1.2.18 constitutes an acceptable basis for certification of the HI-STAR 100 System for the safe transport of spent nuclear fuel. However, no credit is taken for MPC integrity for certification of the HI-STAR 100 System for the transport of intact or damaged fuel assemblies. Credit is only taken for the additional containment boundary of the MPC-68F and MPC-24EF for the transport of fuel classified as fuel debris in order to meet the requirements of 10 CFR 71.63(b).

The HI-STAR 100 System provides a large margin of safety. The third column in Table 1.2.18 summarizes the performance if the MPC is postulated to suffer gross failure in the post-accident | analysis. Even with this postulated failure, the performance of the HI-STAR 100 System is acceptable for the transport of intact and damaged fuel assemblies, showing the defense-in-depth | methodology incorporated into the HI-STAR 100 System.

The containment boundary of the HI-STAR 100 System is shown to satisfy the special requirements of 10CFR71.61 for irradiated nuclear fuel shipments.

To meet the requirements of 10CFR71.63(b) for plutonium shipments, which is considered applicable for the transport of fuel classified as fuel debris, double containment is provided by | the containment boundary of the overpack and the secondary containment boundary of the MPC-68F and MPC-24EF, serving as a separate inner container.

1.2.1.1 Gross Weight

The gross weight of the HI-STAR 100 System depends on which of the MPCs is loaded into the HI-STAR 100 overpack for shipment. Table 2.2.1 summarizes the maximum calculated component weights for the HI-STAR 100 overpack, impact limiters, and each MPC loaded to maximum capacity with design basis SNF. The maximum gross transport weight of the HI-STAR 100 System is to be marked on the packaging nameplate.

1.2.1.2 Materials of Construction, Dimensions, and Fabrication

All materials used to construct the HI-STAR 100 System are ASME Code materials, except the neutron shield, neutron poison, optional aluminum heat conduction elements, thermal expansion foam, seals, pressure relief devices, aluminum honeycomb, pipe couplings, and other material classified as Not Important to Safety. The specified materials of construction along with outline dimensions for important-to-safety items are provided in the drawings in Section 1.4.

The materials of construction and method of fabrication are further detailed in the subsections that follow. Section 1.3 provides the codes applicable to the HI-STAR 100 packaging for materials, design, fabrication, and inspection, including NRC-approved alternatives to the ASME Code.

1.2.1.2.1 HI-STAR 100 Overpack

The HI-STAR 100 overpack is a heavy-walled steel cylindrical vessel. A single overpack design is provided that is capable of transporting each type of MPC. The inner diameter of the overpack is approximately 68-3/4 inches and the height of the internal cavity is approximately 191-1/8 inches. The overpack inner cavity is sized to accommodate the MPCs. The outer diameter of the overpack is approximately 96 inches and the height is approximately 203-1/4 inches.

Figure 1.2.1 provides a cross sectional elevation view of the overpack containment boundary. The overpack containment boundary is formed by a steel inner shell welded at the bottom to a bottom plate and, at the top, to a heavy top flange with a bolted closure plate. Two concentric grooves are machined into the closure plate for the seals. The closure plate is recessed into the top flange and the bolted joint is configured to protect the closure bolts and seals in the event of a drop accident. The closure plate has test and vent ports that are closed by a threaded port plug with a seal. The bottom plate has a drain port that is also closed by a threaded port plug with a seal. The containment boundary forms an internal cylindrical cavity for housing the MPC.

The outer surface of the overpack inner shell is buttressed with intermediate shells of gamma shielding that are installed in a manner to ensure a permanent state of contact between adjacent layers. Besides serving as an effective gamma shield, these layers provide additional strength to the overpack to resist puncture or penetration. Radial channels are vertically welded to the outside surface of the outermost intermediate shell at equal intervals around the circumference. These radial channels act as fins for improved heat conduction to the overpack outer enclosure shell surface and as cavities for retaining and protecting the neutron shielding. The enclosure shell is formed by welding enclosure shell panels between each of the channels to form additional cavities. Neutron shielding material is placed into each of the radial cavity segments formed by the radial channels, the outermost intermediate shell, and the enclosure shell panels. The exterior flats of the radial channels and enclosure shell panels form the overpack outer enclosure shell (Figure 1.2.2). Atop the outer enclosure shell, pressure relief devices (e.g., rupture disks) are positioned in a recessed area. The relief devices relieve internal pressure that may develop as a result of the fire accident and subsequent off-gassing of the neutron shield material. Within each radial channel, a layer of silicone sponge is positioned to act as a thermal expansion foam to compress as the neutron shield expands in the axial direction. Appendix 1.C

provides material information on the thermal expansion foam. Figure 1.2.2 provides a mid-plane cross section view of the overpack, depicting the inner shell, intermediate shells, radial channels, outer enclosure shell, and neutron shield.

The exposed steel surfaces (except seal seating surfaces) of the overpack and the intermediate shell layers are coated to prevent corrosion. Coating materials are chosen based on the expected service conditions, considering the dual purpose certification status of the HI-STAR 100 System under 10 CFR 72 for spent fuel storage as well as transportation. The coatings applied to the overpack exposed exterior and interior surfaces are specified on the drawings in Section 1.4. The material data on the coatings is provided in Appendix 1.C. The inner cavity of the overpack is coated with a material appropriate to its high temperatures and the exterior of the overpack is coated with a material appropriate for fuel pool operations and environmental exposure. The coating applied to the intermediate shells acts as a surface preservative and is not exposed to the fuel pool or ambient environment.

Lifting trunnions are attached to the overpack top flange for lifting and rotating the cask body between vertical and horizontal positions. The lifting trunnions are located 180° apart in the sides of the top flange. On overpack serial numbers 1020-001 through 1020-007, pocket trunnions are welded to the lower side of the overpack 180° apart to provide a pivoting axis for rotation. The pocket trunnions are slightly off-center to ensure proper rotation direction of the overpack. As shown in Figure 1.1.4, the trunnions do no protrude beyond the cylindrical envelope of the overpack outer enclosure shell. This feature reduces the potential for direct impact on a trunnion in the event of an overpack side impact. After fabrication of HI-STAR overpack serial number 1020-007, the pocket trunnions were deleted from the overpack design.

1.2.1.2.2 <u>Multi-Purpose Canisters</u>

1.2.1.2.2.1 General Description

In this subsection, discussion of those attributes applicable to all of the MPC models is provided. Differences among the models are discussed in subsequent subsections. Specifications for the authorized contents of each MPC model, including non-fuel hardware and neutron sources are provided in Section 1.2.3.

The HI-STAR 100 MPCs are welded cylindrical structures with flat ends. Each MPC is an assembly consisting of a honeycombed fuel basket, a baseplate, a canister shell, a lid with vent and drain ports and cover plates, and a closure ring. The outer diameter of all MPCs and cylindrical height of each generic design MPC is fixed (see discussion in Subsection 1.2.1.2.2.3 regarding Trojan plant-specific MPCs). The number of spent nuclear fuel storage locations in each of the MPCs depends on the fuel assembly characteristics. As the generic MPCs are interchangeable, they correspondingly have identical exterior dimensions. The outer dimension of the MPC is nominally 68-3/8 inches and the length is nominally 190-1/4 inches. Figures 1.2.3-1.2.5 depict the cross sectional views of the different MPCs. Drawings of the MPCs are provided in Section 1.4. Key system data for the HI-STAR 100 System are outlined in Tables 1.2.2 and 1.2.3.

The generic MPC-24/24E/24EF and Trojan plant MPC-24E/EF differ in construction from the MPC-32 and MPC-68/68F in one important aspect: the fuel cells are physically separated from one another by a flux trap between each cell for criticality control (Figures 1.2.3 and 1.2.4). All MPC baskets are formed from an array of plates welded to each other, such that a honeycomb structure is created that resembles a multi-flanged, closed-section beam in its structural characteristics.

The MPC fuel basket is positioned and supported within the MPC shell by a series of basket supports welded to the inside of the MPC shell. In the peripheral area created by the basket, the MPC shell, and the basket supports, optional aluminum heat conduction elements are installed in some early production MPC-68 and MPC-68F models (see Figure 1.2.3). These heat conduction elements are fabricated from thin aluminum alloy 1100 in shapes and a design that allow a snug fit in the confined spaces and ease of installation. The heat conduction elements are along the full length of the MPC basket, except at the drain pipe location, to create a nonstructural thermal connection that facilitates heat transfer from the basket to the shell. In their operating condition, the heat conduction elements conform to, and contact the MPC shell and basket walls. In SAR Revision 10, a refined thermal analysis, described in Chapter 3, has allowed the elimination of these heat conduction elements from the MPC design, thus giving this design feature "optional" status.

Lifting lugs attached to the inside surface of the MPC canister shell serve to permit placement of the empty MPC into the overpack, and are considered non-structural, non-pressure retaining attachments to the MPC pressure boundary. The lifting lugs also serve to axially locate the MPC lid prior to welding. These internal lifting lugs are not used to handle a loaded MPC, since the MPC lid blocks access to the lifting lugs.

The top of the HI-STAR 100 MPC incorporates a redundant closure system. Figure 1.2.6 provides a sketch of the MPC closure details. The MPC lid is a circular plate (fabricated from one piece, or two pieces - split top and bottom) that is edge-welded to the MPC shell. If the two-piece lid design is employed, only the top piece is analyzed as part of the enclosure vessel pressure boundary. The bottom piece acts primarily as a radiation shield and is attached to the top piece with a non-structural, non-pressure retaining weld, as depicted on the MPC enclosure vessel drawing in Section 1.4. The MPC lid is equipped with vent and drain ports that are used to remove moisture and gas from the MPC and backfill the MPC with a specified pressure of inert gas (helium). The vent and drain ports are sealed closed by cover plates welded to the MPC lid before the closure ring is installed. The closure ring is a circular ring edge-welded to the MPC shell and MPC lid. The MPC lid provides sufficient rigidity to allow the entire MPC loaded with SNF to be lifted by the threaded holes in the MPC lid during transfer from the storage-only HI-STORM 100 System to the HI-STAR 100 overpack for transportation. Threaded insert plugs are installed to provide shielding when the threaded holes are not in use.

All MPCs are <u>designed</u> to handle intact fuel assemblies, damaged fuel assemblies, and fuel classified as fuel debris. Damaged fuel and fuel debris must be transported in damaged fuel containers or other approved damaged/failed fuel canister. At this time, only BWR damaged fuel and fuel debris from the Dresden Unit 1 and Humboldt Bay plants is certified for transportation in the MPC-68 and the MPC-68F. Similarly, only PWR damaged fuel and fuel debris from the

Trojan plant is certified for transportation in the Trojan plant-specific MPC-24E and the MPC-24EF. The definitions, and applicable specifications for all authorized contents, including the requirements for canning certain fuel, are provided in Subsection 1.2.3.

Intact SNF can be placed directly into the MPC. Damaged SNF and fuel debris must be placed into a Holtec damaged fuel container or other authorized canister for transportation inside the MPC and the HI-STAR 100 overpack. Figures 1.2.10 through 1.2.11 provide sketches of the containers authorized for transportation of damaged fuel and fuel debris in the HI-STAR 100 System. One Dresden Unit 1 Thoria rod canister, shown in Figure 1.2.11A, is also authorized for transportation in HI-STAR 100.

In order to qualify the MPC-68F and MPC-24EF shells as a secondary containment boundary for the transportation of Dresden Unit 1/Humboldt Bay and Trojan plant fuel debris, respectively, the MPC-68 and MPC-24E enclosure vessels have been slightly modified to further strengthen the lid-to-shell joint area. These fuel debris MPCs are given the "F" suffix (hence, MPC-68F and MPC-24EF)[†]. The differences between the standard and "F-model" MPC lid-to-shell joints are shown on Figure 1.2.17, and include a thickened upper shell, a larger lid-to-shell weld size, and a correspondingly smaller lid diameter. The design of the rest of the enclosure vessel is identical between the standard MPC and the "F-model" MPC.

The MPC-68F and MPC-24EF provide the separate inner container per 10CFR71.63(b) for the HI-STAR 100 System transporting fuel classified as fuel debris to ensure double containment. The overpack containment boundary provides the primary containment boundary.

1.2.1.2.2.2 <u>MPC-24/24E/24EF</u>

The MPC-24 is designed to transport up to 24 PWR intact fuel assemblies meeting the limits specified in Subsection 1.2.3. The MPC 24E is designed to transport up to 24 PWR intact and up to four PWR damaged fuel assemblies in damaged fuel containers. The MPC-24EF is designed to transport up to 24 PWR intact fuel assemblies and up to four PWR damaged fuel assemblies or fuel assemblies classified as fuel debris. At this time, however, generic PWR damaged fuel and fuel debris are not authorized for transportation in the MPC-24E/EF.

All MPC-24-series fuel baskets employ the flux trap design for criticality control, as shown in the drawings in Section 1.4. The fuel basket design for the MPC-24E is an enhanced MPC-24 basket layout designed to improve the fuel storage geometry for criticality control. The fuel basket design of the MPC-24EF is identical to the MPC-24E. The MPC-24E/EF basket designs also employ a higher ¹⁰B loading than the MPC-24, as shown in Table 1.2.3. The differences between the MPC-24EF enclosure vessel design and the MPC-24/24E enclosure vessel are discussed in Subsection 1.2.1.2.2.1.

[†] The drawing in Section 1.4 also denotes an MPC-68FF fuel debris canister design. However, the MPC-68FF is not authorized for use in transportation under the HI-STAR 100 10 CFR 71 CoC.
1.2.1.2.2.3 Trojan Plant MPC-24E/EF

The Trojan plant MPC-24E and -24EF models are designs that have been customized for that plant's fuel and the concrete storage cask being used at the Trojan plant Independent Spent Fuel Storage Installation (ISFSI) (Docket 72-0017). The design features that are unique to the Trojan plant MPCs are specifically noted on the MPC enclosure vessel and MPC-24E/EF fuel basket drawings in Section 1.4. These differences include:

- a shorter MPC fuel basket and cavity length to match the shorter Trojan fuel assembly length
- shorter corner fuel storage cell lengths to accommodate the Trojan Failed Fuel Cans
- a different fuel storage cell and flux trap dimension in the corner cells to accommodate the Trojan Failed Fuel Cans
- a different configuration of the flow holes at the bottom of the fuel basket (rectangular vs. semi-circular)

All other design features in the Trojan MPCs are identical to the generic MPC-24E/EF design. The HI-STAR 100 overpack design has not been modified for the Trojan MPC design.

The technical analyses described in this SAR were verified in most cases to bound the Trojanspecific design features. Where necessary, Trojan plant-specific evaluations were performed and are summarized in the appropriate SAR section. To accommodate the shorter Trojan plant MPC length in a standard-length HI-STAR 100 overpack, a spacer was designed for installation into the overpack above the Trojan MPC (see Figure 1.1.5 and the drawing in Section 1.4) for transportation in the standard-length HI-STAR 100 overpack. This spacer prevents the MPC from moving more than the MPC was analyzed to move in the axial direction and serves to transfer the axial loads from the MPC lid to the overpack top closure plate within the limits of the supporting analyses. See Section 2.7.1.1 for additional discussion of the spacer used with the Trojan MPC design. Hereafter in this SAR, the Trojan plant-specific MPC design is only distinguished from the generic MPC-24E/EF design when necessary to describe unique evaluations performed for those MPCs.

1.2.1.2.2.4 <u>MPC-32</u>

NOTE: The MPC-32 is not certified for transportation at this time.

The MPC-32 is designed to transport up 32 PWR intact fuel assemblies meeting the specifications in Subsection 1.2.3. Damaged fuel and fuel debris are not permitted to be transported in the MPC-32. The MPC-32 enclosure vessel design is identical to the MPC-24/24E enclosure vessel design as shown on the drawings in Section 1.4. The MPC-32 fuel basket does not employ flux traps for criticality control. Credit for burnup of the fuel is taken in the criticality analyses for accident conditions and to meet the requirements of 10 CFR 71.55(b). Because the MPC is designed to preclude the intrusion of moderator under all normal and

credible accident conditions of transport, the moderator intrusion condition analyzed as required by 10 CFR 71.55(b) is a non-mechanistic event for the HI-STAR 100 System.

1.2.1.2.2.5 <u>MPC-68/68F</u>

The MPC-68 is designed to transport up to 68 BWR intact fuel assemblies and damaged fuel assemblies meeting the specifications in Subsection 1.2.3. Zircaloy channels are permitted. At this time, only damaged fuel from the Dresden Unit 1 and Humboldt Bay plants is authorized for transportation in the MPC-68. The MPC-68F is designed to transport <u>only</u> fuel and other authorized material from the Dresden Unit 1 and Humboldt Bay plants meeting the specifications in Subsection 1.2.3. The sole difference between the MPC-68 and MPC-68F fuel basket design is a reduction in the required ¹⁰B areal density in the Boral. A reduction in the required ¹⁰B areal density of the Boral is possible for the MPC-68F due to limited types of fuel and low enrichments permitted to be transported in this MPC model. The differences between the MPC-68F enclosure vessel design and the MPC-68 enclosure vessel are discussed in Subsection 1.2.1.2.2.1.

1.2.1.2.2.6 <u>Alloy X</u>

The HI-STAR MPC is constructed entirely from stainless steel alloy materials (except for the neutron absorber and aluminum vent and drain cap seal washers in all MPCs, and the aluminum heat conduction elements in the first several production units of MPC-68 and MPC-68F). No carbon steel parts are used in the design of the HI-STAR 100 MPC. Concerns regarding interaction of coated carbon steel materials and various MPC operating environments [1.2.1] are not applicable to the HI-STAR MPCs. All structural components in a HI-STAR MPC will be fabricated of Alloy X, a designation that warrants further explanation.

Alloy X is a fictitious material that should be acceptable as a Mined Geological Depository System (MGDS) waste package and that meets the thermophysical properties set forth in this document.

At this time, there is considerable uncertainty with respect to the material of construction for an MPC that would be acceptable as a waste package for the MGDS. Candidate materials being | considered for acceptability by the DOE include:

- Type 316
- Type 316LN
- Type 304
- Type 304LN

The DOE material selection process is primarily driven by corrosion resistance in the potential environment of the MGDS. As the decision regarding a suitable material to meet disposal requirements is not imminent, this application requests approval for use of any one of the four Alloy X materials.

For the MPC design and analysis, Alloy X (as defined in this SAR) may be one of the following materials. Any steel part in an MPC may be fabricated from any of the acceptable Alloy X materials listed below, except that all steel pieces comprising the MPC shell (i.e., the 1/2" thick cylinder) must be fabricated from the same Alloy X stainless steel type:

- Type 316
- Type 316LN
- Type 304
- Type 304LN

The Alloy X approach is accomplished by qualifying the MPC for all mechanical, structural, neutronic, radiological, and thermal conditions using material thermophysical properties that are | the least favorable for the entire group for the analysis in question. For example, when calculating the rate of heat rejection to the outside environment, the value of thermal conductivity used is the lowest for the candidate material group. Similarly, the stress analysis calculations use the lowest value of the ASME Code allowable stress intensity for the entire group. Stated differently, we have defined a material, which is referred to as Alloy X, whose thermophysical properties, from the MPC design perspective, are the least favorable of the candidate materials group. The evaluation of the Alloy X constituents to determine the least favorable properties is provided in Appendix 1.A.

The Alloy X approach is conservative because no matter which material is ultimately utilized, the Alloy X approach guarantees that the performance of the MPC will exceed the analytical predictions contained in this document.

1.2.1.3 Impact Limiters

The HI-STAR 100 overpack is fitted with aluminum honeycomb impact limiters, termed AL-STARTM, one at each end, once the overpack is positioned and secured in the transport frame. The impact limiters ensure the inertia loadings during the normal and hypothetical accident conditions of transport are maintained below design levels. The impact limiter design is discussed further in Chapter 2 and drawings are provided in Section 1.4.

1.2.1.4 Shielding

The HI-STAR 100 System is provided with shielding to minimize personnel exposure. The HI-STAR 100 System will be transported by exclusive use shipment to ensure the external radiation requirements of 10CFR71.47 are met. During transport, a personnel barrier is installed to restrict access to the overpack to protect personnel from the HI-STAR 100 exterior surface temperature in accordance with 10CFR71.43(g). The personnel barrier provides a stand-off equal to the exterior radial dimension of the impact limiters. Figure 1.2.8 provides a sketch of the personnel barrier being installed.

The initial attenuation of gamma and neutron radiation emitted by the radioactive spent fuel is provided by the MPC fuel basket structure built from inter-welded plates and Boral neutron poison panels with sheathing attached to the fuel cell walls. The MPC canister shell, baseplate,

and lid provide additional thicknesses of steel to further reduce gamma radiation and, to a smaller extent, neutron radiation at the outer MPC surfaces. No shielding credit is taken for the aluminum heat conduction elements installed in some of the early production MPC-68 and MPC-68F units.

The primary HI-STAR 100 shielding is located in the overpack and consists of neutron shielding and additional layers of steel for gamma shielding. Neutron shielding is provided around the outside circumferential surface of the overpack. Gamma shielding is provided by the overpack inner, intermediate and enclosure shells with additional axial shielding provided by the bottom plate and the top closure plate. During transport, the impact limiters will provide incremental gamma shielding and provide additional distance from the radiation source at the ends of the package. An additional circular segment of neutron shielding is contained within each impact limiter to provide neutron attenuation.

1.2.1.4.1 Boral Neutron Absorber

Boral is a thermal neutron poison material composed of boron carbide and aluminum alloy 1100. Boron carbide is a compound having a high boron content in a physically stable and chemically inert form. The boron carbide contained in Boral is a fine granulated powder that conforms to ASTM C-750-80 nuclear grade Type III. The aluminum alloy 1100 is a lightweight metal with high tensile strength that is protected from corrosion by a highly resistant oxide film. The two | materials, boron carbide and aluminum, are chemically compatible and ideally suited for longterm use in the radiation, thermal, and chemical environment of a nuclear reactor, spent fuel pool, or dry cask.

The documented historical applications of Boral, in environments comparable to those in spent fuel pools and fuel storage casks, dates to the early 1950s (the U.S. Atomic Energy Commission's AE-6 Water-Boiler Reactor [1.2.2]). Technical data on the material was first printed in 1949, when the report "Boral: A New Thermal Neutron Shield" was published [1.2.3]. In 1956, the first edition of the "Reactor Shielding Design Manual" [1.2.4], contains a section on Boral and its properties.

In the research and test reactors built during the 1950s and 1960s, Boral was frequently the material of choice for control blades, thermal-column shutters, and other items requiring very good thermal-neutron absorption properties. It is in these reactors that Boral has seen its longest service in environments comparable to today's applications.

Boral found other uses in the 1960s, one of which was a neutron poison material in baskets used in the shipment of irradiated, enriched fuel rods from Canada's Chalk River laboratories to Savannah River. Use of Boral in shipping containers continues, with Boral serving as the poison in many cask designs.

Boral has been licensed by the NRC for use in numerous BWR and PWR spent fuel storage | racks and has been extensively used in international nuclear installations.

Boral has been exclusively used in fuel storage applications in recent years. Its use in spent fuel pools as the neutron absorbing material can be attributed to its proven performance and several unique characteristics, such as:

- The content and placement of boron carbide provides a very high removal cross section for thermal neutrons.
- Boron carbide, in the form of fine particles, is homogeneously dispersed throughout the central layer of the Boral panels.
- The boron carbide and aluminum materials in Boral do not degrade as a result of longterm exposure to radiation.
- The neutron absorbing central layer of Boral is clad with permanently bonded surfaces of aluminum.
- Boral is stable, strong, durable, and corrosion resistant.

Boral absorbs thermal neutrons without physical change or degradation of any sort from the anticipated exposure to gamma radiation and heat. The material does not suffer loss of neutron attenuation capability when exposed to high levels of radiation dose.

Holtec International's QA Program ensures that Boral is manufactured under the control and surveillance of a Quality Assurance/Quality Control Program that conforms to the requirements of 10CFR71, Subpart H and 10CFR72, Subpart G. Holtec International has procured over 200,000 panels of Boral from AAR Advanced Structures for over 20 projects. Boral has always been purchased with a minimum ¹⁰B loading requirement. Coupons extracted from production runs were tested using the "wet chemistry" procedure. The actual ¹⁰B loading, out of thousands of coupons tested, has <u>never</u> been found to fall below the design specification. The size of this coupon data base is sufficient to provide confidence that all future procurements will continue to yield Boral with full compliance with the stipulated minimum loading. Furthermore, the surveillance, coupon testing, and material tracking processes that have so effectively controlled | the quality of Boral are expected to continue to yield Boral of similar quality in the future. Nevertheless, to add another layer of insurance, only 75% ¹⁰B credit of the fixed neutron absorber is assumed in the criticality analysis.

The oxide layer that is created from the reaction of the outer aluminum cladding and the edges of the Boral panels with air and water provides a barrier to further reaction of the aluminum cladding with air or the spent fuel pool water during loading and unloading operations. However, with extended submergence in an MPC filled with water or in the plant's spent fuel pool, the hydrostatic pressure can drive water into the Boral core (comprised of particulate B_4C and aluminum powder) where previously unexposed aluminum powder may react with the water to create hydrogen. The rate of hydrogen generation and the total hydrogen generated is dependent on several variables:

- <u>Aluminum particle size</u>: Aluminum particle size in the Boral core and associated porosity affects the amount of aluminum available for reaction with water. Larger aluminum particles yield less surface area for reaction, but higher porosity for aluminum-water interaction; smaller aluminum particles yield more surface area for reaction, but lower porosity for aluminum-water reaction.
- <u>Presence of trace impurities:</u> The presence of trace impurities in the Boral core due to the manufacturing process (i.e., sodium hydroxide, boron oxide, and iron-oxide) can affect the rate of hydrogen production, both increasing and suppressing the reaction. Sodium dissolved in the water increases the pH and tends to increase the rate of hydrogen production. This is counteracted by the boron oxide, which hydrolyzes to boric acid (H₃BO₃) and reduces the rate of hydrogen production. Trace impurities do not affect the total amount of hydrogen generated.
- <u>Pool water chemistry:</u> Chemicals in the plant spent fuel pool water (e.g., copper, boron) can affect the rate of hydrogen production, both increasing (copper) and suppressing (boron) the reaction.
- <u>MPC loading operations:</u> Operating needs or preferences by individual utilities as to when, and for how long the MPC is kept at varying water depths in the spent fuel pool, and how long the MPC is kept filled with water outside the spent fuel pool can affect the amount of aluminum in the Boral core that may be exposed to water.

Due to the variability in hydrogen generation from the Boral-water reaction, the operating procedures in Chapter 7 require monitoring for combustible gases and either exhausting or purging the space beneath the MPC lid during loading and unloading operations when an ignition event could occur (i.e., when the space beneath the MPC lid is open to the welding or cutting operation).

1.2.1.4.2 <u>Holtite-A[™] Neutron Shielding</u>

The specification for the overpack and impact limiter neutron shield material is predicated on functional performance criteria. These criteria are:

- Attenuation of neutron radiation and associated neutron capture to appropriate levels;
- Durability of the shielding material under normal conditions, in terms of thermal, chemical, mechanical, and radiation environments;
- Stability of the homogeneous nature of the shielding material matrix;
- Stability of the shielding material in mechanical or thermal accident conditions to the desired performance levels; and

• Predictability of the manufacturing process under adequate procedural control to yield an in-place neutron shield of desired function and uniformity.

Other aspects of a shielding material, such as ease of handling and prior nuclear industry use, are also considered, within the limitations of the main criteria. Final specification of a shield material is a result of optimizing the material properties with respect to the main criteria, along with the design of the shield system, to achieve the desired shielding results.

Holtite-A is the only approved neutron shield material that fulfills the aforementioned criteria. Holtite-A is a poured-in-place solid borated synthetic neutron-absorbing polymer. Holtite-A is specified with a nominal B_4C loading of 1 weight percent for the HI-STAR 100 System. Appendix 1.B provides the Holtite-A material properties germane to its function as a neutron shield. Holtec has performed confirmatory qualification tests on Holtite-A under the company's QA program.

In the following, a brief summary of the performance characteristics and properties of Holtite-A is provided.

Density

The nominal specific gravity of Holtite-A is 1.68 g/cm^3 as specified in Appendix 1.B. To conservatively bound any potential weight loss at the design temperature and any inability to reach the theoretical density, the density is reduced by 4% to 1.61 g/cm^3 . The density used for the shielding analysis is assumed to be 1.61 g/cm^3 to underestimate the shielding capabilities of the neutron shield.

Hydrogen

The nominal weight concentration of hydrogen is 6.0%. However, all shielding analyses | conservatively assume 5.9% hydrogen by weight in the calculations.

Boron Carbide

Boron carbide dispersed within Holtite-A in finely dispersed powder form is present in 1% (nominal) weight concentration. Holtite-A may be specified with a B_4C content of up to 6.5 | weight percent. For the HI-STAR 100 System, Holtite-A is specified with a nominal B_4C weight percent of 1%.

Design Temperature

The design temperature of Holtite-A is set at 300°F. The maximum spatial temperature of Holtite-A under all normal operating conditions must be demonstrated to be below this design temperature.

Thermal Conductivity

It is evident from Figure 1.2.2 that Holtite-A is directly in the path of heat transmission from the inside of the overpack to its outside surface. For conservatism, however, the design basis thermal conductivity of Holtite-A under heat rejection conditions is set equal to zero. The reverse condition occurs under a postulated fire event when the thermal conductivity of Holtite-A aids in the influx of heat to the stored fuel in the fuel basket. The thermal conductivity of Holtite-A is conservatively set at 1 Btu/hr-ft- $^{\circ}$ F for all fire accident analyses.

The Holtite-A neutron shielding material is stable at normal design temperatures over the long term and provides excellent shielding properties for neutrons.

1.2.1.4.3 Gamma Shielding Material

For gamma shielding, HI-STAR 100 utilizes carbon steel in plate stock form. Instead of utilizing a thick forging, the gamma shield design in the HI-STAR 100 overpack borrows from the concept of layered vessels from the field of ultra-high pressure vessel technology. The shielding is made from successive layers of plate stock. The fabrication of the shell begins by rolling the inner shell plate and making the longitudinal weld seam. Each layer of the intermediate shells is constructed from two halves. The two halves of the shell are precision sheared, beveled, and rolled to the required radii. The two halves of the second layer are wrapped around the first shell. Each shell half is positioned in its location and while applying pressure using a specially engineered fixture, the halves are tack welded. The beveled edges to be joined are positioned to make contact or have a slight gap. The second layer is made by joining the two halves using two longitudinal welds. Successive layers are assembled in a like manner. Thus, the welding of every successive shell provides a certain inter-layer contact (Figure 1.2.7).

A thick structural component radiation barrier is thus constructed with four key features, namely:

- The number of layers can be increased as necessary to realize the required design objectives.
- The layered construction is ideal to stop propagation of flaws.
- The thinner plate stock is much more ductile than heavy forgings used in other designs.
- Post-weld heat treatment is not required by the ASME Code, simplifying fabrication.

1.2.1.5 Lifting and Tie-Down Devices

The HI-STAR 100 overpack is equipped with two lifting trunnions located in the top flange. The lifting trunnions are designed in accordance with 10CFR71.45, NUREG-0612 [1.2.11], and ANSI N14.6 [1.3.3], manufactured from a high strength alloy, and are installed in threaded openings. The lifting trunnions may be secured in position by optional locking pads, shaped to | make conformal contact with the curved overpack. Once the locking pad is bolted in position, the inner diameter is sized to restrain the trunnion from backing out. The two off-center pockets

located near the overpack bottom plate on overpack serial numbers 1020-001 through 1020-007 are pocket trunnions. The pocket trunnions were eliminated from the design after serial number 1020-007 was fabricated and are no longer considered qualified tie-down devices. However, the pocket trunnions on these overpacks may still be used for normal handling activities such as upending and downending.

The lifting, upending, and downending of the HI-STAR 100 System requires the use of external handling devices. A lifting yoke is utilized when the cask is to be lifted or set in a vertical orientation. For those overpacks that have been fabricated with the pocket trunnions, transport and rotation cradles may include rotation trunnions that interface with the pocket trunnions to provide a pivot axis. A lift yoke may be connected to the lifting trunnions and the crane hook used for upending or downending the HI-STAR 100 System by rotating on the pocket trunnions for these overpacks. For those overpacks fabricated without pocket trunnions, the overpack must be transferred into the transport saddle with appropriate lift rigging. If an overpack having pocket trunnions is secured to the transport vehicle without engaging the pocket trunnions, plugs are required to be installed in the pocket to provide radiation shielding (see the overpack drawing in Section 1.4).

For transportation, the HI-STAR 100 System is engineered to be mounted on a transport frame secured to the transporter bed. Figure 1.2.8 provides a sketch of the HI-STAR 100 System secured for transport and the drawing in Section 1.4 provides additional details. The transport frame has a lower saddle with attachment points for belly slings around the cask body designed to prevent excessive vertical or lateral movement of the cask during normal transportation. The impact limiters affixed to both ends of the cask are designed to transmit the design basis axial loads into the cradle structure. See Section 2.5 for discussion of the qualification of tie-down devices.

The top of the MPC lid is equipped with four threaded holes that allow lifting of the loaded MPC. These holes allow the loaded MPC to be raised/lowered from the HI-STAR overpack. For users of the HI-STORM 100 Dry Storage System, MPC handling operations are performed using a HI-TRAC transfer cask of the HI-STORM 100 System (Docket No. 72-1014). The HI-TRAC transfer cask allows the sealed MPC loaded with spent fuel to be transferred from the HI-STORM 100 overpack (storage-only) to the HI-STAR 100 overpack, or vice versa. The threaded holes in the MPC lid are designed in accordance with NUREG-0612 and ANSI N14.6 and are plugged during transportation to prevent radiation streaming.

1.2.1.6 <u>Heat Dissipation</u>

The HI-STAR 100 System can safely transport SNF by maintaining the fuel cladding temperature below the limits specified in Table 1.2.3 for normal and accident conditions. These limits have been established consistent with the guidance in NRC Interim Staff Guidance (ISG) document No. 11, Revision 2 (Ref. [1.2.14]). The temperature of the fuel cladding is dependent on the decay heat and the heat dissipation capabilities of the cask. The total heat load per BWR and PWR MPC is identified in Table 1.2.3. The SNF decay heat is passively dissipated without any mechanical or forced cooling.

The HI-STAR 100 System must meet the requirements of 10CFR71.43(g) for the accessible surface temperature limit. To meet this requirement the HI-STAR 100 System is shipped as an exclusive use shipment and includes an engineered personnel barrier during transport.

The primary heat transfer mechanisms in the HI-STAR 100 System are conduction and surface radiation.

The free volume of the MPC and the annulus between the external surface of the MPC and the inside surface of the overpack containment boundary are filled with 99.995% pure helium gas during fuel loading operations. Table 1.2.3 specifies the acceptance criteria for helium fill pressure in the MPC internal cavity. Besides providing an inert dry atmosphere for the fuel cladding, the helium also provides conductive heat transfer across any gaps between the metal surfaces inside the MPC and in the annulus between the MPC and overpack containment boundary. Metal conduction transfers the heat throughout the MPC fuel basket, through the MPC aluminum heat conduction elements (if installed) and shell, through the overpack inner shell, intermediate shells, steel radial connectors and finally, to the outer neutron shield enclosure shell. The most adverse temperature profiles and thermal gradients for the HI-STAR 100 System with each of the MPCs are discussed in detail in Chapter 3. The thermal analysis in Chapter 3 no longer takes credit for the aluminum heat conduction elements and they have been designated as optional equipment.

1.2.1.7 <u>Coolants</u>

There are no coolants utilized in the HI-STAR 100 System. As discussed in Subsection 1.2.1.6 above, helium is sealed within the MPC internal cavity. The annulus between the MPC outer surface and overpack containment boundary is also purged and filled with helium gas.

1.2.1.8 Pressure Relief Systems

No pressure relief system is provided on the HI-STAR 100 packaging containment boundary.

The sole pressure relief devices are provided in the overpack outer enclosure (Figure 1.1.4). The overpack outer enclosure contains the neutron shield material. Normal loadings will not cause the rupture disks to open. The rupture disks are installed to relieve internal pressure in the neutron shield cavities caused by the fire accident. The overpack outer enclosure is not designed as a pressure vessel. Correspondingly, the rupture disks are designed to open at relatively low pressures as stated below.

Relief Device location	Set pressure, psig
Overpack outer enclosure	30, +/- 5

1.2.1.9 Security Seal

The HI-STAR 100 packaging provides a security seal that while intact, provides evidence that the package has not been opened by unauthorized persons. When installed, the impact limiters | cover all penetrations into the HI-STAR 100 packaging containment boundary. Therefore, the security seal is placed to ensure that the impact limiters are not removed which thereby ensures that the package has not been opened. As shown on the HI-STAR transport assembly drawing in Section 1.4, security seals are provided on one impact limiter attachment bolt on the top impact limiter and through two adjacent bolts on the bottom impact limiter. A hole is provided in the head of the bolt and the impact limiter. Lockwire shall be threaded through the hole and joined with a security seal.

1.2.1.10 Design Life

The design life of the HI-STAR 100 System is 40 years. This is accomplished by using materials of construction with a long proven history in the nuclear industry and specifying materials known to withstand their operating environments with little to no degradation. A maintenance program, as specified in Chapter 8, is also implemented to ensure the HI-STAR 100 System will exceed its design life of 40 years. The design considerations that assure the HI-STAR 100 System performs as designed throughout the service life include the following:

HI-STAR Overpack

- Exposure to Environmental Effects
- Material Degradation
- Maintenance and Inspection Provisions

<u>MPC</u>

- Corrosion
- Structural Fatigue Effects
- Maintenance of Helium Atmosphere
- Allowable Fuel Cladding Temperatures
- Neutron Absorber Boron Depletion

1.2.2 Operational Features

Table 1.2.7 provides the sequence of basic operations necessary to load fuel and prepare the HI-STAR 100 System for transport. More detailed guidance for transportation-related loading, unloading, and handling operations is provided in Chapter 7 and is supported by the drawings in Section 1.4. A summary of the loading and unloading operations is provided below. Figures 1.2.9 and 1.2.16 provide a pictorial view of the loading and unloading operations, respectively.

1.2.2.1 Applicability of Operating Procedures for the Dual-Purpose HI-STAR 100 System

The HI-STAR 100 System is a dual-purpose system certified for use as a dry storage cask under 10 CFR 72 and a transportation package under 10 CFR 71. In addition, the MPC is certified for use under 10 CFR 72 in the storage-only HI-STORM 100 System (a ventilated concrete cask system). Therefore, it is possible that the HI-STAR 100 overpack and/or the MPC may be loaded, prepared, and sealed under the operating procedures for storage, delineated in the HI-STAR 100 storage FSAR (Docket 72-1008) or the HI-STORM 100 storage FSAR (Docket 72-1014). In those cases, the operating procedures governing MPC and overpack preparation for storage would apply. The MPC and HI-STAR 100 overpack, as applicable, must be confirmed to meet all requirements of the Part 71 Certificate of Compliance before being released for shipment.

For those instances where the MPC is being loaded and shipped off-site in a HI-STAR 100 overpack under 10 CFR 71 without first being deployed at an ISFSI (known as "load-and-go" operations), the operating procedures in Chapter 7 (and summarized below) apply for preparation of the MPC and HI-STAR overpack. For those cases where the MPC is transferred from storage in a HI-STORM overpack to a HI-STAR overpack for shipment, the operating procedures in Chapter 7 (and summarized below) govern the preparation activities for the HI-STAR overpack.

Loading Operations

At the start of loading operations, the overpack is configured with the closure plate removed. The lift yoke is used to position the overpack in the designated preparation area or setdown area for overpack inspection and MPC insertion. The annulus is filled with plant demineralized water and an inflatable annulus seal is installed. The inflatable seal prevents contact between spent fuel pool water and the MPC shell reducing the possibility of contaminating the outer surfaces of the MPC. The MPC is then filled with spent fuel pool water or plant demineralized water (borated as required for MPC-32). The overpack and MPC are lowered into the spent fuel pool for fuel loading using the lift yoke. Pre-selected assemblies are loaded into the MPC and a visual verification of the assembly identification is performed.

While still underwater, a thick shielding lid (the MPC lid) is installed. The lift yoke is remotely engaged to the overpack lifting trunnions and is used to lift the overpack close to the spent fuel pool surface. The MPC lift bolts (securing the MPC lid to the lift yoke) are removed. As the overpack is removed from the spent fuel pool, the lift yoke and overpack are sprayed with demineralized water to help remove contamination.

The overpack is removed from the pool and placed in the designated preparation area. The top surfaces of the MPC lid and the top flange of the overpack are decontaminated. The inflatable annulus seal is removed, and an annulus shield is installed. The annulus shield provides additional personnel shielding at the top of the annulus and also prevents small items from being dropped into the annulus (foreign material exclusion). If used, the Automated Welding System (AWS) is installed. The MPC water level is lowered slightly and the space under the MPC lid is purged or exhausted and monitoring is performed. The MPC lid is seal-welded using the AWS. Liquid penetrant examinations are performed on the root and final passes and ultrasonic examination is also performed on the MPC lid-to-shell weld or, in place of the ultrasonic examination, the weld may be inspected by multiple-pass liquid penetrant examination at approximately every 3/8 inch of weld depth. Then a small volume of the water is displaced with helium gas. The helium gas is used for leakage testing. A helium leakage rate test is performed on the MPC lid confinement weld (lid-to-shell) to verify weld integrity and to ensure that the leakage rates are within acceptance criteria. The MPC water is displaced from the MPC by blowing pressurized helium or nitrogen gas into the vent port of the MPC, thus displacing the water through the drain line. At the appropriate time in the sequence of activities, based on the type of test performed (hydrostatic or pneumatic), a pressure test of the MPC enclosure vessel is performed.

The Forced Helium Dehydration (FHD) System is connected to the MPC and is used to remove residual water from the MPC and reduce the level of moisture in the MPC to acceptable levels. This is accomplished by recirculating dry, heated helium through the MPC cavity to absorb the moisture. When the helium exiting the MPC is determined to meet the required moisture limit, the MPC is considered sufficiently dried for transportation (see Section 3.4.1.1.16 for a description of the FHD System.

Following MPC drying operations, the MPC is backfilled with a predetermined amount of helium gas. The helium backfill ensures adequate heat transfer, provides an inert atmosphere for fuel cladding integrity, and provides the means of future leakage rate testing of the MPC enclosure vessel boundary welds. Cover plates are installed and seal-welded over the MPC vent and drain ports with liquid penetrant examinations performed on the root and/or final passes, depending on the number of weld passes required. That is, if only a single weld pass is required, only a final liquid penetrant examination is performed. The cover plates are helium leakage tested to confirm that they meet the established leakage rate criteria.

The MPC closure ring is then placed on the MPC, aligned, tacked in place, and seal welded, providing redundant closure of the MPC enclosure vessel closure welds. Tack welds are visually examined, and the root and/or final welds (depending on the number of weld passes required) are inspected using the liquid penetrant examination technique to ensure weld integrity. The annulus shield is removed and the remaining water in the annulus is drained. The AWS is removed. The overpack closure plate is installed and the bolts are torqued. The overpack annulus is dried using the vacuum drying system (VDS).

If the MPC being transported is an "F-model" canister, a helium leakage test on the canister must be performed to confirm the integrity of the secondary containment boundary prior to backfilling the overpack annulus.

The overpack annulus is backfilled with helium gas for heat transfer and seal testing. Concentric metallic seals in the overpack closure plate prevent the leakage of the helium gas from the annulus and provide the containment boundary to the release of radioactive materials. The seals on the overpack vent and drain port plugs are leak tested along with the overpack closure plate inner seal. Cover plates with metallic seals are installed over the overpack vent and drain ports to provide redundant closure of the overpack penetrations. A port plug with a metallic seal is

installed in the overpack closure plate test port to provide fully-redundant closure of all overpack penetrations.

The overpack is surveyed for removable contamination and secured on the transport vehicle with impact limiters installed, the security seals are attached, and the personnel barrier is installed. The HI-STAR 100 packaging is then ready for transport.

Unloading Operations

The HI-STAR 100 System unloading procedures describe the general actions necessary to prepare the MPC for unloading, cool the stored fuel assemblies in the MPC (if necessary), flood | the MPC cavity, remove the lid welds, unload the spent fuel assemblies, and recover the overpack and empty MPC. Special precautions are outlined to ensure personnel safety during the unloading operations, and to prevent the risk of MPC overpressurization and thermal shock to the stored spent fuel assemblies.

After removing the impact limiters, the overpack and MPC are positioned in the designated preparation area. At the site's discretion, a gas sample is drawn from the overpack annulus and analyzed. The gas sample provides an indication of MPC enclosure vessel performance. The annulus is depressurized, the overpack closure plate is removed, and the annulus is filled with plant demineralized water. The annulus shield is installed to protect the annulus from debris produced from the lid removal process. Similarly, overpack top surfaces are covered with a protective fire-retarding blanket.

The Weld Removal System (WRS) is positioned on the MPC lid. The MPC closure ring is core drilled over the locations of the vent and drain port cover plates. The MPC closure ring and vent and drain port cover plates are core drilled to the extent necessary to allow access by the Remote Valve Operating Assemblies (RVOAs). Local ventilation is established around the vent and drain ports. The RVOAs are connected to allow access to the MPC cavity for re-flooding operations.

The MPC cavity gas is verified to be below an appropriate temperature (approximately 200°F) to allow water flooding. Depending on the time since initial fuel loading and the age and burnup of the contained fuel, mechanical cooling of the MPC cavity gas may or may not be required to ensure the cavity gas temperature meets the acceptance criterion. A thermal evaluation should be performed to determine the MPC bulk cavity gas temperature at the time of unloading. Based on that thermal evaluation, if the MPC cavity gas temperature does not already meet the acceptance limit, any appropriate means to cool the cavity gas may be employed to reduce the gas temperature to the acceptance criterion. Typically, this may involve intrusive means, such as recirculation cooling of the MPC cavity helium, or non-intrusive means, such as cooling of the exterior surface of the MPC enclosure vessel with water or air. The thermal evaluation should include an evaluation of the cooling process, if required, to determine the appropriate criteria for the cooling process, such as fluid flow rate(s), fluid temperature(s), and the cooling duration required to meet the acceptance criterion. Following fuel cool-down (if required), the MPC is flooded with water. The WRS is positioned for MPC lid-to-shell weld removal. The WRS is then removed with the MPC lid left in place. The annulus shield is removed and the inflatable annulus seal is installed and pressurized. The MPC lid is rigged to the lift yoke and the lift yoke is engaged to overpack lifting trunnions. The overpack is placed in the spent fuel pool and the MPC lid is removed. All fuel assemblies are returned to the spent fuel storage racks. The overpack and MPC are returned to the designated preparation area. The annulus water is drained and the MPC and overpack are dispositioned for re-use or waste.

1.2.3 Contents of Package

The HI-STAR 100 packaging is classified as a Type B package under 10CFR71. As the HI-STAR 100 System is designed to transport spent nuclear fuel, the maximum activity of the contents requires that the HI-STAR 100 packaging be classified as Category I in accordance with Regulatory Guide 7.11 [1.2.10]. This section delineates the authorized contents permitted for shipment in the HI-STAR 100 System, including fuel assembly types; non-fuel hardware; neutron sources; physical parameter limits for fuel assemblies and sub-components; enrichment, burnup, cooling time, and decay heat limits; location requirements; and requirements for canning the material.

1.2.3.1 Determination of Design Basis Fuel

The HI-STAR 100 package is designed to transport most types of fuel assemblies generated in the commercial U.S. nuclear industry. Boiling-water reactor (BWR) fuel assemblies have been supplied by General Electric (GE), Siemens (SPC), Exxon Nuclear, ANF, UNC, ABB Combustion Engineering, Allis-Chalmers (AC) and Gulf Atomic. Pressurized-water reactor (PWR) fuel assemblies are generally supplied by Westinghouse, Babcock & Wilcox, ANF, and ABB Combustion Engineering. ANF, Exxon, and Siemens are historically the same manufacturing company under different ownership. Within this report, SPC is used to designate fuel manufactured by ANF, Exxon, or Siemens. Publications such as Refs. [1.2.6], [1.2.7], and [1.2.15] provide a comprehensive description of fuel discharged from U.S. reactors. A central object in the design of the HI-STAR 100 System is to ensure that a majority of SNF discharged from the U.S. reactors can be transported in one of the MPCs.

The cell openings in the fuel basket have been sized to accommodate all BWR and PWR assemblies listed in Refs. [1.2.6], [1.2.7], and [1.2.15], except as noted below. Similarly, the cavity length of the MPC has been set at a dimension that permits transportation of most types of PWR fuel assemblies and BWR fuel assemblies with or without fuel channels. The one exception is as follows:

• The South Texas Units 1 & 2 SNF, and CE 16x16 System 80TM SNF are too long to be accommodated in the available MPC cavity length.

In addition to satisfying the cross sectional and length compatibility, the active fuel region of the SNF must be enveloped in the axial direction by the neutron absorber located in the MPC fuel basket. Alignment of the neutron absorber with the active fuel region is ensured by the use of upper and lower fuel spacers suitably designed to support the bottom and restrain the top of the

fuel assembly. The spacers axially position the SNF assembly such that its active fuel region is properly aligned with the neutron absorber in the fuel basket. Figure 1.2.15 provides a pictorial representation of the fuel spacers positioning the fuel assembly active fuel region. Both the upper and lower fuel spacers are designed to perform their function under normal and hypothetical accident conditions of transport. Due to the shorter, custom MPC design for Trojan plant fuel, only lower fuel spacers are needed for certain fuel assemblies that do not contain integral control rod assemblies. This creates the potential for a slight misalignment between the active fuel region of a fuel assembly and the neutron absorber panels affixed to the cell walls of the Trojan MPCs. This condition is addressed in the criticality evaluations described in Chapter 6.

In summary, the geometric compatibility of the SNF with the MPC designs does not require the definition of a design basis fuel assembly. This, however, is not the case for structural, containment, shielding, thermal-hydraulic, and criticality criteria. In fact, the same fuel type in a category (PWR or BWR) may not control the cask design in all of the above-mentioned criteria. To ensure that no SNF listed in Refs. [1.2.6], [1.2.7], and [1.2.15] that is geometrically admissible in the HI-STAR MPC is precluded from loading, it is necessary to determine the governing fuel specification for each analysis criteria. To make the necessary determinations, potential candidate fuel assemblies for each qualification criteria were considered. Table 1.2.8 lists the PWR fuel assemblies evaluated. These fuel assemblies were evaluated to define the governing design criteria for PWR fuel. The BWR fuel assembly designs evaluated are listed in Table 1.2.9. Tables 1.2.10 and 1.2.11 provide the fuel characteristics determined to be acceptable for transport in the HI-STAR 100 System. Each "array/class" listed in these tables represents a bounding set of parameters for one or more fuel assembly types. The array/classes are defined in SAR Section 6.2. Table 1.2.12 lists the BWR and PWR fuel assembly designs that are found to govern for the qualification criteria, namely reactivity, shielding, and thermal. Thermal is broken down into three criteria, namely: 1) fuel assembly effective planar conductivity, 2) fuel basket effective axial conductivity, and 3) MPC density and heat capacity. Substantiating results of analyses for the governing assembly types are presented in the respective chapters dealing with the specific qualification topic. Tables 1.2.10, 1.2.11, and 1.2.21 through 1.2.36 provide the specific limits for all material authorized to be transported in the HI-STAR 100 System. Additional information on the design basis fuel definition is presented in the following subsections.

1.2.3.2 Design Payload for Intact Fuel

Intact fuel assemblies are defined as fuel assemblies without known or suspected cladding defects greater than pinhole leaks and hairline cracks, and which can be handled by normal means. The design payload for intact fuel to be transported in the HI-STAR 100 System is provided in Tables 1.2.10, 1.2.11, and 1.2.22 through 1.2.36. The placement of a single stainless steel clad fuel assembly in an MPC necessitates that all fuel assemblies (stainless steel clad or Zircaloy clad) stored in that MPC meet the maximum heat generation requirements for stainless steel clad fuel. Stainless steel clad fuel assemblies are not authorized for transportation in the MPC-68F or MPC-32.

Fuel assemblies without fuel rods in fuel rod locations cannot be classified as intact fuel unless dummy fuel rods, which occupy a volume equal to or greater than the original fuel rods, replace

the missing rods prior to loading. Any intact fuel assembly that falls within the geometric, | thermal, and nuclear limits established for the design basis intact fuel assembly can be safely transported in the HI-STAR 100 System.

The fuel characteristics specified in Tables 1.2.10, 1.2.11, and 1.2.21 have been evaluated in this SAR and are acceptable for transport in the HI-STAR 100 System.

1.2.3.3 Design Payload for Damaged Fuel and Fuel Debris

Damaged fuel and fuel debris are defined in Table 1.0.1. The only PWR damaged fuel and fuel debris authorized for transportation in the HI-STAR 100 System is that from the Trojan plant. The only BWR damaged fuel and fuel debris authorized for transportation in the HI-STAR 100 System is that from the Dresden Unit 1 and Humboldt Bay plants.

Damaged fuel may only be transported in the MPC-24E, MPC-24EF, MPC-68, or MPC-68F as shown in Tables 1.2.23 through 1.2.26. Fuel debris may only be transported in the MPC-24EF and the MPC-68F as shown in Tables 1.2.24 and 1.2.26. Damaged fuel and fuel debris must be transported in stainless steel Holtec damaged fuel containers (DFCs) or other approved stainless steel damaged/failed fuel canister in the HI-STAR 100 System. The list of approved damaged/failed fuel canisters and associated SAR figures are provided below:

- Holtec-designed Dresden Unit 1 and Humboldt Bay Damaged Fuel Container(Figure 1.2.10)
- Sierra Nuclear-designed Trojan Failed Fuel Can (Figure 1.2.10A) containing Trojan damaged fuel, fuel debris, or Trojan Fuel debris process cans; or containing Trojan Fuel Debris Process Can Capsules (Figure 1.2.10C), which themselves contain Trojan Fuel Debris Process Cans (Figure 1.2.10B).
- Holtec-designed Damaged Fuel Container for Trojan plan fuel (Figure 1.2.10D)
- Dresden Unit 1's TN Damaged Fuel Container (Figure 1.2.11)
- Dresden Unit 1's Thoria Rod Canister (Figure 1.2.11A)

1.2.3.3.1 BWR Damaged Fuel and Fuel Debris

Dresden Unit 1 (UO₂ fuel rods and MOX fuel rods) and Humboldt Bay fuel arrays (Assembly Classes 6x6A, 6x6B, 6x6C, 7x7A, and 8x8A) are authorized for transportation as damaged fuel in the MPC-68 and damaged fuel or fuel debris in the MPC-68F. No other BWR damaged fuel or fuel debris is authorized for transportation.

The limits for transporting Dresden Unit 1 and Humboldt Bay damaged fuel and fuel debris are given in Table 1.2.23 and 1.2.24. The placement of a single damaged fuel assembly in an MPC-68 or MPC-68F, or a single fuel debris damaged fuel container in an MPC-68F necessitates that

all fuel assemblies (intact, damaged, or debris) placed in that MPC meet the maximum heat generation requirements specified in Tables 1.2.23 and 1.2.24.

The fuel characteristics specified in Tables 1.2.11, 1.2.23 and 1.2.24 for Dresden Unit 1 and Humboldt Bay fuel arrays have been evaluated in this SAR and are acceptable for transport as damaged fuel or fuel debris in the HI-STAR 100 System. Because of the long cooling time, small size, and low weight of spent fuel assemblies qualified as damaged fuel or fuel debris, the DFC and its contents are bounded by the structural, thermal, and shielding analyses performed for the intact BWR design basis fuel. Separate criticality analysis of the bounding fuel assembly for the damaged fuel and fuel debris has been performed in Chapter 6.

As Dresden Unit 1 and Humboldt Bay fuel assemblies classified as fuel debris have significant cladding damage, no cladding integrity is assumed. To meet the double containment criteria of 10CFR71.63(b) for plutonium shipments, the MPC-68F provides the secondary containment boundary (separate inner container), while the overpack provides the primary containment boundary.

The fuel characteristics specified in Table 1.2.11 for the Dresden Unit 1 and Humboldt Bay fuel | arrays (Assembly Classes 6x6A, 6x6B, 6x6C, 7x7A, and 8x8A) have been evaluated in this SAR and are acceptable for transport as damaged fuel or fuel debris in the HI-STAR 100 System after being placed in a damaged fuel container.

1.2.3.3.2 PWR Damaged Fuel and Fuel Debris

The PWR damaged fuel and fuel debris authorized for transportation in the HI-STAR 100 System is limited to that from the Trojan plant. The limits for transporting Trojan plant damaged fuel and fuel debris in the Trojan MPC-24E/EF are given in Tables 1.2.10, 1.2.25 and 1.2.26. All Trojan plant damaged fuel, and fuel debris listed below is authorized for transportation in the HI-STAR 100 System [1.2.12]:

- Damaged fuel assemblies in Trojan failed fuel cans
- Damaged fuel assemblies in Holtec's Trojan plant PWR damaged fuel container
- Fuel assemblies classified as fuel debris in Trojan failed fuel cans
- Trojan fuel assemblies classified as fuel debris in Holtec's Trojan damaged fuel container
- Fuel debris consisting of loose fuel pellets, fuel pellet fragments, and fuel assembly metal fragments (portions of fuel rods, portions of grid assemblies, bottom nozzles, etc.) in Trojan failed fuel cans
- Trojan fuel debris process cans loaded into Trojan fuel debris process can capsules and then into Trojan failed fuel cans. The fuel debris process cans contain fuel debris (metal fragments) and were used to process organic media removed from the Trojan spent fuel

pool during cleanup operations in preparation for decommissioning the pool. The fuel debris process cans have metallic filters in the can bottom and lid that allowed removal of water and organic media using high temperature steam, while retaining the solid residue from the processed media and fuel debris inside the process can[†]. Up to five process cans can be loaded into a process can capsule, which is vacuumed, purged, backfilled with helium, and seal-welded closed to provide a sealed containment for the fuel debris.

One Trojan Failed Fuel Can is not completely filled with fuel debris. Therefore, a stainless steel failed fuel can spacer is installed in this FFC to minimize movement of the fuel debris during normal transportation and hypothetical accident conditions. The spacer is a long, square tube with a baseplate that rests atop the fuel debris inside the Trojan FFC. A drawing of the Trojan failed fuel can spacer is provided in Section 1.4. A summary of the structural analysis of the FFC spacer is provided in Section 2.6.1.3.1.3.

1.2.3.4 Structural Payload Parameters

The main physical parameters of an SNF assembly applicable to the structural evaluation are the fuel assembly length, envelope (cross sectional dimensions), and weight. These parameters, which define the mechanical and structural design, are listed in Tables 1.2.22 through 1.2.27 for the various MPC models. The centers of gravity reported in Chapter 2 are based on the maximum fuel assembly weight. Upper and lower fuel spacers (as appropriate) maintain the axial position of the fuel assembly within the MPC basket and, therefore, the location of the center of gravity. The upper and lower spacers are designed to withstand normal and accident conditions of transport. An axial clearance of approximately 2 inches is provided to account for the irradiation and thermal growth of the fuel assemblies. The suggested upper and lower fuel spacer lengths are listed in Tables 1.2.16 and 1.2.17. Due to the custom design of the Trojan MPCs, only lower fuel spacers are required with Trojan plant fuel assemblies not containing non-fuel hardware or neutron sources. In order to qualify for transport in the HI-STAR 100 MPC, the SNF must satisfy the physical parameters listed in Tables 1.2.21 through 1.2.36, as applicable.

1.2.3.5 Thermal Payload Parameters

The principal thermal design parameter for the fuel is the peak fuel cladding temperature, which is a function of the maximum heat generation rate per assembly and the decay heat removal capabilities of the HI-STAR 100 System. The maximum heat generation rate per assembly for the design basis fuel assembly is based on the fuel assembly type with the lowest thermal performance characteristics. The parameters that define this decay heat design basis fuel are listed in Table 1.2.12. The governing thermal parameters to ensure that the range of SNF discussed previously are bounded by the thermal analysis discussed in detail and specified in Chapter 3. By utilizing these bounding thermal parameters, the calculated peak fuel rod cladding temperatures are conservative for the actual spent fuel assemblies, which are apt to have a higher thermal conductivity.

[†] The Trojan Fuel Debris Process Cans were used in the spent fuel pool cleanup effort conducted as part of plant decommissioning. This project is complete and not associated with certification of Trojan fuel debris for transportation in the HI-STAR 100 System under 10 CFR 71.

The peak fuel cladding temperature limit for normal conditions of transport is 400°C (752°F), which is consistent with the guidance in ISG-11, Revision 2 [1.2.14]. Tables 1.2.21 through 1.2.27 provide the maximum heat generation for all fuel assemblies authorized for transportation in the HI-STAR 100 System. The basis for these limits is discussed in Chapter 3.

Finally, the axial variation in the heat emission rate in the design basis fuel is defined based on the axial burnup distribution. For this purpose, the data provided in Refs. [1.2.8], [1.2.9], and [1.2.12] are utilized and summarized in Table 1.2.15 and Figures 1.2.13, 1.2.13A, and 1.2.14, for reference. These distributions are representative of fuel assemblies with the design burnup levels considered. These distributions are used for analysis only, and do not provide a criteria for fuel assembly acceptability for transport in the HI-STAR 100 System.

1.2.3.6 Radiological Payload Parameters

The principal radiological design criteria are the 10CFR71.47 and 10CFR71.51 radiation dose rate and release requirements for the HI-STAR 100 System. The radiation dose rate is directly affected by the gamma and neutron source terms of the SNF assembly.

The gamma and neutron sources are separate and are affected differently by enrichment, burnup, and cool time. It is recognized that, at a given burnup, the radiological source terms increase monotonically as the initial enrichment is reduced. The shielding design basis fuel assembly is, therefore, evaluated for different combinations of maximum burnup, minimum cooling time, and minimum enrichment. The shielding design basis intact fuel assembly thus bounds all other intact fuel assemblies.

The design basis dose rates can be met by a variety of burnup levels, cooling times, and minimum enrichments. Tables 1.2.21 through 1.2.36 include the burnup and cooling time values that meet the radiological dose rate requirements for all authorized contents to be transported in each MPC model. The allowable maximum burnup, minimum cooling time, and minimum enrichment limits were chosen strictly based on the dose rate requirements. All allowable burnup, cooling time, and minimum enrichment combinations result in calculated dose rates less than the regulatory dose rate limits.

Table 1.2.15 and Figures 1.2.13. 1.2.13A, and 1.2.14 provide the axial distribution for the radiological source term for PWR and BWR fuel assemblies, and for Trojan plant-specific fuel, based on the actual burnup distribution. The axial burnup distributions are representative of fuel assemblies with the design basis burnup levels considered. These distributions are used for analysis only, and do not provide criteria for fuel assembly acceptability for transport in the HI-STAR 100 System.

Thoria rods placed in Dresden Unit 1 Thoria Rod Canisters meeting the requirements of Table 1.2.21 and Dresden Unit 1 fuel assemblies with one Antimony-Beryllium neutron source have been qualified for transport. Up to one Dresden Unit 1 Thoria Rod Canister plus any combination of damaged fuel assemblies in damaged fuel containers and intact fuel, up to a total of 68 may be transported.

1.2.3.7 Criticality Payload Parameters

As discussed earlier, the MPC-68/68F and MPC-32 feature a basket without flux traps. In these fuel baskets, there is one panel of neutron absorber between adjacent fuel assemblies. The MPC-24/24E/24EF employs a construction wherein two neighboring fuel assemblies are separated by two panels of neutron absorber with a water gap between them (flux trap construction). The MPC-24 flux trap basket can accept a much higher enrichment fuel than a non-flux trap basket without taking credit for fuel assembly burnup in the criticality analysis. The maximum initial ²³⁵U enrichment for PWR and BWR fuel authorized for transport is specified by fuel array/class in Tables 1.2.10 and 1.2.11, respectively. Trojan plant fuel is limited to a lower maximum initial enrichment of 3.7 wt.% ²³⁵U compared to other fuel in its array/class, based on the specific analysis performed for the custom-designed Trojan MPCs containing only Trojan plant fuel.

The MPC-24 Boral ¹⁰B areal density is specified at a minimum loading of 0.0267 g/cm². The MPC-24E/EF, MPC-32, and MPC-68 Boral ¹⁰B areal density is specified at a minimum loading | of 0.0372 g/cm². The MPC-68F Boral ¹⁰B areal density is specified at a minimum loading of 0.01 g/cm².

For all MPCs, the ¹⁰B loading areal density used for analysis is conservatively established at 75% of the minimum ¹⁰B areal density to demonstrate that the reactivity under the most adverse accumulation of tolerances and biases is less than 0.95. The reduction in ¹⁰B areal density credit meets NUREG-1617 [1.0.5], which requires a 25% reduction in ¹⁰B areal density credit. A large body of sampling data accumulated by Holtec from thousands of manufactured Boral panels indicates the average ¹⁰B areal densities to be approximately 15% greater than the specified minimum.

Credit for burnup of the fuel, in accordance with the intent of the guidance in Interim Staff Guidance Document 8 (ISG-8) [1.2.13], is taken in the criticality analysis to allow the transportation of certain PWR fuel assemblies in MPC-32. Burnup credit is a required input to qualify PWR fuel for transportation in the MPC-32, considering the inleakage of moderator (i.e., unborated water) under accident conditions. This hypothetical event is non-credible given the double barrier design engineered into the HI-STAR 100 System with the fully welded MPC enclosure vessel (designed for 60 g's) surrounded by the sealed overpack, which is designed for deep submersion under water (greater than 650 feet submersion) without breach. The details of the burnup credit analyses are provided in Chapter 6, including detailed discussion of how the recommendations of ISG-8 were implemented. Exceptions to some of the recommendations in ISG-8 were necessary (e.g., partial credit for fission products) in order to develop burnup versus enrichment curves that can be practically implemented at the plants. These exceptions are described in Chapter 6.

1.2.3.8 Non-Fuel Hardware and Neutron Sources

BWR fuel is permitted to be stored with or without Zircaloy channels. Control blades and stainless steel channels are not authorized for transportation in the HI-STAR 100 System. Dresden Unit 1 (D-1) neutron sources are authorized for transportation a shown in Tables 1.2.23 and 1.2.24. The D-1 neutron sources are single, long rods containing Sb-Be source material that fits into a water rod location in a D-1 fuel assembly.

Except for Trojan plant fuel, no PWR non-fuel hardware or neutron sources are authorized for transportation in the HI-STAR 100 System. For Trojan plant fuel only, the following non-fuel hardware and neutron sources are permitted for transportation in specific quantities as shown in Tables 1.2.25 and 1.2.26:

- Rod Cluster Control Assemblies (RCCAs) with cladding made of Type 304 stainless steel and Ag-In-Cd neutron absorber material.
- Burnable Poison Rod Assemblies (BPRAs) with cladding made of Type 304 stainless steel and borosilicate glass tube neutron poison material.
- Thimble Plug Devices made of Type 304 stainless steel.
- Neutron source assemblies with cladding made of Type 304 stainless steel two (2) californium primary source assemblies and four (4) antimony-beryllium secondary source assemblies.

These devices are designed with thin rods of varying length and materials as discussed above, that fit into the fuel assembly guide tubes within the fuel rod lattice. The upper fittings for each device can vary to accommodate the handling tool (grapple) design. During reactor operation, the positions of the RCCAs are controlled by the operator using the control rod drive system, while the BPRAs, TPDs, and neutron sources stay fully inserted.

A complete list of the authorized non-fuel hardware and neutron sources, including appropriate limits on the characteristics of this material, is provided in Tables 1.2.23 through 1.2.36, as applicable.

1.2.3.9 Summary of Authorized Contents

The criticality safety index for the HI-STAR 100 Package is zero. A fuel assembly is acceptable | for transport in a HI-STAR 100 System if it fulfills the following criteria.

- a. It satisfies the physical parameter characteristics listed in Tables 1.2.10 or 1.2.11, as applicable..
- b. It satisfies the cooling time, decay heat, burnup, enrichment, and other limits specified in Tables 1.2.21 through 1.2.36, as applicable.

c. Deleted.

d. Deleted.

A damaged fuel assembly shall be transported in a damaged fuel container or other authorized damaged/failed fuel canister, and shall meet the characteristics specified in Tables 1.2.23 through 1.2.26 for transport in the MPC-68, MPC-68F, MPC-24E, or MPC-24EF. Fuel classified as fuel debris shall be placed in a damaged fuel container or other authorized damaged/failed fuel canister and shall meet the characteristics specified in Tables 1.2.26 for transport in the MPC-68F, MPC-24EF. The classified as fuel debris shall be placed in a damaged fuel container or other authorized damaged/failed fuel canister and shall meet the characteristics specified in Tables 1.2.24 or 1.2.26 for transport in the MPC-68F or MPC-24EF.

Stainless steel clad fuel assemblies shall meet the characteristics specified in Tables 1.2.22 through 1.2.33 for transport in the MPC-24, MPC-24E, MPC-24EF, or MPC-68.

MOX BWR fuel assemblies shall meet the requirements of Tables 1.2.23 or 1.2.24 for intact and damaged fuel/fuel debris.

Thoria rods placed in Dresden Unit 1 Thoria Rod Canisters meeting the requirements of Table 1.2.21 and Dresden Unit 1 fuel assemblies with one Antimony-Beryllium neutron source have been qualified for transport. Up to one Dresden Unit 1 Thoria Rod Canister plus any combination of damaged fuel assemblies in damaged fuel containers and intact fuel, up to a total of 68 may be transported.

Dresden Unit 1 fuel assemblies with one Antimony-Beryllium neutron source are authorized for loading in the MPC-68 or MPC-68F.

Table 1.2.2 summarizes the key system data for the HI-STAR 100 System. Table 1.2.3 summarizes the key parameters and limits for the HI-STAR 100 MPCs. Tables 1.2.10, 1.2.11, and 1.2.21 through 1.2.37 and other tables referenced from these tables provide the limiting conditions for all material to be transported in the HI-STAR 100 System.

TABLE INTENTIONALLY DELETED

ł

PARAMETER		VALUE (Nominal)
Types of MPCs in	6	4 for PWR
this SAR	0	2 for BWR
MPC capacity	MPC-24	Up to 24 intact ZR or stainless steel clad PWR fuel assemblies
	MPC-24E	Up to 24 intact ZR or stainless steel clad PWR fuel assemblies. Up to four (4) Trojan plant fuel assemblies classified as damaged fuel, each in a Trojan Failed Fuel Can or a Holtec damaged fuel container, and the complement intact fuel assemblies.
	MPC-24EF	Up to 24 intact ZR or stainless steel clad PWR fuel assemblies. Up to four (4) Trojan plant fuel assemblies classified as damaged fuel or fuel debris, each in a Trojan Failed Fuel Can or a Holtec damaged fuel container; or other Trojan fuel debris stored in Trojan Process Cans either placed directly into a Trojan Failed Fuel Can or placed inside Trojan Process Can Capsules and then in Trojan Failed Fuel Cans; and the complement intact fuel assemblies.
	MPC-32	Up to 32 intact ZRclad PWR fuel assemblies.
	MPC-68	Up to 68 intact ZR or stainless steel clad BWR fuel assemblies or damaged ZR clad fuel assemblies* in damaged fuel containers within an MPC-68
	MPC-68F	Up to 4 damaged fuel containers with ZR clad BWR fuel debris* and the complement intact or damaged* ZR clad BWR fuel assemblies within an MPC-68F.
		*Only damaged fuel and fuel debris from Dresden Unit 1 or Humboldt Bay is authorized for transportation in the MPC-68 and MPC- 68F.

SUMMARY OF KEY SYSTEM DATA FOR HI-STAR 100

Table 1.2.3							
KEY PARAMETERS FOR HI-STAR 100 MULTI-PURPOSE CANISTERS							

PARAMETER	PWR	BWR
Unloaded MPC weight (lb)	See Table 2.2.1	See Table 2.2.1
Minimum neutron absorber ¹⁰ B loading (g/cm ²)	0.0267 (MPC-24) 0.0372 (MPC-24E/EF) 0.0372 (MPC-32)	0.0372 (MPC-68) 0.01 (MPC-68F)
Pre-disposal service life (years)	40	40
Design temperature, max./min. (°F)	725° [†] /-40° ^{††}	725° [†] /-40° ^{††}
Design Internal pressure (psig)		
Normal Conditions Off-normal Conditions Accident Conditions	100 100 200	100 100 200
Total heat load, max. (kW)	20.0	18.5
Maximum permissible peak fuel cladding temperature (°F)	752° (normal conditions) 1058° (accident conditions)	752° (normal conditions) 1058° (accident conditions)
MPC internal environment Helium filled (psig)	≥ 0 and ≤ 44.8 psig ^{ttt} at a reference temperature of 70°F	\geq 0 and \leq 44.8 psig ^{tt†} at a reference temperature of 70°F
MPC external environment/overpack internal environment Helium filled initial pressure (psig, at STP)	\geq 10 and \leq 14	\geq 10 and \leq 14
Maximum permissible reactivity including all uncertainty and biases	<0.95	<0.95
End closure(s)	Welded	Welded
Fuel handling	Opening compatible with standard grapples	Opening compatible with standard grapples
Heat dissipation	Passive	Passive

^{*} Maximum normal condition design temperature for the MPC fuel basket. A complete listing of design temperatures for all components is provided in Table 2.1.2

^{††} Temperature based on minimum ambient temperature (10CFR71.71(c)(2)) and no fuel decay heat load.

This value represents the nominal backfill value used in the thermal analysis, plus 2 psig operating tolerance. Based on the MPC pressure results in Table 3.4.15 and the pressure limits specified in Table 2.1.1, there is sufficient analysis margin to accommodate this operating tolerance.

Tables 1.2.4 through 1.2.6

INTENTIONALLY DELETED

HI-STAR 100 LOADING OPERATIONS DESCRIPTION

Site-specific handling and operating procedures will be prepared, reviewed, and approved by each owner/user. 1 Overpack and MPC lowered into the fuel pool without closure plate and MPC lid Fuel assemblies transferred to the MPC fuel basket 2 3 MPC lid lowered onto the MPC 4 Overpack/MPC assembly moved to the decon pit and MPC lid welded in place, examined, pressure tested, and leak tested 5 MPC dewatered, dried, backfilled with helium, and the vent/drain port cover plates and closure ring welded 6 Overpack drained and external surfaces decontaminated 7 Overpack seals and closure plate installed and bolts pre-tensioned 8 Overpack cavity dried, backfilled with helium, and helium leak tested 9 HI-STAR 100 System transferred to transport bay 10 HI-STAR 100 placed onto transport saddles, tied down, impact limiters and personnel barrier installed, and package surveyed for release for transport.

PWR FUEL ASSEMBLIES EVALUATED TO DETERMINE DESIGN BASIS SNF

Assembly Class	Array Type
B&W 15x15	All
B&W 17x17	All
CE 14x14	All
CE 16x16	All except System 80 TM
WE 14x14	A11
WE 15x15	All
WE 17x17	All
St. Lucie	All
Ft. Calhoun	All
Haddam Neck (Stainless Steel Clad)	All
San Onofre 1 (Stainless Steel Clad, except MOX)	All
Indian Point 1	All

BWR FUEL ASSEMBLIES EVALUATED TO DETERMINE DESIGN BASIS SNF

Assembly Class	Array Type					
GE BWR/2-3	All 7x7	All 8x8	A11 9x9	All 10x10		
GE BWR/4-6	All 7x7	All 8x8	All 9x9	All 10x10		
Humboldt Bay	All 6x6	All 7x7 (Zircaloy Clad)				
Dresden-1	All 6x6	All 8x8				
LaCrosse (Stainless Steel Clad)	All					

Fuel Assembly Array/Class	14x14A	14x14B	14x14C	14x14D	14x14E
Clad Material (Note 2)	ZR	ZR	ZR	SS	SS
Design Initial U (kg/assy.) (Note 3)	<u>≤</u> 407	<u>≤</u> 407	≤425	<u>≤</u> 400	≤206
Initial Enrichment	≤ 4.6 (24)	≤4.6 (24)	≤4.6 (24)	≤4.0 (24)	< 5.0
(MFC-24, 24E, and 24EF) (wt % ²³⁵ U)	≤ 5.0 (24E/24EF)	≤ 5.0 (24E/24EF)	≤ 5.0 (24E/24EF)	≤ 5.0 (24E/24EF)	≤ 3.0
Initial Enrichment (MPC-32) (wt % ²³⁵ U) (Note 5)	N/A	N/A	N/A	N/A	N/A
No. of Fuel Rod Locations	179	179	176	180	173
Fuel Clad O.D. (in.)	<u>≥ 0.400</u>	<u>≥ 0.417</u>	<u>≥</u> 0.440	<u>≥</u> 0.422	≥ 0.3415
Fuel Clad I.D. (in.)	<u>≤</u> 0.3514	<u>≤</u> 0.3734	≤ 0.3880	<u>≤</u> 0.3890	≤ 0.3175
Fuel Pellet Dia. (in.)	<u>≤</u> 0.3444	<u>≤</u> 0.3659	<u>≤ 0.3805</u>	<u>≤</u> 0.3835	≤ 0.3130
Fuel Rod Pitch (in.)	<u>≤</u> 0.556	<u>≤</u> 0.556	<u>≤</u> 0.580	<u>≤</u> 0.556	Note 6
Active Fuel Length (in.)	<u>≤ 150</u>	<u>≤</u> 150	<u>≤</u> 150	<u>≤</u> 144	≤ 102
No. of Guide and/or Instrument Tubes	17	17	5 (Note 4)	16	0
Guide/Instrument Tube Thickness (in.)	≥ 0.017	<u>≥</u> 0.017	≥ 0.038	≥ 0.0145	N/A

Table 1.2.10 PWR FUEL ASSEMBLY CHARACTERISTICS (Note 1)

-.

Fuel Assembly Array/Class	15x15A	15x15B	15x15C	15x15D	15x15E	15x15F
Clad Material (Note 2)	ZR	ZR	ZR	ZR	ZR	ZR
Design Initial U (kg/assy.) (Note 3)	<u>≤</u> 464	≤464	<u>≤</u> 464	≤ 475	≤475	≤475
Initial Enrichment	≤4.1 (24)	≤4.1 (24)	≤ 4.1 (24)	<u>≤</u> 4.1 (24)	≤4.1 (24)	≤4.1 (24)
(MPC-24, 24E, and 24EF (wt % ²³⁵ U)	≤4.5 (24E/24EF)	≤4.5 (24E/24EF)	≤ 4.5 (24E/24EF)	≤4.5 (24E/24EF)	≤4.5 (24E/24EF)	≤4.5 (24E/24EF)
Initial Enrichment (MPC-32) (wt % ²³⁵ U) (Note 5)	N/A	N/A	N/A	≤5.0	<u>≤</u> 5.0	≤ 5.0
No. of Fuel Rod Locations	204	204	204	208	208	208
Fuel Clad O.D. (in.)	≥0.418	≥ 0.420	≥ 0.417	<u>≥</u> 0.430	≥ 0.428	≥ 0.428
Fuel Clad I.D. (in.)	≤ 0.3660	≤ 0.3736	≤ 0.3640	≤ 0.3800	≤ 0.3790	≤ 0.3820
Fuel Pellet Dia. (in.)	≤0.3580	≤ 0.3671	≤0.3570	≤ 0.3735	≤ 0.3707	≤0.3742
Fuel Rod Pitch (in.)	≤0.550	<u>≤</u> 0.563	≤ 0.563	≤ 0.568	<u>≤</u> 0.568	≤ 0.568
Active Fuel Length (in.)	<u>≤</u> 150	≤150	≤ 150	≤ 150	<u>≤</u> 150	≤ 150
No. of Guide and/or Instrument Tubes	21	21	21	17	17	17
Guide/Instrument Tube Thickness (in.)	≥ 0.0165	≥ 0.015	≥0.0165	≥ 0.0150	≥ 0.0140	≥ 0.0140

Table 1.2.10 (continued) PWR FUEL ASSEMBLY CHARACTERISTICS (Note 1)

Fuel Assembly Array/Class	15x15G	15x15H	16x16A	17x17A	17x17B	17x17C
Clad Material (Note 2)	SS	ZR	ZR	ZR	ZR	ZR
Design Initial U (kg/assy.) (Note 3)	≤ 420	<u>≤</u> 475	<u>≤</u> 443	<u>≤</u> 467	<u>≤</u> 467	<u>≤</u> 474
Initial Enrichment (MPC-24, 24E, and	≤ 4.0 (24)	≤ 3.8 (24)	<u>≤</u> 4.6 (24)	<u>≤</u> 4.0 (24)	≤4.0 (24)	≤ 4.0 (24)
24EF) (wt % ²³⁵ U)	≤ 4.5 (24E/24EF)	≤ 4.2 (24E/24EF)	≤ 5.0 (24E/24EF)	≤4.4 (24E/24EF)	≤ 4.4 (24E/24EF) (Note 7)	≤4.4 (24E/24EF)
Initial Enrichment (MPC-32) (wt % ²³⁵ U) (Note 5)	N/A	<u>≤</u> 5.0	N/A	≤ 5.0	≤5.0	≤ 5.0
No. of Fuel Rod Locations	204	208	236	264	264	264
Fuel Clad O.D. (in.)	≥ 0.422	<u>≥</u> 0.414	≥ 0.382	≥ 0.360	<u>≥ 0.372</u>	<u>≥</u> 0.377
Fuel Clad I.D. (in.)	≤ 0.3890	<u>≤</u> 0.3700	≤ 0.3320	≤ 0.3150	≤ 0.3310	<u>≤</u> 0.3330
Fuel Pellet Dia. (in.)	≤ 0.3825	<u>≤ 0.3622</u>	≤ 0.3255	≤ 0.3088	<u>≤</u> 0.3232	≤ 0.3252
Fuel Rod Pitch (in.)	≥ 0.563	≥ 0.568	≥ 0.50 6	<u>≥</u> 0.496	<u>≥</u> 0.496	<u>≥</u> 0.502
Active Fuel Length (in.)	≤ 144	≤150	≤ 150	≤150	≤150	≤ 150
No. of Guide and/or Instrument Tubes	21	17	5 (Note 4)	25	25	25
Guide/Instrument Tube Thickness (in.)	≥ 0.0145	≥ 0.0140	≥ 0.0400	≥ 0.016	≥ 0.014	≥ 0.020

Table 1.2.10 (continued) PWR FUEL ASSEMBLY CHARACTERISTICS (Note 1)

Table 1.2.10 (continued) PWR FUEL ASSEMBLY CHARACTERISTICS

Notes:

- 1. All dimensions are design nominal values. Maximum and minimum dimensions are specified to bound variations in design nominal values among fuel assemblies within a given array/class.
- 2. ZR designates any zirconium-based fuel cladding material authorized for use in a commercial power reactor.
- 3. Design initial uranium weight is the nominal uranium weight specified for each assembly by the fuel manufacturer or reactor user. For each PWR fuel assembly, the total uranium weight limit specified in this table may be increased up to 2.0 percent for comparison with users' fuel records to account for manufacturer's tolerances.
- 4. Each guide tube replaces four fuel rods.
- 5. Minimum assembly average burnup is required per Table 1.2.34.
- 6. This fuel assembly array/class includes only the Indian Point Unit 1 fuel assembly. This fuel assembly has two pitches in different sectors of the assembly. These pitches are 0.441 inches and 0.453 inches.
- 7. Trojan plant-specific fuel is governed by the limits specified for array/class 17x17B and will be transported in the custom-designed Trojan MPC-24E/EF canisters. The Trojan MPC-24E/EF design is authorized to transport only Trojan plant fuel with a maximum initial enrichment of 3.7 wt.%²³⁵U.

Fuel Assembly Array/Class	6x6A	6x6B	6x6C	7x7A	7x7B	8x8A]
Clad Material (Note 2)	ZR	ZR	ZR	ZR	ZR	ZR]
Design Initial U (kg/assy.) (Note 3)	≤110	≤110	≤110	≤ 100	≤ 195	≤ 120	
Maximum Planar- Average Initial Enrichment (wt % ²³⁵ U)	≤2.7	\leq 2.7 for the UO ₂ rods. See Note 4 for MOX rods.	≤2.7	≤2.7	≤4.2	≤2.7	
Initial Maximum Rod Enrichment (wt % ²³⁵ U)	<u>≤</u> 4.0	≤ 4.0	≤4.0	≤ 5.5	≤ 5.0	≤4.0]
No. of Fuel Rod Locations	35 or 36	35 or 36 (up to 9 MOX rods)	36	49	49	63 or 64	
Fuel Clad O.D. (in.)	≥ 0.5550	≥ 0.5625	≥ 0.5630	≥ 0.486 0	≥ 0.5630	≥ 0.4120	1
Fuel Clad I.D. (in.)	≤ 0.5105	≤ 0.4945	<u>≤ 0.4990</u>	≤ 0.4204	<u>≤ 0.4990</u>	<u>≤</u> 0.3620	1
Fuel Pellet Dia. (in.)	≤ 0.4980	≤ 0.4820	≤ 0.4880	≤ 0.4110	≤ 0.4910	≤ 0.3580	1
Fuel Rod Pitch (in.)	≤ 0.710	≤0.710	<u>≤</u> 0.740	≤ 0.631	≤0.738	≤ 0.523	1
Active Fuel Length (in.)	<u>≤ 120</u>	≤ 120	≤ 77.5	<u>≤</u> 80	<u>≤ 150</u>	<u>≤ 120</u>	1
No. of Water Rods (Note 11)	1 or 0	1 or 0	0	0	0	1 or 0	
Water Rod Thickness (in.)	> 0	> 0	N/A	N/A	N/A	≥0]
Channel Thickness (in.)	≤ 0.060	≤ 0.060	<u>≤ 0.060</u>	≤ 0.060	≤ 0.120	≤ 0.100	

 Table 1.2.11

 BWR FUEL ASSEMBLY CHARACTERISTICS (Note 1)

Fuel Assembly Array/Class	8x8B	8x8C	8x8D	8x8E	8x8F	9x9A
Clad Material (Note 2)	ZR	ZR	ZR	ZR	ZR	ZR
Design Initial U (kg/assy) (Note 3)	<u>≤</u> 185	≤185	≤ 185	<u>≤</u> 185	≤185	≤177
Maximum Planar-Average Initial Enrichment (wt % ²³⁵ U)	<u>≤</u> 4.2	<u>≤</u> 4.2	≤ 4.2	<u>≤</u> 4.2	<u>≤</u> 4.0	≤4.2
Initial Maximum Rod Enrichment (wt % ²³⁵ U)	≤ 5.0	<u>≤</u> 5.0	≤ 5.0	≤ 5.0	≤ 5.0	≤ 5.0
No. of Fuel Rod Locations	63 or 64	62	60 or 61	59	64	74/66 (Note 5)
Fuel Clad O.D. (in.)	≥ 0.4840	≥ 0.4830	≥ 0.4830	<u>≥</u> 0.4930	≥ 0.4576	≥ 0.4400
Fuel Clad I.D. (in.)	<u>≤</u> 0.4295	≤ 0.4250	≤ 0.4230	<u>≤</u> 0.4250	<u>≤</u> 0.3996	≤ 0.3840
Fuel Pellet Dia. (in.)	≤ 0.4195	<u>≤</u> 0.4160	≤ 0.4140	<u>≤</u> 0.4160	≤ 0.3913	≤ 0.3760
Fuel Rod Pitch (in.)	≤ 0.642	≤ 0.641	≤ 0.640	≤ 0.640	≤ 0.609	≤ 0.566
Design Active Fuel Length (in.)	≤ 150	≤ 150	≤ 150	≤150	≤ 150	≤150
No. of Water Rods (Note 11)	1 or 0	2	1 - 4 (Note 7)	5	N/A (Note 12)	2
Water Rod Thickness (in.)	≥ 0.034	> 0.00	> 0.00	≥ 0.034	≥ 0.0315	> 0.00
Channel Thickness (in.)	≤ 0.120	≤ 0.120	<u>≤ 0.120</u>	≤ 0.100	≤ 0.055	≤ 0.120

Table 1.2.11 (continued) BWR FUEL ASSEMBLY CHARACTERISTICS (Note 1)
Fuel Assembly Array/Class	9x9 B	919 C	919 D	9x9 E (Note 13)	919 F (Note 13)	9x9 G
Clad Material (Note 2)	ZR	ZR	ZR	ZR	ZR	ZR
Design Initial U (kg/assy.) (Note 3)	≤177	≤ 177	≤ 177	≤ 177	≤177	≤177
Maximum Planar- Average Initial Enrichment (wt % ²³⁵ U)	<u>≤</u> 4.2	<u>≤4.2</u>	≤4.2	<u>≤</u> 4.0	<u>≤</u> 4.0	≤4.2
Initial Maximum Rod Enrichment (wt % ²³⁵ U)	<u>≤</u> 5.0	≤ 5.0	<u>≤</u> 5.0	≤ 5.0	≤ 5.0	≤ 5.0
No. of Fuel Rod Locations	72	80	79	76	76	72
Fuel Clad O.D. (in.)	≥ 0.4330	≥0.4230	<u>≥</u> 0.4240	≥ 0.4170	<u>≥ 0.4430</u>	≥ 0.4240
Fuel Clad I.D. (in.)	≤ 0.3810	≤0.3640	<u>≤</u> 0.3640	<u>≤</u> 0.3640	≤ 0.3860	<u><</u> 0.3640
Fuel Pellet Dia. (in.)	≤ 0.3740	≤ 0.3565	≤ 0.3565	≤ 0.3530	<u>≤</u> 0.3745	≤ 0.3565
Fuel Rod Pitch (in.)	≤ 0.572	≤ 0.572	≤0.572	≤ 0.572	≤ 0.572	≤ 0.572
Design Active Fuel Length (in.)	≤ 150	≤ 150	≤ 150	≤ 150	≤ 150	≤150
No. of Water Rods (Note 11)	1 (Note 6)	1	2	5	5	1 (Note 6)
Water Rod Thickness (in.)	> 0.00	≥ 0.020	≥ 0.0300	≥ 0.0120	≥ 0.0120	≥ 0.0320
Channel Thickness (in.)	≤ 0.120	≤0.100	≤0.100	≤ 0.120	≤ 0.120	≤ 0.120

Table 1.2.11 (continued) BWR FUEL ASSEMBLY CHARACTERISTICS (Note 1)

---- -

Fuel Assembly Array/Class	10x10 A	10x10 B	10x10 C	10x10 D	10x10 E
Clad Material (Note 2)	ZR	ZR	ZR	SS	SS
Design Initial U (kg/assy.) (Note 3)	≤ 186	<u>≤</u> 186	<u>≤</u> 186	≤125	≤125
Maximum Planar- Average Initial Enrichment (wt % ²³⁵ U)	≤4.2	<u><</u> 4.2	<u>≤</u> 4.2	≤ 4.0	≤4.0
Initial Maximum Rod Enrichment (wt % ²³⁵ U)	≤ 5.0	≤ 5.0	≤ 5.0	≤ 5.0	≤ 5.0
No. of Fuel Rod Locations	92/78 (Note 8)	91/83 (Note 9)	96	100	96
Fuel Clad O.D. (in.)	<u>≥ 0.4040</u>	≥ 0.3957	≥ 0.3780	<u>≥</u> 0.3960	<u>≥</u> 0.3940
Fuel Clad I.D. (in.)	≤ 0.3520	<u>≤ 0.3480</u>	≤ 0.3294	<u>≤</u> 0.3560	≤ 0.3500
Fuel Pellet Dia. (in.)	≤ 0.3455	≤ 0.3420	≤ 0.3224	≤ 0.3500	≤ 0.3430
Fuel Rod Pitch (in.)	<u>≤</u> 0.510	<u>≤</u> 0.510	≤ 0.488	<u>≤</u> 0.565	≤ 0.557
Design Active Fuel Length (in.)	≤ 150	≤ 150	≤ 150	≤ 83	≤ 83
No. of Water Rods (Note 11)	2	1 (Note 6)	5 (Note 10)	0	4
Water Rod Thickness (in.)	≥ 0.030	> 0.00	≥ 0.031	N/A	≥ 0.022
Channel Thickness (in.)	≤ 0.120	≤ 0.120	≤ 0.055	≤ 0.080	≤ 0.080

Table 1.2.11 (continued) BWR FUEL ASSEMBLY CHARACTERISTICS (Note 1)

I

1

Table 1.2.11 (continued) BWR FUEL ASSEMBLY CHARACTERISTICS

NOTES:

- 1. All dimensions are design nominal values. Maximum and minimum dimensions are specified to bound variations in design nominal values among fuel assemblies within a given array/class.
- 2. ZR designates any zirconium-based fuel cladding material authorized for use in a commercial power reactor.
- 3. Design initial uranium weight is the nominal uranium weight specified for each assembly by the fuel manufacturer or reactor user. For each BWR fuel assembly, the total uranium weight limit specified in this table may be increased up to 1.5 percent for comparison with users' fuel records to account for manufacturer tolerances.
- 4. ≤ 0.635 wt. % ²³⁵U and ≤ 1.578 wt. % total fissile plutonium (²³⁹Pu and ²⁴¹Pu), (wt. % of total fuel weight, i.e., UO₂ plus PuO₂).
- 5. This assembly class contains 74 total rods; 66 full length rods and 8 partial length rods.
- 6. Square, replacing nine fuel rods.
- 7. Variable.
- 8. This assembly contains 92 total fuel rods; 78 full length rods and 14 partial length rods.
- 9. This assembly class contains 91 total fuel rods; 83 full length rods and 8 partial length rods.
- 10. One diamond-shaped water rod replacing the four center fuel rods and four rectangular water rods dividing the assembly into four quadrants.
- 11. These rods may also be sealed at both ends and contain ZR material in lieu of water.
- 12. This assembly is known as "QUAD+." It has four rectangular water cross segments dividing the assembly into four quadrants.
- 13. For the SPC 9x9-5 fuel assembly, each fuel rod must meet either the 9x9E or the 9x9F set of limits or clad O.D., clad I.D., and pellet diameter.

DESIGN BASIS FUEL ASSEMBLY FOR EACH DESIGN CRITERION

Criterion	MPC-68/68F	MPC-24/24E/24EF/32
Reactivity		B&W 15x15
-	SPC 9x9-5	(Array/Class 15x15F)
	(Array/Class 9x9E/F)	
Shielding (Source	GE 7x7	B&W 15x15
Term)		
Fuel Assembly		
Effective Planar	GE 11 9x9	<u>W</u> 17x17 OFA
Thermal Conductivity		
Fuel Basket Effective		
Axial Thermal	GE 7x7	<u>W</u> 14x14 OFA
Conductivity		
MPC Density and heat	CE 7.7	W 14-14 OFA
Capacity		

Tables 1.2.13 and 1.2.14

INTENTIONALY DELETED

GENERIC FUEL DISTRIBUTION [†]						
Interval	Axial I Bottom (% of Act	Distance From of Active Fuel tive Fuel Length)	PWR Fuel Ne Distribu	ormalized ation	BWR Fuel Normalized Distribution	
1	0%	% to 4-1/6%	0.548	5	0.2200	
2	4-1/	6% to 8-1/3%	0.847	7	0.7600	
3	8-1/3	3% to 16-2/3%	1.077	0	1.0350	
4	16-2/	3% to 33-1/3%	1.105	0	1.1675	
5	33-	1/3% to 50%	1.098	0	1.1950	
6	50%	% to 66-2/3%	1.079	0	1.1625	
7	66-2/	3% to 83-1/3%	1.050	1	1.0725	
8	83-1/	3% to 91-2/3%	0.960	4	0.8650	
9	91-2/	'3% to 95-5/6%	0.733	8	0.6200	
10	95-5	5/6% to 100%	0.467	0	0.2200	
	<u> </u>	TROJAN PLAN	NT FUEL DIST	RIBUTION	it	
Inte	rval	Axial Distance F Active (% of Active I	rom Bottom of Fuel Juel Length)	Norm	alized Distribution	
1	1	0% to	5%		0.59	
	2	5% to	10%		0.89	
	3	10% to	15%		1.03	
	4	15% to	20%		1.07	
:	5	20% to	25%		1.09	
	6	25% to	45%		1.10	
7 45% to		70%	· · · · · · · · · · · · · · · · · · ·	1.09		
8 70% to		75%		1.07		
9 75% to 3		80%		1.05		
10 80% to 3		85%		1.02		
11 85% to 9		90%		0.96		
1	2	90% to	95 %		0.82	
13 95% to 1		100%		0.56		

NORMALIZED DISTRIBUTION BASED ON BURNUP PROFILE

[†] References [1.2.8] and [1.2.9]

tt Reference [1.2.12]

SUGGESTED PWR UPPER AND LOWER FUEL SPACER LENGTHS (Note 1)

Fuel Assembly Type	Assembly Length w/o NFH [†] (in.)	Location of Active Fuel from Bottom (in.)	Max. Active Fuel Length (in.)	Upper Fuel Spacer Length (in.)	Lower Fuel Spacer Length (in.)
CE 14x14	157	4.1	137	9.5	10
CE 16x16	176.8	4.7	150	0	0
BW 15x15	165.7	8.4	141.8	6.7	4.1
W 17x17 OFA	159.8	3.7	144	8.2	8.5
W 17x17S	159.8	3.7	144	8.2	8.5
W 17x17V5H	160.1	3.7	144	7.9	8.5
W 15x15	159.8	3.7	144	8.2	8.5
W 14x14S	159.8	3.7	145.2	9.2	7.5
W 14x14 OFA	159.8	3.7	144	8.2	8.5
Ft. Calhoun	146	6.6	128	10.25	20.25
St. Lucie 2	158.2	5.2	136.7	10.25	8.05
B&W 15x15 SS	137.1	3.873	120.5	19.25	19.25
W 15x15 SS	137.1	3.7	122	19.25	19.25
W 14x14 SS	137.1	3.7	120	19.25	19.25
Indian Point 1	137.2	17.705	101.5	18.75	20.0

Notes: 1. These fuel spacer lengths are not applicable to Trojan plant fuel. Trojan plant fuel spacer lengths are determined uniquely for the custom-designed Trojan MPC-24E/EF, as necessary, based on the presence of non-fuel hardware. They are sized to maintain the active fuel within the envelope of the neutron absorber affixed to the cell walls and allow for an approximate 2-inch gap between the fuel and the MPC lid. See Chapter 6 for discussion of potential misalignments between the active fuel and the neutron absorber.

^{*} NFH is an abbreviation for non-fuel hardware, including control components. Fuel assemblies with control components may require shorter fuel spacers.

Fuel Assembly Type	Assembly Length (in.)	Location of Active Fuel from Bottom (in.)	Max. Active Fuel Length (in.)	Upper Fuel Spacer Length (in.)	Lower Fuel Spacer Length (in.)
GE/2-3	171.2	7.3	150	4.8	0
GE/4-6	176.2	7.3	150	0	0
Dresden 1	134.4	11.2	110	18	28.0
Humboldt Bay	95	8	79	40.5	40.5
Dresden 1 Damaged Fuel or Fuel Debris	142.1 [†]	11.2	110	17	16.9
Humboldt Bay Damaged Fuel or Fuel Debris	105.5 [†]	8	79	35.25	35.25
LaCrosse	102.5	10.5	83	37	37.5

SUGGESTED BWR UPPER AND LOWER FUEL SPACER LENGTHS (Note 1)

Notes: 1.Each user shall specify the fuel spacer lengths based on their fuel length and allowing an approximate 2-inch gap between the fuel and the MPC lid. See Chapter 6 for discussion of potential misalignments between the active fuel and the neutron absorber.

[†] Fuel length includes the damaged fuel container.

Aspect of Post-Accident Performance	Results with Demonstrated Integrity of MPC	Results with Postulated Gross Failure of MPC Enclosure Vessel
Containment Boundary Integrity	The MPC enclosure vessel is leak tested to 5.0×10^{-6} atm cm ³ /s (helium). The overpack containment boundary is standard air leak tested to 4.3×10^{-6} atm cm ³ /s (helium). Both boundaries are shown to withstand all hypothetical accident conditions. Therefore, there will be no detectable release of radioactive materials.	The overpack containment boundary is leak tested to 4.3×10^{-6} atm cm ³ /s (helium). The overpack containment boundary is shown to withstand all hypothetical accident conditions. Therefore, the overpack containment boundary meets the accident condition leakage rates.
Maintenance of Subcritical Margins (Maximum k _{eff})	The MPC enclosure vessel is seal welded and there is no breach of the MPC. The bolted closure overpack containment boundary has been shown to prevent water immersion. Therefore, the maximum reactivity of the fuel in a dry MPC is less than 0.5.	The bolted closure overpack containment boundary has been shown to prevent water immersion. Therefore, the maximum reactivity of the fuel in a dry MPC is less than 0.5. Assuming the MPC is fully flooded with water, the reactivity is shown to be below the regulatory requirement of 0.95 including uncertainties and bias.
Adequate Shielding	The MPC enclosure vessel boundary has no effect on the dose rates of the HI-STAR 100 System.	Failure of the MPC enclosure vessel to maintain a release boundary has no effect on the dose rates of the HI-STAR 100 System.
Adequate Heat Rejection (Peak Fuel Cladding Temperature)	The MPC enclosure vessel maintains the helium and the peak fuel cladding temperature is demonstrated to remain below 800°F in the post-fire hypothetical accident condition.	Assuming the MPC internal helium fill pressure is released into the overpack containment, the pressure within the small annulus would rise to equalize with the MPC internal pressure. There would be a corresponding slight pressure decrease in the MPC enclosure vessel. The comparatively small volume of the annulus and pressure differential results in the slight pressure change. This will have a negligibly small effect on the peak fuel cladding temperature. The overpack containment boundary is demonstrated to withstand all hypothetical accident conditions. Therefore, there is no credible mechanism for the release of the helium.

Tables 1.2.19 and 1.2.20

INTENTIONALLY DELETED

DESIGN CHARACTERISTICS FOR THORIA RODS IN D-1 THORIA ROD CANISTERS

PARAMETER	MPC-68 or MPC-68F
Cladding Type	ZR
Composition	98.2 wt.% ThO ₂ , 1.8 wt.% UO ₂ with an enrichment of 93.5 wt. % 235 U
Number of Rods Per Thoria Canister	<u>≤ 18</u>
Decay Heat Per Thoria Canister	\leq 115 watts
Post-Irradiation Fuel Cooling Time and Average Burnup Per Thoria Canister	Cooling time ≥ 18 years and average burnup $\leq 16,000$ MWD/MTIHM
Initial Heavy Metal Weight	≤ 27 kg/canister
Fuel Cladding O.D.	\geq 0.412 inches
Fuel Cladding I.D.	\leq 0.362 inches
Fuel Pellet O.D.	\leq 0.358 inches
Active Fuel Length	\leq 111 inches
Canister Weight	\leq 550 lbs., including Thoria Rods
Canister Material	Type 304 SS

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-24

PARAMETER	VALUE		
Fuel Type	Uranium oxide, PWR intact fuel assemblies meeting the limits in Table 1.2.10 for the applicable array/class		
Cladding Type	ZR or Stainless Steel (SS) as specified in Table 1.2.10 for the applicable array/class		
Maximum Initial Enrichment	As specified in Table 1.2.10 for the applicable array/class		
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly	ZR clad: As specified in Table 1.2.28 or Table 1.2.29, as applicable		
	SS clad: As specified in Table 1.2.30		
Decay Heat Per Assembly	ZR clad: \leq 833 Watts		
	SS clad: \leq 488 Watts		
Fuel Assembly Length	\leq 176.8 in. (nominal design)		
Fuel Assembly Width	\leq 8.54 in. (nominal design)		
Fuel Assembly Weight	≤ 1,680 lbs		
Other Limitations	 Quantity is limited to up to 24 PWR intact fuel assemblies. 		
	 Non-fuel hardware and neutron sources not permitted. 		
	 Damaged fuel assemblies and fuel debris not permitted. 		
	 Trojan plant fuel not permitted. 		

PARAMETER		VALUE (Note 1)	
Fuel Type(s)	Uranium oxide, BWR intact fuel assemblies meeting the limits in Table 1.2.11 for the applicable array/class, with or without Zircaloy channels	Uranium oxide, BWR damaged fuel assemblies meeting the limits in Table 1.2.11 for array/class 6x6A, 6x6C, 7x7A, or 8x8A, with or without Zircaloy channels, placed in Damaged Fuel Containers(DFCs)	Mixed Oxide (MOX) BWR intact fuel assemblies meeting the limits in Table 1.2.11 for array/class 6x6B, with or without Zircaloy channels	Mixed Oxide (MOX) BWR damaged fuel assemblies meeting the limits in Table 1.2.11 for array/class 6x6B, with or without Zircaloy channels, placed in Damaged Fuel Containers (DFCs)
Cladding Type	ZR or Stainless Steel (SS) as specified in Table 1.2.11 for the applicable array/class	ZR	ZR	ZR
Maximum Initial Planar-Average and Rod Enrichment	As specified in Table 1.2.11 for the applicable array/class	As specified in Table 1.2.11 for the applicable array/class	As specified in Table 1.2.11 for array/class 6x6B	As specified in Table 1.2.11 for array/class 6x6B
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly	ZR clad: As specified in Table 1.2.31 except as provided in Notes 2 and 3 SS clad: Note 4	Cooling time ≥ 18 years, average burnup $\leq 30,000$ MWD/MTU, and minimum initial enrichment ≥ 1.8 wt. % ²³⁵ U.	Cooling time \geq 18 years, average burnup \leq 30,000 MWD/MTIH M, and minimum initial enrichment \geq 1.8 wt. % ²³⁵ U.	Cooling time ≥ 18 years, average burnup $\leq 30,000$ MWD/MTIHM, and minimum initial enrichment ≥ 1.8 wt. % ²³⁵ U.
Decay Heat Per Assembly	ZR clad: ≤ 272 Watts (Note 5) SS clad: ≤ 83 Watts	≤ 115 Watts	\leq 115 Watts	≤ 115 Watts

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-68

Table 1.2.23 (cont'd)

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-68

PARAMETER	VALUE (Note 1)				
Fuel Assembly Length	\leq 176.2 in. (nominal design)	\leq 135.0 in. (nominal design)	\leq 135.0 in. (nominal design)	\leq 135.0 in. (nominal design)	
Fuel Assembly Width	\leq 5.85 in. (nominal design)	\leq 4.70 in. (nominal design)	\leq 4.70 in. (nominal design)	\leq 4.70 in. (nominal design)	
Fuel Assembly Weight	<pre> < 700 lbs (including channels) </pre>	\leq 550 lbs, (including channels and DFC)	≤ 400 lbs, (including channels)	<u>< 550 lbs</u> , (including channels and DFC)	
Quantity per MPC	Up to 68 BWR intact fuel assemblies	Up to 68 BWR damaged and/or intact fuel assemblies	Up to 68 BWR intact fuel assemblies	Up to 68 BWR damaged and/or intact fuel assemblies	
Other Limitations	 Quantity is limited to up to one (1) Dresden Unit 1 thoria rod canister meeting the specifications listed in Table 1.2.21 plus any combination of Dresden Unit 1 or Humboldt Bay damaged fuel assemblies in DFCs and intact fuel assemblies up to a total of 68. 				
	 Stainless steel channels are not permitted. 				
	 Fuel debris is not permitted. 				
	 Dresden Unit 1 fuel assemblies with one antimony-beryllium neutron source are permitted. The antimony-beryllium neutron source material shall be in a water rod location. 				

Notes:

- 1. A fuel assembly must meet the requirements of any one column and the other limitations to be authorized for transportation.
- Array/class 6x6A, 6x6C, 7x7A, and 8x8A fuel assemblies shall have a cooling time ≥ 18 years, an average burnup ≤ 30,000 MWD/MTU, and a minimum initial enrichment ≥ 1.8 wt. % ²³⁵U.
- Array/class 8x8F fuel assemblies shall have a cooling time ≥ 10 years, an average burnup ≤ 27,500 MWD/MTU, and a minimum initial enrichment ≥ 2.4 wt. %²³⁵U.
- SS-clad fuel assemblies shall have a cooling time ≥ 16 years, an average burnup ≤ 22,500 MWD/MTU, and a minimum initial enrichment ≥ 3.5 wt. % ²³⁵U.
- 5. Array/class 8x8F fuel assemblies shall have a decay heat \leq 183.5 Watts.

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-68F

PARAMETER		VALUE (No	tes 1 and 2)	
Fuel Type(s)	Uranium oxide, BWR intact fuel assemblies meeting the limits in Table 1.2.11 for array/class 6x6A, 6x6C, 7x7A, or 8x8A, with or without Zircaloy channels	Uranium oxide, BWR damaged fuel assemblies or fuel debris meeting the limits in Table 1.2.11 for array/class 6x6A, 6x6C, 7x7A, or 8x8A, with or without Zircaloy channels, placed in Damaged Fuel Containers(DFCs)	Mixed Oxide (MOX) BWR intact fuel assemblies meeting the limits in Table 1.2.11 for array/class 6x6B, with or without Zircaloy channels	Mixed Oxide (MOX) BWR damaged fuel assemblies or fuel debris meeting the limits in Table 1.2.11 for array/class 6x6B, with or without Zircaloy channels, placed in Damaged Fuel Containers (DFCs))
Cladding Type	ZR	ZR	ZR	ZR
Maximum Initial Planar-Average and Rod Enrichment	As specified in Table 1.2.11 for the applicable array/class	As specified in Table 1.2.11 for the applicable array/class	As specified in Table 1.2.11 for array/class 6x6B	As specified in Table 1.2.11 for array/class 6x6B
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly	Cooling time ≥ 18 years, average burnup $\leq 30,000$ MWD/MTU, and minimum initial enrichment ≥ 1.8 wt. % ²³⁵ U.	Cooling time ≥ 18 years, average burnup $\leq 30,000$ MWD/MTU, and minimum initial enrichment ≥ 1.8 wt. % ²³⁵ U.	Cooling time \geq 18 years, average burnup \leq 30,000 MWD/MTIH M, and minimum initial enrichment \geq 1.8 wt. % ²³⁵ U.	Cooling time ≥ 18 years, average burnup $\leq 30,000$ MWD/MTIHM, and minimum initial enrichment ≥ 1.8 wt. % ²³⁵ U.
Decay Heat Per Assembly	\leq 115 Watts	≤115 Watts	<u> < 115 Watts </u>	<u>< 115 Watts</u>

Table 1.2.24 (cont'd)

PARAMETER	VALUE (Note 1)			
Fuel Assembly Length	\leq 135.0 in. (nominal design)	≤ 135.0 in. (nominal design)	\leq 135.0 in. (nominal design)	≤ 135.0 in. (nominal design)
Fuel Assembly Width	\leq 4.70 in. (nominal design)	≤ 4.70 in. (nominal design)	≤ 4.70 in. (nominal design)	\leq 4.70 in. (nominal design)
Fuel Assembly Weight	\leq 400 lbs (including channels)	\leq 550 lbs (including channels and DFC)	<pre> ≤ 400 lbs (including channels) </pre>	\leq 550 lbs (including channels and DFC)
Other Limitations	 Quantity is Humboldt I storage loca 7x7A, and - uranium o MOX BW uranium o MOX BW up to one specificat Stainless st Dresden Un source are p shall be in a 	 channels) channels and DFC) Quantity is limited to up to four (4) DFCs containing Dresden Unit 1 or Humboldt Bay uranium oxide or MOX fuel debris. The remaining fuel storage locations may be filled with array/class 6x6A, 6x6B, 6x6C, 7x7A, and 8x8A fuel assemblies of the following type, as applicable: uranium oxide BWR intact fuel assemblies MOX BWR intact fuel assemblies uranium oxide BWR damaged fuel assemblies in DFCs MOX BWR damaged fuel assemblies in DFCs up to one (1) Dresden Unit 1 thoria rod canister meeting the specifications listed in Table 1.2.21 Stainless steel channels are not permitted. Dresden Unit 1 fuel assemblies with one antimony-beryllium neutron 		

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-68F

Notes:

- 1. A fuel assembly must meet the requirements of any one column and the other limitations to be authorized for transportation.
- 2. Only fuel from Dresden Unit 1 and Humboldt Bay plant are permitted for transportation in the MPC-68F.

PARAMETER	VALUE (Note 1)		
Fuel Type	Uranium oxide PWR intact fuel assemblies meeting the limits in Table 1.2.10 for the applicable array/class	Trojan plant damaged fuel meeting the limits in Table 1.2.10 for array/class 17x17B, placed in a Holtec Damaged Fuel Container (DFC) designed for Trojan plan fuel or a Trojan Failed Fuel Can (FFC)	
Cladding Type	ZR or Stainless Steel (SS) assemblies as specified in Table 1.2.10 for the applicable array/class	ZR	
Maximum Initial Enrichment	As specified in Table 1.2.10 for the applicable array/class	3.7 wt. % ²³⁵ U	
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly (except Trojan plant fuel and non- fuel hardware)	ZR clad: As specified in Table 1.2 28 or 1.2.29, as applicable SS clad: As specified in Table 1.2.30	Not applicable	
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly for Trojan plant fuel	As specified in Table 1.2.35	As specified in Table 1.2.35	
Post-irradiation Cooling Time and Burnup for Trojan plant Non-fuel Hardware and Neutron Sources	As specified in Table 1.2.36	Not applicable	
Decay Heat Per Assembly (except for Trojan plant fuel)	ZR clad: \leq 833 Watts SS clad: \leq 488 Watts	Not applicable	
Decay heat per Assembly for Trojan plant fuel	\leq 725 Watts	≤ 725 Watts	

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-24E

Table 1.2.25 (cont'd)

PARAMETER	VALUE (Note 1)	
Fuel Assembly Length	\leq 176.8 in. (nominal design)	\leq 169.3 in. (nominal design)
Fuel Assembly Width	≤ 8.54 in. (nominal design)	\leq 8.43 in. (nominal design)
Fuel Assembly Weight	<pre></pre>	\leq 1680 lbs (including DFC or Failed Fuel Can)
Other Limitations	 Quantity per MPC: up assemblies. For Trojar damaged fuel assemblies storage locations 3, 6, fuel storage locations intact fuel assemblies. Trojan plant fuel must designed Trojan MPC (see Figure 1.1.5). Fur permitted to be transp Except for Trojan plant not contain non-fuel h assemblies containing transported in any fue Trojan plant damaged transported in a Holta a Trojan plant FFC. One (1) Trojan plant Sources, each in a Tromay be transported in source may be transported in source may be transported in a Tromay be transported in source may be transported in a turk MPC-24E. Trojan plant non-fuel may not be transported in a may not be transported in the transport	to 24 PWR intact fuel a plant fuel only, up to four (4) ies may be stored in fuel 19, and/or 22. The remaining may be filled with Trojan plant t be transported in the custom- s with the MPC spacer installed el from other plants is not orted in the Trojan MPCs. at fuel, the fuel assemblies shall ardware. Trojan intact fuel non-fuel hardware may be 1 storage location. fuel assemblies must be to DFC for Trojan plant fuel or Sb-Be and/or two (2) Cf neutron jan plant intact fuel assembly any one MPC. Each neutron orted in any fuel storage norized for transportation in the hardware and neutron sources d in the same fuel storage d fuel assemblies.

Notes:

1. A fuel assembly must meet the requirements of any one column and the other limitations to be authorized for transportation.

PARAMETER VALUE (Note 1)			
Fuel Type	Uranium oxide PWR intact fuel assemblies meeting the limits in Table 1.2.10 for the applicable array/class	Trojan plant damaged fuel meeting the limits in Table 1.2.10 for array/class 17x17B, placed in a Holtec Damaged Fuel Container (DFC)designed for Trojan plant fuel or a Trojan Failed Fuel Can (FFC)	Trojan plant Fuel Debris Process Can Capsules and/or Trojan plant fuel assemblies classified as fuel debris, for which the original fuel assemblies meet the applicable criteria in Table 1.2.10 for array/class 17x17B, placed in a Holtec Damaged Fuel Container (DFC) designed for Trojan plant fuel or a Trojan Failed Fuel Can (FFC)
Cladding Type	ZR or Stainless Steel (SS) assemblies as specified in Table 1.2.10 for the applicable array/class	ZR	ZR
Maximum Initial Enrichment	As specified in Table 1.2.10 for the applicable array/class	\leq 3.7 wt. % ²³⁵ U	\leq 3.7 wt. % ²³⁵ U
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly (except Trojan plant fuel and non-fuel hardware)	ZR clad: As specified in Table 1.2 28 or 1.2.29, as applicable SS clad: As specified in Table 1.2.30	Not applicable	Not applicable
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly for Trojan plant fuel	As specified in Table 1.2.35	As specified in Table 1.2.35	As specified in Table 1.2.35

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-24EF

Table 1.2.26 (cont'd)

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-24EF

PARAMETER	VALUE (Note 1)		
Post-irradiation Cooling Time and Burnup for Trojan plant Non-fuel Hardware and Neutron Sources	As specified in Table 1.2.36	As specified in Table 1.2.36	As specified in Table 1.2.36
Decay Heat Per Assembly (except for Trojan plant fuel)	ZR clad: ≤ 833 Watts SS clad: ≤ 488 Watts	Not applicable	Not applicable
Decay heat per Assembly for Trojan plant fuel	≤ 725 Watts	≤ 725 Watts	≤ 725 Watts
Fuel Assembly Length	\leq 176.8 in. (nominal design)	\leq 169.3 in. (nominal design)	\leq 169.3 in. (nominal design)
Fuel Assembly Width	\leq 8.54 in. (nominal design)	\leq 8.43 in. (nominal design)	<pre> ≤ 8.43 in. (nominal design) </pre>
Fuel Assembly Weight	≤ 1680 lbs (including non-fuel hardware(≤ 1680 lbs (including DFC or Failed Fuel Can)	≤ 1680 lbs (including DFC or Failed Fuel Can)

,

Other Limitations	 Quantity per MPC: Up to 24 PWR intact fuel assemblies. For Trojan plant fuel only, up to four (4) damaged fuel assemblies, fuel assemblies classified as fuel debris, and/or Trojan Fuel Debris Process Can Capsules may be stored in fuel storage locations 3, 6, 19, and/or 22. The remaining fuel storage locations may be filled with Trojan plant intact fuel assemblies. Trojan plant fuel must be transported in the custom-designed Trojan MPCs with the MPC spacer installed (see Figure 1.1.5). Fuel from other plants is not permitted to be transported in the Trojan MPCs. Except for Trojan plant fuel, the fuel assemblies shall not contain non-fuel hardware or neutron sources. Trojan intact fuel assemblies containing non-fuel hardware may be transported in any fuel storage location. Trojan plant damaged fuel assemblies, fuel assemblies classified as fuel debris, and Fuel Debris Process Can Capsules must be transported in a Trojan Failed Fuel Can or a Holtec DFC for Trojan plant fuel. One (1) Trojan plant Sb-Be and/or two (2) Cf neutron sources, each in a Trojan plant intact fuel assembly may be transported in any fuel storage location.

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-24EF

Notes:

1. A fuel assembly must meet the requirements of any one column and the other limitations to be authorized for transportation.

LIMITS FOR MATERIAL TO BE TRANSPORTED IN MPC-32 (Note 1)

PARAMETER	VALUE	
Fuel Type	Uranium oxide, PWR intact fuel assemblies meeting the limits in Table 1.2.10 for array/classes 15x15D, E, F, and H and 17x17A, B, and C	
Cladding Type	ZR	
Maximum Initial Enrichment	As specified in Table 1.2.10	
Post-irradiation Cooling Time, Average Burnup, and Minimum Initial Enrichment per Assembly	As specified in Table 1.2.32 or Table 1.2.33, as applicable	
Decay Heat Per Assembly	\leq 625 Watts	
Minimum Burnup per Assembly	As specified in Table 1.2.34 for the applicable array/class	
Fuel Assembly Length	\leq 176.8 in. (nominal design)	
Fuel Assembly Width	\leq 8.54 in. (nominal design)	
Fuel Assembly Weight	≤ 1,680 lbs	
Other Limitations	 Quantity is limited to up to 32 PWR intact fuel assemblies in the above- specified array/classes only. 	
	 Non-fuel hardware and neutron sources not permitted. 	
	 Damaged fuel assemblies and fuel debris not permitted. 	
	 Trojan plant fuel not permitted. 	

NOTES:

1. The MPC-32 is not authorized for transportation in the HI-STAR 100 System at this time.

FUEL ASSEMBLY COOLING, AVERAGE BURNUP, AND MINIMUM ENRICHMENT LIMITS FOR TRANSPORTATION IN MPC-24/24E/24EF; PWR FUEL WITH ZR CLADDING AND WITH NON-ZIRCALOY IN-CORE GRID SPACERS

ASSEMBLY POST- IRRADIATION COOLING TIME (years)	ASSEMBLY BURNUP (MWD/MTU)	ASSEMBLY ENRICHMENT (wt. % ²³⁵ U)
<u>≥9</u>	<u>≤</u> 24,500	≥2.3
≥11	≤ 29,500	<u>≥</u> 2.6
<u>≥13</u>	<u>≤</u> 34,500	≥ 2.9
<u>≥</u> 15	<u>≤</u> 39,500	≥ 3.2
<u>≥ 18</u>	<u>≤</u> 44,500	≥3.4

FUEL ASSEMBLY COOLING, AVERAGE BURNUP, AND MINIMUM ENRICHMENT LIMITS FOR TRANSPORTATION IN MPC-24/24E/24EF;PWR FUEL WITH ZR CLADDING AND WITH ZIRCALOY IN-CORE GRID SPACERS

ASSEMBLY POST- IRRADIATION COOLING TIME (years)	ASSEMBLY BURNUP (MWD/MTU)	ASSEMBLY ENRICHMENT (wt. % ²³⁵ U)
<u>≥6</u>	<u>≤</u> 24,500	<u>≥</u> 2.3
<u>≥</u> 7	<u>≤</u> 29,500	<u>≥</u> 2.6
<u>≥9</u>	<u>≤</u> 34,500	<u>≥</u> 2.9
≥11	≤ 39,500	<u>≥ 3.2</u>
<u>≥</u> 14	<u>≤</u> 44,500	≥ 3.4

FUEL ASSEMBLY COOLING, AVERAGE BURNUP, AND MINIMUM ENRICHMENT LIMITS FOR TRANSPORTATION IN MPC-24/24E/24EF; PWR FUEL WITH STAINLESS STEEL CLADDING

ASSEMBLY POST- IRRADIATION COOLING TIME (years)	ASSEMBLY BURNUP (MWD/MTU)	ASSEMBLY ENRICHMENT (wt. % ²³⁵ U)
≥ 19	≤ 30,000	<u>≥</u> 3.1
<u>≥ 24</u>	≤40,000	≥ 3.1

FUEL ASSEMBLY COOLING, AVERAGE BURNUP, AND MINIMUM ENRICHMENT LIMITS FOR TRANSPORTATION IN MPC-68

ASSEMBLY POST- IRRADIATION COOLING TIME (years)	ASSEMBLY BURNUP (MWD/MTU)	ASSEMBLY ENRICHMENT (wt. % ²³⁵ U)
<u>≥ 8</u>	<u>≤</u> 24,500	≥ 2.1
<u>≥ 9</u>	≤ 29,500	≥ 2.4
≥11	<u>≤</u> 34,500	≥ 2.6
<u>≥</u> 14	<u>≤</u> 39,500	<u>≥ 2.9</u>
≥ 19	≤ 44,500	≥ 3.0

FUEL ASSEMBLY COOLING, AVERAGE BURNUP, AND MINIMUM ENRICHMENT LIMITS FOR TRANSPORTATION IN MPC-32; PWR FUEL WITH ZR CLADDING AND WITH NON-ZIRCALOY IN-CORE GRID SPACERS (Note 1)

ASSEMBLY POST- IRRADIATION COOLING TIME (years)	ASSEMBLY BURNUP (MWD/MTU)	ASSEMBLY ENRICHMENT (wt. % ²³⁵ U)
≥ 12	<u>≤</u> 24,500	≥ 2.3
<u>≥</u> 14	<u>≤ 29,500</u>	<u>≥ 2.6</u>
<u>≥</u> 16	<u>≤</u> 34,500	<u>≥ 2.9</u>
<u>≥ 19</u>	<u>≤</u> 39,500	<u>≥ 3.2</u>
≥20	<u>≤</u> 42,500	<u>≥</u> 3.4

NOTES:

1. MPC-32 is not authorized for transportation at this time.

FUEL ASSEMBLY COOLING, AVERAGE BURNUP, AND MINIMUM ENRICHMENT LIMITS FOR TRANSPORTATION IN MPC-32; PWR FUEL WITH ZR CLADDING AND WITH ZIRCALOY IN-CORE GRID SPACERS (Note 1)

ASSEMBLY POST- IRRADIATION COOLING TIME (years)	ASSEMBLY BURNUP (MWD/MTU)	ASSEMBLY ENRICHMENT (wt. % ²³⁵ U)
<u>≥</u> 8	≤24,500	≥ 2.3
≥9	≤ 29,500	≥ 2.6
<u>≥ 12</u>	<u>≤</u> 34,500	≥2.9
≥14	<u>≤</u> 39,500	≥ 3.2
<u>≥ 19</u>	<u>≤</u> 44,500	≥ 3.4

NOTES:

1. MPC-32 is not authorized for transportation at this time.

FUEL ASSEMBLY MINIMUM BURNUP REQUIREMENTS FOR TRANSPORTATION IN MPC-32 (Note 1)

FUEL ASSEMBLY ARRAY/CLASS	MINIMUM BURNUP (B) AS A FUNCTION OF INITIAL ENRICHMENT (E) (Note 2) (GWD/MTU)	
15x15D, E, F, and H	Later	
17x15A, B, and C	Later	

Notes:

- 1. MPC-32 is not authorized for transportation at this time.
- 2. E = Initial enrichment from the fuel vendor's data sheet, i.e., for 4.05wt. %, E = 4.05.

TROJAN PLANT FUEL ASSEMBLY COOLING, AVERAGE BURNUP, AND MINIMUM ENRICHMENT LIMITS (Note 1)

Post-irradiation Cooling Time (years)	Assembly Burnup (MWD/MTU)	Assembly Minimum Enrichment (wt. % ²³⁵ U)
≥16	<u>≤</u> 42,000	≥ 3.09
≥16	≤ 37,500	≥ 2.6
<u>≥ 16</u>	<u>≤</u> 30,000	≥2.1

Notes:

1. Each fuel assembly must only meet one set of limits (i.e., one row).

TROJAN PLANT NON-FUEL HARDWARE AND NEUTRON SOURCE COOLING AND BURNUP LIMITS

Type Of Hardware or Neutron Source	Burnup (MWD/MTU)	Post-irradiation Cooling Time (years)
BPRAs	<u>≤ 15,998</u>	<u>≥ 24</u>
TPDs	<u>≤118,674</u>	≥11
RCCAs	≤ 125,515	<u>≥9</u>
Cf neutron source	<u>≤ 15,998</u>	≥24
Sb-Be neutron source with 4 source rods, 16 burnable poison rods, and 4 thimble plug rods	<u>≤</u> 45,361	≥ 19
Sb-Be neutron source with 4 source rods and 20 thimble plug rods	<u>≤</u> 88,547	≥9


















Figure 1.2.9; Major HI-STAR 100 Loading Operations (Sheet 1 of 3)



Figure 1.2.9; Major HI-STAR 100 Loading Operations (Sheet 2 of 3)



Figure 1.2.9; Major HI-STAR 100 Loading Operations (Sheet 3 of 3)







FIGURE 1.2.12

INTENTIONALLY DELETED

HI-STAR SAR REPORT HI-951251

PWR Axial Burnup Distribution 1.2000 1.0000 Normalized Burnup 0.6000 0.4000 0.2000 0.0000 12 0 24 36 48 60 72 84 96 108 120 132 144 Active Fuel Length (in.)





Figure 1.2.13A; Trojan Plant Fuel Axial Burnup Distribution Profile

Report HI-951251

BWR Axial Burnup Distribution



REPORT HI-951251





Figure 1.2.16; Major HI-STAR 100 Unloading Operations (Sheet 1 of 3)







Figure 1.2.16; Major HI-STAR 100 Unloading Operations (Sheet 3 of 3)



1.3 DESIGN CODE APPLICABILITY

The ASME Boiler and Pressure Vessel Code (ASME Code), 1995 Edition with Addenda through 1997 [1.3.1], is the governing code for the construction of the HI-STAR 100 System, as clarified in Table 1.3.2. The ASME Code is applied to each component consistent with the function of the component. Table 1.3.3 lists each structure, system and component (SSC) of the HI-STAR 100 System that are labeled Important to Safety (ITS), along with its function and governing Code. Some components perform multiple functions and in those cases, the most restrictive Code is applied. In accordance with NUREG/CR-6407, "Classification of Transportation Packaging and Dry Spent Fuel Storage System Components" [1.3.2] and according to importance to safety, components of the HI-STAR 100 System are classified as A, B, C, or NITS (not important to safety) in Table 1.3.3. Table 1.3.3 may not include all NITS items associated with the HI-STAR 100 Package.

Table 1.3.1 lists the applicable ASME Code section and paragraph for material procurement, design, fabrication and inspection of the components of the HI-STAR 100 System that are governed by the ASME Code. The ASME Code section listed in the design column is the section used to define allowable stresses for structural analyses.

Table 1.3.2 lists the alternatives to the ASME Code for the HI-STAR 100 System and the justification for those alternatives.

The MPC is classified as important to safety. The MPC structural components include the internal fuel basket and the enclosure vessel. The fuel basket is designed and fabricated as a core support structure, in accordance with the applicable requirements of Section III, Subsection NG of the ASME Code, with certain NRC-approved alternatives, as discussed in Table 1.3.2. The enclosure vessel is designed and fabricated as a Class 1 component pressure vessel in accordance with Section III, Subsection NB of the ASME Code, with certain NRC-approved alternatives, as discussed in Table 1.3.2. The principal exceptions are the MPC lid, vent and drain cover plates, and closure ring welds to the MPC lid and shell, as discussed in Table 1.3.2. In addition, the threaded holes in the MPC lid are designed in accordance with the requirements of ANSI N14.6 [1.3.3] for critical lifts to facilitate vertical MPC transfer.

The MPC closure welds are partial penetration welds that are structurally qualified by analysis, as presented in Chapter 2. The MPC closure ring welds are inspected by performing a liquid penetrant examination of the root pass (if more than one weld pass is required) and final weld surface, in accordance with the requirements contained in Section 8.1. The MPC lid weld may be examined by either volumetric or multi-layer liquid penetrant examination. If volumetric examination is used, it shall be the ultrasonic method and shall include a liquid penetrant examination of the root and final weld layers. If multi-layer liquid penetrant examination is used alone, at a minimum, it must include the root and final weld layers and each approximately 3/8 inch of weld to detect critical weld flaws. The integrity of the MPC lid weld is further verified by performing a pressure test (hydrostatic or pneumatic) and a helium leak test in accordance with the requirements contained in Section 8.1.

The structural analysis of the MPC, in conjunction with the redundant closures and nondestructive examination, pressure testing, and helium leak testing performed during MPC | fabrication and MPC closure, provides assurance of canister closure integrity in lieu of the specific weld joint requirements of the ASME Code, Section III, Subsection NB.

The HI-STAR overpack is classified as important to safety. The HI-STAR overpack top flange, | closure plate, inner shell, and bottom plate are designed and fabricated in accordance with the requirements of ASME Code, Section III, Subsection NB, to the maximum extent practical (see Table 1.3.2). The remainder of the HI-STAR overpack steel structure is designed and fabricated | in accordance with the requirements of ASME Code, Section III, Subsection III, Subsection NF, to the maximum extent practical (see Table 1.3.2).

Table 1.3.1

· · · · · · · · · · · · · · · · · · ·				
HI-STAR 100 Component	Material Procurement	Design	Fabrication	Inspection
Overpack containment boundary	Section II; and Section III, Subsection NB, NB-2000	Section III, Subsection NB, NB-3200	Section III, Subsection NB, NB-4000	Section III, Subsection NB, NB-5000 and Section V
Overpack intermediate shells, radial channels, outer enclosure	ack Section II; and Section III, Section III, Section III, Section III, Section NF New Section N		Section III, Subsection NF, NF-4000	Section III, Subsection NF, NF-5360 and Section V
MPC helium retention boundary	Section II; and Section III, Subsection NB, NB-2000	Section III, Subsection NB, NB-3200	Section III, Subsection NB, NB-4000	Section III, Subsection NB, NB-5000 and Section V
MPC fuel basket	Section II; and Section III, Subsection NG, NG-2000 for core support structures (NG-1121)	Section III, Subsection NG, NG-3300 and NG-3200 for core support structures (NG-1121)	Section III, Subsection NG, NG-4000 for core support structures (NG-1121)	Section III, Subsection NG, NG-5000 and Section V for core support structures (NG-1121)
Lifting Trunnions	Section II; and Section III, Subsection NF, NF-2000	ANSI N14.6	Section III, Subsection NF, NF-4000	ANSI 14.6 See Chapter 8
MPC Basket Supports (Angled Plates)	Section II, and Section III, Subsection NG, NG-2000 for internal structures (NG-1122)	Section III, Subsection NG, NG-3300 and NG-3200 for internal structures (NG-1122)	Section III, Subsection NG, NG-4000 for internal structures (NG-1122)	Section III, Subsection NG, NG-5000 and Section V for internal structures (NG-1122)
Damaged Fuel Container	Section II, and Section III, Subsection NG, NG-2000	Section III, Subsection NG, NG-3300 and NG-3200	Section III, Subsection NG, NG-4000	Section III, Subsection NG, NG-5000 and Section V

HI-STAR 100 ASME BOILER AND PRESSURE VESSEL CODE APPLICABILITY

Table 1.3.2

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures
MPC, MPC basket assembly, and HI- STAR overpack steel structure.	Subsection NCA	General Requirements. Requires preparation of a Design Specification, Design Report, Overpressure Protection Report, Certification of Construction Report, Data Report, and other administrative controls for an ASME Code stamped vessel.	Because the MPC, overpack, and transfer cask are not ASME Code stamped vessels, none of the specifications, reports, certificates, or other general requirements specified by NCA are required. In lieu of a Design Specification and Design Report, the HI-STAR SAR includes the design criteria, service conditions, and load combinations for the design and operation of the HI-STAR 100 System as well as the results of the stress analyses to demonstrate that applicable Code stress limits are met. Additionally, the fabricator is not required to have an ASME-certified QA program. All important-to-safety activities are governed by the NRC-approved Holtec QA program. Because the cask components are not certified to the Code, the terms "Certificate Holder" and "Inspector" are not germane to the manufacturing of NRC-certified cask components. To eliminate ambiguity, the responsibilities assigned to the Certificate Holder in the various articles of Subsections NB, NG, and NF of the Code, as applicable, shall be interpreted to apply to the NRC Certificate of Compliance (CoC) holder (and by extension, to the component fabricator) if the requirement must be fulfilled. The Code term "Inspector" means the QA/QC personnel of the CoC holder and its vendors assigned to oversee and inspect the manufacturing process.

LIST OF ASME CODE ALTERNATIVES FOR HI-STAR 100 SYSTEM

Notes: 1) There are no known residuals on finished component surfaces.

2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.

- 3) Component nomenclature taken from drawings in Chapter 1.
- 4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.
- 5) Thermaline 450 and Carboline 890 were the product names at the time of initial licensing. Chemically identical products with different names are permitted. For example, Carboline 890 was re-named Carboguard 890 in 2000, with no change to the coating material and is, therefore, acceptable for use where Carboline 890 is specified.

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures
МРС	NB-1100	Statement of requirements for Code stamping of components.	MPC vessel is designed and will be fabricated in accordance with ASME Code, Section III, Subsection NB to the maximum practical extent, but Code stamping is not required.
MPC	NB-2000	Requires materials to be supplied by ASME- approved material supplier.	Materials will be supplied by Holtec approved suppliers with Certified Material Test Reports (CMTRs) in accordance with NB-2000 requirements.
MPC basket supports and lift lugs	NB-1130	NB-1132.2(d) requires that the first connecting weld of a nonpressure-retaining structural attachment to a component shall be considered part of the component unless the weld is more than 2t from the pressure- retaining portion of the component, where t is the nominal thickness of the pressure- retaining material. NB-1132.2(e) requires that the first connecting weld of a welded nonstructural attachment to a component shall conform to NB-4430 if the connecting weld is within 2t from the pressure-retaining portion of the component.	The MPC basket supports (nonpressure-retaining structural attachments) and lift lugs (nonstructural attachments used exclusively for lifting an empty MPC) are welded to the inside of the pressure-retaining MPC shell, but are not designed in accordance with Subsection NB. The basket supports and associated attachment welds are designed to satisfy the stress limits of Subsection NG and the lift lugs and associated attachment welds are designed to satisfy the stress limits of Subsection NF, as a minimum. These attachments and their welds are shown by analysis to meet the respective stress limits for their service conditions. Likewise, non-structural items, such as shield plugs, spacers, etc. if used, can be attached to pressure-retaining parts in the same manner.
MPC, MPC basket assembly, and HI- STAR overpack steel structure.	NB-3100 NG-3100 NF-3100	Provides requirements for determining design loading conditions, such as pressure, temperature, and mechanical loads.	These requirements are not applicable. The HI-STAR SAR, serving as the Design Specification, establishes the service conditions and load combinations for the storage system.

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures
MPC	NB-3350	NB-3352.3 requires, for Category C joints, that the minimum dimensions of the welds and throat thickness shall be as shown in Figure NB-4243-1.	The MPC shell-to-baseplate weld joint design (designated Category C) may not include a reinforcing fillet weld or a bevel in the MPC baseplate, which makes it different than any of the representative configurations depicted in Figure NB-4243-1. The transverse thickness of this weld is equal to the thickness of the adjoining shell (1/2 inch). The weld is designed as a full penetration weld that receives VT and RT or UT, as well as final surface PT examinations. Because the MPC shell design thickness is considerably larger than the minimum thickness required by the Code, a reinforcing fillet weld that would intrude into the MPC cavity space is not included. Not including this fillet weld provides for a higher quality radiographic examination of the full penetration weld.
			From the standpoint of stress analysis, the fillet weld serves to reduce the local bending stress (secondary stress) produced by the gross structural discontinuity defined by the flat plate/shell junction. In the MPC design, the shell and baseplate thicknesses are well beyond that required to meet their respective membrane stress intensity limits.
MPC, MPC basket assembly, and HI- STAR overpack steel structure	NB-4120 NG-4120 NF-4120	NB-4121.2, NG-4121.2, and NF-4121.2 provide requirements for repetition of tensile or impact tests for material subjected to heat treatment during fabrication or installation.	In-shop operations of short duration that apply heat to a component, such as plasma cutting of plate stock, welding, machining, coating, and pouring of Holtite are not, unless explicitly stated by the Code, defined as heat treatment operations.
			For the steel parts in the HI-STAR 100 System components, the duration for which a part exceeds the off-normal temperature limit shall be limited to 24 hours in a particular manufacturing process (such as the Holtite pouring process).

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures
MPC and HI-STAR overpack steel structure	NB-4220 NF-4220	Requires certain forming tolerances to be met for cylindrical, conical, or spherical shells of a vessel.	The cylindricity measurements on the rolled shells are not specifically recorded in the shop travelers, as would be the case for a Code-stamped pressure vessel. Rather, the requirements on inter-component clearances (such as the MPC-to-overpack) are guaranteed through fixture-controlled manufacturing. The fabrication specification and shop procedures ensure that all dimensional design objectives, including inter-component annular clearances are satisfied. The dimensions required to be met in fabrication are chosen to meet the functional requirements of the dry storage components. Thus, although the post-forming Code cylindricity requirements are not evaluated for compliance directly, they are indirectly satisfied (actually exceeded) in the final manufactured components.
MPC Lid and Closure Ring Welds	NB-4243	Full penetration welds required for Category C Joints (flat head to main shell per NB- 3352.3)	MPC lid and closure ring are not full penetration welds. They are welded independently to provide a redundant seal. Additionally, a weld efficiency factor of 0.45 has been applied to the analyses of these welds.
MPC Lid-to-Shell Weld	NB-5230	Radiographic (RT) or ultrasonic (UT) examination required.	Only UT or multi-layer liquid penetrant (PT) examination is permitted. If PT alone is used, at a minimum, it will include the root and final weld layers and each approximately 3/8 inch of weld depth.
MPC Closure Ring, Vent and Drain Cover Plate Welds	NB-5230	Radiographic (RT) or ultrasonic (UT) examination required.	Root (if more than one weld pass is required) and final liquid penetrant examination to be performed in accordance with NB-5245. The MPC vent and drain cover plate welds are leak tested. The closure ring provides independent redundant closure for vent and drain cover plates.

LIST OF ASME CODE ALTERNATIVES FOR HI-STAR 100 SYSTEM

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures		
MPC Enclosure Vessel and Lid	NB-6111	All completed pressure retaining systems shall be pressure tested.	The MPC vessel is seal welded in the field following fuel assembly loading. The MPC vessel shall then be pressure tested as defined in Chapter 8. Accessibility for leakage inspections preclude a Code compliant pressure test. All MPC vessel welds (except closure ring and vent/drain cover plate) are inspected by volumetric examination, except that the MPC lid-to-shell weld shall be verified by volumetric or multi-layer PT examination. If PT alone is used, at a minimum, it must include the root and final layers and each approximately 3/8 inch of weld depth. For either UT or PT, the maximum undetectable flaw size must be demonstrated to be less than the critical flaw size. The critical flaw size must be determined in accordance with ASME Section XI methods. The critical flaw size shall not cause the primary stress limits of NB-3000 to be exceeded. The inspection results, including relevant findings (indications) shall be made a permanent part of the user's records by video, photographic, or other means which provide an equivalent retrievable record of weld integrity. The video or photographic records should be taken during the final interpretation period described in ASME Section V, Article 6, T-676. The vent/drain cover plate welds are confirmed by helium leakage testing and liquid penetrant examination and the closure ring weld is confirmed by liquid penetrant examination. The inspection of the weld must be performed by qualified personnel and shall meet the acceptance requirements of ASME Code Section III, NB-5350 for PT or NB-5332 for UT.		
MPC Enclosure Vessel	NB-7000	Vessels are required to have overpressure protection.	No overpressure protection is provided. The function of MPC vessel is as a helium retention boundary. MPC vessel is designed to withstand maximum internal pressure considering 100% fuel rod failure and maximum accident temperatures.		

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures
MPC Enclosure Vessel	NB-8000	States requirements for nameplates, stamping and reports per NCA-8000.	HI-STAR 100 System to be marked and identified in accordance with 10CFR71 and 10CFR72 requirements. Code stamping is not required. QA data package to be in accordance with Holtec approved QA program.
Overpack Containment Boundary	NB-1100	Statement of requirements for Code stamping of components.	Overpack containment boundary is designed, and will be fabricated in accordance with ASME Code, Section III, Subsection NB to the maximum practical extent, but Code stamping is not required.
Overpack Containment Boundary	NB-2000	Requires materials to be supplied by ASME- approved material supplier.	Materials will be supplied by Holtec approved suppliers with CMTRs per NB-2000.
Overpack Containment Boundary	NB-7000	Vessels are required to have overpressure protection.	No overpressure protection is provided. Function of overpack vessel is as a radionuclide containment boundary under normal and hypothetical accident conditions. Overpack vessel is designed to withstand maximum internal pressure and maximum accident temperatures.
Overpack Containment Boundary	NB-8000	States requirements for nameplates, stamping and reports per NCA-8000.	HI-STAR 100 System to be marked and identified in accordance with 10CFR71 and 10CFR72 requirements. Code stamping is not required. QA data package to be in accordance with Holtec's approved QA program.
MPC Basket Assembly	NG-2000	Requires materials to be supplied by ASME- approved material supplier.	Materials will be supplied by Holtec approved supplier with CMTRs in accordance with NG-2000 requirements.

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures
MPC Basket Assembly	NG-4420	NG-4427(a) allows a fillet weld in any single continuous weld to be less than the specified fillet weld dimension by not more than 1/16 inch, provided that the total undersize portion of the weld does not exceed 10 percent of the length of the weld. Individual undersize weld portions shall not exceed 2 inches in length.	Modify the Code requirement (intended for core support structures) with the following text prepared to accord with the geometry and stress analysis imperatives for the fuel basket: For the longitudinal MPC basket fillet welds, the following criteria apply: 1) The specified fillet weld throat dimension must be maintained over at least 92 percent of the total weld length. All regions of undersized weld must be less than 3 inches long and separated from each other by at least 9 inches. 2) Areas of undercuts and porosity beyond that allowed by the applicable ASME Code shall not exceed 1/2 inch in weld length. The total length of undercut and porosity over any 1- foot length shall not exceed 2 inches. 3) The total weld length in which items (1) and (2) apply shall not exceed a total of 10 percent of the overall weld length. The limited access of the MPC basket panel longitudinal fillet welds makes it difficult to perform effective repairs of these welds and creates the potential for causing additional damage to the basket assembly (e.g., to the neutron absorber and its sheathing) if repairs are attempted. The acceptance criteria provided in the foregoing have been established to comport with the objectives of the basket design and preserve the margins demonstrated in the supporting stress analysis. From the structural standpoint, the weld acceptance criteria are established to ensure that any departure from the ideal, continuous fillet weld seam would not alter the primary bending stresses on which the design of the fuel baskets is predicated. Stated differently, the permitted weld discontinuities are limited in size to ensure that they remain classifiable as local stress elevators ("peak stress", F, in the ASME Code for which specific stress intensity limits do not apply).
MPC Basket Assembly	NG-8000	States requirements for nameplates, stamping and reports per NCA-8000.	The HI-STAR 100 System will be marked and identified in accordance with 10CFR71 and 10CFR72 requirements. No Code stamping is required. The MPC basket data package will be in conformance with Holtec's QA program.
Overpack Intermediate Shells	NF-2000	Requires materials to be supplied by ASME- approved material supplier.	Materials will be supplied by Holtec approved supplier with CMTRs in accordance with NF-2000 requirements.

Component	Reference ASME Code Section/Article	Code Requirement	Alternative, Justification & Compensatory Measures
Overpack Containment Boundary	NB-2330	Defines the methods for determining the T_{NDT} for impact testing of materials.	T_{NDT} shall be defined in accordance with Regulatory Guides 7.11 and 7.12 for the containment boundary components.
Overpack Containment Boundary	NF-3320 NF-4720	NF-3324.6 and NF-4720 provide requirements for bolting.	 These Code requirements are applicable to linear structures wherein bolted joints carry axial, shear, as well as rotational (torsional) loads. The overpack bolted connections in the structural load path are qualified by design based on the design loadings defined in the SAR. Bolted joints in these components see no shear or torsional loads under normal storage conditions. Larger clearances between bolts and holes may be necessary to ensure shear interfaces located elsewhere in the structure engage prior to the bolts experiencing shear loadings (which occur only during side impact scenarios). Bolted joints that are subject to shear loads in accident conditions are qualified by appropriate stress analysis. Larger bolt-to-hole clearances help ensure more efficient operations in making these bolted connections, thereby minimizing time spent by operations personnel in a radiation area. Additionally, larger bolt-to-hole clearances allow interchangeability of the lids from one particular fabricated

Table 1.3.3

MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

OVERPACK ^(1, 2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Matl. (if dissimilar)
Containment	Inner Shell	A	ASME Section III; Subsection NB	SA203-E	Table 2.3.4	Paint inside surface with Thermaline 450 (Note 5). External surface to be coated with a surface preservative.	NA
Containment	Bottom Plate	A	ASME Section III; Subsection NB	SA350-LF3	Table 2.3.4	Paint inside surface with Thermaline 450 (Note 5).	NA
Containment	Top Flange	A	ASME Section III; Subsection NB	SA350-LF3	Table 2.3.4	Paint inside surface with Thermaline 450. Paint outside surface with Carboline 890 (Note 5).	NA
Containment	Closure Plate	A	ASME Section III; Subsection NB	SA350-LF3	Table 2.3.4	Paint inside surface with Thermaline 450. Paint outside surface with Carboline 890 (Note 5).	NA
Containment	Closure Plate Bolts	A	ASME Section III; Subsection NB	SB637- N07718	Table 2.3.5	NA	NA
Containment	Port Plug	A	Non-code	SA193-B8	Not required	NA	NA
Containment	Port Plug Seal	A	Non-code	Alloy X750	Not required	NA	NA

Notes: 1) There are no known residuals on finished component surfaces.

2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.

- 3) Component nomenclature taken from drawings in Chapter 1.
- 4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.
- 5) Thermaline 450 and Carboline 890 were the product names at the time of initial licensing. Chemically identical products with different names are permitted. For example, Carboline 890 was re-named Carboguard 890 in 2000, with no change to the coating material and is, therefore, acceptable for use where Carboline 890 is specified.

TABLE 1.3.3 (continued)

MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

OVERPACK ^(1,2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Matl. (if dissimilar)
Containment	Closure Plate Seal	Α	Non-code	Alloy X750	Not required	NA	NA
Containment	Port Cover Seal	В	Non-code	Alloy X750	Not required	NA	NA
Shielding	Intermediate Shells	В	ASME Section III; Subsection NF	SA516-70	Table 2.3.2	Internal surfaces to be coated with a silicone encapsulant (dow corning SYLGARD 567 or equivalent) for surface preservation. Exposed areas of fifth intermediate shell to be painted with Carboline 890 (Note 5).	NA
Shielding	Neutron Shield	B	Non-code	Holtite-A	Not required	NA	Holtite/CS
Shielding	Plugs for Drilled Holes	NITS	Non-code	SA193-B7	Not required	NA	NA
Shielding	Removable Shear Ring	В	ASME Section III; Subsection NF	SA203-E	Table 2.3.4	Paint external surface with Carboline 890 (Note 5).	NA
Shielding	Pocket Trunnion Plug Plate	с	Non-code	SA240-304	Not required	NA	NA

Notes: 1) There are no known residuals on finished component surfaces.

- 2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.
- 3) Component nomenclature taken from drawings in Chapter 1.
- 4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.
- 5) Thermaline 450 and Carboline 890 were the product names at the time of initial licensing. Chemically identical products with different names are permitted. For example, Carboline 890 was re-named Carboguard 890 in 2000, with no change to the coating material and is, therefore, acceptable for use where Carboline 890 is specified.

TABLE 1.3.3 (continued)

MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

OVERPACK (1,2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Matl. (if dissimilar)
Heat Transfer	Radial Channels	В	ASME Section III; Subsection NF	SA515-70	Table 2.3.3	Paint outside surface with Carboline 890 (Note 5).	NA
Rotation Pivot and Shielding	Pocket Trunnion	В	Non-Code	SA705-630 or SA564-630	Table 2.3.5	NA	NA
Structural Integrity	Lifting Trunnion	Α.	ANSI N14.6	SB637- N07718	Table 2.3.5	NA	NA
Structural Integrity	Backing Strip	В	Non-code	A569	Not required	NA	NA
Structural Integrity	Relief DeviceCoupling	NITS	Non-code	C/S	Not required	NA	NA
Structural Integrity	Relief Device	C	Non-code	Commercial	Not required	NA	Brass-C/S
Structural Integrity	Relief Device Pipe	C	Non-code	SA-106	Not required	NA	NA
Structural Integrity	Relief Device Plate	C	Non-code	A569	Not required	NA	NA
Structural Integrity	Removable Shear Ring Bolt	С	Non-code	SA193-B7	Not required	NA	NA
Structural Integrity	Thermal Expansion Foam	NITS	Non-code	Silicone Foam	Not required	NA	NA
Structural Integrity	Closure Bolt Washer	NITS	Non-code	SA240-304	Not required	NA	NA
Structural Integrity	Enclosure Shell Panels	В	ASME Section III; Subsection NF	SA515-70	Table 2.3.3	Paint outside surface with Carboline 890 (Note 5).	NA

Notes: 1) There are no known residuals on finished component surfaces.

- 2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.
- 3) Component nomenclature taken from drawings in Chapter 1.
- 4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.
- 5) Thermaline 450 and Carboline 890 were the product names at the time of initial licensing. Chemically identical products with different names are permitted. For example, Carboline 890 was re-named Carboguard 890 in 2000, with no change to the coating material and is, therefore, acceptable for use where Carboline 890 is specified.
MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

OVERPACK (1,2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Matl. (if dissimilar)
Structural Integrity	Enclosure Shell Return	В	ASME Section III; Subsection NF	SA515-70	Table 2.3.3	Paint outside surface with Carboline 890 (Note 5).	NA
Structural Integrity	Port Cover	В	ASME Section III; Subsection NF	SA203E	Table 2.3.4	Paint outside surface with Carboline 890 (Note 5).	NA
Structural Integrity	Port Cover Bolt	С	Non-code	SA193-B7	Not required	NA	NA
Operations	Trunnion Locking Pad and End Cap Bolt	С	Non-code	SA193-B7	Not required	NA	NA
Operations	Lifting Trunnion End Cap	С	Non-code	SA516-70 or SA515 Gr. 70	Table 2.3.2	Paint exposed surfaces with Carboline 890 (Note 5).	NA
Operations	Lifting Trunnion Locking Pad	с	Non-code	SA516-70	Table 2.3.2	Paint exposed surfaces with Carboline 890 (Note 5).	NA
Operations	Nameplate	NITS	Non-code	S/S	Not required	NA	NA

Notes: 1) There are no known residuals on finished component surfaces.

- 2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.
- 3) Component nomenclature taken from drawings in Chapter 1.
- 4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.
- 5) Thermaline 450 and Carboline 890 were the product names at the time of initial licensing. Chemically identical products with different names are permitted. For example, Carboline 890 was re-named Carboguard 890 in 2000, with no change to the coating material and is, therefore, acceptable for use where Carboline 890 is specified.

Table 1.3.3 (cont'd)

MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

MPC (1, 2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Matl. (if dissimilar)
Helium Retention (Secondary Containment)	Shell	A	ASME Section III; Subsection NB	Alloy X ⁽⁵⁾	See Appendix 1.A	NA	NA
Helium Retention (Secondary Containment)	Baseplate	A	ASME Section III; Subsection NB	Alloy X	See Appendix 1.A	NA	NA
Helium Retention (Secondary Containment)	Lid (One-piece design and top portion of optional two- piece design)	A	ASME Section III; Subsection NB	Alloy X	See Appendix 1.A	NA	NA
Helium Retention (Secondary Containment)	Closure Ring	A	ASME Section III; Subsection NB	Alloy X	See Appendix 1.A	NA	NA
Helium Retention (Secondary Containment)	Port Cover Plates	A	ASME Section III; Subsection NB	Alloy X	See Appendix 1.A	NA	NA
Criticality Control	Basket Cell Plates	A	ASME Section III; Subsection NG; core support structures (NG-1121)	Alloy X	See Appendix 1.A	NA	NA
Criticality Control	Boral	A	Non-code	NA	NA	NA	Aluminum/SS
Shielding	Drain and Vent Shield Block	С	Non-code	Alloy X	See Appendix 1.A	NA	NA
Shielding	Plugs for Drilled Holes	NITS	Non-code	Alloy X	See Appendix 1.A	NA	NA

Notes: 1) There are no known residuals on finished component surfaces.

2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.

3) Component nomenclature taken from Bill of Materials in Chapter 1.

4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.

5) For details on Alloy X material, see Appendix 1.A.

Rev. 10

MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

MPC^(1,2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Matl. (if dissimilar)
Shielding	Bottom portion of optional two-piece MPC lid design	В	Non-code	Alloy X	See Appendix 1.A	NA	NA
Heat Transfer	Optional Heat Conduction Elements	В	Non-code	Aluminum; Alloy 1100	NA	Sandblast Specified Surfaces	Aluminum/SS
Structural Integrity	Upper Fuel Spacer Column	В	ASME Section III; Subsection NG (only for stress analysis)	Alloy X	See Appendix 1.A	NA	NA
Structural Integrity	Sheathing	Α	Non-code	Alloy X	See Appendix 1.A	Aluminum/SS	NA
Structural Integrity	Shims	NITS	Non-code	Alloy X	See Appendix 1.A	NA	NA
Structural Integrity	Basket Supports (Angled Plates)	А	ASME Section III; Subsection NG; internal structures (NG-1122)	Alloy X	See Appendix 1.A	NA	NA
Structural Form	Basket Supports (Flat Plates)	NITS	Non-code	Alloy X	See Appendix 1.A	NA	NA
Structural Integrity	Upper Fuel Spacer Bolt	NITS	Non-code	A193-B8	Per ASME Section II	NA	NA
Structural Integrity	Upper Fuel Spacer End Plate	В	Non-code	Alloy X	See Appendix 1.A	NA	NA

Notes: 1) There are no known residuals on finished component surfaces.

2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.

3) Component nomenclature taken from Bill of Materials in Chapter 1.

4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.

5) For details on Alloy X material, see Appendix I.A.

MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

MPC^(1,2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Matl. (if dissimilar)
Structural Integrity	Lower Fuel Spacer Column	В	ASME Section III; Subsection NG (only for stress analysis)	S/S	See Appendix 1.A	NA	NA
Structural Integrity	Lower Fuel Spacer End Plate	В	Non-code	Alloy X	See Appendix 1.A	NA	NA
Structural Integrity	Vent Shield Block Spacer	С	Non-code	Alloy X	See Appendix 1.A	NA	NA
Structural Integrity	Trojan MPC Spacer	В	Non-code	304 S/S	Per ASME Section II	NA	NA
Structural Integrity	Trojan Failed Fuel Can Spacer	В	ASME Section III, Subsection NF	304 or 304LN S/S	Per ASME Section	NA	NA
Operations	Vent and Drain Tube	C	Non-code	S/S	Per ASME Section II	Thread area surface hardened	NA
Operations	Vent & Drain Cap	C	Non-code	S/S	Per ASME Section II	NA	NA
Operations	Vent & Drain Cap Seal Washer	NITS	Non-code	Aluminum	NA	NA	Aluminum/SS
Operations	Vent & Drain Cap Seal Washer Bolt	NITS	Non-code	Aluminum	NA	NA	NA
Operations	Reducer	NITS	Non-code	Alloy X	See Appendix 1.A	NA	NA
Operations	Drain Line	NITS	Non-code	Alloy X	See Appendix 1.A	NA	NA

Notes: 1) There are no known residuals on finished component surfaces.

2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.

3) Component nomenclature taken from Bill of Materials in Chapter 1.

4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.

5) For details on Alloy X material, see Appendix 1.A.

MATERIALS AND COMPONENTS OF THE HI-STAR 100 SYSTEM

MPC^(1,2)

Primary Function	Component ⁽³⁾	Safety Class ⁽⁴⁾	Codes/Standards (as applicable to component)	Material	Strength (ksi)	Special Surface Finish/Coating	Contact Mati. (if dissimilar)
Operations	Damaged Fuel Container	C	ASME Section III; Subsection NG	Primarily 304 S/S	See Appendix 1.A	NA	NA
Operations	Trojan Failed Fuel Can	С	ASME Section III; Subsection NG	304 S/S	Per ASME Section II	NA	NA
Operations	Drain Line Guide Tube	NITS	Non-code	S/S	NA	NA	NA

- Notes: 1) There are no known residuals on finished component surfaces.
 - 2) All welding processes used in welding the components shall be qualified in accordance with the requirements of ASME Section IX. All welds shall be made using welders qualified in accordance with ASME Section IX. Weld material shall meet the requirements of ASME Section II and the applicable Subsection of ASME Section III. For parts beyond the purview of ASME Section III, compliance with Section IX and Section II of the Code shall be observed to the extent practicable.

3) Component nomenclature taken from Bill of Materials in Chapter 1.

4) A,B and C denote important to safety classifications as described in NUREG/CR-6407. NITS stands for Not Important To Safety.

5) For details on Alloy X material, see Appendix 1.A.

1.4 DRAWINGS

The following drawings provide sufficient detail to describe the HI-STAR 100 packaging.

The classification of all components important to safety in accordance with Regulatory Guide 7.10 and NUREG/CR-6407 is provided in Table 1.3.3. Operational information, such as bolt torque and pressure-relief specifications are provided in Chapters 7 and 8. The maximum weight of the package and the maximum weight of the contents is provided in Table 2.2.1.

Drawing Number/Sheet	Description	Rev.
3913	Licensing Drawing for HI-STAR 100 Overpack Assembly	5
3923	Licensing Drawing for MPC Enclosure Vessel	8
3925	Licensing Drawing for MPC-24E/EF Fuel Basket Assembly	4
3926	Licensing Drawing for MPC-24 Fuel Basket Assembly	5
3927	Licensing Drawing for MPC-32 Fuel Basket Assembly	6
3928	Licensing Drawing for MPC-68/68F/68FF Fuel Basket Assembly	4
5014-C1765 Sht 1/7†	HI-STAR 100 Impact Limiter	1
5014-C1765 Sht 2/7 [†]	HI-STAR 100 Bottom Impact Limiter	1
5014-C1765 Sht 3/7 [†]	HI-STAR 100 Top Impact Limiter	1
5014-C1765 Sht 4/7 [†]	HI-STAR 100 Top Impact Limiter	1
5014-C1765 Sht 5/7 [†]	HI-STAR 100 Top Impact Limiter Detail of Item #6	1
5014-C1765 Sht 6/7 [†]	HI-STAR 100 Impact Limiter Honeycomb Details	1
5014-C1765 Sht 7/7 [†]	HI-STAR 100 Bottom Impact Limiter	0
3930	HI-STAR 100 Assembly For Transport	1
4111	Licensing Drawing for Trojan MPC Spacer Ring	0
4119	Licensing Drawing for Holtec Damaged Fuel Container for Trojan Plant Fuel	1

The following HI-STAR 100 System design drawings are provided in this section

[†] These drawing titles include the term "CoC No. 9261, Appendix B." Rather than appending the drawings directly to the CoC, they are incorporated into the CoC by reference. The "Appendix B" will be removed from each drawing as part of its next normal revision.

Drawing Number/Sheet	Description	Rev.
4122	Licensing Drawing for Trojan FFC Spacer	0
PFFC-001	Failed Fuel Can Assembly	8
PFFC-002	Failed Fuel Can Shell and Lid Assembly	7

		8	77			6	5 4 1 3
	CL	IENT	GENERAL				LICENSING DRAWING PACKAC
D	PR	OJECT NO. 1	020 P.O.	NO.	N/A		GENERAL NOTES: 1. THE EQUIPMENT DESIGN DOCUMENTED IN THIS DRAWING PACKAGE HAS BEEN CONFIRMED BY HOLTEC INTERNATIONAL TO COMPLY WITH THE SAFETY ANALYSIS DESCRIBED IN THE SAFETY ANALYSIS REPORT.
	DR PA	AWING CKAGE I.D.	3913 тота	_ SHEE1	rs 9		 THE REVISION LEVEL OF EACH INDIVIDUAL SHEET IN THIS PACKAGE IS THE SAME AS THE REVISION LEVEL OF THIS COVER SHEET. A REVISION TO ANY SHEET(S) IN THIS PACKAGE REQUIRES UPDATING OF REVISION NUMBERS OF ALL SHEETS TO THE NEXT REVISION NUMBER. ALL CATEGORY A. B. AND C JOINTS (NB-3351) IN THE CONTAINMENT BOUNDARY SHALL BE FULL
		LICENSING	DRAWING PACKA	GE CO	NTENTS:		4. OVERPACK INNER DIAMETER SHALL BE VERIFIED TO BE AT LEAST 1/16" GREATER THAN THE MAXIMUM OUTER MPC SHELL DIAMETER BY THE USE OF AN APPROPRIATELY SIZED TEST GAGE. GRINDING OR MACHINING THE INNER OVERPACK SURFACES IS PERMITTED TO ENSURE ADEQUATE CLEARANCE.
	F	SHEET	DESCRIPTION]	5. THE ASME BOILER AND PRESSURE VESSEL CODE (ASME CODE), 1995 EDITION WITH ADDENDA THROUGH 1997 IS THE GOVERNING CODE FOR THE HI-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERNATIVES AS LISTED IN SAR TABLE 1.3.2 (TRANSPORTATION) AND FSAR TABLE 2.2.15 (STORAGE). THE OVERPACK IS CONSTRUCTED IN ACCORDANCE WITH ASME SECTION III, SUBSECTION NB FOR THE CONTAINMENT BOUNDARY, AND SUBSECTION NF FOR THE BALANCE OF THE STRUCTURE, AS DESCRIBED IN THE SAR (TRANSPORTATION) AND FSAR (STORAGE).
с		2 ELEVATION VII 3 DETAIL OF TO 4 DETAIL OF CL	EW PFLANGE AT 0° & 180° OSURE PLATE BOLT HOLE & BOL				6. ALL WELDS REQUIRE VISUAL EXAMINATION. ADDITIONAL NDE INSPECTIONS ARE NOTED ON THE DRAWING. NDE TECHNIQUES AND ACCEPTANCE CRITERIA ARE GOVERNED BY ASME SECTIONS V AND III, RESPECTIVELY AS CLARIFIED IN THE APPLICABLE SAFETY ANALYSIS REPORTS.
		5 TOP PLAN VII 6 MID-PLANE S 7 TEST VENT &	EW "D" "D". SECTION "E" "E" DRAIN PORT DETAILS				7. PRESSURE (CONTAINMENT) BOUNDARY COMPONENTS SHALL MEET THE MATERIAL ACCEPTANCE CRITERIA OF NB-2000, ALL SUPPORT STRUCTURE (NON-CONTAINMENT) MATERIAL SHALL MEET THE ACCEPTANCE CRITERIA OF NF-2000.
		8 DETAIL "H" 9 POCKET TRUM	NNION DETAIL				S. WELDING FROEDER SAND WELDER GORDINGENOND BOUNDARY WELDS AND SUBSECTION SECTION III (SUBSECTION NB FOR PRESSURE (CONTAINMENT) BOUNDARY WELDS AND SUBSECTION NF FOR NON-CONTAINMENT BOUNDARY WELDS). 9. POCKET TRUNNION PLUGS SHALL BE INSTALLED DURING TRANSPORT, IF POCKET TRUNNIONS ARE
-		<u></u>					10. THE NEUTRON SHIELD ENCLOSURE SHALL INCLUDE OVERPRESSURE PROTECTION. THE MAXIMUM SET PRESSURE OF THE PRESSURE RELIEF DEVICES IS 35 PSIG.
							11. ALL SEALING SURFACES SHALL BE STAINLESS STEEL OVERLAID.
			REVISION LOG	1			12. DRAWING IS NOT TO SCALE.
							BY THE DESIGNER. HOLES NOT USED FOR TRANSPORTATION SHALL BE PLUGGED WITH STEEL.
в	REV	AFFECTED DRAWING SHEET NUMBERS	SUMMARY OF CHANGES/ AFFECTED ECOs	PREP. BY	APPROVAL DATE	VIR# 1	14. DIMENSIONAL TOLERANCES ON THIS DRAWING ARE PROVIDED TO ENSURE THAT THE EQUIPMENT DESIGN IS CONSISTENT WITH THE SUPPORTING ANALYSES. HARDWARE IS FABRICATED IN ACCORDANCE WITH THE DESIGN DRAWINGS, WHICH MAY HAVE MORE RESTRICTIVE TOLERANCES, TO ENSURE COMPONENT FIT-UP. DO NOT USE WORST-CASE TOLERANCE STACK-UP FROM THIS DRAWING TO DETERMINE COMPONENT FIT-UP.
		INITIAL ISSUE	1020-41	D. Butler	05/28/02	63/07	15. ALL WELD SIZES ARE MINIMUMS. LARGER WELDS ARE PERMITTED, LOCAL AREAS OF UNDERSIZE WELDS ARE ACCEPTABLE WITHIN THE LIMITS SPECIFIED IN THE ASME CODE. AS APPLICABLE.
	$\begin{vmatrix} 1 \\ 2 \end{vmatrix}$	ALL SHEETS SHEETS 1, 2, & 6	1020-42	S. CAIN S. CAIN	11/18/02	16412	16. THIS COMPONENT IS CLASSIFIED AS ITS-A BASED ON THE HIGHEST CLASSIFICATION OF ANY SUBCOMPONENT. SUBCOMPONENT CLASSIFICATIONS ARE PROVIDED IN SAR TABLE 1.3.3 (TRANSPORTATION) AND FSAR TABLE 2.2.6 (STORAGE).
	3	SHEETS 1, 2, 5, 6 & 9	1020-45	T.F.O.	5/19/03	57621	17. HOLTITE-A NEUTRON SHIELDING MATERIAL HAS CRITICAL CHARACTERISTICS OF (ALL VALUES NOMINAL):
	4	SHEETS 1 & 9	1020-46	T.F.O.	7/10/03	78609	1 WT % B C, 6 WT % HYDROGEN, AND A SPECIFIC GRAVITY OF 1.68 GM/CC. OTHER CONSTITUENTS OF HOLTITE-A SHALL BE AS DESCRIBED IN HOLTEC PROPRIETARY REPORT HI-2002396, REV. 3.
	5	SHEETS 9	1020-47	T.F.O.	7/31/03	53374	18. OVERPACK INNER CAVITY TO BE COATED WITH THERMALINE 450 OR IDENTICAL SUBSTITUTE. OVERPACK EXTERNAL SURFACES (EXCEPT THREADED HOLES AND SEALING SURFACES) TO BE COATED WITH CARBOLINE 890 OR IDENTICAL SUBSTITUTE.
							19. FOR NON-CODE WELDS, THE PROVISIONS OF EITHER ASME IX OR AWS MAY BE FOLLOWED.
		_		ļ	.		20. TOLERANCES FOR THE THICKNESS OF ASME CODE OVERPACK MATERIAL ARE SPECIFIED IN ASME SECTION II.
A							21. REFER TO THE COMPONENT COMPLETION RECORD (CCR) FOR THE COMPLETE LIST OF APPROVED DESIGN DEVIATIONS FOR EACH INDIVIDUAL SERIAL NUMBER.
		† THE VALIDATION CONFIRMS THAT	DENTIFICATION RECORD (VIR) NUMBER IS ALL APPROPRIATE REVIEWS OF THIS DRAY	A COMPUTER GEN	I NERATED RANDOM N ITED IN COMPANY'S	UNBER WHICH NETWORK.	22. DIMENSIONS NOTED AS NOMINAL ("NOM.") IN THIS DRAWING ARE FOR INFORMATION ONLY, IN ORDER TO INDICATE THE GENERAL SIZE OF THE COMPONENT OR PART. NOMINAL DIMENSIONS HAVE NO SPECIFIC TOLERANCE, BUT ARE MET THROUGH FABRICATION IN ACCORDANCE WITH OTHER DIMENSIONS THAT ARE TOLERANCED AND INSPECTED. NOMINAL DIMENSIONS ARE NOT SPECIFICALLY VERIFIED DURING THE FABRICATION PROCESS.
	۱	8	7	1		6	5 4 3

(



	CLENT .		GENERAL			
HOLTEC	DESCRIPTION	1	HI-STAR 1	00		
 HOLTEC CENTER 555 LINCOLN DRIVE WEST MARLTON, NJ 08053	ĺ		OVERPAC ELEVATION	K VIEW		
	Ď	Damag	3913		2	5
	SCALE	NTS	FLE ANTH, C:\DRUMMU25\10201	3013		

HOLTEC NTERNATIONAL HOLTEC CENTER SSS LINCOLN DRIVE WEST MARLTON, NJ DB053		™ HI- DE1	GENERAL STAR 100 OVERI FAIL OF TOP FLA AT 0° & 180°	PACK		
Currently Diversion	Ď	Difference	3913	3 5		
	SCALE	NTS	PLE PATHE C:\DRAWHCS\1020\3813			

	SCALE	NTS MER	THE GLORAMONIC	1030/3013					
Provide Decompos	D 3913				sect 4	5			
OLTEC CENTER 55 LINCOLN DRIVE WEST WRILTON, NJ 08053		CLOSURE PLATE BOLT HOLE & BOLT							
NTERNATIONAL	DESCRIPTION	HI-STAR 100 OVERPACK							
	CUENT		GENER	AL					

Π		0.0017		GENERAL				
	HULILU NTERNATIONAL HOLTEC CENTER 555 LINCOLN DRIVE WEST WARLTON, NJ 08053	HI-STAR 100 OVERPACK						
ſ	ufron()n ()Aures()5	Ď	DRAMME	3913	жа 5	5		
ſ.		sou	NTS	FLT PERIN: C:\DAMAGES\1020\3013				





	naus	GENERAL										
INTERNATIONAL	BESCHARTER	u .	HI-STAR 100									
HOLTEC CENTER 555 LINCOLN DRIVE WEST MARLTON, NJ 08053												
CONFINITION DIVIDINGS	Đ	CPMBNING -	3913	8	5							
<u>l</u>	30412	NTS	FILE PATHE & (Distances) + (200), 39+3									

HOLTEC INTERNATIONAL HOLTEC CENTER 555 LINCOLN DRIVE WEST MARLTON, NJ 08053	GENERAL POCKET TRUNNION DETAIL								
COTATING AND	Ď	9	5						
<u> </u>	SCALC	NTS	PLE PATH: 6:\DAWNOS\1020\3013						

CLIENT GE	NERAL		ICENS	ING DRAW	ING	PAC	CKA
PROJECT NO. 5014	P.O. NO. N/A						
DRAWING 3923	TOTAL 5 SHEETS 5			REVISION LOG			
		REV	AFFECTED DRAWING SHEET NUMBERS	SUMMARY OF CHANGES/ AFFECTED ECOs	PREPARED	APPROVAL DATE	VIR# +
		0	INITIAL ISSUE	5014-57	S.CAIN	5/30/02	59761
LICENSING DRAW	NG PACKAGE CONTENTS:	1	SHEETS 2 & 4	5014-62	S.CAIN	6/17/02	13107
		2	ALL	5014-82, REV. 1	S.CAIN	8/1/02	97257
SHEET	DESCRIPTION	3	SHEETS 3 & 4	5014-71	S.CAIN	10/9/02	82001
1 COVER SHEET		4	SHEET 4	1021-50	S CAIN	11/26/02	28576
2 MPC ENCLOSURE VESSEL GE	NERAL ARRANGEMENT		SHEET A	5014.62 PEV 2		12/12/07/2	04724
3 MPC ENCLOSURE VESSEL EL	EVATION DETAILS	5			5.CAIN	12/13/02	
4 MPC ENCLOSURE VESSEL LIE	DETAILS	6	SHEETS 2 & 4	1/21-01, KEV. U	S.CAIN	2/12/03	57439
5 MPC FUEL SPACER DETAILS		1	ALL	5014-87	T.F.O.	5/19/03	30375
		8	SHEETS 1 & 3	5014-94	T.F.O.	7/10/03	26573
			+ THE VALIDATION IDEN	TFICATION RECORD (VIR) NUMBER IS A COMPUTE	R GENERATED RANDO	M NUMBER WHIC	н
			CONFIRMS THAT ALL AF	PROPRIATE REVIEWS OF THIS DRAWING ARE DOC	UMENTED IN COMPAN	IY'S NETWORK.	
		GEN	ERAL NOTES:		······································	<u></u>	
		1. Th	E EQUIPMENT DESIGN DO	CUMENTED IN THIS DRAWING PACKAGE	HAS BEEN CONFI	RMED BY HOLT	EC
		IN a pr		Y WITH THE SAFETY ANALYSES DESCRIBE	D IN THE SAFETY	ANALYSIS REF	PORT.
			ESIGN IS CONSISTENT WIT	H THE SUPPORTING ARE PROVIDED TO E	ARE IS FABRICATE	EQUIPMENT DINACCORDA	
		Fr	T-UP, DO NOT USE WORST	-CASE TOLERANCE STACK-UP FROM THI	S DRAWING TO DE	TERMINE CON	PONENT FIT-
		3. TI TI OI	IE REVISION LEVEL OF EA HIS COVER SHEET. A REVIS FALL SHEETS TO THE NE)	CH INDIVIDUAL SHEET IN THIS PACKAGE SION TO ANY SHEET(S) IN THIS PACKAGE IT REVISION NUMBER.	IS THE SAME AS T REQUIRES UPDAT	HE REVISION L ING OF REVISI	EVEL OF ON NUMBERS
		4. T	HE ASME BOILER AND PRE	SSURE VESSEL CODE (ASME CODE), 199			VED
		AL	TERNATIVES AS LISTED I	N SAR TABLE 1.3.2 (TRANSPORTATION) AN EL IS CONSTRUCTED IN ACCORDANCE W	ID FSAR TABLE 2.2	15 (STORAGE). ION
		NI FS	WITH CERTAIN APPROVE AR (STORAGE). NEW OR I	D CODE ALTERNATIVES AS DESCRIBED I REVISED ASME CODE ALTERNATIVES REC	N THE SAR (TRAN	SPORTATION)	AND FORE
		IM A = a	PLEMENTATION.				
			L MPC ENCLOSURE VESS LOY X IS ANY OF THE FO	EL STRUCTURAL MATERIALS ARE "ALLOY LLOWING STAINLESS STEEL TYPES: 316, 3 ITH ASME SECTION & DART A WEID MAT	X UNLESS OTHE 16LN, 304, AND 30	KWISE NOTED. 4LN. ALLOY X	
		P/ Ti	ART C. MPC ENCLOSURE V	ESSEL WALL (I.E. CYCLINDER SHELL) WIL	L BE FABRICATED	OF PIECES MA	DE FROM
		6.AI	L WELDS REQUIRE VISUA	LEXAMINATION (VT), ADDITIONAL NDF IN	SPECTIONS ARE	NOTED ON THE	DRAWING
<u></u>		A	S REQUIRED. NDE TECHNI ESPECTIVELY, AS CLARIFI	QUES AND ACCEPTANCE CRITERIA ARE C ED IN THE APPLICABLE SAFETY ANALYSIS	OVERNED BY ASM	E SECTIONS V	AND III,
		7. UI	NLESS OTHERWISE NOTE	D, FULL PENETRATION WELDS MAY BE MA	DE FROM EITHER	SIDE OF A CO	MPONENT.
		8. FL	JEL BASKET SUPPORTS A	RE ILLUSTRATIVE. ACTUAL FUEL BASKET	SUPPORT ARRAN	GEMENTS ARE	SHOWN
		ט ות פ	FEERENCES RETWEEN T	NOREI URWINGS. IF GENERIC MOC ENCLOSHEE VESSEL AN			
		J VI	ESSEL ARE SPECIFICALLY	NOTED (FOR PART 71 USE ONLY).			LUUUNE
		10.	ALL WELD SIZES ARE MINI ACCEPTABLE WITHIN THE	MUMS. LARGER WELDS ARE PERMITTED. LIMITS SPECIFIED IN THE ASME CODE. AS	LOCAL AREAS OF APPLICABLE.	UNDERSIZE W	ELDS ARE
	•						







	CLEHT	GENERAL								
HOLTEC INTERNATIONAL HOLTEC CENTER SSELINCOLN DRIVE WEST MARLTON, NJ DODS	CERCHAP	MPC ENCLOSURE VESSEL LID DETAILS								
3925, 3926, 3927, 3928	Đ	3923	4	8						
	N.M.E	NONE RUE ALTE GEDRAWING	30014\3823							

	GENERAL								
HOLTEC INTERNATIONAL MOLTEC CENTER SSS LINCOLN DRIVE WEST WARLTON, NJ 00053	FUEL SPACER DETAILS								
3925, 3926, 3927, 3928	B 3923 Exat NONE	BIGET ATV 5 8							

PROJECT ND. 1022 P.O. NO. N/A D DRAWING PACKAGE I.D. 3925 TOTAL SHEETS 4 REVISION LOC LICENSING DRAWING PACKAGE CONTENTS: Image: Control acett Imac		CLI	ENT	GF	ENERAL			L	ICEN	SING DRAV	VING	PAC	CKA
D DERAWING PACKAGE I.D. 3925 TOTAL SHEETS 4 LICENSING DRAWING PACKAGE CONTENTS: Intermediate and the state of the state o		PR		NO. 1022	P.O. NO. N	I/A				······································			
PACING C. LD. OTHER CONTENTS: LICENSING DRAWING PACKAGE CONTENTS: REV	D	DR	AWING	3925	TOTAL	4				REVISION LOG			
LICENSING DRAWING PACKAGE CONTENTS: SHEET DESCRIPTION 1 LL 2 DESCRIPTION 2 HERT 13.2 3 HERT 13.2 4 HERT 13.2 4 HERT 13.2 5 HERT 13.2 6 HERT 13.2 6 HERT 13.2 7 HUE ANART ARRADELT 2 HUE ANART ARRADELT 2 HUE ANART ARRADELT 2 HUE ANART ARRADELT 3 HERT 13.24 4 HERT 13.24 5 HERT 13.24 6 HERT 13.24 10 HUE ANART ARRADEL ONT DETAILS 11 HUE ANART ARRADEL ONT DETAILS 12 HUE ANART ARRADEL ONT DETAILS 13 HERT 13.24 14 HUE ANART ARRADEL ONT DETAILS 15 HUE ANART ARRADEL ONT DETAILS 16 HUE ANART ARRADEL ONT DETAILS 17 HUE ANART ARRADEL ONT DETAILS 18 HUE ANART ARRADEL ONT DETAILS <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>REV</td><td>AFFECTED DRAW</td><td>ING SUMMARY OF CHANGES/</td><td>PREPARED</td><td></td><td>VIR# +</td></t<>								REV	AFFECTED DRAW	ING SUMMARY OF CHANGES/	PREPARED		VIR# +
ILICENSING DRAWING PACKAGE CONTENTS: ILICENSING DRAWING PACKAGE CONTENTS: INAL AND CONTENT AND								0	INITIAL ISSUE	1022-38	SLC	5/30/02	17385
C 2 SHEET DESCRIPTION 1 COVER SHEET 3 SHEET 12.3.4 102.4 3 FUEL MARET ANAMAGEMENT 3 SHEET 13.4.4 102.4 4 FUEL MARET ANAMAGEMENT 3 SHEET 13.4.4 102.4 5 FUEL MARET ANAMAGEMENT 3 SHEET 13.4.4 102.4 6 FUEL MARET ANAMAGEMENT 3 SHEET 13.4.4 102.4 6 FUEL MARET ANAMAGEMENT 3 SHEET 13.4.4 102.4 1 FUEL MARET ANAMAGEMENT 1 SHEET 14.4 102.4 1 FUEL MARET BURNOW COMMAND ANAMAGEMENT SHEET 14.4 102.4 1 FUEL MARET BURNOW COMMAND ANAMAGEMENT SHEET 14.4 102.4 1 FUEL MARET BURNOW COMMAND ANAMAGEMENT SHEET 14.4 102.4 102.4 1 FUEL MARET DESIGN COMMAND ANAMAGEMENT SHEET 14.4 102.4 102.4 1 FUEL MARET DESIGN COMMAND ANAMAGEMENT SHEET 14.4 102.4 102.4 1 FUEL MARET DESIGN COMMAND ANAMAGEMENT SHEET 14.4 102.4 102.4 1 FUEL MARET DESIGN COMMAND ANAMAGEMENT SHEET 14.4 102.4 102.4			1	ICENSING DRAV	WING PACKAGE	CONTENTS:		1	ALL	1022-38, REV. 1	SLC	8/28/02	69345
SHEET DESCRIPTION 1 COMEN SHEET 2 Fuel MARCE ANTRAMEMENT 3 FUEL MARCE ANTRAMEMENT 4 Fuel MARCE ANTRAMEMENT 6 Fuel MARCE ANTRAMEMENT 6 Fuel MARCE ANTRAMEMENT 6 Fuel MARCE ANTRAMEMENT 6 Fuel MARCE ANTRAMEMENT 7 Fuel MARCE ANTRAMEMENT 8 Fuel MARCE ANTRAMEMENT 9 Fuel MARCE ANTRAMEMENT 1 Fuel MARCE ANTRAME			2					2	SHEETS 2 & 3	1022-30, REV. 0 & 5014-82, REV. 0	SLC	4/23/03	61822
SHEET DESURPTION 1 COVER SHEET 2 PALE MARKT RARMAGENENT 3 PALE MARKT RARMAGENENT 4 PALE MARKT RARMAGENENT BETMANN PALEXAMENT REPORT RARMAGENENT			[1		2		1022.48	SLC	5/19/03	40190
1 DUCK SHEET ALL 3 PRUE BANKET ANDANCE DETAILS 4 DRUE BANKET ANDARD MED DETAILS 4 DRUE BANKET ANDARD MED DETAILS 4 DRUE BANKET ANDARD DETAILS 4 DRUE BANKET ANDARD MED DETAILS 4 DRUE BANKET ANDARD DETAILS 5 DRUE BANKET ANDARD DETAILS 6 DRUE BANKET AND PRESSURE VESSEL CODER (ANDARD DETAILS ANDARD DETAILS ANDARD DETAILS AND PRESSURE VESSEL CODER (ANDARD DETAILS ANDARD DETAILS AND PRESSURE VESSEL CODER (ANDARD DETAILS AND PRESSURE VESSEL CODER (ANDARD DETAILS AND PRESSURE VESSEL CODER (ANDARD DETAILS ANDARD DETAILS AND PRESSURE VESSEL CODER (ANDARD ANDARD DETAILS AND PRESSURE			SHEET		DESCRIPTION			3		1022.64		7/15/03	24545
3 PUEL BASKET LAYOUT AND WELD DETAILS 4 FUEL BASKET LAYOUT AND WELD DETAILS 1 FUEL BASKET LAYOUT AND WELD DETAILS 1 THE ALBASKET LAYOUT DETAILS 1 THE ALBASKET AND DETERMANCE DETAILS DETAILS 1 THE ALBASKET AND			2	FUEL BASKET ARRANGEMEN	тт			4	SHEETS 1, 3, & 4	10-2201		// 1/03	34515
4 HEE BASKET SUPPORT DETAILS 0			3	FUEL BASKET LAYOUT AND W	VELD DETAILS								
C Image: Comparison of the comparison			4	FUEL BASKET SUPPORT DET	AILS								
B Image: Control of the second s	c									·			
B 						· · · ·							
B It Is VALENTRY DEPUTY REPORT AND ADDRESS AS OUT OF THE BALANNO AN COMPLEX It IS VALENT ALL AND CALL THE VALENT OF THE BALANNO AN COMPLEX It IS COMPLEX COMPLEX It IS COMPLEX <tr< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></tr<>													
B GENERAL NOTES: 1. THE EQUIPMENT DESIGN DOCUMENTED IN THIS BRAWING PACKAGE HAX TO COMPLY WITH THE SUBJECT ON THIS DRAWING PACKAGE HAX TO COMPLY WITH THE SUBJECT ON THIS DRAWING ARE PROVIDED TO ENSU WITH THE REPORT OF THE PACKAGES ON THIS DRAWING ARE PROVIDED TO ENSU WITH THE REPORT OF THE PACKAGES ON THIS DRAWING ARE PROVIDED TO ENSU WITH THE REPORT OF THE PACKAGES ON THIS DRAWING ARE PROVIDED TO ENSU WITH THE REPORT OF THE PACKAGES ON THIS DRAWING ARE PROVIDED TO ENSU WITH THE REPORT OF THE PACKAGES ON THIS DRAWING ARE PROVIDED TO ENSU WITH THE REPORT OF THE PACKAGES ON THIS DRAWING TO DETERMINE COMPONENT FTLJP FOOD THIS DRAWING TO DETERMINE COMPONENT FTLJP OF THE PROVIDED TO ENSU WITH THE REPORT OF THE PACKAGES ON THE PACKAGES ON THE CODE FOR THE H-STAR TOO SYSTEM. WITH CERTINATION OF DATION OF NUMBER. 8 CODE FOR THE H-STAR TOO SYSTEM. WITH CERTINATION OF DATION OF NUMBER. THE ASWRE PROVIDED TO ENSU WITH CERTINATION OF THE PACKAGES ON THE PACKAGES ON NUMPEOR THE STAR TOO SYSTEM. WITH CERTINATION OF DATION OF NUMPER. 9 CODE FOR THE H-STAR TOO SYSTEM. WITH CERTINATION OF NUMPEOR THE PACKAGES ON WITH THE REPOURSE ON THE ODDE SWITH THE REQUIREMENTS OF ASMRE SECTION II, PART CURRENCES DOWN THE STAR THE PACKAGES ON WITH THE REPOURSE ON THE ODDE SWITH THE REQUIREMENTS OF ASMRE SECTION II, PART CURRENCES DOWN THE STAR THE PACKAGES ON THE WITH REPOURSE ON THE ODDE SWITH THE REPORT TO THE PACKAGES ON THE PACKAGES ON THE PACKAGES ON THE PACKAGES ARE TAND THE SWITH AND AND THE SWITH AND AND AND AND SWITH THE REPORT TO THE PACKAGES ON THE WITH THE REPORT OF THE PACKAGES ON THE WITH AND AND AND AND AND SWITH THE REPORT TO THE PACKAGES ON THE WITH AND AND AND SWITH THE REPORT THE OTHER THAN AND AND AND AND SWITH THE REPORT THE OTHER THANK THE AND THE THAN AND AND SWITH THE REPORT AND AND AND AND AND AND SWITH THE REPORT AND AND									+ THE VALIDATION CONFIRMS THAT	I DENTIFICATION RECORD (VIR) NUMBER IS A COMPU ALL APPROPRIATE REVIEWS OF THIS DRAWING ARE DO	TER GENERATED RANDO	M NUMBER WHIC	н
General NOTES: Inte coursent to Ession Documented in this Drawing PackAge Has To COMPLY WITH THE SAFETY ANALYSES DESCRIBED IN THE SAFETY AN DIMENSIONAL TOLERANCES TO THIS DRAWING APE ROYOTED TO ENSURE OUTDED TO ENSURE COMPONENT FITUP: DIMENSIONAL TOLERANCES TO THIS DRAWING APE ROYOTED TO ENSURE OUTDED TO ENSURE COMPONENT FITUP: DIMENSIONAL TOLERANCES TO ENSURE COMPONENT FITUP: DIMENSIONAL STOLEMANCES TO ENSURE COMPONENT FITUP: DIMENSIONAL STOLEMANCES TO ENSURE COMPONENT FITUP: DIMENSION TO ANY SHEET IN THIS PACKAGE IS TO REVISION TO ANY SHEET IN THIS PACKAGE IS TO REVISION TO ANY SHEET IN THIS PACKAGE IS TO REVISION TO ANY SHEET IN THIS PACKAGE IS TO REVISION TO ANY SHEET IN THIS PACKAGE IS TO REVISION TO ANY SHEET IN THIS PACKAGE IS TO REVISION TO ANY SHEET IN THIS PACKAGE IS COME (ASME COOL) STIEM, WITH CHERALS AND PRESSURE VESSEL CODE (ASME COOL) STIEM, WITH SHE RESULTE REGULARE MARE THE COURSE OF THE MARE TO SHEET IN THIS PACKAGE IS CONSTRUE STIEME THE ASTRONOL BETORE THE MELLENANTIA THE ASME BOLIER AND PRESSURE VESSEL CODE (ASME COOL) STIEM, WITH THE REALTING OF INTERNATION OF ADDITIONAL NDE INSEE STIEME SEQURE PROVIDE AT TERM STIEME SEQURE PROVIDE SEQURE PROVIDE AT TERM STIEME THE RESULTEMENT ON OF ADDITIONAL NDE INSEE STIEME SEQURE PROVIDE SEQURE PROVIDE AT THEME INFORMATION ON A DOTIONAL NDE INSEE STIEME STIEMET THE RESULTEMENT ON AND ADDITIONAL NDE INSEE STIEME STIEMET THE RESULTEMENT ON A DOTIONAL NDE INSEE STIEME STIEMET TO THE RESULTEMENT ON AND ADDITIONAL NDE INSEE STIEME THE STIEMET THE STI							-						
B "TO COMPLY WITH THE SAFETY ANALYSES DESCRIED IN THE SAFETY AN C DIMENSIONAL TO LESSING COMPONENT FLY AN ACCOUNT AND THE SAFETY AND ACCOUNT AND A									IES: MENT DESIGN DOCHI	MENTED IN THIS DRAWING PACKAGE HAS BE	EN CONFIRMED BY	HOLTEC INTER	
B 2. DIMENSIONAL TO LERANCES ON THE DRAWING ARE PROVIDED TO FASH WITH THE SUPPORTING ANALYSIS. HARDWARE IS FARRICATED IN ACCOUNT WITH THE SUPPORTING TO DIFFERINE COMPONENT FLUP. B 3. THE REVISION LEVEL OF EACH INDIVIDUAL SHEET IN THIS PROCKAGE IS UNMEER. B 3. THE REVISION LEVEL OF EACH INDIVIDUAL SHEET IN THIS PROCKAGE IS UNMEER. B 3. THE REVISION LEVEL OF EACH INDIVIDUAL SHEET IN THIS PROCKAGE IS UNMEER. B 3. THE REVISION LEVEL OF EACH INDIVIDUAL SHEET IN THIS PROCKAGE IS UNMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH CERTAIN APPROVED ALTERN NUMEER. CODE FOR THE H-STAR 100 SYSTEM, WITH AFTOR THE EMELS NUMER SECTION THE STAR 100 SYSTEM, WITH CERTAIN APPROVED AT THE STAR NUMER SECTION THE STAR 100 SYSTEM, WITH AND THE STAR NUMER SECTION THE STAR 100 SYSTEM, WITH AND THE STAR NUMER SECTION THE STAR 100 SYSTEM, WITH AND THE STAR 100 SYSTEM, WITH APPROVED AT THE STAR NUMER SECTION THE STAR 100 SYSTEM, WIT	⊢ →							TO COMPLY	WITH THE SAFETY	NALYSES DESCRIBED IN THE SAFETY ANAL	YSIS REPORT.		
A HAVE MORE RESTRICTIVE TO LERANCES, 10 ENSURE COMPONENT FITUP. ROM THIS DRAWING COMPONENT FITUP. PROM THIS DRAWING COMPONENT FITUP.								2. DIMENSION WITH THE S	AL TOLERANCES ON UPPORTING ANALYS	THIS DRAWING ARE PROVIDED TO ENSURE SIS. HARDWARE IS FABRICATED IN ACCORDA	THAT THE EQUIPME NCE WITH THE DES	IGN DRAWING	CONSISTEN S, WHICH M
B 3. THE REVISION LEVEL OF EACH INDIVIDUAL SHEET IN THIS PACKAGE IS T B 3. THE REVISION TARY SHEET(S) IN THIS PACKAGE IS T NUMBER 4. THE RAYSION TARY SHEET(S) IN THIS PACKAGE IS TO NUMBER CODE FOR THE HI-STAR 100 SYSTEM. WITH CERTAIN APPROVED ALTERN A CODE FOR THE HI-STAR 100 SYSTEM. WITH CERTAIN APPROVED ALTERN A CODE FOR THE HI-STAR 100 SYSTEM. WITH CERTAIN APPROVED ALTERN A CODE FOR THE HI-STAR 100 SYSTEM. WITH CERTAIN APPROVED ALTERN A CODE FOR THE HI-STAR 100 SYSTEM. WITH CERTAIN APPROVED ALTERN A CODE FOR THE HI-STAR 100 SYSTEM. WITH CERTAIN APPROVED ALTERN A CODE FOR THE HI-STAR TARE SOLVED. B CODE FOR THE CERTAIN THE SACKAGE. A CODE FOR THE CERTAIN THE SACKAGE. B CODE FOR THE CERTAIN THE SACKAGE. C ALL MECD SACKET STRUCTURAL MATERNALS. C ALL WEDS REQUIRE VISUAL EXAMINATION (T), ADDITIONAL NDE INSPECTION IS ASKET STRUCTURAL MATERNALS. C ALL WEDS REQUIRE VISUAL EXAMINATION (T), ADDITIONAL NDE INSPECTION IS ASKET STRUCTURAL MATERNALS. C ALL WEDS REQUIRE VISUAL EXAMINATION (T), ADDITIONAL NDE INSPECTION IS ASKET STRUCTURAL MATERNALS. C ALL WEDS REQUIRE VISUAL EXAMINATION (T), ADDITIONAL NDE INSPECTION IS ASKET FORTS. C								FROM THIS	DRAWING TO DETER	RANCES, TO ENSURE COMPONENT FIT-UP. L MINE COMPONENT FIT-UP.	JO NOT USE WORST	-CASE TULERA	ANCE STACK
B NUMBER. A THE ASME BOILER AND PRESSURE VESSEL CODE (ASME CODE), 1995 ED CODE FOR THE LASTAR IDD SYSTEM, WITH CERTAIN APPROVED ALTERN AND FSART TABLE 22.15 ISTORED IN CAPPROVED ALTERN AND FSART TABLE 22.15 ISTORED IN CAPPROVAL BEFORE IN CAPPROVED ALTERN AND FSART TABLE 22.15 ISTORED IN CAPPROVAL BEFORE INCOMENTION OF THE PARELS INTO THE BAPACITY IN CAPPROVAL AND CAPPROVAL BEFORE INCOMENTION OF THE PARELS INTO THE MACHTER INCOMENT IN CAPPROVAL AND CAPPROVALIND AND CAPPROVALIND AND CAPPROVAL AND CAPPROVAL								3. THE REVISI REVISION	ON LEVEL OF EACH I	NDIVIDUAL SHEET IN THIS PACKAGE IS THE : THIS PACKAGE REQUIRES UPDATING OF REV	SAME AS THE REVIS	NON LEVEL OF	THIS COVER
B 4. THE ASME BOLER AND PRESSURE VESSEL CODE (ASME CODE). AT EACH AND PRESSURE VESSEL CODE (ASME CODE). AND PRESSURE VESSURE VESSEL CODE (ASME CODE). AND PRESSURE VESSEL CODE (ASME CODE).								NUMBER.					
B SUBSECTION INCLESS DISCRIPTION AND AS DESCRIPTION AS DESCRIPTION AND AS DESCRIPTION AS DESCRIP								4. THE ASME I CODE FOR	BOILER AND PRESSU THE HI-STAR 100 SYS	RE VESSEL CODE (ASME CODE), 1995 EDITIC STEM, WITH CERTAIN APPROVED ALTERNATI	IVES AS LISTED IN S	AR TABLE 1.3.2	(TRANSPOR
A 5. ALL MPC BASKET STRUCTURAL MATERIALS COMPLY WITH THE REQUIRE COMPLIES WITH THE REQUIREMENTS OF ASME SECTION II, PART C. 6. ALL MPC BASKET STRUCTURAL MATERIALS COMPLY WITH THE REQUIRE COMPLIES WITH THE REQUIREMENTS OF ASME SECTION II, PART C. 6. ALL WELDS REQUIRE VISUAL EXAMINATION (VT). ADDITIONAL NDE INSPE COMPLIES AND ACCEPTANCE CRITERIA ARE GOVERNED BY ASME SEC APPLICABLE SAFETY ANALYSIS REPORTS. 7. FABRICATOR MAY ADD WELDS TO STITCH WELDS AT THEIR DISCRETION 8. DO NOT MAKE A CONTINUOUS SEAL WELD BETWEEN THE SHEATHING AN 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 10. DIFFERENCES BETWEEN THE GENERIC MPC-24E/FF BASKET AND THE TI 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOW OCCURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL DIAMETER HOLE IN EACH PANEL INS BEEN ANALYZED AND FOUND TO DE ALONG ONCE DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL DIAMETER HOLE IN EACH PANEL HAS BEEN ANALYZED AND FOUND TO DE ALONG ONCE DEDGE OF A PANEL ONLY BEEN MANLYZED AND FOUND TO DE ALONG ONCE DEDGE OF A PANEL ONLY BEEN ANALYZED AND FOUND TO DE ALONG ONCE THAN EIGHT PANELS WITHIN THE TOP OR BOTTOM 27" OF TH C) THE DAMAGE DOES NOT EXTEND FARTHER THAN 1" FROM THE ED PANEL AND; D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CON 12 THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEFT CLASURE DIAMETER AND;	P							SUBSECTIC ALTERNATI	N NG AS DESCRIBED	D IN THE SAR (TRANSPORTATION) AND FSAR	(STORAGE). NEW O	R REVISED AS	ME CODE
A Complies with the Requirements of ASME SECtion II, PART C. 6 ALL WELDS REQUIRE VISUAL EXAMINATION (VT). ADDITIONAL NDE INSPECTION III, PART C. 7 FABRICATOR MAY ADD WELDS TO STITCH WELDS AT THEIR DISCRETION. 8 DO NOT MAKE A CONTINUOUS SEAL WELD BETWEEN THE SHEATHING AN 9 ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 10 DIFFERENCES BETWEEN THE GENERIC MPC-24E/EF BASKET AND THE T 11 BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS, HOWE 0CCURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL DIAMETER HOLE IN EACH PANEL HAS BEEN ANALYZED AND FOUND TO E ALONG ONE EDGE OF A PANEL ONLY IS PERMITTED IF: A								5. ALL MPC BA	SKET STRUCTURAL	MATERIALS COMPLY WITH THE REQUIREMEN	NTS OF ASME SECT	ION II, PART A.	WELD MATE
A 6. ALL WELDS REQUIRE VISUAL EXAMINATION (VI) ADDITIONAL NOE INSPECTIVE CARLES ARE OVERNED BY ASME SET APPLICABLE SAFETY ANALYSIS REPORTS. C 7. FABRICATOR MAY ADD WELDS TO STITCH WELDS AT THEIR DISCRETION. 8. DO NOT MAKE A CONTINUOUS SEAL WELD BETWEEN THE SHEATHING AM 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 10. DIFFERENCES BETWEEN THE GENERIC MPC-24E/EF BASKET AND THE T 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOWE 0CCURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOWE 0CCURING DURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOWE 0CCURING DURING DURING BURING BUSTALLATION OF THE PANELS INTO THE MPC FUEL 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOWE 0CCURING DURING DURING BURING BURING DURING TO THE MPC FUEL 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOWE 0CCURING DURING DURING BURING BURING DURING FOUND FOUND TO F 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOWE 0CCURING DURING DURING BURING DURING FOUND FOU					· · · · · · · · · · · · · · · · · · ·			COMPLIES	WITH THE REQUIRE				
A 7. FABRICATOR MAY ADD WELDS TO STITCH WELDS AT THEIR DISCRETION 8. DO NOT MAKE A CONTINUOUS SEAL WELD BETWEEN THE SHEATHING AN 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 10. DIFFERENCES BETWEEN THE GENERIC MPC-24E/EF BASKET AND THE T 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS, HOWE 0CCURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL DIAMETER HOLE IN EACH PANEL HAS BEEN ANALYZED AND FOUND TO I A B) THE DAMAGE DOES NOT EXCEED 5 SQ. IN. AND; B) THE DAMAGE DOES NOT EXCEED 5 SQ. IN. AND; D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CONI 12. THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLAND.									REQUIRE VISUAL EX S AND ACCEPTANCE E SAFETY ANALYSIS	CRITERIA ARE GOVERNED BY ASME SECTION E CRITERIA ARE GOVERNED BY ASME SECTION REPORTS	ONS V AND III, RESP	ECTIVELY, AS C	
A 8. DO NOT MAKE A CONTINUOUS SEAL WELD BETWEEN THE SHEATHING AN 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 10. DIFFERENCES BETWEEN THE GENERIC MPC-24E/EF BASKET AND THE T 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS, HOWE 0.CCURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL 11. BORAL PANEL AREA OF DAMAGE DOES NOT EXCEED 5 SQ. IN. AND; 12. DIFFERENCES IS LOCATED WITHIN THE TOP OR BOTTOM 27" OF TH 13. DI THE DAMAGE IS LOCATED WITHIN THE TOP OR BOTTOM 27" OF TH 13. DO NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CONI 14. THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA								7. FABRICATO	R MAY ADD WELDS 1	TO STITCH WELDS AT THEIR DISCRETION			
A 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE "ALLOY X" UNLESS OTHERWISE NOTE 316, 316LN, 304, AND 304LN. 9. ALL STRUCTURAL MATERIALS ARE SET AND THE MICH STOP AND FOUND TO FARTHER THAN 1" FROM THE FIGHT OF A PANEL AND; 9. NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CONID IN COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA						·		8. DO NOT MA	KE A CONTINUOUS S	EAL WELD BETWEEN THE SHEATHING AND	THE CELL WALL.		
A 310, 310LN, 304, AND SUGLN. A 10. DIFFERENCES BETWEEN THE GENERIC MPC-24E/EF BASKET AND THE T 11. BORAL PANELS ARE INTENDED TO HAVE NO SIGNIFICANT FLAWS. HOW: 0CCURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL DIAMETER HOLE IN EACH PANEL AND YOUND TO I ALONG ONE EDGE OF A PANEL ONLY IS PERMITTED IF: A				<u> </u>				9. ALL STRUC	TURAL MATERIALS A	RE "ALLOY X" UNLESS OTHERWISE NOTED. A	ALLOY X IS ANY OF 1	THE FOLLOWING	G STAINLES
A 11. BORAL PANELS ARE INTELECTION OF THE PANELS INTO THE WASHET FAMOLET IN OCCURING UNING INSTALLATION OF THE PANELS INTO THE WASHES INTO TO IN ALONG ONE EDGE OF A PANEL ONLY IS PERMITTED IF: A A) THE TOTAL AREA OF DAMAGE DOES NOT EXCEED 5 SQ. IN. AND; B) THE DAMAGE IS LOCATED WITHIN THE TOP OR BOTTOM 27" OF TH C) THE DAMAGE DOES NOT EXTEND FARTHER THAN 1" FROM THE ED D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CONT 12. THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA				+	14 - 19 - 19 - 19 - 19 - 19 - 19 - 19 -			SID, STOLN,	CES BETWEEN THE	GENERIC MPC-24F/EF RASKET AND THE TRO	JAN PLANT MPC-24F	EF BASKET AF	RE SPECIFIC
A OCCURING DURING INSTALLATION OF THE PANELS INTO THE MPC FUEL DIAMETER HOLE IN EACH PANEL HAS BEEN ANALYZED AND FOUND TO I A ALONG ONE EDGE OF A PANEL ONLY IS PERMITTED IF: A) A) THE TOTAL AREA OF DAMAGE DOES NOT EXCEED 5 SQ. IN. AND; B) THE DAMAGE IS LOCATED WITHIN THE TOP OR BOTTOM 27" OF TH C) C) D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CONT D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CONT 12 THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA		ł						11. BORAL PA	NELS ARE INTENDED	TO HAVE NO SIGNIFICANT FLAWS. HOWEVE	R, TO ACCOUNT FO	R MANUFACTU	IRING DEVIA
A A A A A A A A A A A A A A A A A A A								OCCURING	B DURING INSTALLAT	TON OF THE PANELS INTO THE MPC FUEL BA EL HAS BEEN ANALYZED AND FOUND TO BE	ASKET, BORAL DAMA ACCEPTABLE. ADDI	GE OF UP TO T TIONALLY, LOC	ALIZED BOR
A) THE TOTAL AREA OF DAMAGE DOES NOT EXCEED \$ SQ. IN. AND; B) THE DAMAGE IS LOCATED WITHIN THE TOP OR BOTTOM 27" OF TH C) THE DAMAGE DOES NOT EXTEND FARTHER THAN 1" FROM THE ED PANEL AND; D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CON 12 THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA													
C) THE DAMAGE DOES NOT EXTEND FARTHER THAN 1" FROM THE EL PANEL AND; D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS COM 12 THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA	A	ļ						B) THF	DAMAGE IS LOCATE	D WITHIN THE TOP OR BOTTOM 27" OF THE P	ANEL AND:		
D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CON 12 THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA			-			<u> </u>		C) THE	DAMAGE DOES NOT	EXTEND FARTHER THAN 1" FROM THE EDGE	OF THE		
D) NO MORE THAN EIGHT PANELS WITHIN A BASKET HAVE THIS CON 12 THIS COMPONENT IS CLASSIFIED AS ITS A BASED ON THE HIGHEST CLA								PAN	IEL AND;		1011		
								D) NO M	PONENT IS CLASSIFI	ED AS ITS-A BASED ON THE HIGHEST CLASS		UBCOMPONEN	IT. SUBCOM
CLASSIFICATIONS ARE PROVIDED IN SAR TABLE 1.3.3 (TRANSPORTATIO								CLASSIFIC	ATIONS ARE PROVID	DED IN SAR TABLE 1.3.3 (TRANSPORTATION)	AND FSAR TABLE 2.	2.6 (STORAGE).	











		CLIENT	8	7		6	· [5 ICENS				
			GE	INERAL			- L.				FAL	
(PROJECT	NO. 1022	P.O. NO.	N/A							
	D	DRAWING PACKAGE	3926	TOTAL SHEETS	4				REVISION LOG			
							REV	AFFECTED DRAWING SHEET NUMBERS	SUMMARY OF CHANGES/ AFFECTED ECOs	PREPARED	APPROVAL DATE	VIR# +
					CONTENTO.		0		1022-38	SLC	5/30/02	74916
		<u>L</u>	CENSING DRAW	ING PACKAGE (JUNIENIS:		1	ALL	1022-38, REV. 1	SLC	8/28/02	37611
	-						2	SHEET 1	1022-43, REV. 0	SLC	10/9/02	84086
		SHEET		DESCRIPTION			3	SHEET 3	5014-82, REV. 0	SLC	3/25/03	70654
		1	COVER SHEET				4	SHEETS 1, 2, & 4	1022-48	SLC	5/19/03	13075
		2	FUEL BASKET ARRANGEMENT				5	SHEETS 1, 3, & 4	1022-51	SLC	7/11/03	81349
		4	FUEL BASKET SUPPORTS									
	c											
					,,							
								THE VALIDATION IDENT	TFICATION RECORD (VIR) NUMBER IS A COMPUT PROPRIATE REVIEWS OF THIS DRAWING ARE DO	ER GENERATED RANDO	M NUMBER WHIC	ж
							CENERA	I NOTES				
(' ?						1. THE E	QUIPMENT DESIGN DOCU	MENTED IN THIS DRAWING PACKAGE H	AS BEEN CONFIRM	ED BY HOLTEC	INTERNATIO
								OMPLY WITH THE SAFETY	ANALYSES DESCRIBED IN THE SAFETY	ANALYSIS REPORT		
							2. DIMEN WITH HAVE	THE SUPPORTING ANALYS MORE RESTRICTIVE TOLE	SIS. HARDWARE IS FABRICATED IN ACCO RANCES. TO ENSURE COMPONENT FIT	ORDANCE WITH TH	E DESIGN DRA	WINGS, WHI
							FROM	THIS DRAWING TO DETER	RMINE COMPONENT FIT-UP.			
							3. THE R SHEET	EVISION LEVEL OF EACH I. A REVISION TO ANY SHI REVISION NUMBER	INDIVIDUAL SHEET IN THIS PACKAGE IS EET(S) IN THIS PACKAGE REQUIRES UP(THE SAME AS THE DATING OF REVISIO	N NUMBERS O	F ALL SHEET
	в						4. THE A	SME BOILER AND PRESSU	IRE VESSEL CODE (ASME CODE), 1995 E	DITION WITH ADDE	NDA THROUGH	4 1997, IS TH
							CODE AND F	FOR THE HI-STAR 100 SY SAR TABLE 2.2.15 (STORA	STEM, WITH CERTAIN APPROVED ALTER GE). THE MPC FUEL BASKET IS CONSTR	NATIVES AS LISTE	D IN SAR TABL	E 1.3.2 (TRAI
							ALTER	INATIVES REQUIRE PRIOF	NRC APPROVAL BEFORE IMPLEMENTA	TION.	VEW OR REVIS	ed asme co
							5. ALL M COMP	PC BASKET STRUCTURAL LIES WITH THE REQUIREN	MATERIALS COMPLY WITH THE REQUIR MENTS OF ASME SECTION II, PART C.	EMENTS OF ASME	SECTION II, PA	RT A. WELD
							6. ALL W	ELDS REQUIRE VISUAL E	(AMINATION (VT). ADDITIONAL NDE EXA ANCE CRITERIA ARE GOVERNED BY AS	MINATIONS ARE NO	TED ON THE D	RAWING, AS
		· · · · · · · · · · · · · · · · · · ·			<u>Na pone ana amin'ny sola dia mampia amin</u>		THE A	PPLICABLE SAFETY ANAL	YSIS REPORTS.			
							7. FABRI	CATOR MAY ADD WELDS	TO STITCH WELDS AT THEIR DISCRETIC	N .		
							8. DO NO	T MAKE A CONTINUOUS	SEAL WELD BETWEEN THE SHEATHING	AND THE CELL WAL	L.	
							9. ALL S TYPES	TRUCTURAL MATERIALS A 5: 316, 316 LN, 304, AND 30	RE "ALLOY X" UNLESS OTHERWISE NOT 4 LN.	ED. ALLOY X IS AN	Y of the foll	OWING STA
				······································			10. BOR	AL PANELS ARE INTENDED URING DURING INSTALLA	TO HAVE NO SIGNIFICANT FLAWS. HO	VEVER, TO ACCOU	NT FOR MANU	FACTURING
							A 1*1	DIAMETER HOLE IN EACH	PANEL HAS BEEN ANALYZED AND FOUN	D TO BE ACCEPTA	BLE.	
	A						CLAS	SIFICATIONS ARE PROVID	DED IN SAR TABLE 1.3.3 (TRANSPORTAT	ION) AND FSAR TAE	BLE 2.2.6 (STOP	AGE).
							12. FOR	HI-STAR STORAGE, ALUM	INUM HEAT CONDUCTION ELEMENTS AF	RE REQUIRED.		
7							(4) 13. DELE	TED.				
(
				······································		L		·····				
			8	7	I	6		5	T 4	1	3	









	CLIENT	GENERAL								
HOLTEC INTERNATIONAL HOLTEC CENTER BELINCOLN DRIVE WEST MARLTON, NJ 000053	MPC 24 FUEL BASKET SUPPORTS									
3923	Ď	3926	6 4							
		NONE	IDRAWINGB 1022	878						

	CLIENT	/ IERAL	<u>I</u>	0	L	ICENS	ING DRAW	ING	PAC	CKA	'G
	PROJECT NO. 1023	P.O. NO. N//	4								
D	DRAWING PACKAGE I.D. 3927	TOTAL SHEETS	4				REVISION LOG				13
		.l.,,,	·		REV	AFFECTED DRAWING	SUMMARY OF CHANGES/ AFFECTED ECOs	PREPARED	APPROVAL DATE	VIR# +	יי 14
					0	INITIAL ISSUE	1023-10	S.CAIN	5/28/02	94474	1
	LICENSING DRAW	ING PACKAGE CO	ONTENTS:		1	ALL	1023-10, REV. 1	S.CAIN	8/27/02	25046	11
					2	SHEET 3	1023-3. REV. 1	SCAIN	9/6/02	62849	1.
]		3	CULET 2 # 3	5014.82 REV 0 1023-13 REV 1 1023-14 REV 1		3/0/02	12016	- "
	SHEET	DESCRIPTION			5			S.CAIN	3/18/03	12910	- 17
	2 FUEL BASKET ARRANGEMENT				4	SHEET 3	5014-86, REV. 0	S.CAIN	3/25/03	46879	-
	3 FUEL BASKET LAYOUT AND WEL	LD DETAILS			5	ALL	1023-18, 1023-21	S.CAIN	5/19/03	60888	-
	4 FUEL BASKET SUPPORTS				6	ALL	1023-24	S.CAIN	7/11/03	67743	
c											
		······									
		· •····				• THE VALIDATION IDEN	TIFICATION RECORD (VIR) NUMBER IS A COMPUTER (SENERATED RANDO	NUMBER WHICH	н]
						CONFIRMS THAT ALL A	PPROPRIATE REVIEWS OF THIS DRAWING ARE DOCU!	ENTED IN COMPAN	Y'S NETWORK.		
								····			-
В					 THE EQUIP TO COMPL DIMENSION WITH THE HAVE MOR FROM THIS THE REVIS A REVISIO NUMBER. THE ASME CODE FOR AND FSAR SUBSECTIO ALTERNAT ALL MPC B COMPLIES 	MENT DESIGN DOCUMEN Y WITH THE SAFETY ANA VAL TOLERANCES ON THI SUPPORTING ANALYSIS. E RESTRICTIVE TOLERAN S DRAWING TO DETERMIN ION LEVEL OF EACH INDI IN TO ANY SHEET(S) IN THE BOILER AND PRESSURE THE HI-STAR 100 SYSTE TABLE 2.2.15 (STORAGE) ON NG AS DESCIBED IN T TVES REQUIRE PRIOR NR ASKET STRUCTURAL MA	NTED IN THIS DRAWING PACKAGE HAS BEEL LYSES DESCRIBED IN THE SAFETY ANALYS IS DRAWING ARE PROVIDED TO ENSURE TH HARDWARE IS FABRICATED IN ACCORDANI NCES, TO ENSURE COMPONENT FIT-UP. DO VE COMPONENT FIT-UP. WIDUAL SHEET IN THIS PACKAGE IS THE SA HIS PACKAGE REQUIRES UPDATING OF REV VESSEL CODE (ASME CODE), 1995 EDITION M, WITH CERTAIN APPROVED ALTERNATIVE , THE MPC FUEL BASKET IS CONSTRUCTED HE SAR (TRANSPORTATION) AND FSAR (ST IC APPROVAL BEFORE IMPLEMENTATION. TERIALS COMPLY WITH THE REQUIREMENT TS OF ASME SECTION II, PART C.	N CONFIRMED BY IS REPORT. IAT THE EQUIPM CE WITH THE DES NOT USE WORS ME AS THE REVI- ISION NUMBERS WITH ADDENDA IS AS LISTED IN 3 IN ACCORDANC ORAGE). NEW O S OF ASME SECT	HOLTEC INTE SIGN DESIGN IS SIGN DRAWING T-CASE TOLER SION LEVEL OF OF ALL SHEET THROUGH 199 SAR TABLE 1.3. E WITH ASME S R REVISED AS TON II, PART A	ERNATIONAL CONSISTEN 35, WHICH M/ CANCE STACK F THIS COVER TS TO THE NE 7, IS THE GO 2 (TRANSPOI SECTION III, ME CODE	IT AY K-UP R SHE EXT RI WERNI WTATI
			·		6. ALL WELD: NDE TECH	S REQUIRE VISUAL EXAM	IN OF ASME SECTION II, FART C. IINATION (VT). ADDITIONAL NDE INSPECTION CE CRITERIA ARE PROVIDED IN THE APPLIC	IS ARE NOTED O ABLE CODES AS	N THE DRAWIN CLARIFIED IN 1	NG AS REQUI	IRED. \BLE
-					SAFETY A	NALISIS REPURTS. OR MAY AND WEI DO TO 4	STITCH WELDS AT THEIR DISCRETION				
					8. DO NOT M	AKE A CONTINUOUS SEA	L WELD BETWEEN THE SHEATHING AND TH	E CELL WALL.			
					9. ALL STRUC	TURAL MATERIALS ARE	ALLOY X" UNLESS OTHERWISE NOTED. ALI	OY X IS ANY OF	THE FOLLOW	NG STAINLES	SS STE
					TYPES: 31	6, 316 LN, 304, AND 304 LN ANELS ARE INTENDED TO	N. NAVENO SIGNIFICANT FLAWS HOWEVER	TO ACCOUNT FO			
					OCCURIN A 1" DIAM	G DURING INSTALLATION ETER HOLE IN EACH PAN	N OF THE PANELS INTO THE MPC FUEL BASI IEL HAS BEEN ANALYZED AND FOUND TO B	E ACCEPTABLE.	AGE OF UP TO	THE EQUIVA	LENT
^					11. THIS CON CLASSIFI	PONENT IS CLASSIFIED	AS ITS-A BASED ON THE HIGHEST CLASSIFI IN SAR TABLE 1.3.3 (TRANSPORTATION) AN	CATION OF ANY	SUBCOMPONE 2.6 (STORAGE	NT. SUBCOM	IPONE
		· · · · · · · · · · · · · · · · · · ·			12 DELETED	l.			-	-	
		·····									
		······································									
	i i i				1						









		8		7			6		5		4		3	
	CLIENT		GEN	ERAL				L	ICENS	ING	DRAW	NG	PAC	CKA
	PROJEC	TN	O. 1021	P.O. NO.	N/A									
D	DRAWIN	G E I.I	_{D.} 3928	TOTAL SHEETS	4	ļ .				RE	VISION LOG			
ł								REV	AFFECTED DRAWING SHEET NUMBERS	SUM	MARY OF CHANGES/	PREPARED	APPROVAL DATE	VIR# +
								0	INITIAL ISSUE	1021-41		S.CAIN	5/28/02	67360
			CENSING DRAWIN	NG PACKAGE	E CONT	ENTS:		1	ALL	1021-41, REV. 1		S.CAIN	8/27/02	76629
								2	SHEETS 2 & 3	5014-82, REV. 0		S.CAIN	3/25/03	24329
	SHEE	T	D	ESCRIPTION	<u>.</u>			3	SHEETS 1, 2, 3, & 4	1021-52, 1021-54		S.CAIN	5/19/03	72132
	1		COVER SHEET					4	ALL	1021-56		S.CAIN	7/11/03	26801
	2		FUEL BASKET ARRANGEMENT											
	3		FUEL BASKET LAYOUT AND WELD	DETAILS										
	4		FUEL BASKET SUPPORTS											
			······								·····			
						_							· ·	
									THE VALIDATION IDEN CONFIRMS THAT ALL AND	TIFICATION RECOR	D (VIR) NUMBER IS A COMPUTER (WS OF THIS DRAWING ARE DOCUM	ENERATED RANDO	om number whi NY'S Network.	CH
								<u></u>						
┛			· · · · · · · · · · · · · · · · · · ·					GENERAL NO	DTES:					
7								1. THE EQUI TO COMPL	PMENT DESIGN DOCUME Y WITH THE SAFETY ANA	NTED IN THIS DE ALYSES DESCRIE	AWING PACKAGE HAS BEEN BED IN THE SAFETY ANALYS	I CONFIRMED B IS REPORT.	IY HOLTEC INT	ERNATIONAL
					· · · · ·			2. DIMENSIO	NAL TOLERANCES ON TH SUPPORTING ANALYSIS.	IS DRAWING AR	E PROVIDED TO ENSURE TH FABRICATED IN ACCORDANC	AT THE EQUIPM	MENT DESIGN I	IS CONSISTEN
				<u> </u>				HAVE MOR	RE RESTRICTIVE TOLERA S DRAWING TO DETERMIN	NCES, TO ENSU NE COMPONENT	RE COMPONENT FIT-UP. DO	NOT USE WOR	ST-CASE TOLE	RANCE STAC
								3. THE REVIS	SION LEVEL OF EACH IND	IVIDUAL SHEET	IN THIS PACKAGE IS THE SA	ME AS THE REV	SOF ALL SHEE	
				<u></u>				REVISION	NUMBER.					
3								4. THE ASME CODE FOR	BOILER AND PRESSURE THE HI-STAR 100 SYSTE	VESSEL CODE (EM, WITH CERTA	ASME CODE), 1995 EDITION IN APPROVED ALTERNATIVE	WITH ADDENDA S AS LISTED IN	SAR TABLE 1.	97, IS THE GO 3.2 (TRANSPO SECTION III
					· · · · ·			SUBSECTI AL TERNAT	ON NG AS DESCRIBED IN	THE SAR (TRAN	ISPORTATION) AND FSAR (S	TORAGE). NEW	OR REVISED	ASME CODE
								5. ALL MPC E	BASKET STRUCTURAL MA	TERIALS COMPL	Y WITH THE REQUIREMENT	S OF ASME		
								SECTION	I, PART A. WELD MATERIA			ASME SECTION	N II, PART C.	ING
								AS REQUI	RED. NDE TECHNIQUES A	ND ACCEPTANC	SAFETY ANALYSIS REPORT	BY ASME SEC	TIONS V AND I	ii,
								7. FABRICAT	OR MAY ADD WELDS TO	STITCH WELDS	AT THEIR DISCRETION.			
			······································					8. DO NOT M	AKE A CONTINUOUS SEA	L WELD BETWE	EN THE SHEATHING AND TH	E CELL WALL.		
								9. ALL STRU	CTURAL MATERIALS ARE	"ALLOY X" UNLE	SS OTHERWISE NOTED. ALL	OY X IS ANY OF	THE FOLLOW	ING
								10. BORAL P	ANELS ARE INTENDED TO	D HAVE NO SIGN	FICANT FLAWS. HOWEVER,	TO ACCOUNT P	OR MANUFAC	TURING
								DEVIATIO TO THE E	ONS OCCURING DURING I EQUIVALENT OF A 1" DIAM	INSTALLATION O	F THE PANELS INTO THE MP EACH PANEL HAS BEEN ANA	C FUEL BASKET	I, BORAL DAM	CEPTABLE.
								11. THIS COL	MPONENT IS CLASSIFIED	AS ITS-A BASED	ON THE HIGHEST CLASSIFIC	CATION OF ANY		ENT. 2.6 (STORAGE
A								A 12. DELETER						
				·· <u>·</u> ·····										
		+	·····											
	L	 Я	1	7		T	6		5	1	4		3	
				-			-		-					










		BILL OF MATER	IAL (FULL SCALE)
ITEM ND.	UTY.	MATERIAL	DESCRIPTION
1	N/A	ALUMINUM HONEYCOMB	<u>2E 24EEL 6</u>
2		SEE NOTE 4	1 3/8 THK. INNER STIFFENING CYLINDER
3	1	SEE NOTE 2	1/8 THK. ENELLISLRE END DESK
4	1	SEE NOTE 2	1/8 THK. ENCLOSIRE SIDE PLATE FOR B.I. OR 1/8 THK. ENCLOSURE SIDE PLATE FOR T.I.
5	1	SEE NOTE 2	L/8 THK. ANNULAR DISK
6	1	SEE NOTE 4	1 7/8 THK. QUTER STIFFENING CAL. FOR B.I. OR 1 7/8 THK. QUTER STIFF. CAL. FOR T.I.
7	1	SEE NOTE 4	I THK. X END PLATE/SHIELDING DISK FOR B.L. DR 1/4 THK. X END PLATE/SHIELDING DISK FOR T.I.
8	16	SEE NOTE 4	1/2 THK. DUTER OUSSET
9	16	SEE NOTE 4	1/2 THK. INNER GLOSSET
10	1	HOLTITE - A	2 1/2 THK. HELICIE-A NITH L VT. Z BAC (SEE KITE 15)
11	1	SEE NUTE 2	L/4 THK. DISK
12	16	288 -E £1-A2	1 3/4"- 8 LNC X 25 1/4 LG. SDICKET HEAD CAP SCREW WITH 2 5/16" LG. USEABLE THREAD AT END FOR B.I. DNLY; W/STD. WASHER (PLLS TEFLON GASKET(S)).
13	N/A	ALLIMINUM HONEYCOMB	SEE SHEET 6
14	1	SEE NOTE 4	I THK. PLATE
ITEM	15, 16,	17 & 18 DELETED	
19	1	A569 or A366	1/8 THK. CYLINDER
20	1	SEE NOTE 4	1 5/8 THK. PLATE FOR T.I.
21	40	SA-193-87	2 1/4 DIA. BAR FOR T.I.
22A	10	288-EQI-A2	11 - B UNC X 2 5/8 LG. SUCKET HEAD SCREW WITH 1 5/16" LG. THREADED END FOR T.T.; W/2 3/4 0.D. WASHER & 2 0.D. X 1 1/4 15, RIPPER BUSHING, (70 SHORE A DURAMETER)
228	10	28 8- 691-42	I - B UNC X 2 5/8 LG, SOCKET HEAD SCREW WITH I 5/16" LG. THREADED END FOR T.I.; W/2 3/4 0.0. WASHER.
23	N/A_	ALUMINUM HONEYCOMB	SEE SHEET 6
24	1	SEE NOTE 2	1/8 THK, DISK
25	1	SEE NOTE 2	1/18 THK. CILLINDER
26	1	SEE NOTE 2	1/8 THK. CYLINDER
27	N/A	ALUMINUM HONEYCOMB	SEE SHEET 6
28			N/A
29	8	SA-479-304	PER DET. DVG.C1765 SHT.2
30	8	SA516-70	support cap

LEGEND: T.I. = TOP IMPACT LIMITER AND B.I. = BOTTOM IMPACT LIMITER

Δ

NOTES: 1) HATCH LINES IN ALIMINUM HONEYCOMB SHOW ORIENTATION OF PRINCIPAL CRUSH AXES. 2) PLATE AND SHEET MATERIAL IS ASME SECTION II SA-240 TYPE 304 (UNS NO. S30400) STAINLESS STEEL 3) ALL DIMENSIONS ARE NOMINAL. 4) ALL SHEET (GAGED) METAL TO BE ASS9 AND ALL PLATE TO BE ASME SECTION II SA516-70. 5) UNLESS OTHERVISE NOTED ALL DIMENSIONING UNITS ARE INCHES. 6) HOLTITE-A IS A NEUTRON SHIELD MATERIAL WITH 1 WT NOM. 2 84C, 6.0 NOM. WT.Z H, AND A DENSITY OF 1.68 9/cm³ NOM.

7) CRITICAL DIMENSIONS (INDICATED BY AN "+") ARE ESSENTIALL TO ASSURE EQUIPMENT FUNCTIONALITY.

10V 19-50	t	REVISED AS SHOWN	2.000			
	0	FOR APPROVAL	C.Y. 2-18-99	8.G.J. 군(만원	8.6. 2-18-69	¥.G 2-1
GATE	REY .	(IESORIPTION	ARP.	DEDED	P.H.	0
			871	9Y1	APPR	OVE
		REV]	SID	1		

1	
THIS DRAVING IS AN INTELLECTUAL PROPERTY OF FOLTE	ic
THE EXPLICIT CONDITION THAY IT VILL NOT HE DUPLIC IN VHOLE OF IN PART, OR CIVEN TO OTHERS WITHOUT V	IVZK AVED RUTTEN
CONSERT OF HOLIEC INTERNATIONAL.	IGN
ANALYSIS	
INTERNATIONAL CONSULTING	
DESCRIPTION HI-STAR 100 IMPACT LIMITER	
CLIENT VARIOUS	
COMPANION DRAWINGS	REV.
PROJECT No.5014 DRAWING No.	
e\DRAVINGS\5014\1MPACT\C1755-	1.RL

HOLTEC INTERNATIONAL CONSULTING	IGN
DESCRIPTION	
HI-STAR 100 BUTTOM IMPACT LIMITER Coc No. 9261, APPENDIX B	
CLIENT	
VARIOUS	
COMPANION DRAWINGS	REV.
PROJECT No.5014 DRAWING No.	
P.O. No. VARIOUS C1765 SHT. :	2 0F 7

HOLTEC	EQUIPMENT DESIGN ANALYSIS CONSULTING
DESCRIPTION	
HI-STAR 100 TOP	IMPACT LIMITER
CoC No. 9261,	APPENDIX B
CLIENT	
VARID	2L
COMPANION DRAWINGS	REV.
PROJECT No.5014	DRAWING No.
P.O. No. VARIOUS	С1765 <u>знг. э ся 7</u>

HOLTEC	QUIFMENT DES ANALYSIS CONSULTING	IGN
DESCRIPTION		
HI-STAR 100 TOP I	MPACT LIMITE	R
CoC No.9261, A	PPENDIX B	_
CLIENT	······	
VARIOUS	5	
COMPANION DRAVINGS		REV. 1
PROJECT No .5014	DRAWING No.	
P.O. No. VARIOUS	C1765 SH.	4 DF 7



HOLTEC INTERNATIONAL	QUIPMENT DES ANALYSIS CONSULTING	IGN				
DESCRIPTION HI-STAR 100 IMPACT LIMITER HONEYCOMB DETAILS Coc. No. 9261, APPENDIX B						
CLIENT	CLIENT					
VARIOUS	:					
COMPANION DRAVINGS REV.						
PROJECT No.5014 P.O. No. VARIDUS	DRAWING No. C1765 SHT.	6 DF 7				

	8 7 6	5 4 3
	CLIENT GENERAL	LICENSING DRAWING PACKAG
D	PROJECT NO. 1020 P.O. NO. N/A	REVISION LOG
	DRAWING PACKAGE I.D. 3930 TOTAL SHEETS 3 A	REV. AFFECTED DRAWING SHEET NUMBERS SUMMARY OF CHANGES/ AFFECTED ECOs PREP. BY APPROVAL DATE VIR# 1 0 INITIAL ISSUE 1020-41 T.F.O. 5/28/02 32824
	LICENSING DRAWING PACKAGE CONTENTS:	
	SHEET DESCRIPTION	
С	1 COVER SHEET 2 HI-STAR 100 ASSEMBLY FOR TRANSPORT 3 HI-STAR 100 TRANSPORT CONFIGURATION	
		THE VALUATION IDENTIFICATION RECORD (VIR) NUMBER IS A COMPUTER GENERATED RANDOM NUMBER VEICH CONTRIGET THAT ALL ARREPORTING OF THIS RANDOW FOR THIS ROTATION ROUTING AND THE RECORD
		GENERAL NOTES:
В		1. THE EQUIPMENT DESIGN DOCUMENTED IN THIS DRAWING PACKAGE HAS BEEN CONFIRMED BY HOLTEC INTERNATIONAL TO COMPLY WITH THE SAFETY ANALYSES AS DESCRIBED IN THE SAFETY ANALYSIS REPORT.
		2. ALL DIMENSIONS ARE NOMINAL EXCEPT AS NOTED. NOMINAL DIMENSIONS HAVE NO SPECIFIC TOLERANCE, BUT ARE MET THROUGH FABRICATION IN ACCORDANCE WITH OTHER DIMENSIONS THAT ARE TOLERANCED AND INSPECTED. NOMINAL DIMENSIONS ARE NOT SPECIFICALLY VERIFIED DURING THE FABRICATION PROCESS.
		3. THE REVISION LEVEL OF EACH INDIVIDUAL SHEET IN THIS PACKAGE IS THE SAME AS THE REVISION LEVEL OF THIS COVER SHEET. A REVISION TO ANY SHEET(S) IN THIS PACKAGE REQUIRES UPDATING OF REVISION NUMBERS OF ALL SHEETS TO THE NEXT REVISION NUMBER.
		- · · ·
A		
	8 7 6	5 4 3



	CUIDIT		GENERAL		
HOLE C INTERNATIONAL HOLTEC GENTER 555 LINCOLN DRIVE WEST MARLTON, NJ 08053	HI-STAR 100 ASSEMBLY FOR TRANSPORT				
CONF-ANDIO DRAMMES	D	-	3930	2	1
	SONT 1	NTS .	THE PATTE C:\Disamers\1028\3830		



	CLII	ENT	POI GENER/	RTLAND AL ELECTRIC		L	ICENS	ING DRAW	/IN
	PRO	DJECT N	IO. 1135	P.O. NO.	N/A				
D	DR/ PAC	AWING CKAGE I	. _{D.} 4111	TOTAL SHEETS	2			REVISION LOG	
						REV	AFFECTED DRAWING SHEET NUMBERS	SUMMARY OF CHANGES/ AFFECTED ECOs	PR
		ב	DESIGN DRAWIN	<u>G PACKAGE CC</u>	ONTENTS:	0	INITIAL ISSUE	ECO-1135-39	
-		SHEET		DESCRIPTION					
		3HEET	COVER SHEET						
		2	ASSEMBLY DRAWING	AND DETAILS					
								· · · · · · · · · · · · · · · · · · ·	
				······································			• THE VALIDATION IDEN	TIFICATION RECORD (VIR) NUMBER IS A COMPUT	ER GENERAT
							CONFIRMS THAT ALL A	PPROPRIATE REVIEWS OF THIS DRAWING ARE DO	CUMENTED I
ר									
						GENERAL	NOTES:		
						1. THE EQU	JIPMENT DESIGN DOCUM	ENTED IN THIS DRAWING PACKAGE HAS	SEEN CON
						I I COM	PLY WITH THE SAFETY AN	IALYSIS DESCRIBED IN THE SAFETY ANA	YSIS REP
						2. THE REV SHEET.	PLY WITH THE SAFETY AN /ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT DEVISION NUMBER	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA	LYSIS REP E SAME AS TING OF R
					······	2. THE REV SHEET. TO THE I 3. THE ASM	PLY WITH THE SAFETY AN /ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. /IE BOILER AND PRESSUR	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT	YSIS REP E SAME AS TING OF R
в					······	2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS COMS	PLY WITH THE SAFETY AN IISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. IE BOILER AND PRESSUR IING CODE FOR THE HI-ST 3.2 (TRANSPORTATION) / THI CTED IN ACCORDAN	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT FAR 100 SYSTEM, WITH CERTAIN APPROV IND FSAR TABLE 2.2.15 (STORAGE), THE I SE WITH ASME SECTION III SUBSECTION	LYSIS REP E SAME AS TING OF R TON WITH 'ED ALTER HOLTEC TI NG AS DE
в					······	2. THE REV SHEET. TO THE I 3. THE ASK GOVERN TABLE 1. IS CONS REVISED	PLY WITH THE SAFETY AN /ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. //E BOILER AND PRESSUR IING CODE FOR THE HI-SI 3.2 (TRANSPORTATION) / TRUCTED IN ACCORDANC D ASME CODE ALTERNATI	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT IAR 100 SYSTEM, WITH CERTAIN APPROV IND FSAR TABLE 2.2.15 (STORAGE). THE I CE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE	LYSIS REP E SAME AS TING OF R TON WITH TON WITH
в					· · · · · · · · · · · · · · · · · · ·	2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE THE S	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ING CODE FOR THE HI-SI 3.2 (TRANSPORTATION) TRUCTED IN ACCORDANC ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTA SAFETY ANALYSIS REPOR	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT IAR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE). THE I CE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT), ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME	LYSIS REP E SAME AS TING OF R TON WITH TED ALTER HOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS
В					· · · · · · · · · · · · · · · · · · ·	2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ING CODE FOR THE HI-SI 3.2 (TRANSPORTATION) ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL SAFETY ANALYSIS REPOR MPONENT IS CLASSIFIED	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT TAR 100 SYSTEM, WITH CERTAIN APPROV WD FSAR TABLE 2.2.15 (STORAGE), THE CE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B.	LYSIS REP E SAME AS TING OF R TON WITH ED ALTER HOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS
в						2. THE REV SHEET. TO THE I 3. THE ASK GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIM	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ME BOILER AND PRESSUR ING CODE FOR THE HI-ST 3.2 (TRANSPORTATION) / TRUCTED IN ACCORDANC D ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTA AFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT (AR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE). THE I SE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL WE CODE. AS APPLICABLE	YSIS REP E SAME AS TING OF R TON WITH ED ALTER HOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS AREAS O
B						2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ME BOILER AND PRESSUR JING CODE FOR THE HI-ST 3.2 (TRANSPORTATION) TRUCTED IN ACCORDANC D ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL SAFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT FAR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE). THE I CE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL WE CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT	YSIS REP E SAME AS TING OF R TON WITH 'ED ALTER FOR IMPI TIONS AR SECTIONS AREAS OF
В						2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ING CODE FOR THE HI-SI 3.2 (TRANSPORTATION / TRUCTED IN ACCORDANC ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTA CHNIQUES AND ACCEPTA CHNIQUES AND ACCEPTA CHNIQUES AND ACCEPTA CHNIQUES AND ACCEPTA IS SPECIFIED IN THE ASI ITS SPECIFIED IN THE ASI INCES ON THICKNESS OF TERIAL IS TYPE 304 STAIN	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT TAR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE). THE I SE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL ME CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL.	YSIS REP E SAME AS TING OF R TON WITH ED ALTER FORE IMPI TIONS AR SECTIONS AREAS OF
В						2. THE REV SHEET. TO THE I 3. THE ARM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9. DIMENSI SIZE OF	PLY WITH THE SAFETY AN VISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. WE BOILER AND PRESSUR UNG CODE FOR THE HIS 3.2 (TRANSPORTATION) A SME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL AFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAIN ONS NOTED AS NOMINAL THE COMPONENT OR PAR	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT (AR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE). THE I SE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL ME CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ('NOM.') IN THIS DRAWING ARE FOR INFO	YSIS REP E SAME AS TING OF R TON WITH ED ALTER FOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS AREAS OF TION II.
В						2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9. DIMENSII SIZE OF IN ACCO VERIFIED	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ME BOILER AND PRESSUR JING CODE FOR THE HI-ST 3.2 (TRANSPORTATION) TRUCTED IN ACCORDANC D ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL SAFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAINA ONS NOTED AS NOMINAL THE COMPONENT OR PAR RDANCE WITH OTHER DIM D DURING THE FABRICATION	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA TAR 100 SYSTEM, WITH CERTAIN APPROV IND FSAR TABLE 2.2.15 (STORAGE). THE I CE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL ME CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ("NOM.") IN THIS DRAWING ARE FOR INFO AT. NOMINAL DIMENSIONS HAVE NO SPE MENSIONS THAT ARE TOLERANCED AND I ON PROCESS.	YSIS REP E SAME AS TING OF R TON WITH ED ALTER FORE IMPI TIONS AR SECTIONS AREAS OF ION II.
В						2. THE REV SHEET. TO THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9.DIMENSII SIZE OF IN ACCOL VERIFIED	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ING CODE FOR THE HI-SI 3.2 (TRANSPORTATION / TRUCTED IN ACCORDANC ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL CHNIQUES AND ACCEPTAL CHNIQUES AND ACCEPTAL CHNIQUES AND ACCEPTAL IS SPECIFIED IN THE ASI MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAINE ONS NOTED AS NOMINAL THE COMPONENT OR PAF ROANCE WITH OTHER DIN D DURING THE FABRICATION	IALYSIS DESCRIBED IN THE SAFETY ANAL DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT (AR 100 SYSTEM, WITH CERTAIN APPROV IND FSAR TABLE 2.2.15 (STORAGE). THE I DE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL WE CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ('NOM.') IN THIS DRAWING ARE FOR INFO (T. NOMINAL DIMENSIONS HAVE NO SPE MENSIONS THAT ARE TOLERANCED AND I ON PROCESS.	AREAS OF TION II.
B 						2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9. DIMENSI SIZE OF IN ACCOL VERIFIED	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ING CODE FOR THE HISI 3.2 (TRANSPORTATION) / TRUCTED IN ACCORDANC D ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL SAFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAIN ONS NOTED AS NOMINAL THE COMPONENT OR PAF RDANCE WITH OTHER DIN D DURING THE FABRICATH	IALYSIS DESCRIBED IN THE SAFETY ANAL DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT (AR 100 SYSTEM, WITH CERTAIN APPROV IND FSAR TABLE 2.2.15 (STORAGE). THE I SE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL ME CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ('NOM.') IN THIS DRAWING ARE FOR INFO T. NOMINAL DIMENSIONS HAVE NO SPE MENSIONS THAT ARE TOLERANCED AND I ON PROCESS.	YSIS REP E SAME AS TING OF R TON WITH ED ALTER FOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS AREAS OF ION II.
B 						2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9. DIMENSII SIZE OF IN ACCOL VERIFIED	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ME BOILER AND PRESSUR JING CODE FOR THE HI-SI 3.2 (TRANSPORTATION) TRUCTED IN ACCORDANC D ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL SAFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAINA ONS NOTED AS NOMINAL THE COMPONENT OR PAR RDANCE WITH OTHER DIM D DURING THE FABRICATION	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT FAR 100 SYSTEM, WITH CERTAIN APPROV IND FSAR TABLE 2.2.15 (STORAGE). THE I CE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL ME CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ("NOM.") IN THIS DRAWING ARE FOR INFO AT NOMINAL DIMENSIONS HAVE NO SPE MENSIONS THAT ARE TOLERANCED AND I ON PROCESS.	YSIS REP E SAME AS TING OF R TON WITH ED ALTER FORE IMPI TIONS AR SECTIONS AREAS OF ION II.
B 						2. THE REV SHEET. TO THE I 3. THE ASK GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9. DIMENSI SIZE OF IN ACCOL VERIFIED	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ME BOILER AND PRESSUR ING CODE FOR THE HI-ST 3.2 (TRANSPORTATION) TRUCTED IN ACCORDANC D ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL AFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAINE ONS NOTED AS NOMINAL THE COMPONENT OR PAF RDANCE WITH OTHER DIM D DURING THE FABRICATH	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT (AR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE), THE I SE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL ME CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ('NOM.'') IN THIS DRAWING ARE FOR INFO AT NOMINAL DIMENSIONS HAVE NO SPE MENSIONS THAT ARE TOLERANCED AND I ON PROCESS.	YSIS REP E SAME AS TING OF R TON WITH TON WITH TON TED ALTER FOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS AREAS OF ION II.
B 						2. THE REV SHEET. TO THE I 3. THE ASM GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9.DIMENSII SIZE OF IN ACCO VERIFIED	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ME BOILER AND PRESSUR JING CODE FOR THE HI-ST 3.2 (TRANSPORTATION) TRUCTED IN ACCORDANC DASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL AFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAIN ONS NOTED AS NOMINAL THE COMPONENT OR PAR RDANCE WITH OTHER DIM D DURING THE FABRICATION	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT FAR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE). THE I CE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL WE CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ("NOM.") IN THIS DRAWING ARE FOR INFO T. NOMINAL DIMENSIONS HAVE NO SPE MENSIONS THAT ARE TOLERANCED AND I ON PROCESS.	YSIS REP E SAME AS TING OF R TON WITH TON WITH TON WITH TON ALTER TOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS AREAS OF ION II.
B 						2. THE REV SHEET. TO THE I 3. THE ASK GOVERN TABLE 1. IS CONS REVISED 4. ALL WEL NDE TEC IN THE S 5. THIS CO 6. ALL WEL THE LIMI 7. TOLERA 8. ALL MAT 9. DIMENSII SIZE OF IN ACCOL VERIFIED	PLY WITH THE SAFETY AN ISION LEVEL OF EACH IN A REVISION TO ANY SHE NEXT REVISION NUMBER. ME BOILER AND PRESSUR ING CODE FOR THE HI-SI 3.2 (TRANSPORTATION)/ TRUCTED IN ACCORDANC D ASME CODE ALTERNATI DS REQUIRE VISUAL EXA CHNIQUES AND ACCEPTAL AFETY ANALYSIS REPOR MPONENT IS CLASSIFIED D SIZES ARE MINIMUMS. ITS SPECIFIED IN THE ASI NCES ON THICKNESS OF TERIAL IS TYPE 304 STAINE ONS NOTED AS NOMINAL THE COMPONENT OR PAF RDANCE WITH OTHER DIM D DURING THE FABRICATH	IALYSIS DESCRIBED IN THE SAFETY ANA DIVIDUAL SHEET IN THIS PACKAGE IS TH ET(S) IN THIS PACKAGE REQUIRES UPDA E VESSEL CODE (ASME CODE), 1995 EDIT (AR 100 SYSTEM, WITH CERTAIN APPROV ND FSAR TABLE 2.2.15 (STORAGE), THE I SE WITH ASME SECTION III, SUBSECTION VES REQUIRE PRIOR NRC APPROVAL BE MINATION (VT). ADDITIONAL NDE INSPEC BLE CRITERIA ARE GOVERNED BY ASME T. AS ITS-B. LARGER WELDS ARE PERMITTED. LOCAL ME CODE, AS APPLICABLE. MATERIAL ARE SPECIFIED IN ASME SECT LESS STEEL. ('NOM.'') IN THIS DRAWING ARE FOR INFO ATSIONS THAT ARE TOLERANCED AND I ON PROCESS.	YSIS REP E SAME AS TING OF R TON WITH FO ATER FOLTEC TI NG AS DE FORE IMPI TIONS AR SECTIONS AREAS OF ION II.

3

3

REV	AFFECTED DRAWING SHEET NUMBERS	SUMMARY OF CHANGES/ AFFECTED ECOs	PREPARED BY	APPROVAL DATE	VIR# +
0	INITIAL ISSUE	ECO-1135-39	T.F.O.	5/19/03	42662
		· · · · · · · · · · · · · · · · · · ·			
		· · · · · · · · · · · · · · · · · · ·			
	THE VALIDATION IDENTI CONFIRMS THAT ALL APP	FICATION RECORD (VIR) NUMBER IS A COMPUTER GEN ROPRIATE REVIEWS OF THIS DRAWING ARE DOCUMEN	ERATED RANDO	M NUMBER WHI Y'S NETWORK.	СН

- NFIRMED BY HOLTEC INTERNATIONAL PORT.
- S THE REVISION LEVEL OF THIS COVER REVISION NUMBERS OF ALL SHEETS
- H ADDENDA THROUGH 1997 IS THE RNATIVES AS LISTED IN SAR TROJAN MPC-24E/EF SPACER RING ESCRIBED IN THE SAR. NEW OR PLEMENTATION.
- RE NOTED ON THE DRAWING AS REQUIRED. S V AND III, RESPECTIVELY, AS CLARIFIED
- F UNDERSIZE WELDS ARE ACCEPTABLE WITH

N ONLY, IN ORDER TO INDICATE THE GENERAL LERANCE, BUT ARE MET THROUGH FABRICATION ED. NOMINAL DIMENSIONS ARE NOT SPECIFICALLY





	8	7	6		5		4	3	
CLIENT	PORTLAN	ID GENERAL C COMPANY		L	ICENS	ING DR	RAWING	PAC	CKA
PROJECT	NO. 1135	P.O. NO. NQ182902				·			
DRAWING	. 4119	TOTAL 4				REVISI	ION LOG		
				REV	AFFECTED DRAWING	SUMMARY OF CH	ANGES/ PREPARE	ED APPROVAL	VIR#+
				0	INITIAL ISSUE	1135-39	SLC	5/19/03	92815
	LICENSING DRAW	ING PACKAGE CONTEN	r <u>S:</u>	1	SHEET 3	1135-40	SLC	7/10/03	77740
-									
SHEE	r	DESCRIPTION							
1	COVER SHEET								
2	ASSEMBLY DRAWING AND BILL O	DF MATERIALS							
	LID AND LOCKING ASSEMBLY OF	DAMAGED FUEL CONTAINER				· · · · · · · · · · · · · · · · · · ·			
						<u></u>			
	· · · · · · · · · · · · · · · · · · ·				· · · · · · · · · · · · · · · · · · ·				
									~
					CONFIRMS THAT ALL A	PPROPRIATE REVIEWS OF THIS DR.	AWING ARE DOCUMENTED IN COMP.	ANY'S NETWORK	~n
				L					
						<u> </u>			
				GENERAL NO	TES:				
				1. THE EQUIP	MENT DESIGN DOCUMEN	TED IN THE DRAWING PACK	AGE HAS BEEN CONFIRMED		
				SAFETY AN	ALYSIS REPORT.				_
				2. THE REVISI THIS COVE	ION LEVEL OF EACH INDIV R SHEET. A REVISION TO SETS TO THE NEXT PEVIS	ANY SHEET IN THIS PACK ANY SHEET(S) IN THIS PACK	AGE IS THE SAME AS THE REV (AGE REQUIRES UPDATING O	F REVISION NUM	BERS
·				3. THE ITS CA	TEGORY OF A COMPONE	NT IS THE HIGHEST ITS LEVE	EL OF ALL PARTS THAT MAKE	UP THE COMPO	NENT.
		· · · · · · · · · · · · · · · · · · ·		4. THE ASME	BOILER AND PRESSURE V	ESSEL CODE (ASME CODE)	, 1995 EDITION WITH ADDEND	A THROUGH 199	7. =D
				IN SAR TAE CONTAINE	BLE 1.3.2 (TRANSPORTATI R FOR TROJAN PLANT SN	ON) AND FSAR TABLE 2.2.15 IF IS CONSTRUCTED IN ACC	5 (STORAGE). THE HOLTEC DA ORDANCE WITH ASME SECTIO	MAGED FUEL	
				SUBSECTION PRIOR NRC	ON NG AS DESCRIBED IN CAPPROVAL BEFORE IMP	THE SAR. NEW OR REVISED PLEMENTATION.	ASME CODE ALIERNATIVES	REQUIRE	
				5. WELDING P SUBSECTIO	ROCEDURES AND WELDE	ER QUALIFICATIONS SHALL	BE PER ASME SECTION IX AND	ASME SECTION	4 111,
		·····		6. MATERIALS	USED IN FABRICATION S	HALL MEET THE REQUIREM	ENTS OF ASME SECTION II PA	RT A. WELD MA	TERIAL SHALL
		······		7. ALL WELDS	REQUIRE VISUAL EXAMI				
		· · · · · · · · · · · · · · · · · · ·		8. ALL MATER	HAL IS TYPE 304 STAINLES	SS STEEL, EXCEPT BOLTS (S	SA 193 B8-CLASS 2) HEX NUTS	(SA 194 Gr. 8); A	ND WASHERS
				9. ALL WELD	SIZES ARE MINIMUMS. LAI	RGER WELDS ARE PERMITTI	ED. LOCAL AREAS OF UNDERS	SIZED WELDS AF	RE ACCEPTABL
					E LIMITS OF THE ASME CO	DDE, AS APPLICABLE.			
		······							
		······································							
		······································							
L	<u>l</u>	······		l					









	CLIE	NT	PORTLA	7 ND GENERAL]	6	اــــــــــــــــــــــــــــــــــــ		ING DR	AWING
	PRO		NO. 1135	P.O. NO.	N/A						
D	DRA PAC	WING KAGE I	. _{D.} 4122	TOTAL SHEETS	2	2				REVISION	1 LOG
F								RE	AFFECTED DRAWING	SUMMARY OF CHAN	NGES/ PREP
								0	INITIAL ISSUE	1135-39	SI
_	-					<u></u>					·
		SHEET		DESCRIPTION							
	ŀ	1	COVER SHEET			, -	4				
	-	2	SPACER FOR FAILED FUEL C	AN							
	ŀ								<u> </u>		
-	-		· · · ·	······································	•						
				·····							
									+ THE VALIDATION IDEN CONFIRMS THAT ALL A	TIFICATION RECORD (VIR) NUMBER I PPROPRIATE REVIEWS OF THIS DRAW	S A COMPUTER GENERATED ING ARE DOCUMENTED IN C
	Ļ						-	L		· · · · · · · · · · · · · · · · · · ·	
•				#							
			<u> </u>					GENERAL NO	DTES: PMENT DESIGN DOCUMEN	TED IN THIS DRAWING PACKAG	SE HAS BEEN CONFIRM
	-							INTERNAT	IONAL TO COMPLY WITH T PONENT IS ITS, CATEGOR'	HE SAFETY ANALYSES DESCR (B.	IBED IN THE SAFETY AN
	ļ							3. THE REVIS REVISION	SION LEVEL OF EACH INDIVITO ANY SHEET(S) IN THIS	IDUAL SHEET IN THIS PACKAG PACKAGE REQUIRES UPDATIN	E IS THE SAME AS THE
в	F						$\left\{ \right\}$	NUMBER. 4. THE ASME	BOILER AND PRESSURE	ESSEL CODE (ASME CODE), 19	95 EDITION WITH ADDE
								CODE FOF AND FSAR SUBSECTI	R THE HI-STAR 100 SYSTEM TABLE 2.2.15 (STORAGE). ON NF AS DESCRIBED IN T	I, WITH CERTAIN APPROVED A THE TROJAN FAILED FUEL CAN HE SAR (TRANSPORTATION) A	LTERNATIVES AS LISTE N SPACER IS CONSTRU(ND FSAR (STORAGE), N
								5. FINAL LEN	PRIOR NRC APPROVAL BE	FORE IMPLEMENTATION. E ESTABLISHED AFTER THE FF	C IS FILLED, ALLOWING
								INSIDE TH	E FFC. S USED IN FABRICATION S		TS OF ASME SECTION I
					-			THE REQU	SPACER COLUMN SHALL	ION II, PART C.	
-								= 30 ksi, A	ND CHEMICALS PER ASTM	A554. THE FUEL SPACER PLAT	E MAY BE ANY TYPE ST
		<u>,</u>						9. TOLERAN	CES ON THICKNESS OF AS	ME CODE MATERIAL ARE SPEC	CIFIED IN ASME SECTION
							-				
A]				
					·····.		4				
					· · · · ·						
]				
ļl							<u> </u>			X	· · · · · · · · · · · · · · · · · · ·

3

3

1

REV	AFFECTED DRAWING SHEET NUMBERS	SUMMARY OF CHANGES/ AFFECTED ECOs	PREPARED BY	APPROVAL DATE	VIR# +
0	INITIAL ISSUE	1135-39	SLC	5/19/03	20482
	···· ··· ··· ···				
		L		l	
	THE VALIDATION IDENTI CONFIRMS THAT ALL APP	FICATION RECORD (VIR) NUMBER IS A COMPUTER GE ROPRIATE REVIEWS OF THIS DRAWING ARE DOCUME	NERATED RANDON	M NUMBER WHICH	ł

- ED BY HOLTEC NALYSIS REPORT.
- REVISION LEVEL OF THIS COVER SHEET. A ERS OF ALL SHEETS TO THE NEXT REVISION
- ENDA THROUGH 1997, IS THE GOVERNING ED IN SAR TABLE 1.3.2 (TRANSPORTATION) ICTED IN ACCORDANCE WITH ASME SECTION III, NEW OR REVISED ASME CODE ALTERNATIVES
- AN APPROXIMATE 2" AXIAL CLEARANCE
- , PART A. WELD MATERIAL SHALL MEET
- LE STRENGTH = 75 ksi, YIELD STRENGTH TAINLESS STEEL.
- ND III, RESPECTIVELY.
- **1** II.





UNLISS OTHERWISE SPECIFIED DWENSDAG AME AN INCHES TOLERANCES AME: FRACTOWAL ± 1/8 DEETWALS ANGLES X = ±125 ± 1' . K = ±03 . X = ±01	TOW THE WALL AND THE WORK AND THE WALLY
.xxx = ±.005	NONE PFFC-001 1/1 8

UNLESS OTHERWISE SPECIFIED DAMENSIONS ARE IN INCHES	SCORT 265 LB5 PGE-C	SATC
TOLERANCES ARE FRACTIONAL ± 1/8 DECIMUS ANGLES X = ±.125 ± 1' x = ±.03	TON SANTOS 6/18/	AND IV C SCOTTS WALLY CONTINUES
.XX = 1.01 .XXX = 2.005	Tustin / Ha	9 FALED FUEL CAN
	The survey of the	49 SHELL AND LID 49 ASSEMBLY
NEXT ADDALLY MTC-001	NONE PFF	C-002 1/2 7

	H/A	12-01	NOM NOW
DINENSIONS ARE IN INCHES	PFFC0202R7	en.	CNC
FRACTIONAL ± 1/1	TON SANTOS	5/18/99	
X = ±.125 ± 1	150/	72/44	California -
.X = 2.03 .XX = 2.01	1.3 172	4= 41	WEIWE IN SALLY
.222 = 2.003	With the	444	FAILED FUEL CAN
•	Kouver	ohs	SHELL AND UD
	Ligitus /		ASSEMBLY
ST AND PFEC-001	NONE	PFFC-	002 2/2 7

1.5 Compliance with 10CFR71

The HI-STAR 100 packaging complies with the requirements of 10CFR71 for a Type B(U)F-85 package. Analyses which demonstrate that the HI-STAR 100 packaging complies with the requirements of Subparts E and F of 10CFR71 are provided in this SAR. Specific reference to each section of the SAR that is used to specifically address compliance is provided in Table 1.0.2. The HI-STAR 100 packaging complies with the general standards for all packages, 10CFR71.43, as demonstrated in Section 2.4. Under the tests specified in 10CFR71.71 (normal conditions of transport) the HI-STAR 100 packaging is demonstrated to sustain no degradation in its safety function allowing the HI-STAR 100 packaging to meet the requirements of 10CFR71, Paragraphs 71.45, 71.51, and 71.55. Under the tests specified in 10CFR71.73 (hypothetical accident conditions) and 10CFR71.61 (special requirement for irradiated nuclear fuel shipments), the degradation sustained by the HI-STAR 100 packaging is shown not to cause the HI-STAR 100 packaging to exceed the requirements of 10CFR71, Paragraphs 71.51, 71.55, and 71.63(b).

The HI-STAR 100 packaging meets the structural, thermal, containment, shielding and criticality requirements of 10CFR71, as described in Chapters 2 through 6. The operational procedures and acceptance tests and maintenance program provided in Chapters 7 and 8 ensure compliance with the requirements of 10CFR71.

The following is a summary of the information provided in Chapter 1 that is directly applicable to verifying compliance with 10CFR71:

- The HI-STAR 100 packaging has been described in sufficient detail to provide an adequate basis for its evaluation.
- Drawings provided in Section 1.4 contain information that provides an adequate basis for evaluation of the HI-STAR 100 packaging against the 10CFR71 requirements. Each drawing is identified, consistent with the text of the SAR, and contains keys or annotation to explain and clarify information on the drawing.
- Section 1.0 includes a reference to the NRC-approved Holtec International quality assurance program for the HI-STAR 100 packaging.
- Section 1.3 identifies the applicable codes and standards for the HI-STAR 100 packaging design, fabrication, assembly, and testing.
- The HI-STAR 100 packaging meets the general requirements of 10CFR71.43(a) and 10CFR71.43(b), as demonstrated by the drawings provided in Section 1.4 and the discussion | provided in Subsection 1.2.1.9, respectively.
- The drawings provided in Section 1.4 provide a detailed packaging description that can be evaluated for compliance with 10CFR71 for each technical discipline.

• Any restrictions on the use of the HI-STAR 100 packaging are specified in Subsection 1.2.3 and Chapter 7.

1.6 <u>REFERENCES</u>

- [1.0.1] 10CFR Part 71, "Packaging and Transportation of Radioactive Materials", Title 10 of the Code of Federal Regulations, Office of the Federal Register, Washington, D.C.
- [1.0.2] 49CFR173, "Shippers General Requirements For Shipments and Packagings", Title 49 of the Code of Federal Regulations, Office of the Federal Register, Washington, D.C.
- [1.0.3] Regulatory Guide 7.9, "Standard Format and Content of Part 71 Applications for Approval of Packaging for Radioactive Material", Proposed Revision 2, USNRC, May 1986.
- [1.0.4] 10CFR Part 72, "Licensing Requirements for the Storage of Spent Fuel in an Independent Spent Fuel Storage Installation", Title 10 of the Code of Federal Regulations, Office of the Federal Register, Washington, D.C.
- [1.0.5] NUREG-1617, "Standard Review Plan for Transportation Packages for Spent Nuclear Fuel", J.S. Nuclear Regulatory Commission, March 2000.
- [1.0.6] HI-STAR 100 Final Safety Analysis Report, Holtec Report No. HI-2012610, Revision 1, Docket No. 72-1008.
- [1.0.7] HI-STORM 100 Final Safety Analysis Report, Holtec Report No. HI-2002444, Revision 1, Docket No. 72-1014.
- [1.1.1] U.S. Department of Energy, "Multi-Purpose Canister (MPC) Subsystem Design Procurement Specification", Document No. DBG000000-01717-6300-00001, Rev. 5, January 11, 1996.
- [1.1.2] U.S. Department of Energy, "MPC Transportation Cask Subsystem Design Procurement Specification", Document No. DBF 000000-01717-6300-00001, Rev. 5, January 11, 1996.
- [1.2.1] U.S. NRC Information Notice 96-34, "Hydrogen Gas Ignition During Closure Welding of a VSC-24 Multi-Assembly Scale Basket".
- [1.2.2] Directory of Nuclear Reactors, Vol. II, Research, Test & Experimental Reactors, International Atomic Energy Agency, Vienna, 1959.
- [1.2.3] V.L. McKinney and T. Rockwell III, Boral: A New Thermal-Neutron Shield, USAEC Report AECD-3625, August 29, 1949.

- [1.2.4] <u>Reactor Shielding Design Manual</u>, USAEC Report TID-7004, March 1956.
- [1.2.5] Deleted.
- [1.2.6] ORNL/TM-10902, "Physical Characteristics of GE BWR Fuel Assemblies", by R.S. Moore and K.J. Notz, Martin Marietta (1989).
- [1.2.7] U.S. DOE SRC/CNEAF/95-01, Spent Nuclear Fuel Discharges from U.S. Reactors 1993, Feb. 1995.
- [1.2.8] S.E. Turner, "Uncertainty Analysis Axial Burnup Distribution Effects," presented in "Proceedings of a Workshop on the Use of Burnup Credit in Spent Fuel Transport Casks", SAND-89-0018, Sandia National Laboratory, Oct., 1989.
- [1.2.9] Commonwealth Edison Company, Report No. NFS-BND-95-083, Chicago, Illinois.
- [1.2.10] Regulatory Guide 7.11, "Fracture Toughness Criteria of Base Material for Ferritic Steel Shipping Cask Containment Vessels with a Maximum Wall Thickness of 4 Inches (0.1m)", U.S. Nuclear Regulatory Commission, Washington, D.C., June 1991.
- [1.2.11] NUREG-0612, "Control of Heavy Loads at Nuclear Power Plants", U.S. Nuclear Regulatory Commission, Washington, D.C., July 1980.
- [1.2.12] Trojan ISFSI Safety Analysis Report, Revision 3, USNRC Docket 72-0017.
- [1.2.13] NRC Interim Staff Guidance Document No. 8, "Burnup Credit in the Criticality Safety Analyses of PWR Spent Fuel in Transportation and Storage Casks", Revision 2.
- [1.2.14] NRC Interim Staff Guidance Document No. 11, "Cladding Considerations for the Transportation and Storage of Spent Fuel", Revision 2.
- [1.2.15] DOE/RW-0184, Volume 3, "Characteristics of Spent Fuel, High Level Waste, and Other Radioactive Wastes Which May Require Long-Term Isolation," Appendix 2.A, "Physical Descriptions of LWR Fuel Assemblies," U.S. Department of Energy, Office of Civilian Radioactive Waste Management, December 1987.
- [1.3.1] American Society of Mechanical Engineers, "Boiler and Pressure Vessel Code", 1995 with Addenda through 1997.

- [1.3.2] NUREG/CR-6407, "Classification of Transportation Packaging and Dry Spent Fuel Storage System Components According to Importance to Safety", U.S. Nuclear Regulatory Commission, Washington D.C., February 1996.
- [1.3.3] ANSI N14.6-1993, "Special Lifting Devices for Shipping Containers Weighing 10,000 Pounds (4500 Kg) or More", June 1993.

APPENDIX 1.A: ALLOY X DESCRIPTION

1.A ALLOY X DESCRIPTION

1.A.1 <u>Alloy X Introduction</u>

Alloy X is used within this licensing application to designate a group of stainless steel alloys. Alloy X can be any one of the following alloys:

- Type 316
- Type 316LN
- Type 304
- Type 304LN

Qualification of structures made of Alloy X is accomplished by using the least favorable mechanical and thermal properties of the entire group for all MPC mechanical, structural, neutronic, radiological, and thermal conditions. The Alloy X approach is conservative because no matter which material is ultimately utilized, the Alloy X approach guarantees that the performance of the MPC will meet or exceed the analytical predictions.

This appendix defines the least favorable material properties of Alloy X.

1.A.2 Alloy X Common Material Properties

Several material properties do not vary significantly from one Alloy X constituent to the next. These common material properties are as follows:

- density
- specific heat
- Young's Modulus (Modulus of Elasticity)
- Poisson's Ratio

The values utilized for this licensing application are provided in their appropriate chapters.

1.A.3 <u>Alloy X Least Favorable Material Properties</u>

The following material properties vary between the Alloy X constituents:

- Design Stress Intensity (S_m)
- Tensile (Ultimate) Strength (S_u)
- Yield Strength (S_y)
- Coefficient of Thermal Expansion (α)
- Coefficient of Thermal Conductivity (k)

Each of these material properties are provided in the ASME Code Section II [1.A.1]. Tables 1.A.1

through 1.A.5 provide the ASME Code values for each constituent of Alloy X along with the least favorable value utilized in this licensing application. The ASME Code only provides values to -20° F. The design temperature of the MPC is -40° F to 725° F as stated in Table 1.2.3. Most of the above-mentioned properties become increasingly favorable as the temperature drops. Conservatively, the values at the lowest design temperature for the HI-STAR 100 System have been assumed to be equal to the lowest value stated in the ASME Code. The lone exception is the thermal conductivity. The thermal conductivity decreases with the decreasing temperature. The thermal conductivity value for -40° F is linearly extrapolated from the 70°F value using the difference from 70°F to 100°F.

The Alloy X material properties are the minimum values of the group for the design stress intensity, tensile strength, yield strength, and coefficient of thermal conductivity. Using minimum values of design stress intensity is conservative because lower design stress intensities lead to lower allowables that are based on design stress intensity. Similarly, using minimum values of tensile strength and yield strength is conservative because lower values of tensile strength and yield strength lead to lower allowables that are based on tensile strength and yield strength. When compared to calculated values, these lower allowables result in factors of safety that are conservative for any of the constituent materials of Alloy X. Further discussion of the justification for using the minimum values are used for the coefficient of thermal expansion of Alloy X. The maximum and minimum coefficients of thermal expansion are used as appropriate in this submittal. Figures 1.A.1-1.A.5 provide a graphical representation of the varying material properties with temperature for the Alloy X materials.

1.A.4 <u>References</u>

[1.A.1] ASME Boiler & Pressure Vessel Code Section II, 1995 ed. with Addenda through 1997.

Temp. (^O F)	Туре 304	Type 304LN	Туре 316	Type 316LN	Alloy X (minimum of constituent values)
-40	20.0	20.0	20.0	20.0	20.0
100	20.0	20.0	20.0	20.0	20.0
200	20.0	20.0	20.0	20.0	20.0
300	20.0	20.0	20.0	20.0	20.0
400	18.7	18.7	19.3	18.9	18.7
500	17.5	17.5	18.0	17.5	17.5
600	16.4	16.4	17.0	16.5	16.4
650	16.2	16.2	16.7	16.0	16.0
700	16.0	16.0	16.3	15.6	15.6
750	15.6	15.6	16.1	15.2	15.2
800	15.2	15.2	15.9	14.9	14.9

ALLOY X AND CONSTITUENT DESIGN STRESS INTENSITY (S_m) vs. TEMPERATURE

Notes:

- 1. Source: Table 2A on pages 314, 318, 326, and 330 of [1.A.1].
- 2. Units of design stress intensity values are ksi.

Temp. (⁰ F)	Туре 304	Type 304LN	Туре 316	Type 316LN	Alloy X (minimum of constituent values)
-40	75.0 (70.0)	75.0 (70.0)	75.0 (70.0)	75.0 (70.0)	75.0 (70.0)
100	75.0 (70.0)	75.0 (70.0)	75.0 (70.0)	75.0 (70.0)	75.0 (70.0)
200	71.0 (66.2)	71.0 (66.2)	75.0 (70.0)	75.0 (70.0)	71.0 (66.2)
300	66.0 (61.5)	66.0 (61.5)	73.4 (68.5)	70.9 (66.0)	66.0 (61.5)
400	64.4 (60.0)	64.4 (60.0)	71.8 (67.0)	67.1 (62.6)	64.4 (60.0)
500	63.5 (59.3)	63.5 (59.3)	71.8 (67.0)	64.6 (60.3)	63.5 (59.3)
600	63.5 (59.3)	63.5 (59.3)	71.8 (67.0)	63.1 (58.9)	63.1 (58.9)
650	63.5 (59.3)	63.5 (59.3)	71.8 (67.0)	62.8 (58.6)	62.8 (58.6)
700	63.5 (59.3)	63.5 (59.3)	71.8 (67.0)	62.5 (58.4)	62.5 (58.4)
750	63.1 (58.9)	63.1 (58.9)	71.4 (66.5)	62.2 (58.1)	62.2 (58.1)
800	62.7 (58.5)	62.7 (58.5)	70.9 (66.2)	61.7 (57.6)	61.7 (57.6)

ALLOY X AND CONSTITUENT TENSILE STRENGTH (S_D) vs. TEMPERATURE

Notes:

- 1. Source: Table U on pages 437, 439, 441, and 443 of [1.A.1].
- 2. Units of tensile strength are ksi.
- 3. Values in parentheses are for SA-336 forging material (Types F304, F304LN, F316, and F316LN) that are used solely for the one-piece MPC lids. Other values correspond to SA-240 plate material.

Temp. (°F)	Туре 304	Type 304LN	Туре 316	Type 316LN	Alloy X (minimum of constituent values)
-40	30.0	30.0	30.0	30.0	30.0
100	30.0	30.0	30.0	30.0	30.0
200	25.0	25.0	25.8	25.5	25.0
300	22.5	22.5	23.3	22.9	22.5
400	20.7	20.7	21.4	21.0	20.7
500	19.4	19.4	19.9	19.4	19.4
600	18.2	18.2	18.8	18.3	18.2
650	17.9	17.9	18.5	17.8	17.8
700	17.7	17.7	18.1	17.3	17.3
750	17.3	17.3	17.8	16.9	16.9
800	16.8	16.8	17.6	16.6	16.6

ALLOY X AND CONSTITUENT YIELD STRESSES (Sy) vs. TEMPERATURE

Notes:

1. Source: Table Y-1 on pages 518, 519, 522, 523, 530, 531, 534, and 535 of [1.A.1].

2. Units of yield stress are ksi.

Temp. (°F)	Type 304 and Type 304LN	Type 316 and Type 316LN	Alloy X Maximum	Alloy X Minimum
-40	8.55	8.54	8.55	8.54
100	8.55	8.54	8.55	8.54
150	8.67	8.64	8.67	8.64
200	8.79	8.76	8.79	8.76
250	8.90	8.88	8.90	8.88
300	9.00	8.97	9.00	8.97
350	9.10	9.11	9.11	9.10
400	9.19	9.21	9.21	9.19
450	9.28	9.32	9.32	9.28
500	9.37	9.42	9.42	9.37
550	9.45	9.50	9.50	9.45
600	9.53	9.60	9.60	9.53
650	9.61	9.69	9.69	9.61
700	9.69	9.76	9.76	9.69
750	9.76	9.81	9.81	9.76
800	9.82	9.90	9.90	9.82

ALLOY X AND CONSTITUENT COEFFICIENT OF THERMAL EXPANSION vs. TEMPERATURE

Notes:

1. Source: Table TE-1 on pages 590 and 591 of [1.A.1].

2. Units of coefficient of thermal expansion are in./in.- $^{\circ}$ F x 10⁻⁶.

Temp. (°F)	Type 304 and Type 304LN	Type 316 and Type 316LN	Alloy X (minimum of constituent values)
-40	8.23	6.96	6.96
70	8.6	7.7	7.7
100	8.7	7.9	7.9
150	9.0	8.2	8.2
200	9.3	8.4	8.4
250	9.6	8.7	8.7
300	9.8	9.0	9.0
350	10.1	9.2	9.2
400	10.4	9.5	9.5
450	10.6	9.8	9.8
500	10.9	10.0	10.0
550	11.1	10.3	10.3
600	11.3	10.5	10.5
650	11.6	10.7	10.7
700	11.8	11.0	11.0
750	12.0	11.2	11.2
800	12.2	11.5	11.5

ALLOY X AND CONSTITUENT THERMAL CONDUCTIVITY vs. TEMPERATURE

Notes:

- 1.
- Source: Table TCD on page 606 of [1.A.1]. Units of thermal conductivity are Btu/hr-ft-°F. 2.

DESIGN STRESS INTENSITY VS. TEMPERATURE



HI-STAR SAR REPORT HI-951251 **TENSILE STRENGTH VS. TEMPERATURE**



REPORT HI-951251
YIELD STRESS VS. TEMPERATURE





COEFFICIENT OF THERMAL EXPANSION VS. TEMPERATURE

HI-STAR SAR REPORT HI-951251

Rev. 4

THERMAL CONDUCTIVITY VS. TEMPERATURE



HI-STAR SAR REPORT HI-951251 Rev. 4

APPENDIX 1.B: HOLTITE TM MATERIAL DATA

The information provided in this appendix describes the neutron absorber material, Holtite-A for the purpose of confirming its suitability for use as a neutron shield material in spent fuel storage casks. Holtite-A is one of a family of Holtite neutron shield materials denoted by the generic name HoltiteTM. It is currently the only neutron shield material approved for installation in the HI-STAR 100 cask. It is chemically identical to NS-4-FR which was originally developed by Bisco, Inc. and used for many years as a shield material with B₄C or Pb added.

Holtite-A contains aluminum hydroxide $(Al(OH)_3)$ in an epoxy resin binder. Aluminum | hydroxide is also known by the industrial trade name of aluminum tri-hydrate or ATH. ATH is often used commercially as a fire-retardant. Holtite-A contains approximately 62% ATH supported in a typical 2-part epoxy resin as a binder. Holtite-A contains 1% (nominal) by weight B₄C, a chemically inert material added to enhance the neutron absorption property. Pertinent properties of Holtite-A are listed in Table 1.B.1.

The essential properties of Holtite-A are:

- 1. the hydrogen density (needed to thermalize neutrons),
- 2. thermal stability of the hydrogen density, and
- 3. the uniformity in distribution of B_4C needed to absorb the thermalized neutrons.

ATH and the resin binder contain nearly the same hydrogen density so that the hydrogen density of the mixture is not sensitive to the proportion of ATH and resin in the Holtite-A mixture. B_4C is added as a finely divided powder and does not settle out during the resin curing process. Once the resin is cured (polymerized), the ATH and B_4C are physically retained in the hardened resin. Qualification testing for B_4C throughout a column of Holtite-A has confirmed that the B_4C is uniformly distributed with no evidence of settling or non-uniformity. Furthermore, an excess of B_4C is specified in the Holtite-A mixing and pouring procedure as a precaution to assure that the B_4C concentration is always adequate throughout the mixture.

The specific gravity specified in Table 1.B.1 does not include an allowance for weight loss. The specific gravity assumed in the shielding analysis includes a 4% reduction to conservatively account for potential weight loss at the design temperature of 300°F or an inability to reach theoretical density. Tests on the stability of Holtite-A were performed by Holtec International. The results of the tests are summarized in Holtec Proprietary Reports HI-2002396, "Holtite-A Development History and Thermal Performance Data" and HI-2002420, "Results of Pre- and Post-irradiation Test Measurements." The information provided in these reports demonstrates that Holtite-A possesses the necessary thermal and radiation stability characteristics to function as a reliable shielding material in the HI-STAR 100 overpack.

The Holtite-A is encapsulated in the HI-STAR 100 overpack and therefore should experience a very small weight reduction during the design life of the HI-STAR 100 System.

The data and test results confirm that Holtite-A remains stable under design thermal and radiation conditions, the material properties meet or exceed that assumed in the shielding analysis, and the B_4C remains uniformly distributed with no evidence of settling or non-uniformity.

Based on the information described above, Holtite-A meets all of the requirements for an acceptable neutron shield material.

Table 1.B.1

REFERENCE PROPERTIES OF HOLTITE-A NEUTRON SHIELD MATERIAL

PHYSICAL PROPERTIES		
% ATH	62 (nominal)	
Specific Gravity	1.68 g/cc (nominal)	
Max. Continuous Operating Temperature	300°F	
Hydrogen Density	0.096 g/cc minimum	
Radiation Resistance	Excellent	
CHEMICAL PROPERTIES (Nominal)		
wt% Aluminum	21.5	
wt% Hydrogen	6.0 (nominal)	
wt% Carbon	27.7	
wt% Oxygen	42.8	
wt% Nitrogen	2.0	
wt% B ₄ C	1.0 (nominal)	

PAGES 1.B-4 THROUGH 1.B-20

INTENTIONALLY DELETED

APPENDIX 1.C: MISCELLANEOUS MATERIAL DATA

The information provided in this appendix specifies the thermal expansion foam (silicone sponge), paint, and anti-seize lubricant properties and demonstrates their suitability for use in spent nuclear fuel storage casks. The following is a listing of the information provided.

- HT-800 Series, Silicone Sponge, Bisco Products Technical Data Sheet
- Thermaline 450, Carboline, Product Data Sheet and Application Instructions
- Carboline 890, Carboline, Product Data Sheet and Application Instructions
- FEL-PRO Technical Bulletin, N-5000 Nickel Based-Nuclear Grade Anti-Seize Lubricant

HT-870 silicone sponge is specified as a thermal expansion foam to be placed in the overpack outer enclosure with the neutron shield. Due to differing thermal expansion of the neutron shield and outer enclosure carbon steel, the silicone sponge is provided to compress and allow the neutron shield material to expand. The compression-deflection physical properties are provided for the silicone sponge.

Silicone has a long and proven history in the nuclear industry. Silicone is highly resistant to degradation as a result of radiation at the levels required for the HI-STAR 100 System. Silicone is inherently inert and stable and will not react with the metal surfaces or neutron shield material. Additionally, typical operating temperatures for silicone sponges range from -50° F to 400° F.

Thermaline 450 is specified to coat the inner cavity of the overpack and Carboline 890 is specified to coat the external surfaces of the overpack. As can be seen from the product data sheets, the paints are suitable for the design temperatures (see Table 2.2.3) and chemical environment. Chemically identical substitutes are permitted (i.e., Carboguard 890 in lieu of Carboline 890).

Nuclear grade anti-seize lubricant, N-5000, from FEL-PRO is specified as the lubricant for the overpack closure bolts. The lubricant is formulated to have the lowest practical levels of halogens, sulfur, and heavy metals. NEVER-SEEZ NGBT provides equivalent properties to FEL-PRO N-5000 and is also acceptable for use on the HI-STAR 100 System.



Technical Data

HT-800 SERIES

Specification Grade Silicone Sponge

PHYSICAL PROPERTIES

	SPECIFICATION			•
PROPERTY	HT-870 (Soft)	HT-800 (Medium)	HT-820 (Firm)	TEST METHOD
Density	12 - 24 pcf	16-28 pcf	20 - 32 pcf	ASTM D-3574
Compression Force @ 25% Deflection	2 - 7 psi	6 - 14 psi	12 - 20 psi	ASTM D-1056
Compression Set (Maximum)	10%	10%	10%	ASTM D-1056 (Compressed 50%for 22 hrs. @ 100°C)
Water Absorption (Maximum)	10%	5%	· 5%	ASTM D-1056

Available Industry Specifications:

AMS-3195 (HT-800) AMS-3196 (HT-820)

UL-94 (Limited to specific classes, densities, thicknesses and colors)

2300 East Devon Avenue

should run independent train to confirm material suffability i

Bless Fraduens and these Carsing achiber ra

Date

Elk Grove Village, Illinois 60007-6120

Toll Free: 800/237-2068

Sile application.



SELECTION DATA

GENERIC TYPE: A glass flake filled, phenolic modified, amine cured epoxy novalac.

GENERAL PROPERTIES: A dense cross-linked polymer which exhibits outstanding barrier protection against a variety of chemical exposures. Excellent resistance to wet/dry cycling conditions at elevated temperatures. Designed to coat the exterior of insulated piping. It is also suitable for coating non-insulated piping and equipment exposed to chemical attack. The glass flakes help provide excellent abrasion resistance, permeation resistance and internal reinforcement.

- Temperature resistance to 450°F
- Excellent abrasion resistance
- Excellent overall chemical resistance
- Excellent thermal shock resistance

RECOMMENDED USES: Typically used as a one coat system to coat pipes and tanks that will be insulated. May also be used to coat non-insulated pipe, structural steel, equipment or concrete that may be subjected to severe chemical attack, abrasion or other abuse typical of a chemical plant environment.

TYPICAL CHEMICAL RESISTANCE:

Exposure	<u>Splash & Spillage</u>	Fumes
Acids	Excellent	Excellent
Alkalies	Excellent	Excellent
Solvents	Excellent	Excellent
Salt	Excellent	Excellent
Water	Excellent	Excellent

 TEMPERATURE
 RESISTANCE
 (Under insulation):

 Continuous:
 425°F (218°C)

 Excursions to:
 450°F (232°C)

At-200°F (93°C) coating discoloration may be observed without loss of film integrity.

SUBSTRATES: Apply over properly prepared steel.

COMPATIBLE COATINGS: Normally applied directly to substrate. May be applied over epoxies and phenolics as recommended. May be topcoated with epoxies, polyurethanes or other finish coats as recommended.

July 96 Replaces September 95

SPECIFICATION DATA

THEORETICAL SOLIDS CONTENT OF MIXED MATERIAL:

THERMALINE 450

<u>By Volume</u> 70 ± 2% 0928

VOLATILE ORGANIC CONTENT (VOC):

The following are nominal values: As supplied: 2.13 lbs./gal. (255 gm./liter).

	Fluid	Pounds/	Grams/
Thinner	Ounces/Gal.	Gallon	Liter
213	13	2.56	307

RECOMMENDED DRY FILM THICKNESS:

8-10 mils (200-250 microns) to be achieved in 1 or 2 coats.

THEORETICAL COVERAGE PER MIXED GALLON:* 1,117 mil sq. ft. (27.9 sq.m/l at 25 microns) 139 sq. ft at 8 mils (3.5 sq. m/l at 200 microns) 111 sq. ft at 10 mils (2.8 sq.m/l at 250 microns)

*Mixing and application losses will vary and must be taken into consideration when estimating job requirements.

STORAGE CONDITIONS: Store indoors. Temperature: 40-110°F (4-43°C) Humidity: 0-90%

SHELF LIFE: 24 months when stored indoors at 75°F (24°C)

COLOR: Red (0500) and Gray (5742)

GLOSS: Low (Epoxies lose gloss, discolor and eventually chalk in sunlight exposure.)

ORDERING INFORMATION

Prices may be obtained from your Carboline Sales Representative or Carboline Customer Service Department.

APPROXIMATE SHIPPING WEIGHT:

THERMALINE 450	<u>1's</u> 12 lbs. (5.5 kg)	<u>5's</u> 58 ibs. (26.3 kg)
Thinner 213	8.4 lbs. (3.8 kg)	41 lbs. (18.6 kg)

FLASH POINT: (Setaflash)		
THERMALINE 450 Part A:	53°F	(12°C)
THERMALINE 450 Part B:	>200°F	(>93°C)
Thinner 213	22°F	(-6℃)

To the best of our knowledge the technical data contained herain are true and accurate at the date of issuance and are subject to change without prior notice. User must contact Carboline Company to verify correctness before specifying or ordering. No guarantee of accuracy is given or implied. We guarantee our products to conform to Carboline quality control. We assume no responsibility for coverage, performance or injuries resulting from use. Liability, if any, is limited to replacement of products. Prices and cost data, if shown, are subject to change without prior notice. NO OTHER WARRANTY OR GUARANTEE OF ANY KIND IS MADE BY CARBOLINE, EXPRESS OR IMPLIED, STATUTORY, BY OPERATION OF LAW, OR OTHERWISE, INCLUDING MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

APPLICATION INSTRUCTIONS THERMALINE 450

These instructions are not intended to show product recommendations for specific service. They are issued as an aid in determining correct surface proparation, mixing instructions and application procedure. It is assumed that the proper product recommendations have been made. These instructions should be followed closely to obtain the maximum service from the materials.

SURFACE PREPARATION: Remove all oil or grease from surface to be coated with Thinner 2 or Surface Cleaner 3 (refer to Surface Cleaner 3 instructions) in accordance with SSPC-SP 1.

STEEL:

Not Insulated: Abrasive blast to a Commercial Finish in accordance with SSPC-SP 6 and obtain a 2-3 mil (50-75 micron) blast profile.

Under Insulation: Abrasive blast to a Near White Finish in accordance with SSPC-SP 10 and obtain a 2-3 (50-75 micron) blast profile.

MIXING: Power mix each component separately, then combine and power mix in the following proportions.

Allow 30 minutes induction time at 75°F (24°C) prior to use.

	<u>т сан. кл</u>	<u>5 Gai, Ki</u>
THERMALINE 450 Part A:	0.8 gals.	4.0 gals.
THERMALINE 450 Part B:	0.2 gals.	1.0 gals.

THINNING: May be thinned up to 13 oz/gal with Thinner 213.

Use of thinners other than those supplied or approved by Carboline may adversely affect product performance and void product warranty, whether express or implied.

POT LIFE: Three hours at 75°F (24°C) and less at higher temperatures. Pot life ends when coating loses body and begins to sag.

APPLICATION CONDITIONS:

	<u>Material</u>	<u>Surfaces</u>	<u>Ambient</u>	Humidity
Normal	65-85°F	65-85°F	65-85°F	30-60%
	(18-29°C)	(18-29°C)	(18-29°C)	
Minimum	55°F (13°C)	50°F (10°C)	50°F (10°C)	0%
Maximum	90°F (32°C)	110°F (43°C)	100°F (38°C)	85%

Do not apply when the surface temperature is less than 5°F or 3°C above the dew point.

Special thinning and application techniques may be required above or below normal conditions.

SPRAY: The following spray equipment has been found suitable and is available from manufacturers such as Binks, DeVilbiss and Graco.

Conventional: Pressure pot equipped with dual regulators, 1/2" I.D. minimum material hose, .110" I.D. fluid tip and appropriate air cap.

July 96 Replaces September 95

Altess:	
Pump Ratio:	30:1 (min)*
GPM Output:	3.0 (min)
Material Hose:	1/2" I.D. (min)
Tip Size:	.035"041"
Dutput psi:	2200-2500

*Teflon packings are recommended and are available from the pump manufacturer.

BRUSH: For striping of welds, touch-up of small areas only. Use a natural bristle brush, applying full strokes. Avoid rebrushing.

ROLLER: Not recommended.

DRYING TIMES: These times are based on a dry film thickness of 10 mils (250 microns). Higher film thickness, insufficient ventilation or cooler temperatures will require longer cure times and could result in solvent entrapment and premature failure.

Surface

Temperature	Dry To Handle	Dry to Topcoat	Final Cure
50°F (10°C)	18 hours	48 hours	21 days
50°F (16°C)	12 hours	32 hours	14 days
75°F (24°C)	6 hours	18 hours	7 days
90°F (32°C)	3 hours	8 hours	4 days

If the final cure time has been exceeded, the surface must be abraded by sweep blasting prior to the application of any additional coats.

EXCESSIVE HUMIDITY OR CONDENSATION ON THE SURFACE DURING CURING MAY RESULT IN A SUR-FACE HAZE OR BLUSH; ANY HAZE OR BLUSH MUST BE REMOVED BY WATER WASHING BEFORE RE-COATING.

VENTILATION & SAFETY: WARNING: VAPORS MAY CAUSE EXPLOSION. When used in enclosed areas, thorough air circulation must be used during and after application until the coating is cured. The ventilation system should be capable of preventing the solvent vapor concentration from reaching the lower explosion limit for the solvents used. In addition to insuring proper ventilation, fresh air respirators or fresh air hoods must be used by all application personnel. Where flammable solvents exist, explosion-proof lighting must be used. Hypersensitive persons should wear clean, protective clothing, gloves and/or protective cream on face, hands and all exposed areas.

CLEANUP: Use Thinner 2.

CAUTION: READ AND FOLLOW ALL CAUTION STATE-MENTS ON THIS PRODUCT DATA SHEET AND ON THE MATERIAL SAFETY DATA SHEET FOR THIS PRODUCT.

CAUTION: CONTAINS FLAMMABLE SOLVENTS, KEEP AWAY FROM SPARKS AND OPEN FLAMES. WORKMEN IN CONFINED AREAS MUST WEAR FRESH AIRLINE RESPIRATORS. HYPERSENSITIVE PERSONS SHOULD WEAR GLOVES OR USE PROTECTIVE CREAM. ALL ELECTRICAL EQUIPMENT AND INSTALLATIONS SHOULD BE MADE IN ACCORDANCE WITH THE NATIONAL ELECTRICAL CODE. IN AREAS WHERE EXPLOSION HAZARDS EXIST, WORKMEN SHOULD BE REQUIRED TO USE NONFERROUS TOOLS AND TO WEAR CONDUCTIVE AND NONSPARKING SHOES.



Page 1.C-4





SELECTION DATA

GENERIC TYPE: Cross-linked epoxy.

GENERAL PROPERTIES: CARBOLINE 890 is a self priming, high solids, high gloss, high build epoxy mastic. It can be applied by spray, brush, or roller over hand or power tool cleaned steel and is compatible with most existing coatings and tightly adhered rust. The cured film provides a tough, cleanable surface and is available in a wide variety of colors.

- Single coat corrosion protection.
- Excellent chemical resistance.
- Good flexibility and lower stress upon curing then most epoxy coatings.
- Excellent tolerance of damp (not wet) substrates.
- Very good abrasion resistance.
- Suitable replacement for Carbornastic 801.

RECOMMENDED USES: Recommended where a high performance, chemically resistant epoxy coating is desired. Offers outstanding protection for interior floors, walls, piping, equipment and structural steel or as an exterior coating for railcars, structural steel and equipment in various corrosive environ-ments. Industrial environments include Chemical Processing, Offshore Oil and Gas, Food Processing, Pharmaceutical, Water and Waste Water Treatment, Pulp and Paper and Power Generation among others. May be used as a two coat system direct to metal or concrete for Water and Municipal Waste Water immersion. Acceptable for use in incidental food contact areas and as a lining for hopper cars carrying food grade plastic pellets when processed according to FDA criteria (ref: FDA 21 CFR 175.300). Consult Carboline Technical Service Department for other specific uses.

NOT RECOMMENDED FOR: Strong acid or solvent exposures, immersion service other than water, exterior weathering where color retention is desired, such as a finish for tank exteriors or over chlorinated rubber and latex coatings,

TYPICAL CHEMICAL RESISTANCE:

Exposure	Immersion	Splash & Spillage	_ Furnes
Acids	NR	Very Good	Very Good
Alkalias	NR	Excellent	Excellent
Solvents	NR	Very Good	Excellent
Salt Solutions	Excellent	Excellent	Excellent
Water	Excellent	Excellent	Excellent
TEMPERATURE I	RESISTANCE:	(Non-Immersion)	• • •
Continuous:	250°F (121°C)	
Non-continuous:	300°F (149°C)	

At temperatures above 225°F, coating discoloration and loss of gloss can be observed, without loss of film integrity.

SUBSTRATES: Apply over suitably prepared metal, concrete, or other surfaces as recommended.

COMPATIBLE COATINGS: May be applied directly over inorganic zincs, weathered galvanizing, epoxies, phenolics or other coatings as recommended. A test patch is recommended before use over existing coatings. A mist coat of CARBOLINE 890 is required when applied over inorganic zincs to minimize bubbling. May be topcoated with polyurethanes or acrylics to upgrade weathering resistance. Not recommended over chlorinated rubber or latex coatings. Consult Carboline Technical Service Department for specific recommendations.

June 96 Replaces December 95

SPECIFICATION DATA

THEORETICAL SOLIDS CONTENT OF MIXED MATERIAL." By Volume 75% ± 2%

0986

VOLATILE ORGANIC CONTENT:* As Supplied: 1.78 lbs./gal. (214 grams/liter)

Thinned

CARROLINE 890

	Fluid	Pounds/	Grams/
<u>Thinner</u>	Ounces/Gal.	Gallon	Liter
2	8	2.08	250
2	13	2.26	271
33	16	2.38	285
laries with a	alar		

Varies with color

RECOMMENDED DRY FILM THICKNESS PER COAT: 4-6 mils (100-150 microns).

6-8 mils (150-200 microns) DFT for a more uniform gloss over inorganic zincs, or for use over light rust.

In more severe environments a second coat of 4-6 mils (100-150 microns) is recommended.

Dry film thickness in excess of 10 mils (250 microns) per coat is not recommended. Excessive film thickness over inorganic zinc may increase damage during shipping or erection.

THEORETICAL COVERAGE PER MIXED GALLON:

1203 mil sq. ft. (30 sq. m/l at 25 microns) 241 sq. ft. at 5 mils (6.0 sq. m/l at 125 microns)

Mixing and application losses will vary and must be taken into consideration when estimating job requirements.

STORAGE CONDITIONS: Store Indoors Temperature: 40-110°F (4-43°C) Hurnidity: 0-100%

SHELF LIFE: 36 months when stored at 75°F (24°C).

COLORS: Available in Carboline Color Chart colors. Some colors may require two coats for adequate hiding.

GLOSS: High gloss (Epoxies lose gloss, discolor and eventually chalk in sunlight exposure).

ORDERING INFORMATION

Prices may be obtained from your Carboline Sales Representative or Carboline Customer Service Department. **APPROXIMATE SHIPPING WEIGHT:**

CARBOLINE 890	<u>2 Gal. Kit</u> 29 Ibs. (13 kg)	<u> 10 Gal. Kit </u> 145 lbs. (66 kg)
	1'=	. 5's
THINNER #2	8 lbs.	39 lbs.
· ·	(4 kg)	(18 kg)
THINNER #33	9 lbs.	45 lbs.
	(4 kg)	(20 kg)
FLASH POINT: (Setafla	sh)	
CARBOLINE 890 Part A		89°F (32°C)
CARBOLINE 890 Part B		71°F (22°C)
THINNER #2		24°F (-5°C)
THINNER #33		89°F (32°C)

To the best of our knowledge the technical date contained herein are true and accurate at the date of issuance and are subject to change without prior notice. User must contact Carboline Company to verify correctness before specifying or ordering. No guarantee of accuracy is given or implied. We guarantee ar products to conform to Carboline quality control. We assume no responsibility for coverage, performance or injuries resulting from use. Liability, if any, is limited to replacement of products. Prices and cost date, if shown, sie subject to change without prior notice. NO OTHER WARRANTY OR GUARANTEE OF ANY KIND IS MADE BY CARBOLINE, EXPRESS OR IMPLIED, STATUTORY, BY OPERATION OF LAW, OR OTHERWISE, INCLUDING MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

APPLICATIONS INSTRUCTIONS CARBOLINE[®] 890

These instructions are not intended to show product recommendations for specific service. They are issued as an aid in determining correct surface preparation, mixing instructions and application procedure. It is assumed that the proper product recommendations have been made. These instructions should be followed closely to obtain the maximum service from the materials.

A 1-1 - - - -

SURFACE PREPARATION: Remove all oil or grease from surface to be coated with Thinner #2 or Surface Cleaner #3 (refer to Surface Cleaner #3 instructions) in accordance with SSPC-SP 1.

Steel: For mild environments Hand Tool or Power Tool Clean in accordance with SSPC-SP 2, SSPC-SP 3 or SSPC-SP 11 to produce a rust-scale free surface.

For more severe environments, abrasive blast to a Commercial Finish in accordance with SSPC-SP 6 and obtain a 1% - 3 mil (40-75 micron) blast profile.

For immersion service, abrasive blast to a Near White Metal Finish in accordance with SSPC-SP10 and obtain a 1½ - 3 mil (40-75 micron) blast profile.

Concrete: Must be cured at least 28 days at 70°F (21°C) and 50% R.H. or equivalent time. Remove fins and other protrusions by stoning, sanding or grinding. Abrasiva blast to open all surface voids and remove all form oils, incompatible curing agents, hardeners, laitance and other foreign matter and produce a surface texture similar to that of a medium grit sandpaper. Voids in the concrete may require surfacing. Blow or vacuum off sand and dust.

MIXING: Power mix separately, then combine and power mix

in the readwing proportions:	2 Gal. Kit	10 Gal. Kit
CARBOLINE 890 Part A	1 gallon	5 gallons
CARBOLINE 890 Part B	1 gallon	5 gallons

THINNING: For spray applications, may be thinned up to 13 oz./gal. with Thinner #2. For hot and windy conditions, or for brush and roller application, may be thinned up to 16 oz./gal. with Thinner #33.

Use of thinners other than those supplied or approved by Carboline may adversely affect product performance and void product warranty, whether express or implied.

POT LIFE: Three hours at 75°F (24°C) and less at higher temperatures. Pot life ends when material loses film build.

APPLICATION CONDITIONS:

	<u>Material</u>	Surfaces	_ Ambient_	Humidity
Normal	60-85°F	60-85°F	60-90°F	0-80%
	(16-29°C)	(16-29°C)	(16-32°C)	
Minimum	50°F (10°C)	50°F (10°C)	50°F (10°C)	0%
Maximum	90°F (32°C)	125°F (52°C)	110°F (43°C)	90%

Do not apply or cure the material when the surface temperature is less than 5°F or 3°C above the dew point.

Special thinning and application techniques may be required above or below normal conditions.

SPRAY: This is a high solids coating and may require slight adjustments in spray techniques. Wet film thicknesses are easily and quickly achieved. The following spray equipment has been found suitable and is available from manufacturers such as Binks, DeVilbiss and Graco.

Conventional: Pressure pot equipped with dual regulators, 3/8" I.D. minimum material hose, .070" I.D. fluid tip and appropriate air cap.

June 96 Replaces December 95

Airiess;	
Pump Ratio:	30:1 (min.)*
GPM Output:	3.0 (min.)
Material Hose:	3/8" 1.D. (min.)
Tip Size:	.017021
Output psi:	2100-2300
Filter Size:	60 mesh

*Teflon packings are recommended and are available from the pump manufacturer.

BRUSH OR ROLLER: Use medium bristle brush, or good quality short nap roller. Avoid excessive rebrushing and rerolling. Two coats may be required to obtain desired appearance, hiding and recommended DFT. For best results, tie-in within 10 minutes at 75°F (24°C).

DRYING TIMES: These times are based on a 5 mils (125 microns) dry film thickness. Higher film thicknesses, insufficient ventilation or cooler temperatures will require longer cure times and could result in solvent entrapment and premature failure.

Dry to Touch 2 1/2 hours at 75°F (24°C) Dry to Handle 6 1/2 hours at 75°F (24°C)

Surface	Recoating	Dry to		
Temperature	With Itself	Topcoat	Final Cure	
50°F (10°C)	12 hours	24 hours	3 days	
60°F (16°C)	8 hours	16 hours	2 days	
75°F (24°C)	4 hours	8 hours	1 day	
90°F (32°C)	2 hours	4 hours	16 hours	

Excessive humidity or condensation on the surface during curing can interfere with the cure, can cause discoloration and may result in a surface haze or blush. Any haze or blush <u>must</u> be removed by water washing before recoating. During high humidity conditions, it is recommended that the application be done while temperatures are increasing. For best results over "damp" surfaces, apply by brush or roller.

Maximum Recoat or Topcoat Times at 75°F (24°C):

With Epoxies - 30 days

With Polyurathanes - 90 days

If the maximum recoat time has been exceeded, surface must be abraded by sweep blasting prior to the application of any additional coats.

Minimum cure time before immersion service is 5 days at 75°F (24°C) surface temperature. Cure at temperatures below 60°F (16°C) is not recommended for immersion service.

VENTILATION & SAFETY: WARNING: VAPORS MAY CAUSE EXPLOSION. When used as a tank lining or in enclosed areas, thorough air circulation must be used during and after application until the coating is cured. The ventilation system should be capable of preventing the solvent vapor concentration from reaching the lower explosion limit for the solvents used. In addition to ensuring proper ventilation, fresh air respirators or fresh air hoods must be used by all application personnel. Where flammable solvents exist, explosion-proof lighting must be used. Hypersensitive persons should wear clean, protective clothing, gloves and/or protective cream on face, hands and all exposed areas.

CLEANUP: Use Thinner # 2.

CAUTION: READ AND FOLLOW ALL CAUTION STATEMENTS ON THIS PRODUCT DATA SHEET AND ON THE MATERIAL SAFETY DATA SHEET FOR THIS PRODUCT.

CAUTION: CONTAINS FLAMMABLE SOLVENTS. KEEP AWAY FROM SPARKS AND OPEN FLAMES. IN CONFINED AREAS, WORKMEN MUST WEAR FRESH AIRLINE RESPIRATORS. HYPERSENSITIVE PERSONS SHOULD WEAR GLOVES OR USE PROTECTIVE CREAM. ALL ELECTRIC EQUIPMENT AND INSTALLATIONS SHOULD BE MADE AND GROUNDED IN ACCORDANCE WITH THE NATIONAL ELECTRICAL CODE. IN AREAS WHERE EXPLOSION HAZARDS EXIST, WORKMEN SHOULD BE REQUIRED TO USE NONFERROUS TOOLS AND TO WEAR CONDUCTIVE AND NONSPARKING SHOES.



Page 1.C-6



N-5000 NICKEL BASED - NUCLEAR GRADE ANTI-SEIZE LUBRICANT

N-5000 is a nickel based nuclear grade anti-seize lubricant produced under 100% controlled conditions for highest purity and traceability. It is formulated to have the lowest practical levels of halogens, sulfur, and heavy metals, including copper. N-5000 has a general composition of nickel and graphite fiake in petroleum carrier. All ingredients are selected for extreme purity. It meets or exceeds the following specifications, appendix A of NEDE-31295P, "BWR Operator's Manual for Materials and Processes", Westinghouse Material Specification 53701WQ, and 10CFR Ch1, Part 21, and Part 50, appendix B.

Special Features:

Mercun

- High purity- made from highest purity ingredients.
- Traceability- each can marked.
 Free from copper- less than 50 ppm copper.
- _____Testing- each batch tested before packaging.
- Certifications- 3 copies with each case.

Recommended applications:

 Bolts, studs, valves, pipe fittings, slip fits and press fits in electric power generating plants, chemical plants, pharmaceutical plants, paper mills, and other locations where stainless steel fasteners are used.

Operational Benefits:

- Before assembly certifications and traceability.
- During assembly prevents high friction, galling,
- and seizing. Promotes uniform and predictable clamping.
- During operation high purity prevents stress corrosion.
- Disassembly prevents seizing, galling, destruction of threads.

Typical Physical Prope	rues:			
Composition - Service			Nickel	nd Graphic in Petroleum Dit
Appearance				Silver-Gray paste
Specification				
Flash point (ASTM D 92	-85)			424°F/218°C
statorelie/coefficient/ic/Su	elinets and Bolts)			
(Type 304 Stainless)				0.18
w.Maximum ase temperati				1172 - 118003E/982 EF
Quality Control Physical	Properties:			
2 WeldhupergeilleritASIN	MD (1475-85) 2011 2012			Range 915 Ed 014
Penetration (ASTM D 21	7-88 unworked)			300 - 380
Purity:			·	
Purity: Impurities - Elemental	Test Method Type	References	Controlled	Average Values
Purity: Impurities - Elemental and Combined	Test Method Type	References ASTM OR (SM16)	Controlled Maximum	Average Values
Purity: Impurities - Elemental and Combined Halogens: Chlonnes	Test Method Type	References ASTM OR (SM16)	Controlled Maximum	Average Values
Purity: Impurities - Elemental and Combined Halogens: Chlorine Biomine Iodine 1944	Test Method Type	References ASTM OR (SM16) D808-87, C69979	Controlled Maximum	Average Values
Purity: Impurities - Elemental and Combined Halogens: Chlonne Halogens: Chlonne Halogens: Chlonne Halogens: Chlonne Halogens: Chlonne Halogens: Chlonne	Test Method Type Parr Bomb, Turbidimetric Parr Bomb, Specific ION	References ASTM OR (SM16) D808287, C69979 D3761-84	Controlled Maximum 450 ppm 200 ppm	Average Values
Purity: Impurities - Elemental and Combined Halogens: Chlonne Biomine Ilodine: 14 Fluorine	Test Method Type Parr Bomb, Turbidimetric Parr Bomb, Specific ION Electrode	References ASTM OR (SM16) D808-87 C69979 D3761-84	Controlled Maximum 50 ppm 200 ppm	Average Values
Purity: Impurities - Elemental and Combined Halogens: Chlonne Biomine Ilodine Fluorine	Test Method Type Parr Bomb, Turbidimetric Parr Bomb, Specific ION Electrode Parr Bomb, Turbidimetric	References ASTM OR (SM16) D808:87, C69979 D3761-84 D129-64, D1266-87	Controlled Maximum 50 ppm 200 ppm	Average Values 18 ppm 7 ppm 9 ppm
Purity: Impurities - Elemental and Combined Halogens: Chlonne Biomine Iodine Fluorine Sultur Lead	Test Method Type Parr Bomb, Turbidimetric Parr Bomb, Specific ION Electrode Parr Bomb, Turbidimetric Wet Digestion, AAS	References ASTM OR (SM16) D808/87, C69979 D3761-84 D129-64 D1266:87 (302D), D3559-84	Controlled Maximum 50 ppm 200 ppm 400 ppm 25 ppm	Average Values 18 ppm 7 ppm 9 ppm 1ppm
Purity: Impurities - Elemental and Combined Halogens: Chlonne Biomine Hodine Fluorine Sultur Lead	Test Method Type Parr Bomb, Turbidimetric Parr Bomb, Specific ION Electrode Parr, Bomb, Turbidimetric Wet Digestion, AAS Wet Digestion, AAS	References ASTM OR (SM16) D808-87, C6997/9 D3761-84 D129-64, D1266-87, (302D), D3559-84 (302D), D3557-84	Controlled Maximum 450 ppm 200 ppm 100 ppm 25 ppm 2 ppm	Average Values 18 ppm 7 ppm 9 ppm 1ppm 02 ppm
Purity: Impurities - Elemental and Combined Halogens: Chlonnes Biomine Ilodine Fluorine Sultur Lead Admium	Test Method Type Parr Bomb Turbidimetric Parr Bomb, Specific ION Electrode Parr Bomb Turbidimetric Wet Digestion, AAS Wet Digestion, AAS	References ASTM OR (SM16) D808/87/C6997/9 D3761-84 D129-64/D1266/87/ (302D), D3559-84 (302D), D3557-84 (302D), E37-76	Controlled Maximum 350 ppm 200 ppm 200 ppm 2100 ppm 25 ppm 22 ppm 25 ppm	Average Values 18 ppm 7 ppm 9 ppm 1ppm 02 ppm 9 ppm
Purity: Impurities - Elemental and Combined Halogens: Chlonne Biomine Ilodine Fluorine Sultur Lead Idmium	Test Method Type Parr Bomb, Turbidimetric Parr Bomb, Specific ION Electrode Parr, Bomb, Turbidimetric Wet Digestion, AAS Wet Digestion, AAS Wet Digestion, AAS Wet Digestion, AAS	References ASTM OR (SM16) D808:87, C6997/9 D3761-84 D129:64, D1266:87 (302D), D3559-84 (302D), D3557-84 (302D), E37-76 (302D), D1691-84	Controlled Maximum 350 ppm 200 ppm 200 ppm 2100 ppm 25 ppm 25 ppm 25 ppm	Average Values 18 ppm 7 ppm 9 ppm 1ppm 02 ppm 9 ppm 1 ppm 1 ppm

Digestion & Gold Vapors

Directions for use:

- Before or during assembly, wipe brush onto threads and other joint surfaces needing protection.
- Do not overuse, as excess will be pushed off.
- Use full strength, do not thin.

Packaging:

Part Number	Net Contents is	Type Container	Units/Case	Shipping Wti Case
51243	8 oz. (227 g)	Can-brush top	12	9 lb. (4. Kg.)
51245	8 lb. (3.6 kg)	Can	2	18 lb. (8. Kg.)
51246	2 lb. (908 g)	Can	12	29 lb. (13. Kg.)
51269	1 lb. (454 g)	Can-brush top	· 12	16 lb. (7. Kg.)
51346	1 oz. (28 g)	Tube	48	6 lb. (2.7 Kg.)

N-5000 has an unlimited shelf life when stored at room temperature in the original unopened container.

FOR INDUSTRIAL USE ONLY.

WASH THOROUGHLY AFTER HANDLING. KEEP OUT OF REACH OF CHILDREN. SEE MATERIAL SAFETY DATA For immediate answers to your technical questions, in the United States or Canada call the Technical Support Line at 1-800-992-9799. 'nternational customers call (303) 289-5651, or fax (303) 289-5283

For a Material Safety Data Sheet or Technical Bulletin on this or any Fel-Pro product call our toll-free FAX FOR THE INFO line 24 hours a day, 7 days a week, in the United States or Canada call 800-583-3069. International customers call (303) 289-5651, or fax (303) 289-5283.

Except as expressly stipulated, Fel-Pro's liability, expressed or implied, is limited to the stated selling price . of any defective goods.

N-5000 8/97

412 W. Touty Ave. olnwood, 45 U.S.A 568-2820 Fay 847-574-0019

6120 E. 58th Ave Commerce City, CO 80022 U.S.A 800-992-9799 Fax 303-289-5283

ro of Canada, Ltd 6105 Kastral Road Mississauga, Onti L5T 1Y8 Canada Fax 905-564-1534

Fel-Pro Ltd. 4 Arkwright Way North Newmoor North Newmoor, Irvine KA11 4JU Scotland Fax 44-1294-218157

FEL-PRO CHEMICAL PRODUCTS, L.P. Fel-Pro Chamical Products Latin America L.P. Bodega No. 12, Zona Franca Palmaseca Asropuento Internacional Bonilla Aragon Cali, Colombia 67-2-551-1168 57-2-651-1168 Fax 57-2-651-1179

Fel-Pro Chamical Products, Chile S.A. Av. Pote. Eduardo Frei M. 9231 Quilicura Casilla (P.O. Box) 14325 Santiago, Chile 56-2-623-9216 Fax 56-2-623-9216

APPENDIX 1.D

COMMENT RESOLUTION LETTERS

(83 TOTAL PAGES INCLUDING THIS PAGE)

5014



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND FEDEX

January 15, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter A

References: 1. Holtec Project No. 5014

2. Meeting Between NRC/SFPO and Holtec International held January 13, 1999

Dear Mr. Delligatti,

We are pleased to note that the Spent Fuel Project Office (SFPO) staff has completed it acceptance review of Revision 7 of Holtec International's Safety Analysis Report (SAR) submitted in support of our application for a 10CFR71 Certificate of Compliance (CoC) for the HI-STAR 100 System. This Comment Resolution Letter (CRL), first in the expected series of letters to document the responses to the SFPO's questions in an expedited and interactive mode, provides a synopsis of the January 13, 1999 meeting commitments. The SFPO's requests and Holtec's commitments thereto are listed below by reference to the appropriate SAR Chapters.

Chapter 1 – General Information

A.1.1 Fuel data in Tables 1.2.10 and 1.2.11 must be reviewed for consistency with HI-STAR storage Technical Specification Tables 2.1-2 and 2.1-3. Examples of differences include the absence of maximum planar average enrichments limits in SAR Table 1.2.1 and the U-235 mass for BWR assembly array/class 6X6C fuel is different.

Commitment

SAR Tables 1.2.10 and 1.2.11 will be reviewed against the HI-STAR storage technical specifications and appropriate corrections will be made. Draft Revision 8 SAR pages will be provided to the NRC by February 1, 1999.

Telephone (609) 797-0900 Fax (609) 797-0909

.2

HOLTEC INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 2 of 17

A.1.2 Throughout the SAR, reference is made to NUREG-1536. The correct Standard Review plan to reference for transportation is draft NUREG-1617.

Commitment

Holtec will thoroughly review all chapters of the SAR for references to NUREG-1536. If the reference is correct in context for the dual purpose system, it may be left in the SAR with an additional entry made to provide the appropriate reference to draft NUREG-1617. These changes will be incorporated in Revision 8 of the SAR, to be submitted to the NRC by February 22, 1999.

A.1.3 Several drawings included with SAR Revision 7 are a different revision level than those in Revision 9 of the HI-STAR 100 TSAR.

Commitment and Comment

The drawing revisions in Revision 8 of the SAR are correct. A small number of HI-STAR 100 drawings have been revised since the submittal of Revision 8 of the HI-STAR 100 TSAR in August, 1998. These revisions were made to incorporate lessons learned from the HI-STAR 100 prototype fabrication, clarify a number of Bill-of-Material items, and to reflect the current weld inspection requirements for the MPC lid-to-shell weld (i.e., allowing a multi-layer PT examination in lieu of volumetric examination). Upon receipt of the HI-STAR 100 storage CoC, Holtec will submit a conforming revision to the TSAR which will include updating any drawings affected by these revisions. This TSAR revision will be submitted within 90 days of receipt of the HI-STAR 100 storage CoC. Additionally, a list of the differences between the drawings in the HI-STAR TSAR and the HI-STAR SAR will be provided in a comment resolution letter to be submitted by February 8, 1999.

A.1.4 Table 1.2.1 lists the maximum weight of a loaded HI-STAR 100 as 254,617 lbs. This is greater than the design load for the overpack's lifting trunnions. Please clarify how the general licensees are to ensure the trunnions are not loaded beyond their design limit when lifting a loaded HI-STAR during any phase of loading operations.

HOLTEC INTERNATIONAL Ir. Mark Delligatti

Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 3 of 17

Commitment

A review of this SAR table revealed that 3,600 lbs of the total weight is from the lift rig itself. While the lift rig weight may be subtracted from the load on the trunnions, the user's crane will be required to have the capacity to lift the rigging as well as the loaded HI-STAR. Holtec will review SAR Chapters 1 and 2 and the operating procedures in SAR Chapter 7 to ensure appropriate guidance is provided to licensees to ensure neither the HI-STAR trunnions nor the cranes lifting the HI-STAR are loaded beyond their design capacity. Draft Revision 8 of SAR Table 1.2.1 and other affected SAR sections will be provided to the NRC by February 3, 1999.

A.1.5 In Table 1.3.3, specify the surface preservative to be authorized for use as a corrosion inhibitor between the overpack intermediate shells.

Commitment

}

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.1.6 On SAR page 1.2-20, the description of the mesh size for the screen in the Damaged Fuel Container is inconsistent with the Bill-of-Material.

Commitment

The SAR text will be revised to match the Bill-of-Material. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.1.7 Drawing 1765, Sheet 6 shows the nominal impact limiter crush strengths. Holtec needs to ensure the nominal crush strength, including tolerances does not cause make the impact limiter crush strength to exceed the values used in the ¼ scale drop tests.

Commitment

Nominal values of impact limiter crush strength (F_c) will be specified in Drawing 1765 such that the range of F_c (sum of nominal plus tolerance) will not exceed the actual

Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 4 of 17

values of F_c used in the $\frac{1}{4}$ scale drop tests. Draft Revision 8 SAR pages and draft drawings changes will be submitted to the NRC by February 3, 1999.

A.1.8 On SAR page 1.3-1, the MPC lid-to-shell weld inspection specifies that sufficient intermediate layers are to be inspected by liquid penetrant (PT) examination to detect critical weld flaws. Specify the critical weld flaw size and the required number of weld passes to be inspected by the PT method.

Commitment

See the commitment under item A.2.6.

Chapter 2 - Structural Evaluation

A.2.1 Some allowable stresses in SAR Tables 2.1.3, 2.1.11, and 2.1.12 exceed the ultimate stresses for the materials. The SAR needs to be reviewed and revised where appropriate to ensure the stress criteria of Regulatory Guide (RG) 7.6 are met and all loading combinations of RG 7.8 have been considered.

Commitment

The SAR tables were reviewed with the NRC reviewer at the January 13, 1999 meeting and all stresses were found to be below the ultimate stresses for the material. No further action is required. Chapter 2 of the SAR will be reviewed against RG 7.6 and 7.8 to ensure the appropriate stress limits and loading combinations have been used in the HI-STAR 100 structural analyses. The revised SAR will make explicit commitments to Reg. Guide 7.6 for the containment boundary and document compliance with the stress limits and brittle fracture criteria therein. The adherence of the SAR to Reg. Guide 7.8 load combinations will be documented in the revised SAR with sufficient explanatory material to demonstrate explicit compliance. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.2.2 The outer two shells of the HI-STAR 100 overpack provide protection for the containment boundary. As such, these two shells should meet the criteria of ASME, Section III, Subsection NF, Class 1 rather than Class 3.



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 5 of 17

Commitment

The two outermost intermediate shells (which are welded with single bevel full penetration weld), shall meet the stress intensity limits and brittle fracture criteria for Section III, Subsection NF, Class 1 plate and shell structures. Affected pages of the SAR will be revised. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.2.3 Figure 2.1.10 does not contain sufficient information describing how the loads are applied for the CG-over-corner drop analysis. Details are needed on loads, load application, and resultant stresses for the HI-STAR 100 structural components. (MPC shell and baseplate and the overpack forgings, endplates, bolts, etc).

Commitment

)

A complete stress analysis of the HI-STAR 100 overpack under the C.G.-Over-Corner drop scenario will be presented in the SAR. The maximum values of primary, membrane, local membrane, and primary bending stress intensities in the overpack will be summarized in the SAR. The axial stress in the lid bolts will be shown to remain below the bolt material's nominal yield strength. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.2.4 More information is required on the applied load distributions for side drop analyses.

Commitment

The presentation of the result of the side drop analyses in the SAR will be reviewed and changes made where necessary to add additional information. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

Telephone (609) 797-0900 Fax (609) 797-0909



Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 6 of 17

A.2.5 The justification for not analyzing the oblique drop is inadequate.

<u>Commitment</u>

The value of the peak deceleration (a_{max}) as a function of the oblique drop angle will be computed at 15° intervals up to the C.G.-Over-Corner orientation. The maximum predicted value of the a_{max} from the parametric study shall be used in the stress analysis of item A.2.3 above to bound all oblique drop scenarios. Draft Revision 8 SAR pages will be submitted to the NRC by February 5, 1999.

A.2.6 Figures 2.H.5.14 through 2.H.5.21 in Appendix 2.H showing the drop test data are not in the appropriate time scale to show the data necessary for evaluating the drop test. The time scale should be of shorter duration.

Commitment

Acceleration test data shall be re-plotted to improve its readability during the time when the impact load is maximum (before its sharp attenuation). Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.2.7 The actual location of the accelerometer placement on the overpack ¼ scale model needs to be provided. The filtered data should be superimposed over the unfiltered data.

Commitment

The precise locations of accelerometers will be shown in Appendix 2.H of the SAR. The filtered and unfiltered data will be superimposed on the same plot. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.2.8 The sensitivity of the test data to filter frequency should be determined for the end drop. A minimum filter frequency of 500 Hz or the preferred filter of 550 Hz should be used.



Telephone (609) 797-0900 Fax (609) 797-0909

Commitment

Page 7 of 17

١

Filtering will be performed at 550 Hz for the case of the end drop to evaluate the sensitivity of the predicted peak acceleration to the filter frequency. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.2.9 The critical flaw size for the MPC lid-to-shell weld must be defined. A fracture mechanics analysis would be acceptable.

<u>Complitment</u>

A computation for the acceptable crack flaw size in the MPC lid weld will be prepared which determines the critical flaw size and number of passes for the multi-layer PT examination of the weld. A telephone call with the reviewer will be required to discuss the approach to such an analysis considering the material under consideration is stainless steel. The calculation and appropriate draft revisions to the SAR text will be submitted to the NRC by February 10, 1999.

A.2.10 The thermal stresses for a transient condition where a cask at steady state conditions is moved to a relatively cold ambient location should be analyzed.

Commitment

An integrity evaluation of the containment boundary under the postulate of a sudden drop in the ambient temperature (e.g., from 100°F to -40°F in one hour) will be performed. The thermal stresses in the containment boundary will be shown to not exceed Level A service limits for Section III Class 1 components. Appropriate revisions to the SAR text will be submitted to the NRC by February 10, 1999.

Chapter 3 – Thermal Evaluation

A.3.1 In a fire accident, the Holtite-A will overheat. The offgassing created by the overheating of the neutron shield material needs to be addressed. (This may affect the structural chapter also).

Telephone (609) 797-0900 Fax (609) 797-0909



Commitment

The HI-STAR 100 overpack is equipped with two rupture disks to relieve any internal pressure above the set limit. The text of SAR Chapters 2 and 3 will be revised to address this issue. Draft Revision 8 SAR pages will be submitted to the NRC by February 8, 1999.

A.3.2 Provide a summary of cask hot normal transportation condition temperatures after eliminating credit for the "Rayleigh Effect" (Nussault Number) in the MPC basket periphery.

Commitment

The analysis will be performed as requested and the results provided to the NRC in a comment resolution letter by February 8, 1999.

A.3.3 Provide clarification in the SAR as to the fraction of heat transmitted from the overpack through the top and bottom.

Commitment

The HI-STAR 100 thermal model takes no credit for heat transfer through the overpack impact limiters. The SAR text will be reviewed and clarification provided. Draft Revision 8 SAR pages will be submitted to the NRC by February 8, 1999.

A.3.4 What is the maximum temperature of the accessible surfaces of the casks? 10CFR71.43(g) requires the maximum temperature at all accessible surfaces of the package to be less than 185° F.

Commitment

Access to the sides of the HI-STAR 100 overpack is restricted through the use of a personnel barrier. Access is possible to the impact limiters at the top and bottom of the cask. A specific analysis will be run to determine the impact limiter surface temperature by allowing heat transfer through the top and bottom of the cask. The text of the SAR

INTERNATIONAL

Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 9 of 17

> will be revised to include this information. Draft Revision 8 SAR pages will be submitted to the NRC by February 8, 1999.

A.3.5 EPRI Report TR-106440 should be used for the short term fuel cladding temperature limit for stainless steel clad fuel as the basis.

Commitment

SAR Chapters 2 and 3 will be changed to incorporate this reference and the new temperature limit. Draft Revision 8 SAR pages will be submitted to the NRC by February 8, 1999.

)

A.3.6 The pressure value tables need to be revised to include the average bulk gas temperatures used in the pressure calculation.

Commitment

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

Chapter 4 - Containment

A.4.1 The SAR text references the 1987 edition of ANSI N 14.5. This should be the 1997 edition.

Commitment

The 1997 edition was used. Draft Revision 8 of the HI-STAR SAR pages are provided in Attachment 2 to this letter.

A.4.2 In section 4.1.5, the description of fuel debris should be revised to match that used in the HI-STAR 100 storage Technical Specifications.

Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 10 of 17

Commitment

The SAR will be revised to be consistent with the HI-STAR 100 storage Technical Specifications. Draft Revision 8 SAR pages are provided in Attachment 2 to this letter.

A.4.3 On SAR page 4.1-5, the description of the mesh size for the screen in the Damaged Fuel Container is inconsistent with the Bill-of-Material.

<u>Commitment</u>

The SAR text will be revised to match the Bill-of-Material. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

Chapter 5 – Shielding Evaluation

A.5.1 Provide the coordinates and dimensions of the dose point locations depicted in Figures 5.1.1 and 5.1.2.

Commitment

The coordinates and dimensions of the dose point locations for the shielding model used in SAR Chapter 5 are provided in Attachment 1 to this letter.

A.5.2 Provide a calculation showing the azimuthal variation in dose rate around the circumference of the cask on contact and at 2 meters to account for the fact that the fuel basket is not cylindrical.

<u>Commitment</u>

The calculation will be provided to the NRC by close of business January 20, 1999. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.



Telephone (609) 797-0900 Fax (609) 797-0909

A.5.3 Demonstrate that gamma source term energies outside of the range of 0.7 to 3.0 MeV do not add a significant contribution to the total dose rate. Address the full range of cooling times.

Commitment

The calculation will be provided to the NRC by close of business January 20, 1999. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.5.4 Provide dose rate values for 11, 13, and 14 years cooling time in Chapter 5 for the MPC-24 and MPC-68.

Commitment

ì

Chapter 5 will be revised to add dose rates for the additional cooling times shown in SAR Chapter 1 for the MPC-24 and MPC-68. Draft Revision 8 SAR pages will be submitted to the NRC by January 27, 1999.

A.5.5 Describe how the impact limiters were modeled in the shielding analyses. The text on the top of SAR page 5.3-1 appears to conflict with the text on page 5.3-2.

Commitment

The SAR text will be clarified to accurately describe the modeling of the impact limiters for both normal and accident conditions. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

A.5.6 It is not clear from SAR Figure 5.3.10 and the text of Section 5.3 which types of steel make up which portions of the model.

Commitment

The figure and text will be reviewed and clarified as necessary to clearly describe the steel types assumed in the model. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.



Telephone (609) 797-0900 Fax (609) 797-0909

A.5.7 The dose rate at the pocket trunnions in Table 5.4.15 is above 10 mrem/hr.

Commitment

The dose rate at this point will be re-analyzed. The modeling assumptions will be discussed with the NRC reviewer before the analysis is performed. A proposed refinement in the calculations will be provided in a Comment Resolution Letter by January 20, 1999. A conference call will be held to discuss this issue with the reviewer by January 21, 1999. Upon agreement on the assumptions from the conference call, the calculation will be performed and the results transmitted to the NRC by January 25, 1999. Draft Revision 8 SAR pages will be submitted to the NRC February 3, 1999.

A.5.8 Provide a discussion of the dose rate gradient an the surface of the overpack at dose location 3.

Commitment

The discussion will be provided in a Comment Resolution Letter by January 20, 1999. A conference call will be held to discuss this issue with the reviewer by January 21, 1999. Changes to the SAR, if necessary will be provided by February 3, 1999.

Chapter 6 – Criticality Evaluation

A.6.1 SAR Section 6.1 needs to clearly summarize the maximum k_{eff} considering all biases and uncertainties from the calculations used to demonstrate compliance with Part 71 (71.55(b), (d), and (e) and 71.59(a).1 and (a).2). Refer to NUREG 1617, Section 6.5.1.3.

Commitment

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by January 27, 1999.

A.6.2 The single package evaluation in SAR section 6.4.2.1.1 needs to be revised to show a single package is subcritical when the 2 ½ in inner shell is closely and fully reflected by



Telephone (609) 797-0900 Fax (609) 797-0909

1.

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 13 of 17

water or justify that the reflection provided by the surrounding 6 inches of steel is greater than or equivalent to water.

Commitment

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by January 27, 1999.

A.6.3 Remove the suggestion in SAR page 6.1-5 that flooding or reflection with unborated water is a conservatism.

Commitment

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by January 27, 1999.

Chapter 7 – Operating Procedures

A.7.1 The operating procedures don't appear to address the items required to be verified per 10CFR71.87 prior to shipment. In particular, 10CFR71.87(g) requires verification that the neutron absorber is present and in proper condition. Note that both immediate shipment and shipment after some time in storage need to be addressed with unique requirements for shipment after storage clearly described.

Commitment

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by February 5, 1999.

A.7.2 There is no recognition in the operating procedures that the package may be shipped after some time in storage. Consider what extra steps are needed if shipping takes place after the package is in storage for some period of time (e.g., leakage testing requirements for the MPC-68F)

Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 14 of 17

Commitment

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by February 5, 1999.

A.7.3 Step 25.k of the operating procedures needs to specify the minimum number of multilayer PT exams required based on the minimum flaw size calculated.

<u>Commitment</u>

The requested information will be provided based on the results of item A.2.6 above. Draft Revision 8 SAR pages will be submitted to the NRC by February 10, 1999.

A.7.4 The warning at step 12 in Section 7.1.5 needs to be clarified to address the maximum trunnion loading issue (see item A.1.4).

Commitment

The operating procedures will be revised appropriately to address this issue based on the response to item A.1.4. Draft Revision 8 SAR pages will be submitted to the NRC by February 5, 1999.

A.7.5 Step 26.k in the operating procedures should be revised to include PT as an additional test to be re-performed after a weld is repaired.

Commitment

The requested information will be provided. Draft Revision 8 SAR pages will be submitted to the NRC by February 5, 1999.



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 15 of 17

Chapter 8 – Acceptance Criteria and Maintenance Program

A.8.1 Section 8.1.1.1, item 3 needs to specify the critical flaw size.

Commitment

The requested information will be provided based on the results of item A.2.6 above. Draft Revision 8 SAR pages will be submitted to the NRC by February 10, 1999.

A.8.2 The text of Section 8.1.2.1 may be affected by the resolution of the trunnion loading issue (item A.1.4).

Commitment

I

The Chapter 8 SAR text will be revised as necessary to address this issue based on the response to item A.1.4. If revisions are necessary, draft Revision 8 SAR pages, will be submitted to the NRC by February 3, 1999.

A.8.3 Table 8.1.3 may need revision to specify the number of intermediate PTs required for the multi-layer PT examinations.

Response

Table 8.3.1 is a summary of the required NDE inspections showing the applicable Code requirements. A more appropriate location for this information is Section 8.1.1.1, item 3. The affected text will be revised based on the results of item A.2.6. Draft Revision 8 SAR pages will be submitted to the NRC by February 10, 1999.

If you have any questions or comments, please contact us.

Sincerely,

ORIGINAL SIGNED BY BG

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 16 of 17

Document I.D.: 5014251

Attachments: 1. Coordinates and dimensions of dose point locations.

2. Draft SAR Revision 8 pages 4.1-4, 4.1-5, 4.2-15, and 4.4-1

Approvals:

Brian Gutherman Licensing Manager

Technical Concurrence:

Dr. Everett Redmond (Shielding)

Dr. John Wagner (Criticality)

Mr. Stephen Agace (Operations)

Dr. Alan Soler (Structural Analysis)

Dr. Indresh Rampall (Thermal Analysis)

Ms. Joy Russell (Containment)

KACingh

K. P. Singh, Ph.D. President and CEO

ussell



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 15, 1999 Page 17 of 17

Distribution:

Project

1

<u>Utility</u>

Holtec Project

71188

50438

71178

70651

70851

80518

Mr. David Bland
Mr. J. Nathan Leech
Mr. Bruce Patton
Dr. Max DeLong
Mr. Rodney Pickard
Mr. Ken Phy
Mr. David Larkin
Mr. Eric Meils
Mr. Paul Plante
Mr. Stan Miller
Mr. Jim Clark
Mr. Ray Kellar

Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Co. Private Fuel Storage, LLC American Electric Power New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One

#5014



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND FEDEX

January 16, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter A (errata)

References: 1. Holtec Project No. 5014

2. Letter, B. Gilligan, Holtec to M. Delligatti, NRC, dated January 15, 1999, Document ID 5014251

Dear Mr. Delligatti,

Upon reviewing the above-referenced letter to the NRC regarding HI-STAR 100 commitments, we have discovered an editorial error for which a correction is provided here. Commitments A.7.3, A.8.1, and A.8.3 cross-reference an incorrect commitment number from Chapter 2. Each of these commitments should cross-reference Commitment A.2.9, not A.2.6. Attached are markups of the affected pages from the Reference 2 letter showing the correct cross-reference. We regret any inconvenience this error may have caused.

If you have any questions or comments, please contact us.

Sincerely,

Brian Gutherman Licensing Manager

cc: B. Gilligan S. Agace

_____)ocument I.D.: 5014252

Attachment: Corrected pages from Holtec Letter of January 15, 1999 (Document ID 5014251)



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligati U. S. Nuclear Regulatory Commission January 16, 1999 Page 2 of 2

Distribution:

Project

١

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Cark Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar

<u>Utility</u>

Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Co. Private Fuel Storage, LLC American Electric Power New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One

Holtec Project



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

January 20, 1999

÷£

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter B

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter A from Holtec to NRC dated January 15, 1999

3. Errata to Comment Resolution Letter A, dated January 16, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide information in accordance with commitments A.2.9, A.5.2, A.5.3, A.5.7, and A.5.8 cited in Comment Resolution Letter (CRL) A (Reference 2). In addition, this CRL documents additional questions and commitments recently discussed with the Spent Fuel Project Office staff.

Commitment A.2.9

A position paper describing the proposed Holtec methodology for determining the critical weld flaw size in the MPC lid-to-shell weld and the results are provided as Attachment 1 to this letter.

Commitment A.5.2

The calculation of the azimuthal variation in dose rate is provided in Attachment 2 to this letter.
Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 20, 1999 Page 2 of 4

Commitment A.5.3

The evaluation of gamma source energies outside the range of 0.7 MeV to 3.0 MeV is provided in Attachment 2 to this letter.

Commitment A.5.7

A description of the proposed refinement of the calculation of the dose rate at the pocket trunnions is provided in Attachment 2 to this letter.

Commitment A.5.8

A discussion of the dose rate gradient at the surface of the overpack at dose location 3 is provided in Attachment 2 to this letter.

As requested in our Reference 2 letter, Holtec wishes to discuss the shielding information provided herein with the SFPO staff reviewer before completing the resolution of these comments. A conference call as soon as possible would be appreciated.

NEW ITEMS

- General: In the Reference 3 letter, three examples of an incorrect cross-reference in the Reference 2 letter were identified. Upon further review, one additional example of the same erroneous cross-reference was discovered. Commitment A.1.8 incorrectly references Commitment A.2.6. This cross-reference should be A.2.9.
- B.1.1 The footnote marker in the "BWR" column, "Design Temperature" row of Table 1.2.3 should be a double dagger, not a triple dagger.

Commitment

This item will be corrected and submitted with Revision 8 of the SAR to be submitted to the NRC by February 22, 1999.

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 20, 1999 Page 3 of 4

B.2.1 Provide additional information in the Safety Analysis report (SAR) to verify compliance with 10CFR71.45(a) and 10CFR71.45(b)(3) regarding the design of lifting attachments and tie-down devices under excessive loads.

Commitment

~~

Additional information will be added to the SAR to demonstrate compliance with the cited regulations. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

B.2.1 Provide additional information in the SAR Section 2.5.2.7.3 to clarify the loading, boundary conditions, and stress distribution for the pocket trunnion recess (refer to drawing 1399, Sheet 3)

Commitment

Additional information will be added to the SAR to clarify the information as requested. Draft Revision 8 SAR pages will be submitted to the NRC by February 3, 1999.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014253

- Attachments: 1. Holtec International Proprietary Position Paper DS-213, "Acceptable Flaw Size in MPC Lid-to-Shell Welds".
 - 2. Shielding Information Related to Commitments A.5.2, A.5.3, A.5.7, and A.5.8
 - 3. Affidavit Pursuant to 10CFR2.790

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 20, 1999 Page 4 of 4

Approvals:

FOR BRIAN OUTHERMAN

Brian Gutherman Licensing Manager

Blingh K. P. Singh, Ph.D.

President and CEO

Technical Concurrence:

Dr. Everett Redmond (Shielding)

Dr. Alan Soler (Structural Analysis)

Distribution:

Project

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Ken Phy Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar

<u>Utility</u>

Holtec Project



Telephone (609) 797-0900 Fax (609) 797-0909

 \mathcal{O}

BY FAX AND OVERNIGHT MAIL

January 22, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter C

References: 1. Holtec Project No. 5014

2. Phone Calls Between SFPO Staff and Holtec, held January 21 and 22, 1999

3. Comment Resolution Letter A, dated January 15, 1999 .

4. Comment Resolution Letter B, dated January 16, 1999

Dear Mr. Delligatti,

The purpose of this letter is to document questions from the Spent Fuel Project Office staff regarding the Holtec HI-STAR 100 System Safety Analysis Report (SAR) and Holtec's commitments to respond to those questions. These questions were received over the course of several telephone calls held January 21 and 22, 1999. The submittal dates for Commitment C.5.3 in this letter supersede the draft SAR page submittal dates for Commitments A.5.2 through A.5.8 made in Comment Resolution Letter A (Ref. 3). We are also providing an editorial correction to Comment Resolution Letter B (Ref. 4).

C.3.6 The bulk gas temperatures for the pressure values to be provided under commitment A.3.6 need to be submitted by February 1, 1999.

Commitment

The draft Revision 8 SAR pages will be provided by February 1, 1999

1

C.5.1 The staff requested verification that in Table 5.4.15 the dose rate at two meters for the neutron shield and the channel were the same.



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 22, 1999 Page 2 of 4

Response

The dose rates shown in Table 5.4.15 at two meters for the neutron shield and the channel are the same. No further action is required.

C.5.2 The staff requested that Inconel grid spacers be included in the dose rate assessment. Justification for the cobalt impurity level assumed in the Inconel grid spacers must be provided. A value of 4700 ppm is acceptable based on DOE data, or another value may be used with adequate supporting justification.

Commitment

Holtec will provide the NRC with its proposed cobalt impurity level in Inconel grid spacers via telephone call at 10:00 AM on January 26, 1999 with a confirming letter to follow.

C.5.3 The staff requested that minimum enrichments be specified in the CoC.

Commitment

Holtec will provide minimum enrichments, burnups, and cooling times used in the shielding analyses via draft Revision 8 SAR pages by noon January 29, 1999. Draft Revision 8 SAR pages showing dose rates at all dose locations for all burnups and cooling times from SAR Chapter 1 will be provided to the NRC by February 8, 1999. SAR Tables 5.5.1 and 5.5.3 will include peak dose rates. The balance of the dose rate tables will present average dose rates. Draft Revision 8 SAR text pages reflecting these changes and those listed below will be provided by February 10, 1999.

- Azimuthal variation of dose rate around the circumference of the cask (Commitment A.5.2)
- Source strength data (Commitment A.5.3)
- Description of how the impact limiters were modeled in the normal and hypothetical accident condition shielding analyses (Commitment A.5.5)
- Description of steel types used in the shielding model (Commitment A.5.6)



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 22, 1999 Page 3 of 4

- Dose rates at the pocket trunnions (Commitment A.5.7)
- Discussion of the dose rate gradient at dose location 3 (Commitment A.5.8)
- Location of the hypothetical accident two-meter dose location

Correction

Commitment B.2.1 is listed twice in Comment Resolution Letter B (Ref. 4). The second B.2.1 should be B.2.2.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing Document I.D.: 5014254

Approvals:

Brian Guthefman Licensing Manager

Technical Concurrence:

Dr. Everett Redmond (Shielding Evaluation)

Dr. Indresh Rampall (Thermal Evaluation)

K. P. Singh, Ph.D. President and CEO



INTERNATIONAL

Holtec Center, 555 Lincoln Drive West, Marlton, NJ 08053

Telephone (609) 797-0900 Fax (609) 797-0909

Holtec Project

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 22, 1999 Page 4 of 4

Distribution:

Project

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Can Phy Mr. David Larkin Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar

<u>Utility</u>

Southern Nuclear Operating Company71188Commonwealth Edison50438Pacific Gas & Electric Co.71178Private Fuel Storage, LLC70651American Electric Power70851New York Power Authority80518Washington Public Power Supply SystemWisconsin Electric Power CompanyMaine Yankee Atomic Power Company

Entergy Operations - Arkansas Nuclear One

Vermont Yankee Corporation

Southern California Edison

.



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

January 27, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter D

References: 1. Holtec Project No. 5014

- 2. Phone Calls Between SFPO Staff and Holtec, held January 26 and 27, 1999
- 3. Holtec Comment Resolution Letter "A", B. Gilligan to NRC, M. Delligatti, dated January 15, 1999
- 4. Holtec Comment Resolution Letter "B", B. Gilligan to NRC, M. Delligatti, dated January 22, 1999
- 5. Holtec Comment Resolution Letter "C", B. Gilligan to NRC, M. Delligatti, dated January 22, 1999

Dear Mr. Delligatti,

Ì

The purpose of this letter is to provide information in accordance with previous commitments to the Spent Fuel Project Office staff regarding the Holtec HI-STAR 100 System Safety Analysis Report (SAR). Where draft Revision 8 SAR pages are provided, the commitment number to which they apply is noted on the top of each page. In addition, new questions and commitments arising from telephone calls held January 26 and 27, 1999 are provided. Please refer to Comment Resolution Letters (CRL) A, B, and C (Refs 3 – 5) for the detailed description of each question and commitment statement for which the responses are provided below. This letter meets commitment C.5.2 of CRL "C" (Ref. 5). Note that Commitment D.5.1 modifies one commitment date for Commitment C.5.3 based on discussions with the SFPO staff and NRC Project Manager.



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 27, 1999 Page 2 of 8

Commitment A.2.1

Enclosed please find draft Revision 8 of SAR Section 2.1.2 which has been revised to clarify stress limits and load combinations as a result of our review of Regulatory Guides 7.6 and 7.8

Commitments A.2.6, A.2.7, and A.2.8

Enclosed please find draft Revision 8 of SAR pages 2.H-14 through 2.H-18 and Figures 2.H.5.14 through 2.H.5.21A which discuss and depict the cask drop test data, respectively. Rather than revise the previous figures, new Figures 2.H.5.15A, 15B, 15C, 17A, 19A, and 21A were developed to address the issues which are the subjects of these commitments. The entire series of figures is enclosed for continuity.

Commitment B.2.1

Enclosed please find draft Revision 8 of SAR Section 2.5.3 and Appendix 2.B which address compliance with 10CFR71.45(a) and 10CFR71.45(b)(3) regarding the design of lifting attachments and tie down devices under excessive loads.

Commitment B.2.2

Enclosed please find the following draft Revision 8 of SAR documents which clarify the loading, boundary conditions, and stress distributions for the pocket trunnion recess:

- Revised SAR Section 2.5.2.7
- Revised SAR Section 2.5.2.7.3
- New SAR Section 2.5.2.7.4
- New Figures 2.5.10 and 2.5.11
- New Appendix 2.AN

Commitment A.3.6

Enclosed please find draft Revision 8 of SAR Table 3.4.15 which provides the MPC cavity bulk gas temperatures added to compliment the temperature data previously presented in the table.



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 27, 1999 Page 3 of 8

Commitment A.4.3

Enclosed please find draft Revision 8 of SAR page 4.1-5 which clarifies the description of the screen mesh for the Damaged Fuel Container.

Commitment A.6.1

Enclosed please find draft Revision 8 of SAR pages 6.1-6, 6.1-11, and 6.4-4 which summarize the maximum k_{eff} considering all biases and uncertainties from the calculations to demonstrate compliance with 10CFR71.55 and 71.59.

Commitment A.6.2

Enclosed please find draft Revision 8 of SAR pages 6.4-3 and 6.4-17 which discuss the criticality evaluation considering water surrounding the containment boundary of the cask.

Commitment A.6.3

Enclosed please find draft Revision 8 of SAR pages 6.1-4, 6.1-5, 6.4-2, and 6.4-6 which have been revised to remove the assertion that flooding or reflection with unborated water is a conservatism.

NEW ITEMS

D.5.1: The use of a cobalt-59 impurity level less than 4700 ppm in Inconel grid spacers must be justified or the value of 4700 ppm must be used. Alternatives include a condition of the Certificate of Compliance to require users to verify that the cobalt impurity level in their fuel is less than that assumed in the analysis prior to loading fuel, or two sets of analyses can be presented for fuel with, and without Inconel grid spacers, respectively.

Commitment

A cobalt-59 impurity level of 4700 ppm will be used as a bounding case for modeling Inconel grid spacers in the shielding analysis. Holtec is currently considering the feasibility of providing two sets of shielding data for fuel assemblies with and without



January 27, 1999

Page 4 of 8

Telephone (609) 797-0900 Fax (609) 797-0909

Inconel grid spacers. Minimum enrichments, burnups, and cooling times used in the BWR shielding analyses will be submitted to the NRC by noon January 29, 1999.

Minimum enrichments, burnups, and cooling times used in the PWR shielding analyses will be submitted to the NRC, and Holtec will inform the NRC of its decision on whether to pursue two sets of shielding analyses by the close of business February 2, 1999. If we proceed with the second set of shielding analyses for fuel without Inconel grid spacers, minimum enrichments, burnups, and cooling times used in those PWR shielding analyses will be submitted to the NRC by noon February 5, 1999. Other commitment dates for Commitment C.5.3 remain the same.

D.5.2: Clarify SAR Section 5.2.3 regarding the distribution of uranium mass over the 144-in assumed active fuel length.

Commitment

SAR Section 5.2.3 will be revised to clarify this information. Draft Revision 8 of SAR pages reflecting these changes will be submitted by February 10, 1999.

D.5.3: For the stainless steel fuel, provide dose rates at dose locations other than dose point 2.

Commitment

Dose rates at additional dose point locations will be provided in a Comment Resolution Letter by February 8, 1999. Draft Revision 8 of SAR pages reflecting these changes will be submitted by February 10, 1999.

D.5.4: Revise Figure 5.3.10 or add a new figure to provide the dimensions of the impact limiter modeled in the shielding analysis.

Commitment

The dimensions will be provided in a Comment Resolution Letter by noon January 28, 1999. Final SAR Revision 8, to be submitted by February 22, 1999 will incorporate these changes.



3

Holtec Center, 555 Lincoln Drive West, Marlton, NJ 08053

Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 27, 1999 Page 5 of 8

D.5.5: The fuel mixtures on page 5.C-20 (MCNP input data) are different than the data in Table 5.3.2.

Commitment

The table is incorrect. A corrected draft Revision 8 of SAR Table 5.3.2 will be submitted to the NRC by February 10, 1999.

D.7.1: Revise SAR Section 7.3 to refer to 49CFR173.428 instead of §173.443

Commitment

The affected SAR pages will be revised and submitted to the NRC with Revision 8 of the SAR by February 22, 1999.

D.8.1: The shielding effectiveness test described in SAR Section 8.1.5.2, which is taken from the HI-STAR 100 storage TSAR may not be comprehensive enough for a transportation package. Because the regulations impose dose rate limits "at any point on the package", taking measurements at twelve locations around the circumference may not be adequate. Other applications have been approved with license conditions for shielding tests performed in 6" X 6" test grids.

Commitment

Holtec will research other Certificates of Compliance for this information. The NRC reviewer will discuss this issue with management and advise Holtec. If more stringent shielding test requirements are necessary, Holtec will submit draft revised Revision 8 SAR pages to the NRC by February 10, 1999.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing Document I.D.: 5014255



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 27, 1999 Page 6 of 8

Approvals:

Brian Gutherman Licensing Manager

Enclosures: As Stated

Technical Concurrence:

Dr. Everett Redmond (Shielding Evaluation)

Dr. Indresh Rampall (Thermal Evaluation)

Ms. Joy Russell (Containment Evaluation)

Dr. John Wagner (Criticality Evaluation)

Dr. Alan Soler (Structural Evaluation)

Mr. Steve Agace (Operations)

Distribution:

Project

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Ken Phy Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Co. Private Fuel Storage, LLC American Electric Power New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company

K.P. Sergh /mP

K. P. Singh, Ph.D. President and CEO

hund

FOR DE. INDRESH to Stre Agrice

Holtec Project



Telephone (609) 797-0900 Fax (609) 797-0909

Distribution (cont'd):

÷.,

Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar

January 27, 1999

Page 7 of 8

2

)

Ì

Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 27, 1999 Page 8 of 8

ENCLOSURES TO DOCUMENT 5014255

DRAFT REVISION 8 SAR PAGES

62 PAGES TOTAL



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

January 28, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter E

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter D, dated January 27, 1999

Dear Mr. Delligatti,

Attached please find three pages showing the dimensions of the HI-STAR 100 System impact limiter as modeled in the shielding analysis. This information is provided in accordance with Commitment D.5.4. Figure 5.3.10 will be revised to include this information in Revision 8 of the HI-STAR 100 System Safety Analysis Report to be submitted to the NRC by February 22, 1999. If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing Document I.D.: 5014256 Attachment: As Stated

Approvals:

Brian Gutherman Licensing Manager

rich IMP

K. P. Singh, Ph.D. President and CEO



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 28, 1999 Page 2 of 2

Technical Concurrence:

Dr. Everett Redmond (Shielding Evaluation)

. Tha

Distribution:

Project

<u>Utility</u>

Holtec Project

71188

50438

71178

70651

70851

80518

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Ken Phy Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar

Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Co. Private Fuel Storage, LLC American Electric Power New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations - Arkansas Nuclear One



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

January 29, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter F

References: 1. Holtec Project No. 5014

- 2. Comment Resolution Letter C, dated January 22, 1999
- 3. Comment Resolution Letter D, dated January 27, 1999

Dear Mr. Delligatti,

The Boiling Water Reactor (BWR) minimum enrichments, burnups, and cooling times used in the HI-STAR 100 System shielding analysis are provided in the table below. These enrichment, burnup, and cooling time combinations are applicable to all BWR assembly classes with the exception of LaCrosse, Dresden 1 6x6, and Humboldt Bay 6x6 assembly classes. This information is provided in accordance with Commitment C.5.3 (Ref. 2) as modified by Commitment D.5.1 (Ref. 3). This information will be included in Revision 8 of the HI-STAR 100 System Safety Analysis Report to be submitted to the NRC by February 22, 1999.

BWR Enrichment, Burnup, and Cooling Times

Minimum Enrichment (wt% U-235)	Maximum Burnup (MWD/MTU)	Minimum Cooling Time (years)
2.1	24,500	8
2.4	29,500	9
2.6	34,500	12
2.9	39,500*	15

* The maximum allowed burnup for this enrichment in SAR Chapter 1 will be specified slightly lower based on decay heat considerations.



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 29, 1999 Page 2 of 3

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014257

Approvals:

Brian Gutherman Licensing Manager

Technical Concurrence:

Dr. Everett Redmond (Shielding Evaluation)

1 P Pine4

K. P. Singh, Ph.D. President and CEO

- Ihome

Distribution:

Project

Utility

Holtec Project

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Ken Phy Mr. David Larkin

Southern Nuclear Operating Company 71188 50438 Commonwealth Edison 71178 Pacific Gas & Electric Co. Private Fuel Storage, LLC 70651 70851 American Electric Power New York Power Authority 80518 Washington Public Power Supply System



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission January 29, 1999 Page 3 of 3

Distribution (cont'd):

Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 1, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter G

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter A, dated January 15, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide the NRC with information in accordance with Holtec International's Commitment A.1.1 from Comment Resolution Letter "A" (Ref. 2). Attached please find draft Revision 8 HI-STAR 100 Safety Analysis Report (SAR) pages 1.2-34 through 1.2-40. These changes have been made to align the fuel description in SAR Chapter 1 with the descriptions in the Technical Specifications contained in the proposed Certificate of Compliance for HI-STAR 100 storage, currently in proposed rulemaking. In addition, generic note 1 for these tables has been modified based on feedback from the Holtec Users' Group to provide clarification for implementation. With the NRC's concurrence, Holtec will propose the same change for the HI-STAR 100 storage application through the public comment process.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014259



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 1, 1999 Page 2 of 2

Approvals:

Brian Gutherman Licensing Manager

) Pin H

K. P. Singh, Ph.D. President and CEO

Distribution (w/o attach.) :

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Ken Phy Mr. David Larkin Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Private Fuel Storage, LLC American Electric Power New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One GPUN – Oyster Creek Nuclear Power Station IES Utilities Nebraska Public Power District

Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 2, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

-

INTERNATIONAL

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter H

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter A, dated January 15, 1999

3. Comment Resolution Letter B, dated January 20, 1999

4. Comment Resolution Letter D, dated January 27, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide information to the NRC in accordance with prior commitments made by Holtec International regarding the Spent Fuel Project Office's (SFPO) ongoing review of our HI-STAR 100 System under 10CFR71. In addition, one new item resulting from a conversation held between Holtec and the SFPO staff today is included.

Commitment A.2.9

The following update to Commitment A.2.9 is applicable. Holtec's Position Paper D-213 regarding critical weld flaw size in the MPC lid-to-shell weld, submitted to the NRC in Comment Resolution Letter B (Ref. 3), was discussed with the SFPO staff reviewers. The staff asked Holtec for information to supplement the linear fracture mechanics evaluation presented in the position paper with an analysis methodology similar to the one used in the NEI document dated October 20, 1998. Additionally, a specific dimension for, and number of liquid penetrant examinations (i.e., root, final, and intermediate weld layers) needs to be included in the conclusions of the position paper. Appropriate editorial changes to the SAR need to be made.

HOLTEC INTERNATIONAL

Holtec Center, 555 Lincoln Drive West, Marlton, NJ 08053

Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 2, 1999 Page 2 of 5

Position Paper DS-213 will be revised to consider the methodology outlined in the NEI position paper and submitted to the NRC by February 8, 1999. As advised by the SFPO staff, draft revised SAR pages are not necessary to resolve this issue and will not be submitted per Commitments A.1.8, A.2.9, A.7.3, A.8.1, and A.8.3 (Ref. 2). Instead, affected SAR pages will be revised and submitted with Revision 8 of the SAR by February 22, 1999.

Commitment D.5.1

)

The Pressurized Water Reactor (PWR) minimum enrichments, burnups, and cooling times used in the HI-STAR 100 System shielding analysis are provided in the table below. These enrichment, burnup, and cooling time combinations are applicable to all PWR assembly classes with the exception of Haddam Neck and San Onofre 1 assembly classes. This information satisfies Commitment D.5.1 (Ref. 4).

Minimum Enrichment (wt% U-235)	Maximum Burnup (MWD/MTU)	Minimum Cooling Time (years)
2.3	24,500	10
2.6	29,500	12
2.9	34,500	14
3.2	37,500	15

PWR Enrichment, Burnup, and Cooling Times for Fuel with Inconel Grid Spacers in the Active Fuel Region



February 2, 1999 Page 3 of 5 Holtec Center, 555 Lincoln Drive West, Marlton, NJ 08053

Telephone (609) 797-0900 Fax (609) 797-0909

PWR Enrichment, Burnup, and Cooling Times for Fuel with Zircaloy Grid Spacers in the Active Fuel Region

Minimum Enrichment (wt% U-235)	Minimum Enrichment (wt% U-235)Maximum Burnup (MWD/MTU)	
2.3	24,500	
2.6	29,500	8
2.9	34,500	10
3.2	39,500	12
3.4	44,500*	15

* The maximum allowed burnup for this enrichment in SAR Chapter 1 will be specified slightly lower based on decay heat considerations.

NEW ITEM

- H.5.1: SAR Figure 5.3.10 shows a length of 170.125 inches for the vertical dimension of the outside of the enclosure shell housing the neutron shield. The design drawings show 173.125 for the same dimension. The 170.125 inch length appears to be the height of the neutron shield, excluding the thermal expansion foam. Where do the design drawings show the thermal expansion foam material dimension? Please clarify.
 - <u>Commitment</u>

The vertical dimension of the outside of the enclosure shell housing the neutron shield is 173.125 inches per Design Drawing 1397, Sheet 1. The top and bottom plates (enclosure shell returns, BOM item 17) are one half inch thick and the thermal expansion foam depth is two inches. This creates the three inch difference between the outside dimension of the enclosure shell and the neutron shield material vertical height. The 2-inch depth of foam material is shown on Design Drawing 1398, Sheet 2 at Section D-D. The editorial change to SAR Figure 5.3.10 will be made and submitted with Revision 8 of the SAR by February 22, 1999.



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 2, 1999 Page 4 of 5

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014260

Approvals:

Brian Gutherman Licensing Manager

Technical Concurrence:

Dr. Everett Redmond (Shielding Evaluation)

Dr. Alan Soler (Structural Evaluation)

Distribution:

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Co. Private Fuel Storage, LLC American Electric Power

44

K. P. Singh, Ph.D. President and CEO



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 2, 1999 Page 5 of 5

Distribution (cont'd):

Mr. Ken Phy Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 3, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter I

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter A, dated January 15, 1999

Dear Mr. Delligatti,

ł

The purpose of this letter is to provide on-time responses per our undertaking in Comment Resolution Letter A (Ref. 2) to the Spent Fuel Project Office (SFPO) in support of the ongoing review of our HI-STAR 100 System under 10CFR71. In addition, one new item resulting from a conversation held between Holtec and the SFPO staff today is included.

Commitment A.1.4

Draft Revision 8 Safety Analysis Report (SAR) Tables 1.2.1 and 2.2.3 are provided as Attachment 1 to this letter. These changes demonstrate that the maximum weight *applied to the trunnions* is less than 250,000 lbs. Previously, in addition to the weight of the lift yoke (3,600 lbs.) being included in the maximum lifted weight on crane hook over fuel pool, the MPC weight used was that of the MPC-32 (89,057 lbs.) which bounds the weight of the MPC-24 and MPC-68. Since the MPC-32 has been removed from the application, the weight has been revised to reflect the heaviest MPC in the SAR which is the MPC-68 at 87,241 lbs. The resulting weight on the trunnions is 249,460 as shown on the revised Tables 1.2.1 and 2.2.3.

The total lifted weight on the crane hook over the fuel pool is 253,060 lbs. This weight is clearly specified in Tables 1.2.1 and 2.2.3. In SAR Chapter 7, Section 7.1.5, Step 13, a warning is given to alert the user that the weight of the system being removed from the fuel pool may exceed 250,000 lbs and that users may elect to pump a measured quantity of water from the MPC prior to removal from the fuel pool. The warning also refers to Tables 7.1.1 and 7.1.2 where the weights are described for the various components in each major operations sequence. Draft





Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 3, 1999 Page 2 of 5

Revision 8 SAR Tables 7.1.1 and 7.1.2 are also provided in Attachment 1 to this letter to correspond with the Table 1.2.1 and 2.3.3 revisions.

Commitment A.1.5

Draft Revision 8 SAR Table 1.3.3 is provided as Attachment 2 to this letter to specify the surface preservative between to the intermediate shells. The surface preservative is a silicone encapsulant, Dow Corning SYLGARD 567 or equivalent.

Commitment A.1.6

Draft Revision 8 SAR page 1.2-20 is provided as Attachment 3 to this letter to read "250 x 250 fine mesh screens" in lieu of "250 micron fine mesh screens."

Commitment A.1.7

- Design Drawing 1765, Sheet 6 specifies the nominal crush strength for the aluminum honeycomb in the impact limiter. SAR Subsection 8.1.4.3 states that the tolerance on the crush strength is \pm 7%. The nominal crush strength was specified such that the maximum crush strength (i.e., nominal plus 7%) would be approximately 95% of the crush strength used in the ¹/₄ scale drop tests. Therefore, the nominal crush strength, including tolerances, will not exceed the crush strength values used in the ¹/₄ scale drop tests.

To illustrate this, the following table provides the nominal crush strength specified on Drawing 1765, Sheet 6, the maximum crush strength including tolerances, and the average crush strength used in the ¼ scale drop tests.

Impact Limiter	Nominal Crush	Maximum Crush	Average Crush Strength
Section Type from	Strength for Full-	Strength for Full-	for ¼ Scale Drop Tests
Drawing 1765, Sheet	Scale (psi)	Scale (psi)	(psi)
6			
Section Type 1	1700	1819	1943
Section Type 2	700	749	783
Section Type 3	2300	2461	2583
Section Type 4	1100	1177	1350
Section Type 5	2300	2461	2583



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 3, 1999 Page 3 of 5

As depicted above, the maximum crush strength for the full scale impact limiter will not exceed the crush strength used in the ¼ scale drop tests. No changes are necessary to the drawings or SAR as a result of this commitment.

Commitment A.2.2

The outermost two layers of the intermediate shells have been re-classified to meet the stress intensity limits of ASME Section III, Subsection NF, Class 1 in lieu of Class 3. Draft Revision 8 SAR Tables 2.1.5 and 2.1.15 are provided as Attachment 4 to this letter to show the allowable stress intensities for the outermost two intermediate shells under normal conditions. Note that the allowable limits for hypothetical accidents for NF components are the same for both Class 1 and Class 3. Therefore, no change in safety factors occurs for any accident evaluation. Also, draft Revision 8 SAR Table 2.1.23 is provided in Attachment 4 to specify the appropriate Charpy requirements for the outermost two intermediate shells.

Large margins currently exist for the intermediate shells under normal conditions. SAR Section 2.6 is being reviewed and updated to meet other commitments; any page changes required as a result of changing the outermost two intermediate shells to Class 1 from Class 3 will be provided by February 12, 1999.

Commitment A.8.2

The calculated maximum weight on the HI-STAR 100 trunnions has been revised per Commitment A.1.4 to be less than 250,000 lbs. Therefore, no changes to SAR Section 8.1.2.1 are required.

NEW ITEM

I.5.1: The NRC staff requested additional clarification be provided in the SAR regarding fuel spacers and the axial position of fuel assemblies. In addition, the staff requested additional discussion on the density of Holtite assumed during normal conditions and the depletion of ¹⁰B in the Holtite over the life of the cask.

Commitment

Additional clarification will be added to the SAR as requested. Draft Revision 8 SAR pages will be submitted to the NRC by February 12, 1999.



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 3, 1999 Page 4 of 5

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014261

Approvals:

Brian Gutherman Licensing Manager

Technical Concurrence:

Dr. Everett Redmond II (Shielding Evaluation)

Dr. Alan Soler (Structural Evaluation)

Mr. Steve Agace (Operations)

K. P. Singh, Ph.D. President and CEO



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 3, 1999 Page 5 of 5

Distribution:

Recipient

Ì

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Can Phy Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar

<u>Utility</u>

Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Co. Private Fuel Storage, LLC American Electric Power New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One



Telephone (609) 797-0900 Fax (609) 797-0909

BY OVERNIGHT MAIL

February 5, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter J

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter A, dated January 15, 1999

3. Comment Resolution Letter C, dated January 22, 1999

4. Comment Resolution Letter I, dated February 3, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide responses per our previous commitments in Comment Resolution Letters A, C and I (Refs. 2, 3, and 4) to the Spent Fuel Project Office (SFPO) in support of the ongoing review of our HI-STAR 100 System under 10CFR71. In addition, new items resulting from a conversation held between Holtec and the SFPO staff on February 4, 1999 are included.

Commitments A.2.3 and A.2.4

Draft Revision 8 SAR Section 2.6 and 2.7 excerpts are provided as Attachment 1 to this letter. These revisions provide an expanded discussion of the finite element models with text and figures in SAR Section 2.6. This revised text serves as a lead-in to the discussion in SAR Section 2.7. In Section 2.7, we are providing an expanded description of the applied loads for the side drop and oblique drop events. Additional figures to compliment the text are provided so that the impact limiter applied and reaction loads are clearly described.

Please note that SAR Revision 7 already includes stress analysis results for the CG-over-corner case. We will expand the discussion (e.g., stress results, safety factors) for specific areas of the overpack in draft Revision 8 SAR pages to be submitted by February 12, 1999. We will also provide a complete set of new results for the 30-degree (from horizontal) oblique drop event. As



Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 5, 1999 Page 2 of 5

demonstrated in the discussion of Commitment A.2.5 below, the 30-degree oblique drop is the worst case. Draft Revision 8 SAR Table 2.1.9 (included in Attachment 1) includes two additional cases covering this new drop condition.

Commitment A.2.5

Revised SAR pages 2.H-23 through 2.H-25 and new Figures 2.H.7.1 through 2.H.7.3 are provided as Attachment 2 to this letter. This information provides a comparison of the deceleration, maximum total crush, and available crush stroke of three variations of an oblique drop of the overpack (30, 45, and 60 degrees from horizontal) to determine the worst case for further stress analysis. The results show the 30-degree drop to be the worst case, warranting a detailed stress analysis.

Commitment C.5.3

Draft Revision 8 Safety Analysis Report pages 5.1-6 through 5.1-14, 5.2-12 through 5.2-14, 5.2-17 through 5.2-22, 5.4-9 through 5.4-20, and 5.4-23 through 5.4-28 are provided as Attachment 3 to this letter. These revisions show the calculated dose rates for the HI-STAR 100 System considering recent revisions made to fuel burnup, cooling time, and source terms in the shielding analyses.

Commitments A.7.1 and A.7.2

SAR Section 7.4 has been revised to address the procedural requirements for preparing the HI-STAR 100 overpack for transport following a period of storage. Draft Revision 8 SAR Section 7.4 is provided as Attachment 4 to this letter. Former SAR Sections 7.4 and 7.5 will be renumbered to 7.5 and 7.6, respectively as a result of this change. Draft Revision 8 SAR page 8.1-15 is also provided to specifically address the requirement in 10CFR71.87(g) to ensure the neutron absorber is present and in proper condition. The wording used here is similar to that used in a recently approved exemption to the requirements of 10CFR72.124(b) for the HI-STAR 100 storage application to show that this check is not necessary.

Commitment A.7.4

Draft Revision 8 SAR pages 7.1-13 and 7.2-7 are provided as Attachment 5 to this letter. The existing warning has been modified to instruct the users to ensure that plant-specific lifting devices are qualified to lift the expected load. A previous resolution to Commitment A.1.4 (Ref.

HOLTEC INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission

Telephone (609) 797-0900 Fax (609) 797-0909

4) established that the maximum calculated weight applied to the trunnions is less than 250,000 lbs.

Commitment A.7.5

February 5, 1999 Page 3 of 5

Draft Revision 8 SAR page 7.1-19 is provided as Attachment 6 to this letter. Loading operations Step 26.k now requires a liquid penetrant examination after weld repairs.

Commitment A.8.1

Draft Revision 8 SAR pages 8.1-14 and 8.1-23 are provided as Attachment 7 to this letter. New Subsection 8.1.5.2.1, "Pre-Transport Shielding Tests" has been added to address shielding effectiveness testing required to meet 10 CFR 71.47. These tests are to be performed prior to transporting HI-STAR 100 package to confirm the effectiveness of the neutron and gamma shielding.

NEW ITEMS

J.2.1: Perform an analysis to demonstrate that the HI-STAR 100 design basis deceleration of 60 g's for an end drop is within the buckling limit for the fuel rods being licensed for transport. The analysis should demonstrate that the fuel being licensed is bounded by the analysis.

<u>Commitment</u>

Holtec will perform an analysis to define the worst case fuel assembly or assemblies with respect to the buckling limit under an end drop hypothetical accident. Using the worst case assembly or assemblies, analysis will be performed to demonstrate that under the design basis deceleration of 60 g's, the worst case fuel assembly rods will not buckle. Draft Revision 8 SAR pages will be submitted to the NRC by February 12, 1999.

J.7.1: On SAR pages 7.1-14 (warning) and 7.1-29 (note), dose rates are specified to verify the correct fuel assemblies have been loaded. These actions may not be appropriate for a transport license.



Telephone (609) 797-0900 Fax (609) 797-0909

Commitment

Holtec will evaluate the validity of these statements for the procedures in the SAR. Changes, if required, will be incorporated into Revision 8 of the SAR to be submitted the NRC by February 22, 1999. Please note that the proposed Certificate of Compliance Holtec intends to submit for the HI-STAR 100 Transportation certification will include a requirement that licensee loading procedures include a loading plan and independent verification that only authorized fuel assemblies have been loaded into the MPC in accordance with the CoC.

J.7.2: Revise the operating procedures to include a discussion of the 2 mrem/hr requirement for any normally occupied space (10CFR71.47(b)(4)).

Commitment

Holtec will revise the affected operating procedures and incorporate the changes into Revision 8 of the SAR to be submitted to the NRC by February 22, 1999.

J.7.3: SAR page 7.1-30, Step 7.f should be revised to specify that the radiation surveys should check potential streaming paths and areas of higher dose rates.

Commitment

The affected step will be revised and incorporated into Revision 8 of the SAR to be submitted to the NRC by February 22, 1999.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014262


Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 5, 1999 Page 5 of 5

Approvals:

Brian Guthérman Licensing Manager

Singh.

E. REDMOND

President and CEO

FOR

Technical Concurrence:

Dr. Everett Redmond II (Shielding Evaluation)

Dr. Alan Soler (Structural Evaluation)

Mr. Steve Agace (Operations)

Mr. Mark Soler (Acceptance Criteria and Maintenance)

Distribution (w/o attach.) :

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Can Phy Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar

Southern Nuclear Operating Company Commonwealth Edison Pacific Gas & Electric Co. Private Fuel Storage, LLC American Electric Power New York Power Authority Washington Public Power Supply System Wisconsin Electric Power Company Maine Yankee Atomic Power Company Vermont Yankee Corporation Southern California Edison Entergy Operations – Arkansas Nuclear One



Telèphone (609) 797-0900 Fax (609) 797-0909

BY OVERNIGHT MAIL

February 8, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter K

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter A, dated January 15, 1999

3. Comment Resolution Letter C, dated January 22, 1999

4. Comment Resolution Letter H, dated February 2, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide responses per our previous commitments in Comment Resolution Letters A, C, and H (Refs. 2, 3 and 4) to the Spent Fuel Project Office (SFPO) in support of the ongoing review of our HI-STAR 100 System under 10CFR71.

Commitment A.1.3

To incorporate lessons learned in the prototype fabrication of the HI-STAR 100 overpack and MPC, revisions to the Design Drawings have been made between the HI-STAR 100 Topical Safety Analysis Report (TSAR), Revision 9 and the HI-STAR 100 Safety Analysis Report (SAR), Revision 7. Provided below is a detail description of each of the changes.

Dwg. 1396, Sheet 1 has been revised from Revision 9 to 11 to include the following changes:

- At location E-5, "see note 2 on this sheet" has been added to the inspection criteria of the MPC lid-to-shell weld.
- Note 2 has been added to read "As an alternative to volumetric examination (UT) of the MPC lid-to-shell weld, a multi-layer liquid penetrant (PT) examination may be performed. Multi-layer PT shall include examination of the root and final weld layers, and sufficient



Page 2 of 6

Telephone (609) 797-0900 Fax (609) 797-0909

intermediate layers to detect critical weld flaws." (Please note that this drawing note will be revised again for SAR Revision 8 to reflect the resolution of Commitment A.2.9 regarding the critical weld flaw size and number of intermediate PT examinations).

• The tolerance on the chamfer of the MPC baseplate has been changed from +/- 0.5 degrees to "(REF)".

Dwg. 1397, Sheet 1 has been revised from Revision 13 to 14 to include the following changes:

• Added "(see dwg. 1765, sht. 7 of 7 for orientation)" to the call-out for the impact limiter attachment holes and impact limiter alignment holes. Previously, there was no orientation for the holes given.

Dwg. 1402, Sheet 1 has been revised from Revision 10 to 12 to include the following changes:

- At location E-5, "see note 2 on this sheet" has been added to the inspection criteria of the MPC lid-to-shell weld.
- Note 2 has been added to read "As an alternative to volumetric examination (UT) of the MPC lid-to-shell weld, a multi-layer liquid penetrant (PT) examination may be performed. Multi-layer PT shall include examination of the root and final weld layers, and sufficient intermediate layers to detect critical weld flaws." (Please note that this drawing note will be revised again for SAR Revision 8 to reflect the resolution of Commitment A.2.9 regarding the critical weld flaw size and number of intermediate PT examinations).
- The tolerance on the chamfer of the MPC baseplate has been changed from +/- 0.5 degrees to "(REF)".

BM-1476, Sheets 1 and 2 have been revised from revision 11 to 12 to include the following changes:

- Deleted the term "MIN." after 2 ½" in the description of Item 2, inner shell, because the thickness of plate in the HI-STAR 100 overpack is to be governed by the ASME Code.
- Revised the description of Item 24 to read "Holtite-A" in lieu of "NS-4-FR" to remain consistent with the HI-STAR 100 TSAR and SAR nomenclature.



Telephone (609) 797-0900 Fax (609) 797-0909

• Added "or SA-53" to the material description of Item 38 which is an equivalent material description for the pipe.

Commitment A.2.9

Holtec International Position Paper DS-213, Revision1, is provided as Attachment 2 to this letter. This revised position paper adds a supplemental analysis using a methodology similar to the one discussed in the October 20, 1998 NEI position paper regarding dry storage cask weld cracks. The results of the analysis demonstrate that an assumed 3/8 inch weld flaw size, around the entire circumference of the MPC lid, will not propagate. Therefore, only one intermediate weld layer liquid penetrant (PT) examination is required in addition to the root and final layer PTs. This information will be added to the applicable drawings and text sections in SAR Revision 8, to be submitted to the NRC by February 22, 1999.

Commitment A.2.10

New SAR Section 3.4.3.1 has been added and existing SAR Section 2.6.2.3 has been revised to address the thermal and structural effects, respectively, of a scenario where a loaded overpack is transported from a hot ambient environment to a cold ambient environment. Draft Revision 8 SAR pages for the affected SAR sections are provided as Attachment 3 to this letter.

Commitment A.3.1

The offgassing created by the overheating of the Holtite-A neutron shield material during a fire accident has been evaluated. Draft Revision 8 SAR pages discussing this accident are provided as Attachment 4 to this letter. Please note that the Bill-of-Materials will be revised in SAR Revision 8 to specify rupture disks with larger diameter openings to provide sufficient relieving capacity.

Commitment A.3.2

The HI-STAR 100 SAR in Subsection 3.4.1.1.5 describes the "Rayleigh Effect" in the large peripheral regions of the MPC basket. This effect is caused by local convection cells formed as a result of gravity acting on heat flow-induced temperature gradients in these fluid filled regions when the HI-STAR cask is postulated to be in a perfectly horizontal condition. An additional thermal analysis has been performed wherein the gravity is postulated to be completely absent. In this condition, the fluid in these peripheral regions will be completely motionless (i.e. "Rayleigh Effect" is absent) and the means of heat dissipation is limited to conduction and



Telephone (609) 797-0900 Fax (609) 797-0909

radiation. Tables A.3.2.1 and A.3.2.2 for the MPC-24 basket and MPC-68 basket designs, respectively, are provided as Attachment 5 to this letter. These tables summarize the normal transport condition maximum temperatures under this condition and compare the results with normal condition limits. The basket, cladding and overpack containment boundary temperatures are all within stipulated limits under the postulated zero gravity condition.

Commitment A.3.3

Page 4 of 6

The HI-STAR 100 thermal model takes no credit for heat transfer through the overpack impact limiters. A statement clarifying this conservative assumption has been added to Section 3.4.2 of the HI-STAR 100 SAR. Draft Revision 8 SAR page is provided as Attachment 6 to this letter.

Commitment A.3.4

Access to the HI-STAR 100 overpack enclosure shell is restricted by the use of a personnel barrier (See Holtec Drawing 1809 in Chapter 1 of HI-STAR SAR). Access is possible to the impact limiter exposed surfaces. A new Subsection 3.4.2.1 has been added to Chapter 3 of the SAR to address this issue. In this subsection, the HI-STAR 100 System thermal model for normal transport is solved by including heat dissipation through the impact limiters. A conservatively bounding analysis has been performed by applying the thermal conductivity of aluminum to the honeycomb material encased in the impact limiters. In this manner heat transport to the exposed surfaces from the hot ends of the cask is maximized and accessible surface temperatures over estimated. The draft Revision 8 SAR pages are provided as Attachment 7 to this letter. The maximum exposed surface temperatures are 142°F and 139°F for the MPC-24 and MPC-68 basket designs.

Commitment A.3.5

High temperature creep rupture data presented in EPRI Report TR-106440 has been referenced as the basis for short term temperature limits for stainless steel cladding. The draft Revision 8 SAR pages are provided as Attachment 8 to this letter.

Commitment C.5.3

Draft Revision 8 SAR Tables 5.5.1 and 5.5.3 are provided as Attachment 9 to this letter. These tables present the maximum normal and accident dose rates for the MPC-24 and MPC-68, respectively.

Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 8, 1999 Page 5 of 6

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014263

Attachments: 1. Affidavit Pursuant to 10 CFR 2.790

Others As Stated

Approvals:

Brian Gutherman Licensing Manager

Technical Concurrence:

Dr. Indresh Rampall (Thermal Evaluation)

Dr. Alan Soler (Structural Evaluation)

Dr. Everett Redmond II (Shielding Evaluation)

K. P. Singh, Ph.D. President and CEO

Indrella

Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 8, 1999 Page 6 of 6

Distribution (w/o attach.) :

<u>Recipient</u>

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. David Larkin Mr. David Larkin Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 10, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter L

References: 1. Holtec Project No. 5014

2. Comment Resolution Letter D, dated January 27, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide responses per our previous commitments in Comment Resolution Letter D (Ref. 2) to the Spent Fuel Project Office (SFPO) in support of the ongoing review of our HI-STAR 100 System under 10CFR71. In addition, new commitments resulting from a phone conversation held this morning are included.

Commitments D.5.2 and D.5.3

Draft Revision 8 SAR Sections 5.2.3 and 5.4.4 are provided in Attachment 1 to this letter. These sections discuss the evaluation of stainless steel clad fuel. Draft Revision 8 SAR Tables 5.2.18 through 5.2.22 and 5.4.22 through 5.4.24 are also provided in Attachment 1. These tables provide the calculated source terms and dose rates for the stainless steel clad fuel.

The minimum ²³⁵U enrichment for the damaged and MOX fuel has been lowered to 1.8 wt.% ²³⁵U. Draft Revision 8 SAR Sections 5.4.2 and 5.4.3 and Tables 5.2.2, 5.2.6, and 5.2.14 through 5.2.17 are provided in Attachment 1. These sections and tables provide the discussion and calculated results for the damaged and MOX fuel.



Telephone (609) 797-0900 Fax (609) 797-0909

NEW ITEMS

L.1.1: Bolt torque values need to be on the Design Drawings or added as a license condition.

Commitment

February 10, 1999

Page 2 of 3

The bolt torques will be added to the Design Drawings included in the proposed Certificate of Compliance to be submitted to the NRC by February 22, 1999.

L.1.2: Drawing 1765, Sheet 7 has a reference to a pin and cap detail to Drawing 1546, which is not included in the SAR.

Commitment

There are two references to the pin and cap details to which this note is referring. The details of the full scale pin and cap may be found on Sheet 2 of Drawing 1765, which can be found in the SAR. The details for the ¼ scale model are shown on Drawing 1546, Sheet 1. The reference on Drawing 1765, Sheet 7 will be clarified and submitted with SAR Revision 8 by February 22, 1999.

L.1.3: The minimum 6 foot distance between the bottom impact limiter and the edge of the transport vehicle should be specified in the Certificate of Compliance as a condition of transportation.

Commitment

The 6 foot distance will be added as a condition in the proposed Certificate of Compliance to be submitted to the NRC by February 22, 1999.

If you have any questions or comments, please contact us.

Sincerely, (

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

HOLTEC INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 10, 1999 Page 3 of 3

Telephone (609) 797-0900 Fax (609) 797-0909

Document I.D.: 5014264

Attachment: As Stated

Approvals:

Brian Gutherman Licensing Manager

Technical Concurrence:

Dr. Everett Redmond II (Shielding Evaluation)

Distribution (w/o attach.) :

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. David Larkin Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz

ch K. P. Singh. Ph.D

President and CEO



Telephone (609) 797-0900 Fax (609) 797-0909

BY OVERNIGHT MAIL

February 11, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter M

Reference: Holtec Project No. 5014

Dear Mr. Delligatti,

The purpose of this letter is to provide information and to document commitments resulting from phone conversations held between Holtec International and the Spent Fuel Project Office (SFPO) over the past two days. Attachment 1 to this letter provides source term inventories from the ORIGEN computer code for the MPC-24, -68, and -68F used in the containment analyses. They are identified at the top of each inventory list by "BW15X15", "7X7", and "6X6", respectively. Please note that this computer output is information which is commercially sensitive to Holtec International and is treated by us with strict confidentiality. This information is of the type described in 10CFR2.790(b)(4). Its proprietary status is noted at the bottom of each page of output data. The affidavit provided as Attachment 2 to this letter sets forth the bases for which the information is required to be withheld by the NRC from further disclosure, consistent with these considerations and pursuant to the provisions of 10CFR2.790(b)(1). It is therefore requested that the proprietary information enclosed be withheld from public disclosure in accordance with applicable NRC regulations.

NEW ITEMS

M.4.1: Safety Analysis Report (SAR) Table 4.2.5 appears to have a different A₂ value for crud (Co-60) than provided in Table A-1 of Appendix A to 10 CFR 71.

Commitment

The A_2 value for crud in SAR Table 4.2.5 will be modified to match the value used in the containment evaluation (10.8), which agrees with the value in Table A-1 of Appendix A to 10 CFR 71. Draft Revision 8 SAR pages will be provided to the NRC by February 12, 1999.



Telephone (609) 797-0900 Fax (609) 797-0909

M.4.2: SAR Table 4.2.2 includes isotopes not included in Table A-1 of Appendix A to 10 CFR
71. Please provide clarification of how the isotopic inventory used in the containment analysis was developed and what value of A₂ was used for each isotope.

Commitment

As requested by the NRC, the following methodology will be used to develop the radionuclide inventories for the containment analyses:

- 1. All isotopes constituting 0.01% of inventory or greater (per NUREG-6487);
- 2. Isotopes from the ORIGEN output which have an A₂ value from Table A-1 of Appendix A to 10 CFR 71 less than 1.0;
- 3. Isotopes having half-lives less than 10 days may be neglected since they are included with the parent isotope in Table A-1; and
- 4. For those isotopes not having an A₂ value in Table A-1, the default values in Table A-2 of Appendix A to 10 CFR 71 will be used.

Draft Revision 8 SAR pages will be submitted to the NRC by February 12, 1999.

M.4.3: The containment analyses must include Kr-85 with other gaseous isotopes. Kr-85 should not be evaluated on its own. Additionally, the A₂ value for Kr-85 must be ten times the A₂ value presented in Table A-1 of Appendix A to 10 CFR 71.

Commitment

The containment analysis will be re-evaluated with Kr-85 included in the gaseous mixture. The effective A_2 value for the mixture will be determined as requested. Draft Revision 8 SAR pages will be submitted to the NRC by February 12, 1999.

M.4.4: Provide the equivalent diameter and volumetric leak rate used in the containment analyses.



Telephone (609) 797-0900 Fax (609) 797-0909

Commitment

Page 3 of 4

The additional information requested will be added to the SAR Draft Revision 8 SAR pages will be submitted to the NRC by February 12, 1999.

M.4.5: SAR Appendix 4A appears to have a typographical error in some of the equations (missing pi symbol).

Commitment

Appendix 4A will be reviewed and any editorial changes made. These changes will be included in the submittal of SAR Revision 8 to be submitted to the NRC by February 22, 1999.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014265

Attachments: 1. ORIGEN Computer Output of Isotopic Inventories

2. Affidavit Pursuant to 10 CFR 2.790

Approvals:

Brian Gutherman Licensing Manager

Jemel

K. P. Singh, Ph.D. President and CEO

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 11, 1999 Page 4 of 4

Holtec Center, 555 Lincoln Drive West, Marlton, NJ 08053

Telephone (609) 797-0900 Fax (609) 797-0909

Technical Concurrence:

Dr. Everett Redmond II (Shielding Evaluation)

Ms. Joy Russell (Containment Evaluation)

Joy Russell

Distribution (w/o attach.):

Recipient

Utility

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Ken Phy Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz



Telephone (609) 797-0900 Fax (609) 797-0909

BY OVERNIGHT MAIL

February 12, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter N

References: 1. Holtec Project No. 5014

2. Holtec Comment Resolution Letter A, dated January 15, 1999

3. Holtec Comment Resolution Letter D, dated January 27, 1999

4. Holtec Comment Resolution Letter I, dated February 3, 1999

5. Holtec Comment Resolution Letter J, dated February 5, 1999

6. Holtec Comment Resolution Letter M, dated February 11, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide information in accordance with Holtec International's commitments in Comment Resolution Letters (CRL) "A", "D", "T", "J", and "M" (Refs. 2 through 6). In addition, one new commitment resulting from a phone conversation with the Spent Fuel Project Office today is included.

Commitments A.2.3 and A.2.4

Draft Revision 8 Safety Analysis Report (SAR) Section 2.7 excerpts are provided as Attachment 1 to this letter. The revisions to Section 2.7 provide expanded discussion (e.g., stress results, safety factors) for the CG-over-corner drop analyses. These revisions complement the information provided in CRL "J" (Ref. 5).



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 11, 1999 Page 2 of 4

Commitment J.2.1

Draft Revision 8 SAR Section 2.9 is provided as Attachment 2 to this letter. This section has been revised to add discussion on fuel buckling under the hypothetical accident drop loading.

Commitments M.4.1 through M.4.4

Draft Revision 8 SAR pages are provided as Attachment 3 to this letter. These changes address the requested revisions to the isotopic inventories used in the containment analysis.

Chapter 5, Shielding Evaluation

Draft Revision 8 SAR pages from Chapter 5 are provided as Attachment 4 to this letter. The following specific items have changes and are provided in Attachment 4.

Sections 5.1, 5.1.1, and 5.1.2 - minor changes (table numbers etc.) Figure 5.1.1 - minor change to bottom impact limiter Sections 5.2.1 and 5.2.2 Sections 5.2.5.1 and 5.2.5.2 - removed reference to Table 5.2.23 Tables 5.2.1, 5.2.2, 5.2.15, 5.2.18, and 5.2.23 Sections 5.3.1, 5.3.1.1, and 5.3.2 Parts of Table 5.3.2 and Table 5.3.3 Figures 5.3.11 and 5.3.12 Section 5.4 - minor changes Section 5.4.1 Tables 5.4.14 and 5.4.15 Figure 5.4.1 Section 5.5 - minor change to last paragraph

Commitment D.7.1

Draft Revision 8 SAR Section 7.3 is provided as Attachment 5 to this letter. This section has been revised to implement the requirements of 49 CFR 173.428.

HOLTEC INTERNATIONAL

Holtec Center, 555 Lincoln Drive West, Marlton, NJ 08053

Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 11, 1999 Page 3 of 4

NEW ITEM

N.2.1: Provide additional information on the slap-down analysis which clarifies the loadings, stresses, etc.

Commitment

The requested clarifications will be provided as draft Revision 8 SAR pages by February 15, 1999.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014267

Attachments: As Stated

Approvals:

Brian Gutherman Licensing Manager

Technical Concurrence:

Dr. Everett Redmond II (Shielding Evaluation)

Ms. Joy Russell (Containment Evaluation)

Mr. Steve Agace (Operations)

Dr. Alan Soler (Structural Evaluation)

eing 4

K. P. Singh, Ph.D. President and CEO

2h0n00]



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 11, 1999 Page 4 of 4

Distribution (w/o attach.) :

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Cavid Larkin Mr. David Larkin Mr. David Larkin Mr. Paul Plante Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 15, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter O

References: 1. Holtec Project No. 5014

2. Holtec Comment Resolution Letter N, dated February 12, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide information in accordance with Holtec International's commitment in Comment Resolution Letters (CRL) "N" (Ref. 2).

Commitment N.2.1

Draft Revision 8 Safety Analysis Report (SAR) Section 2.7 excerpts; Tables 2.1.9, 2.7.3, 2.7.5, and 2.AE.27; and Figure 2.7.18 are provided as Attachment 1 to this letter. These revisions provide the additional information regarding the slapdown analysis.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014269

Attachment: As Stated

Telephone (609) 797-0900 Fax (609) 797-0909

HOLLEC INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 15, 1999 Page 2 of 2

Approvals:

Brian Gutherman Licensing Manager K.P. Singlmp

K. P. Singh, Ph.D. President and CEO

Technical Concurrence:

Dr. Alan Soler (Structural Evaluation)

Distribution (w/o attach.) :

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. David Larkin Mr. David Larkin Mr. Eric Meils Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 17, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

- Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter P
- References: 1. Holtec Project No. 5014
 - 2. Holtec Comment Resolution Letter O, dated February 15, 1999

Dear Mr. Delligatti,

The purpose of this letter is to provide additional clarifying information to supplement the information provided in the above-referenced Comment Resolution Letter (CRL) and to document a commitment made in a telephone conversation with the Spent Fuel Project Office (SFPO) held this morning. As discussed with the SFPO during a telephone conversation yesterday, all of the finite element analyses for the HI-STAR 100 hypothetical accident conditions were re-run to include an enhancement to the computer model which more accurately predicts the response of the overpack during the drop accidents. The results of these runs were presented in draft Revision 8 Safety Analysis Report (SAR) Table 2.7.5 submitted with the above-referenced CRL.

The reason for re-running the hypothetical accident finite element analyses is twofold:

- 1. In performing the new analyses of oblique drops, a minor discrepancy in the element generation portion of the ANSYS model was discovered. This led to spurious reactions being developed in directions that were not physically meaningful in the case of oblique impact simulation. To ensure absolute quality assurance and precision in the results data provided in the SAR, all hypothetical drop accident cases were re-run using the corrected model, and all affected SAR tables updated.
- 2. In the simulations performed for SAR Revision 7, the secondary stress intensities due to thermal gradients (in the "heat" condition) were <u>retained</u> in the simulation and reported as if they were primary stress intensities. In the Revision 8 tabular results, these secondary thermal stress intensities were eliminated in the calculation of safety factors since only

Telephone (609) 797-0900 Fax (609) 797-0909

INTERNATIONAL Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 17, 1999 Page 2 of 3

primary stress intensities are required. This is noted in the text of SAR Section 2.7 where the models are discussed. Specifically, in subsection 2.7.1 (last paragraph before subsection 2.7.1.1), it is noted that stress intensities arising from the thermal gradients in the "heat" condition of transport are not included in the safety factor evaluation.

NEW ITEM

P.8.1: The proposed requirements for verifying neutron shield integrity include reference to compliance with 10 CFR 71.47. Compliance with this regulation is already addressed in the operating procedures in SAR Chapter 7. The acceptance test in Chapter 8 should address verification of the integrity of the poured Holtite-A neutron shield one time prior to first use (shipment).

Commitment

Proposed new SAR Section 8.1.5.2.1 and reference to 10 CFR 71.47 will be deleted. A one-time neutron shield verification test has been added to subsection 8.1.5.2 and Table 8.1.2 has been revised accordingly. The SAR now requires users to implement procedures to verify the integrity of the poured neutron shield. Draft Revision 8 SAR pages showing these changes are provided as Attachment 1 to this letter.

If you have any questions or comments, please contact us.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014270

Attachment: As Stated



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 17, 1999 Page 3 of 3

Approvals:

Brian Guthérman Licensing Manager

1 ing

K. P. Singh, Ph.D. President and CEO

Technical Concurrence:

Dr. Everett Redmond II (Shielding Evaluation)

Dr. Alan Soler (Structural Evaluation)

Distribution (w/o attach.) :

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. David Larkin Mr. David Larkin Mr. Paul Plante Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 18, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter Q

References: 1. Holtec Project No. 5014

Dear Mr. Delligatti,

The purpose of this letter is to provide information discussed in telephone conversations with the Spent Fuel Project Office (SFPO) held yesterday afternoon and this morning. Attachment 1 to this letter is draft Revision 8 changes to Safety Analysis Report (SAR) Section 2.7 which show the MPC-68F calculations for the MPC lid-to-shell (LTS) weld stresses under a hypothetical corner drop event.

These calculations were performed using the existing 0.75 inch LTS weld depth and 0.5 inch MPC shell thickness. We believe these calculations show sufficient margin to failure for the weld to ensure continued lid-to-shell weld joint integrity in the wake of a top-down corner drop event. Please consider the review of this information as a starting point for further discussions for achieving the SFPO's acceptance of the MPC-68F lid-to-shell weld configuration.

We highly appreciate the diligence and sense of purpose exhibited by the SFPO staff in this matter.

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014271

Attachment: As Stated



Telephone (609) 797-0900 Fax (609) 797-0909



Approvals:

Brian Gutherman Licensing Manager

lugy

K. P. Singh, Ph.D. President and CEO

Distribution (w/o attach.) :

<u>Recipient</u>

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. David Larkin Mr. David Larkin Mr. David Larkin Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz



Telephone (609) 797-0900 Fax (609) 797-0909

BY FAX AND OVERNIGHT MAIL

February 19, 1999

Mr. Mark S. Delligatti Senior Project Manager Spent Fuel Licensing Section, SFPO, NMSS U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852

Subject: USNRC Docket No. 71-9261 HI-STAR 100 Safety Analysis Report, TAC No. L22085 Comment Resolution Letter R

References: 1. Holtec Project No. 5014

Dear Mr. Delligatti,

The purpose of this letter is to provide information in response to our telephone conversation with the Spent Fuel Project Office (SFPO) held this morning. The MPC-68F shell and lid-to-shell weld designs have been slightly modified to reduce the stresses resulting from a CG-over-corner top-end drop accident. For the MPC-68F <u>only</u>, the thickness of the top of the MPC shell has been increased to 1.0 inch and the lid-to-shell groove weld thickness has been increased to 1 1/4 inches. Attachment 1 to this letter includes updated draft Revision 8 changes to Safety Analysis Report (SAR) Section 2.7 which show the MPC-68F calculations for the MPC lid-to-shell (LTS) weld stresses under this hypothetical corner drop event.

These changes were made to strengthen the weld and shell to meet the stress acceptance limits for this event suggested by the SFPO structural reviewer. The design change is illustrated in SAR Figure 2.7.22 (attached), which clearly indicates that the required modification is conceptually quite straightforward. We have discussed the change with each of the SAR chapter authors and have established that only the structural evaluation (Section 2.7) needs to be updated. Upon SFPO concurrence with the draft material presented herein, the affected Design Drawings and Safety Analysis Report section will be revised and submitted within 24 hours.

Please note that we recognize the schedule burden the SFPO staff is under in preparing the Safety Evaluation Report for this application. We deeply appreciate your continued efforts in reaching this crucial milestone in the licensing process.



Telephone (609) 797-0900 Fax (609) 797-0909

Mr. Mark Delligatti U. S. Nuclear Regulatory Commission February 19, 1999 Page 2 of 2

Sincerely,

Bernard Gilligan Project Manager, HI-STAR/HI-STORM Licensing

Document I.D.: 5014272

Attachment: As Stated

Approvals:

Brian Guthørman Licensing Manager

K. P. Singh, Ph.D. President and CEO

Distribution (w/o attach.) :

Recipient

<u>Utility</u>

Mr. David Bland Mr. J. Nathan Leech Mr. Bruce Patton Dr. Max DeLong Mr. Rodney Pickard Mr. Rodney Pickard Mr. Cand Larkin Mr. David Larkin Mr. David Larkin Mr. Paul Plante Mr. Paul Plante Mr. Stan Miller Mr. Jim Clark Mr. Ray Kellar Mr. Joe Andrescavage Mr. Ron Bowker Mr. William Swantz